UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-Q

(Mark One)

© QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended June 30, 2024

OR

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

to

For the transition period from

Commission File Number: 001-39346

MoneyLion Inc.

(Exact name of registrant as specified in its charter)

Delaware		85-0849243						
(State or other jurisdiction of incorporation or organization)		(I.R.S. Employer Identification No.)						
249-245 West 17 th Street, 4 th Floc New York, New York)r	10011						
(Address of principal executive offic	ces)		(Zip Code)					
	(212) 300 (Registrant's telephone num Securities registered pursuant	ber, including area code)						
Title of each class	Trading Sy	mbol(s)	Name of each exchange on which registered					
Class A common stock, par value \$0.0001 per share	ML		The New York Stock Exchange					
Redeemable warrants, each whole warrant exercisable for 1/30th of one share of Class A common stock	ML W	VS	The New York Stock Exchange					
Indicate by check mark whether the registrant (1) has filed 12 months (or for such shorter period that the registrant was								
Indicate by check mark whether the registrant has submit 232.405 of this chapter) during the preceding 12 months (or								
Indicate by check mark whether the registrant is a large a company. See the definitions of "large accelerated filer," "a	,	, , , , , , , , , , , , , , , , , , , ,						
Large accelerated filer		accelerated filer						
Non-accelerated filer	□ S	maller reporting company	図					
	E	merging growth company						
If an emerging growth company, indicate by check mark if accounting standards provided pursuant to Section 13(a) of	6	to use the extended transition p	eriod for complying with any new or revised financial					
Indicate by check mark whether the registrant is a shell com	pany (as defined in Rule 12b-	2 of the Exchange Act). Yes \Box N	No 🖂					
There were 10,968,455 shares of the registrant's Class A co	mmon stock, par value \$0.000	1 per share, outstanding as of A	ugust 1, 2024.					

MoneyLion Inc. TABLE OF CONTENTS QUARTERLY REPORT ON FORM 10-Q For the Quarterly Period Ended June 30, 2024

PART I – <u>FINANCIAL INFORMATION</u>

Item 1.	Financial Statements	1
	Unaudited Consolidated Balance Sheets	1
	Unaudited Consolidated Statements of Operations	2
	Unaudited Consolidated Statements of Redeemable Convertible Preferred Stock and Stockholders' Equity	3
	Unaudited Consolidated Statements of Cash Flows	5
	Notes to Unaudited Consolidated Financial Statements	6
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	23
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	39
Item 4.	Controls and Procedures	39
PART II -	OTHER INFORMATION	
Item 1	Legal Proceedings	41

Item 1.	Legal Proceedings	41
Item 1A.	<u>Risk Factors</u>	43
Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	45
Item 3.	Defaults Upon Senior Securities	45
Item 4.	Mine Safety Disclosures	45
Item 5.	Other Information	45
Item 6.	Exhibits	47
Signatures		49

i

Signatures

INTRODUCTORY NOTE

General

As used in this Quarterly Report on Form 10-Q, unless the context requires otherwise, references to "MoneyLion," the "Company," "we," "us," "our" and similar references refer to MoneyLion Inc. and, as context requires, its consolidated subsidiaries. "MALKA" refers to Malka Media Group LLC, a wholly-owned subsidiary of MoneyLion Technologies Inc., and "Engine" refers to ML Enterprise Inc., doing business as the brand "Engine by MoneyLion," a wholly-owned subsidiary of MoneyLion Technologies Inc. which was previously named "Even Financial Inc." and subsequently renamed in February 2023.

For convenience, the trademarks and service marks referred to in this Quarterly Report on Form 10-Q are listed without the [®], TM and SM symbols, but we intend to assert, and notify others of, our rights in and to these trademarks and service marks to the fullest extent under applicable law.

Reverse Stock Split

On April 24, 2023, the Company amended the Company's Fourth Amended and Restated Certificate of Incorporation (as amended from time to time, the "Certificate of Incorporation") to effect, effective as of 5:01 p.m. Eastern Time on April 24, 2023, a 1-for-30 reverse stock split (the "Reverse Stock Split") of the Class A common stock, par value \$0.0001 per share (the "Class A Common Stock"). At the effective time of the Reverse Stock Split, every 30 shares of Class A Common Stock either issued and outstanding or held as treasury stock were automatically reclassified into one new share of Class A Common Stock, and the total number of shares of Class A Common Stock authorized for issuance was reduced by a corresponding proportion from 2,000,000 shares to 66,666,666 shares. The Reverse Stock Split was approved by the Company's stockholders at a Special Meeting of Stockholders on April 19, 2023 and approved by the Board of Directors on April 21, 2023. The primary goal of the Reverse Stock Split was to increase the per share price of the Class A Common Stock in order to meet the minimum per share price requirement for continued listing on the New York Stock Exchange (the "NYSE"). The Class A Common Stock began trading on the NYSE on an as-adjusted basis on April 25, 2023 under the existing trading symbol "ML."

In addition, as a result of the Reverse Stock Split, proportionate adjustments were made to the number of shares of Class A Common Stock underlying the Company's outstanding equity awards, the number of shares issuable upon the exercise of the Company's outstanding warrants and the number of shares issuable under the Company's equity incentive plans and certain existing agreements, as well as the exercise, grant and acquisition prices of such equity awards and warrants, as applicable. Furthermore, proportionate adjustments were made to the conversion factor at which the Company's previously outstanding Series A Convertible Preferred Stock, par value \$0.0001 per share (the "Series A Preferred Stock"), converted to Class A Common Stock. The total number of shares of preferred stock of the Company authorized for issuance remained at 200,000,000. Stockholders who would have been entitled to receive fractional shares as a result of the Reverse Stock Split were instead entitled to a cash payment in lieu thereof at a price equal to the fraction of one share to which the stockholder was otherwise entitled multiplied by the closing price per share of the Class A Common Stock on the NYSE on the effective date of the Reverse Stock Split.

The effects of the Reverse Stock Split have been reflected in this Quarterly Report on Form 10-Q for all periods presented.

ii

Cautionary Statement Regarding Forward-Looking Statements

This Quarterly Report on Form 10-Q, including the information incorporated herein by reference, contains forward-looking statements regarding, among other things, the plans, strategies and prospects, both business and financial, of MoneyLion. These statements are based on the beliefs and assumptions of the management of MoneyLion. Although MoneyLion believes that its respective plans, intentions and expectations reflected in or suggested by these forward-looking statements are reasonable, MoneyLion cannot assure you that it will achieve or realize these plans, intentions or expectations. Generally, statements that are not historical facts, including statements concerning possible or assumed future actions, business strategies, events or results of operations, are forward-looking statements. These statements may be preceded by, followed by or include the words "believes," "estimates," "expects," "projects," "forecasts," "may," "will," "should," "seeks," "plans," "scheduled," "anticipates," or "intends" or similar expressions. The forward-looking statements are based on projections prepared by, and are the responsibility of, MoneyLion's management.

Forward-looking statements are inherently subject to known and unknown risks and uncertainties, many of which may be beyond MoneyLion's control. Forward-looking statements are not guarantees of future performance or outcomes, and MoneyLion's actual performance and outcomes, including, without limitation, actual results of operations, financial condition and liquidity, and the development of the market in which MoneyLion operates, may differ materially from those made in or suggested by the forward-looking statements contained in this Quarterly Report on Form 10-Q. Factors that could cause actual results and outcomes to differ from those reflected in forward-looking statements include, without limitation:

- factors relating to the business, operations and financial performance of MoneyLion, including market conditions and global and economic factors beyond MoneyLion's control;
- MoneyLion's ability to acquire, engage and retain customers and clients and sell or develop additional functionality, products and services to them on the MoneyLion platform;
- MoneyLion's reliance on third-party partners, service providers and vendors, including its ability to comply with applicable requirements of such third parties;
- demand for and consumer confidence in MoneyLion's products and services, including as a result of any adverse publicity concerning MoneyLion;
- any inaccurate or fraudulent information provided to MoneyLion by customers or other third parties;
- MoneyLion's ability to realize strategic objectives and avoid difficulties and risks of any acquisitions, strategic investments, entries into new businesses, joint ventures, divestitures and other transactions;
- MoneyLion's success in attracting, retaining and motivating its senior management and other key personnel;
- MoneyLion's ability to renew or replace its existing funding arrangements and raise financing in the future, to comply with restrictive covenants related to its long-term indebtedness and to manage the effects of changes in the cost of capital;
- MoneyLion's ability to achieve or maintain profitability in the future;
- intense and increasing competition in the industries in which MoneyLion and its subsidiaries operate;
- risks related to the proper functioning of MoneyLion's information technology systems and data storage, including as a result of cyberattacks, data security breaches or other similar incidents or disruptions suffered by MoneyLion or third parties upon which it relies;
- MoneyLion's ability to protect its intellectual property and other proprietary rights and its ability to obtain or maintain intellectual property, proprietary rights and technology licensed from third parties;

- MoneyLion's ability to comply with extensive and evolving laws and regulations applicable to its business and the outcome of any legal or governmental proceedings that may be instituted against MoneyLion;
- MoneyLion's ability to establish and maintain an effective system of internal controls over financial reporting;
- MoneyLion's ability to maintain the listing of its Class A Common Stock and its publicly traded warrants to purchase Class A Common Stock (the "Public Warrants") on the NYSE and any volatility in the market price of MoneyLion's securities; and
- other factors detailed under Part II, Item 1A "Risk Factors" in this Quarterly Report on Form 10-Q.

These and other factors are more fully discussed in our filings with the U.S. Securities and Exchange Commission (the "SEC"), including the "Risk Factors" section in our Annual Report on Form 10-K for the year ended December 31, 2023, and Part I, Item 2 "Management's Discussion and Analysis of Financial Condition and Results of Operations" in this Quarterly Report on Form 10-Q.

These forward-looking statements are based on information available as of the date of this Quarterly Report on Form 10-Q and our management's current expectations, forecasts and assumptions, and involve a number of judgments, risks and uncertainties. Accordingly, forward-looking statements should not be relied upon as representing our views as of any subsequent date. We undertake no obligation to update forward-looking statements to reflect events or circumstances after the date they were made, whether as a result of new information, future events or otherwise, except as may be required under applicable securities laws.

iv

Risk Factor Summary

Our business is subject to numerous risks and uncertainties, including those we face in connection with the successful implementation of our strategy and the growth of our business. The following considerations, among others, may offset our competitive strengths or have a negative effect on our business strategy, which could cause a decline in the price of shares of our securities and result in a loss of all or a portion of your investment:

- If we are unable to acquire new customers and clients, engage and retain our existing customers and clients or sell additional functionality, products and services to them on our platform, our business will be adversely affected.
- Any failure to effectively match consumer leads from our Channel Partners with product offerings from our Product Partners or any reduced marketing spend by such Product Partners on our Enterprise platform could have a material adverse effect on our business, financial condition, results of operations and cash flows.
- We depend on various third-party partners, service providers and vendors, and any adverse changes in our relationships with these third parties could materially and adversely affect our business, including if we fail to comply with applicable requirements of such third parties.
- Adverse publicity concerning us, our business or our personnel or our failure to maintain our brand in a cost-effective manner could materially and adversely affect our business.
- Demand for our products or services may decline if we do not continue to innovate or respond to evolving technology or other changes.
- If the information provided to us by customers or other third parties is incorrect or fraudulent, we may misjudge a customer's qualifications to receive our products and services and our results of operations may be harmed and could subject us to regulatory scrutiny or penalties.
- Any acquisitions, strategic investments, entries into new businesses, joint ventures, divestitures and other transactions could fail to achieve strategic objectives, disrupt our ongoing operations or result in operating difficulties, liabilities and expenses, harm its business and negatively impact our results of operations.
- We depend on our senior management team and other key personnel, and if we fail to attract, retain and motivate our personnel, our business, financial condition and results of operations could be adversely affected.
- If our existing funding arrangements are not renewed or replaced or our existing funding sources are unwilling or unable to provide funding to us on terms acceptable to us, or at all, it could have a material adverse effect on our business, financial condition, results of operations and cash flows. We may be unsuccessful in managing the effects of changes in the cost of capital on our business.
- We have a history of losses and may not achieve or maintain profitability in the future.
- Our risk management processes and procedures may not be effective.
- We operate in highly competitive industries, and our inability to compete successfully would materially and adversely affect our business, financial condition, results of operations and cash flows.
- Our business may be adversely affected by economic conditions and other factors, including adverse developments affecting financial institutions or the financial services industry generally, that we cannot control.
- Cyberattacks, data security breaches or other similar incidents or disruptions suffered by us or third parties upon which we rely could have a material adverse effect on our business, harm our reputation and expose us to public scrutiny or liability.

- Defects, failures or disruptions in our systems or those of third parties upon which we rely and resulting interruptions in the availability of our platform could harm our business and financial condition, harm our reputation, result in significant costs to us and expose us to substantial liability.
- We may be unable to sufficiently obtain, maintain, protect or enforce our intellectual property and other proprietary rights, or we may be unable to obtain or maintain intellectual property, proprietary rights and technology licensed from third parties, which could reduce the value of our platform, products, services and brand, impair our competitive position and cause reputational harm.
- We have in the past, and continue to be, subject to inquiries, subpoenas, exams, pending investigations, enforcement matters and litigation by state and federal regulators, the outcomes of which are uncertain and could cause reputational and financial harm to our business, financial condition, results of operations and cash flows.
- Unfavorable outcomes in legal proceedings may harm our business, financial condition, results of operations and cash flows.
- If we are unable to develop and maintain an effective system of disclosure controls and internal control over financial reporting, our ability to produce timely and accurate financial statements or comply with applicable regulations could be impaired.
- Our business is subject to extensive regulation, examination and oversight in a variety of areas, including registration and licensing requirements under federal, state and local laws and regulations. The legal and regulatory regimes governing certain of our products and services are uncertain and evolving.
- If we fail to operate in compliance with state or local licensing requirements, it could adversely affect our business, financial condition, results of operations and cash flows.
- The highly regulated environment in which our third-party financial institution partners operate may subject us to regulation and could have an adverse effect on our business, financial condition, results of operations and cash flows.
- The collection, processing, use, storage, sharing and transmission of personally identifiable information ("PII") and other sensitive data is subject to stringent and changing state and federal laws, regulations, standards and policies and could give rise to liabilities as a result of our failure or perceived failure to protect such data, comply with privacy and data protection laws and regulations or adhere to the privacy and data protection practices that we articulate to our customers.
- The market price of our securities, including the Class A Common Stock, may be volatile. Our failure to meet the continued listing requirements of the NYSE could result in a delisting of our securities.

The risks described above should be read together with the "Cautionary Statement Regarding Forward-Looking Statements" herein, any other risk factors set forth under Part II, Item 1A "Risk Factors" in this Quarterly Report on Form 10-Q, the "Risk Factors" section in the Annual Report on Form 10-K for the year ended December 31, 2023, our consolidated financial statements and the related notes presented in Part I, Item 1 "Financial Statements" in this Quarterly Report on Form 10-Q and the other documents that we file with the SEC. Our business, prospects, financial condition or operating results could be harmed by any of these risks, as well as other risks not currently known to us or that we currently consider immaterial.

vi

PART I – FINANCIAL INFORMATION

Item 1. Financial Statements

MONEYLION INC. CONSOLIDATED BALANCE SHEETS

(dollar amounts in thousands, except per share amounts) (Unaudited)

	June 30, 2024	December 31, 2023		
Assets				
Cash	\$ 98,361	\$	92,195	
Restricted cash, including amounts held by variable interest entities (VIEs) of \$1,585 and \$128	4,749		2,284	
Consumer receivables	237,358		208,167	
Allowance for credit losses on consumer receivables	 (40,523)		(35,329)	
Consumer receivables, net, including amounts held by VIEs of \$150,174 and \$131,283	196,835		172,838	
Enterprise receivables, net	23,045		15,978	
Property and equipment, net	1,961		1,864	
Intangible assets, net	168,302		176,541	
Other assets	52,490		53,559	
Total assets	\$ 545,743	\$	515,259	
Liabilities and Stockholders' Equity				
Liabilities:				
Secured loans, net	\$ 64,482	\$	64,334	
Accounts payable and accrued liabilities	48,984		52,396	
Warrant liability	810		810	
Other debt, net, including amounts held by VIEs of \$129,932 and \$125,419	129,932		125,419	
Other liabilities	 23,079		15,077	
Total liabilities	267,287		258,036	
Commitments and contingencies (Note 15)				
Stockholders' equity:				
Class A Common Stock, \$0.0001 par value; 66,666,666 shares authorized as of June 30, 2024 and December 31, 2023, 10,977,991 and 10,945,658 issued and outstanding, respectively, as of June 30, 2024 and 10,444,627 and 10,412,294 issued and outstanding, respectively, as of December 31, 2023	1		1	
Additional paid-in capital	980,662		969,641	
Accumulated deficit	(692,507)		(702,719)	
Treasury stock at cost, 32,333 shares as of June 30, 2024 and December 31, 2023	(9,700)		(9,700)	
Total stockholders' equity	 278,456		257,223	
Total liabilities and stockholders' equity	\$ 545,743	\$	515,259	

The accompanying notes are an integral part of these consolidated financial statements.

MONEYLION INC. CONSOLIDATED STATEMENTS OF OPERATIONS

(dollar amounts in thousands, except per share amounts) (Unaudited)

		Three Months	Ended	June 30,	Six Months E	nded Jur	ie 30,
		2024		2023	 2024		2023
Revenue							
Service and subscription revenue	\$	127,890	\$	103,237	\$ 245,963	\$	193,978
Net interest income on loan receivables		2,956		3,304	5,889		6,232
Total revenue, net		130,846	_	106,541	251,852		200,210
Operating expenses							
Provision for credit losses on consumer receivables		33,431		25,562	53,661		42,073
Compensation and benefits		24,852		22,572	49,638		46,980
Marketing		10,530		6,549	21,396		12,941
Direct costs		34,449		32,230	65,838		62,032
Professional services		11,007		4,518	16,773		9,517
Technology-related costs		6,512		5,611	13,098		11,649
Other operating expenses		4,338		11,219	14,658		20,214
Total operating expenses		125,119		108,261	235,062		205,406
Net income (loss) before other (expense) income and income taxes		5,727		(1,720)	16,790		(5,196)
Interest expense		(6,714)		(7,330)	(13,531)		(14,841)
Change in fair value of warrant liability		(81)		162	_		13
Change in fair value of contingent consideration from mergers and acquisitions		_		6,367	_		6,613
Goodwill impairment loss		_		(26,721)	_		(26,721)
Other income		2,381		1,257	4,740		2,906
Net income (loss) before income taxes		1,313		(27,985)	 7,999		(37,226)
Income tax benefit		(1,824)	-	(262)	 (2,213)		(286)
Net income (loss)		3,137		(27,723)	 10,212		(36,940)
Reversal of previously accrued dividends on preferred stock		_		2,667	_		690
Net income (loss) attributable to common shareholders	\$	3,137	\$	(25,056)	\$ 10,212	\$	(36,250)
Net income (loss) per share, basic	\$	0.29	\$	(2.71)	\$ 0.95	\$	(4.05)
Net income (loss) per share, diluted	\$	0.26	\$	(2.71)	\$ 0.85	\$	(4.05)
Weighted average shares used in computing net income (loss) per shabasic	are,	10,862,853		9,234,238	 10,694,635		8,944,836
Weighted average shares used in computing net income (loss) per sha diluted	are,	12,091,421		9,234,238	 11,954,975		8,944,836

The accompanying notes are an integral part of these consolidated financial statements.

MONEYLION INC. CONSOLIDATED STATEMENTS OF REDEEMABLE CONVERTIBLE PREFERRED STOCK AND STOCKHOLDERS' EQUITY (amounts in thousands, except share amounts) (Unaudited)

								Total
	Class A Con	nmon Ste	ock		Additional	Accumulated	Treasury	Stockholders'
	Shares		Amount	P	aid-in Capital	Deficit	Stock	Equity
Balances at April 1, 2024	10,787,923	\$	1	\$	975,801	\$ (695,644)	\$ (9,700)	\$ 270,458
Stock-based compensation	_		—		7,531	_	_	7,531
Exercise of stock options and warrants and vesting of RSUs and PSUs, net of tax withholdings	157,735		_		(2,670)	_	_	(2,670)
Net income			_			 3,137	 _	 3,137
Balances at June 30, 2024	10,945,658	\$	1	\$	980,662	\$ (692,507)	\$ (9,700)	\$ 278,456

							Total
	Class A Con	nmon S	tock	Additional	Accumulated	Treasury	Stockholders'
	Shares		Amount	Paid-in Capital	Deficit	Stock	Equity
Balances at January 1, 2024	10,412,294	\$	1	\$ 969,641	\$ (702,719)	\$ (9,700)	\$ 257,223
Stock-based compensation	_		_	14,028	_	_	14,028
Exercise of stock options and warrants and vesting of RSUs and PSUs, net of tax withholdings	533,364		_	(3,007)	_	_	(3,007)
Net income	_		—	—	10,212	_	10,212
Balances at June 30, 2024	10,945,658	\$	1	\$ 980,662	\$ (692,507)	\$ (9,700)	\$ 278,456

MONEYLION INC.

CONSOLIDATED STATEMENTS OF REDEEMABLE CONVERTIBLE PREFERRED STOCK AND STOCKHOLDERS' EQUITY

(amounts in thousands, except share amounts)

(Unaudited)

-	Redeemable (Preferred Stor	ek (Series A)	Class A Com		Additional	Accumulated	Treasury	Total Stockholders'	
Balanan at Amil 1, 2022	Shares	Amount \$ 173,328	Shares 8,799,069	Amount	Paid-in Capital \$ 771,881	Deficit \$ (666,691)	Stock (9,700)	Equity	
Balances at April 1, 2023	25,701,595		8,799,009	\$ 1		\$ (000,091)	\$ (9,700)	\$ 95,491	
Stock-based compensation	_	_	_	_	5,250	-	-	5,250	
Exercise of stock options and warrants and vesting of RSUs and PSUs, net of tax withholdings	_	_	72,128	_	(183)	_	_	(183)	
Issuance of options and preferred stock in connection with Engine Acquisition and the related contingent consideration, net of working capital adjustments	4,354,092	1,440	23,453	_	304	_	_	304	
Voluntary conversion of preferred stock to common stock	(6,634)	(45)	228	_	45	_	_	45	
Reversal of previously accrued dividends on preferred stock	_	_	_	_	2,667	_	_	2,667	
Settlement of accrued dividends on preferred stock	_	_	229,605	_	2,976	_	_	2,976	
Automatic conversion of redeemable convertible preferred stock (Series A)	(30,049,053)	(174,723)	1,012,093	_	174,849	_	_	174,849	
Other	_	_	(901)	_	(11)	_	-	(11)	
Net loss	_	_	_	_	_	(27,723)	_	(27,723)	
Balances at June 30, 2023	_	\$	10,135,675	\$ 1	\$ 957,778	\$ (694,414)	\$ (9,700)	\$ 253,665	

	Redeemable Preferred Stor		Class A Com	mon Stock	Additional	Accumulated	Treasury	Total Stockholders'
	Shares	Amount	Shares (1)	Amount (1)	Paid-in Capital ⁽¹⁾	Deficit	Stock	Equity
Balances at January 1, 2023	25,655,579	\$ 173,208	8,587,345	\$ 1	\$ 766,839	\$ (657,979)	\$ (9,700)	\$ 99,161
Stock-based compensation	_	-	_	_	10,955	_	_	10,955
Exercise of stock options and warrants and vesting of RSUs and PSUs, net of tax withholdings	_	_	172,925	_	(782)	_	_	(782)
Issuance of common stock in connection with earnout and make-whole provisions related to the acquisition of Malka Media Group LLC	_	_	110,925	_	1,914	_	_	1,914
Issuance of equity in connection with Engine Acquisition and the related contingent consideration, net of working capital adjustments	4,400,172	1,560	23,453	_	304	_	_	304
Voluntary conversion of preferred stock to common stock	(6,698)	(45)	230	_	45	_	_	45
Reversal of previously accrued dividends on preferred stock	_	_	_	_	690	_	_	690
Settlement of accrued dividends on preferred stock	—	—	229,605	_	2,976	—	—	2,976
Automatic conversion of redeemable convertible preferred stock (Series A)	(30,049,053)	(174,723)	1,012,093	_	174,849	_	_	174,849
Other	—	_	(901)	_	(12)	505	—	493
Net loss						(36,940)		(36,940)
Balances at June 30, 2023		\$	10,135,675	<u>\$ 1</u>	\$ 957,778	\$ (694,414)	\$ (9,700)	\$ 253,665

Prior period results have been adjusted to reflect the Reverse Stock Split of the Class A Common Stock at a ratio of 1-for-30 that became effective April 24, 2023. (1) See Note 1, "Description of Business and Basis of Presentation," for details.

The accompanying notes are an integral part of these consolidated financial statements.

MONEYLION INC. CONSOLIDATED STATEMENTS OF CASH FLOWS (amounts in thousands) (Unaudited)

	1	hree Months	Ended Ju	· · · · · · · · · · · · · · · · · · ·		Six Months E	nded Jur	
	202	4		2023		2024		2023
Cash flows from operating activities:								
Net income (loss)	\$	3,137	\$	(27,723)	\$	10,212	\$	(36,940
Adjustments to reconcile net income (loss) to net cash from operating activities:								
Provision for losses on receivables		33,431		25,562		53,661		42,073
Depreciation and amortization expense		6,331		6,113		12,543		12,297
Change in deferred fees and costs, net		322		782		678		1,398
Change in fair value of warrants		81		(162)		—		(13
Change in fair value of contingent consideration from mergers and acquisitions		_		(6,367)		—		(6,613
Loss (gain) on foreign currency translation		44		(171)		(53)		(178
Goodwill impairment loss		—		26,721		_		26,721
Stock compensation expense		7,531		5,250		14,028		10,955
Deferred income taxes		235		(400)		471		(493
Changes in assets and liabilities:								
Accrued interest receivable		(38)		(211)		(76)		(238
Enterprise receivables, net		(5,527)		1,708		(7,067)		(2,422
Other assets		3,117		5,610		1,753		4,360
Accounts payable and accrued liabilities		(1,103)		1,156		(3,359)		(8,649
Other liabilities		237		(2,505)		(1,354)		(4,215
Net cash provided by operating activities		47,798		35,363		81,437		38,043
Cash flows from investing activities:		<u> </u>	-	<u> </u>	-	· · · · ·	-	
Net originations and collections of finance receivables		(41,676)		(33,185)		(69,398)		(52,832
Purchase of property and equipment and software development		(2,244)		(1,638)		(4,401)		(2,675
Settlement of contingent consideration related to mergers and acquisitions		(_,)		(766)		(,,)		(1,116
Net cash used in investing activities		(43,920)		(35,589)		(73,799)		(56,623
Cash flows from financing activities:		(10,720)		(55,565)		(13,177)		(50,025
Net proceeds from (repayments to) special purpose vehicle credit facilities						4,000		(24,000
Repayments to secured/senior lenders				(5,000)		4,000		(5,000
		_		(154)		_		
Payment of deferred financing costs Payments related to the automatic conversion of redeemable convertible preferred stock (Series A) in lieu of fractional		_		(134)				(154
shares of common stock and dividends on preferred stock		—		(3,007)		—		(3,007
Payments related to issuance of common stock related to exercise of stock options and warrants, net of tax withholdings related to vesting of stock-based compensation	5	(2,670)		(183)		(3,007)		(782
Other		_		(12)		-		(12
Net cash (used in) provided by financing activities		(2,670)		(8,356)		993		(32,955
Net change in cash and restricted cash		1,208		(8,582)		8,631		(51,535
Cash and restricted cash, beginning of period		101,902		110,756		94,479		153,709
	\$	103,110	s	102,174	S	103,110	\$	102,174
Cash and restricted cash, end of period	\$	105,110	3	102,174	æ	105,110	φ	102,174
Supplemental disclosure of cash flow information:								
Cash paid for interest	\$	6,400	\$	7,043	\$	12,848	\$	14,508
Supplemental disclosure of non-cash investing and financing activities:								
Voluntary conversion of preferred stock to common stock	\$		S	45	\$	_	\$	45
Automatic conversion of redeemable convertible preferred stock (Series A) to common stock	\$	_	s	174,849	\$	_	\$	174,849
Reversal of previously accrued dividends on preferred stock	\$		s	2,667	\$	_	\$	690
Issuance of common stock to settle accrued dividends on preferred stock and Preferred Stock Equivalents	\$	_	s	3,280	\$	_	\$	3,280
Equity issued as consideration for mergers and acquisitions	\$	_	s	1,744	\$		\$	1,864
Equity issued as settlement of contingent consideration related to Malka Acquisition	3 S		S	1,/44	s S	_	\$	1,804
Equity issued as settlement of contingent consideration related to Marka Acquisition	\$ \$	_	5	1,440	\$ \$	_	\$ \$	1,914

The accompanying notes are an integral part of these consolidated financial statements.

MONEYLION INC. NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(amounts in thousands, except share and per share amounts or as otherwise indicated)

(Unaudited)

1. DESCRIPTION OF BUSINESS AND BASIS OF PRESENTATION

MoneyLion Inc. ("MoneyLion" or the "Company") was founded in 2013 and is headquartered in New York, New York. On September 22, 2021, MoneyLion Inc., formerly known as Fusion Acquisition Corp., consummated a business combination (the "Business Combination") with MoneyLion Technologies Inc., formerly known as MoneyLion Inc. Following the Business Combination, MoneyLion Inc. became a publicly traded company, with MoneyLion Technologies Inc. continuing the existing business operations as a subsidiary of MoneyLion Inc. MoneyLion Inc.'s Class A common stock, par value \$0.0001 per share (the "Class A Common Stock"), is listed on the New York Stock Exchange (the "NYSE") under the ticker symbol "ML." "MALKA" refers to Malka Media Group LLC, a wholly-owned subsidiary of MoneyLion Technologies Inc., and "Engine" refers to ML Enterprise Inc., doing business as the brand "Engine by MoneyLion," a wholly-owned subsidiary of MoneyLion Technologies Inc. which was previously named "Even Financial Inc." and subsequently renamed in February 2023.

MoneyLion is a leader in financial technology, powering the next generation of personalized products and financial content for American consumers. MoneyLion designs and offers modern personal finance products, tools and features and curate money-related content that delivers actionable insights and guidance to its users. MoneyLion also operates and distributes embedded finance marketplace solutions that match consumers with personalized third-party offers from its partners, providing convenient access to an expansive breadth of financial solutions that enable consumers to borrow, spend, save and achieve better financial outcomes. In addition, MoneyLion provides creative media and brand content services to clients across industries through its media division and leverages its adaptive, in-house content studio to produce and deliver engaging and dynamic content in support of MoneyLion's product and service offerings.

Basis of Presentation—The consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America ("U.S. GAAP") and the rules and regulations of the Securities and Exchange Commission ("SEC"). The consolidated financial statements include the accounts of MoneyLion Inc. and its wholly owned subsidiaries and consolidated variable interest entities ("VIEs") for which the Company is the primary beneficiary. All intercompany transactions and balances have been eliminated in consolidation. The Company does not have any items of other comprehensive income (loss); therefore, there is no difference between net income (loss) and comprehensive income (loss) for the three and six months ended June 30, 2024 and 2023.

Reverse Stock Split—On April 24, 2023, the Company amended the Company's Fourth Amended and Restated Certificate of Incorporation (as amended from time to time, the "Certificate of Incorporation") to effect, effective as of 5:01 p.m. Eastern Time on April 24, 2023, a 1-for-30 reverse stock split (the "Reverse Stock Split") of the Class A Common Stock. At the effective time of the Reverse Stock Split, every 30 shares of Class A Common Stock either issued and outstanding or held as treasury stock were automatically reclassified into one new share of Class A Common Stock, and the total number of shares of Class A Common Stock authorized for issuance was reduced by a corresponding proportion from 2,000,000 shares to 66,666,666 shares. The Reverse Stock Split was approved by the Company's stockholders at a Special Meeting of Stockholders on April 19, 2023 and approved by the Board of Directors on April 21, 2023. The primary goal of the Reverse Stock Split was to increase the per share price of the Class A Common Stock in order to meet the minimum per share price requirement for continued listing on the NYSE. The Class A Common Stock began trading on the NYSE on an as-adjusted basis on April 25, 2023 under the existing trading symbol "ML."

In addition, as a result of the Reverse Stock Split, proportionate adjustments were made to the number of shares of Class A Common Stock underlying the Company's outstanding equity awards, the number of shares issuable upon the exercise of the Company's outstanding warrants and the number of shares issuable under the Company's equity incentive plans and certain existing agreements, as well as the exercise, grant and acquisition prices of such equity awards and warrants, as applicable. Furthermore, proportionate adjustments were made to the conversion factor at which the Company's previously outstanding Series A Convertible Preferred Stock, par value \$0.0001 per share (the "Series A Preferred Stock"), were converted to Class A Common Stock. The total number of shares of preferred stock of the Company authorized for issuance remained at 200,000,000. Stockholders who would have been entitled to receive fractional shares as a result of the Reverse Stock Split were instead entitled to a cash payment in lieu thereof at a price equal to the fraction of one share to which the stockholder was otherwise entitled multiplied by the closing price per share of the Class A Common Stock on the NYSE on the effective date of the Reverse Stock Split.

The effects of the Reverse Stock Split have been reflected in these consolidated financial statements and the accompanying footnotes for all periods presented, which includes adjusting the description of any activity that may have been transacted on a pre-Reverse Stock Split basis.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

In the opinion of the Company, the accompanying consolidated financial statements contain all adjustments, consisting of normal recurring adjustments and adjustments to eliminate intercompany transactions and balances, necessary for a fair presentation of its financial position and its results of operations, changes in redeemable convertible preferred stock and stockholders' equity and cash flows.

The Company's accounting policies are set forth in Note 2, "Summary of Significant Accounting Policies" of the Company's Notes to Consolidated Financial Statements included in the Company's Annual Report on Form 10-K for the year ended December 31, 2023. Included herein are certain updates to those policies and the related disclosures.

Revenue Recognition and Related Receivables—The following table summarizes revenue by type for the three and six months ended June 30, 2024 and 2023:

	Т	hree Months	Ended	June 30,		Six Months E	nded June 30,	
	2024		2023		2024			2023
Consumer revenues								
Service and subscription fees	\$	89,444	\$	66,268	\$	174,653	\$	128,706
Net interest income on finance receivables		2,956		3,304		5,889		6,232
Total consumer revenues		92,400		69,572		180,542		134,938
Enterprise service revenues		38,446		36,969		71,310		65,272
Total revenue, net	\$	130,846	\$	106,541	\$	251,852	\$	200,210

Fair Value of Financial Instruments—Accounting Standards Codification ("ASC") 820, *Fair Value Measurement* ("ASC 820"), provides a single definition of fair value and a common framework for measuring fair value as well as disclosure requirements for fair value measurements used in financial statements. Under ASC 820, fair value is determined based upon the exit price that would be received by a company to sell an asset or paid by a company to transfer a liability in an orderly transaction between market participants, exclusive of any transaction costs. Fair value measurements are determined by either the principal market or the most advantageous market. The principal market is the market with the greatest level of activity and volume for the asset or liability. Absent a principal market to measure fair value, the Company uses the most advantageous market, which is the market from which the Company would receive the highest selling price for the asset or pay the lowest price to settle the liability, after considering transaction costs. However, when using the most advantageous market, transaction costs are only considered to determine which market is the most advantageous and these costs are then excluded when applying a fair value measurement. ASC 820 creates a three-level hierarchy to prioritize the inputs used in the valuation techniques to derive fair values. The basis for fair value measurements for each level within the hierarchy is described below, with Level 1 having the highest priority and Level 3 having the lowest.

Level 1 - Quoted prices in active markets for identical assets or liabilities.

Level 2 – Observable inputs other than Level 1 quoted prices, such as quoted prices for similar assets and liabilities in active markets, quoted prices in markets that are not active for identical or similar assets and liabilities, or other inputs that are observable or can be corroborated by observable market data for substantially the full term of the assets or liabilities.

Level 3 – Valuations are based on inputs that are unobservable and significant to the overall fair value measurement of the assets or liabilities. Inputs reflect management's best estimate of what market participants would use in pricing the asset or liability at the measurement date. Consideration is given to the risk inherent in the valuation technique and the risk inherent in the inputs to the model.

The Company had no assets measured at fair value on a recurring or non-recurring basis as of June 30, 2024 or December 31, 2023. The Private Placement Warrants (as defined herein) were measured at fair value on a recurring basis as of June 30, 2024 and December 31, 2023 and are further described in Note 13, "Stock Warrants." The Company had no liabilities measured at fair value on a non-recurring basis as of June 30, 2024 or December 31, 2023. There have been no transfers between levels during the six months ended June 30, 2024 and 2023.

The Company also has financial instruments which are not measured at fair value. The Company has evaluated cash (Level 1), restricted cash (Level 1) and consumer receivables, net (Level 3) and believes the carrying value approximates the fair value due to the short-term nature of these balances. The carrying value of the secured loans approximates their fair value based on the relatively short duration these instruments have been outstanding and the secured loans' variable interest rate based on market rates. The carrying value of other debt approximates its fair value based on the relatively short duration these instruments have been outstanding and availability of alternative financing sources at similar interest rates with the same terms. The fair value of secured loans and other debt would be based on Level 2 fair value measurements.

Recently Adopted Accounting Pronouncements—The Company adopted ASU 2016-13, Financial Instruments – Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments, which, along with subsequent related ASUs, creates a new credit impairment standard for financial assets measured at amortized cost and available-for-sale debt securities. The ASU requires financial assets measured at amortized cost (including loans, trade receivables and held-to-maturity debt securities) to be presented at the net amount expected to be collected, through an allowance for credit losses that are expected to occur over the remaining life of the asset, rather than incurred losses. The ASU requires that credit losses on available-for-sale debt securities be presented as an allowance rather than as a direct write-down. The measurement of credit losses for newly recognized financial assets (other than certain purchased assets) and subsequent changes in the allowance for credit losses are recorded in the statement of operations as the amounts expected to be collected change. The Company adopted ASU 2016-13 and the related subsequent ASUs effective January 1, 2023, and applied the changes prospectively, recognizing a cumulative-effect adjustment to the beginning balance of retained earnings as of the adoption date. Upon adoption, the Company increased consumer receivables, net by \$692, decreased enterprise receivables, net by \$187 and reduced accumulated deficit by \$505. The adoption of the new guidance did not impact the Company's unaudited consolidated interim statements of operations or cash flows.

Recently Issued Accounting Pronouncements Not Yet Adopted—The Company currently qualifies as an "emerging growth company" under the Jumpstart Our Business Startups Act of 2012. Accordingly, the Company has the option to adopt new or revised accounting guidance either (i) within the same periods as those otherwise applicable to non-emerging growth companies or (ii) within the same time periods applicable to private companies. The Company has elected to adopt new or revised accounting guidance within the same time periods as private companies, as indicated below, management determines it is preferable to take advantage of early adoption provisions offered within the applicable guidance.

In November 2023, the FASB issued ASU No. 2023-07, Segment Reporting (Topic 280): Improvements to Reportable Segment Disclosures. The guidance expands the disclosures required for reportable segments in the Company's annual and interim consolidated financial statements, primarily through enhanced disclosures about significant segment expenses. The standard will be effective for the Company beginning with the Company's annual reporting for fiscal year 2025 and interim periods thereafter, with early adoption permitted. The Company is currently evaluating the impact of this standard on its disclosures.

In December 2023, the FASB issued ASU No. 2023-09, *Income Taxes (Topic 740): Improvements to Income Tax Disclosures*. The guidance requires disclosure of disaggregated income taxes paid, prescribes standardized categories for the components of the effective tax rate reconciliation and modifies other income tax-related disclosures. The standard will be effective for the Company beginning with the Company's annual reporting for fiscal year 2026 and interim periods thereafter, with early adoption permitted. The Company is currently evaluating the impact of this standard on its income tax disclosures.

3. CONSUMER RECEIVABLES

The Company's finance receivables consist of secured personal loans and principal amounts of Instacash advances. Secured loan principal balances are either partially or fully deposited into an escrow account upon origination with any remaining balance being given to the borrower. The funds in the escrow account may be used to pay the secured personal loan in full or can be released to the borrower once the secured personal loan is paid in full. Until such time, the funds in the escrow account may be collected by the Company in the event the borrower becomes contractually past due. Accrued interest receivables represent the interest accrued on the loan receivables based upon the daily principal amount outstanding except for loans that are on nonaccrual status.

The Company's policy is to suspend recognition of interest income on secured personal loans and place the secured personal loan on nonaccrual status when the account is more than 60 days past due on a contractual basis or when, in the Company's estimation, the collectability of the account is uncertain, and the account is less than 90 days contractually past due. The Company has elected to not measure an allowance for losses on accrued interest receivable. Any accrued interest receivable that becomes 90 days past due on a contractual basis is charged-off by reversing net interest income on loan receivables. Net charge-offs of accrued interest income were \$272 and \$366 for the three months ended June 30, 2024 and 2023 and \$509 and \$673 for the six months ended June 30, 2024 and 2023.

Fees receivable represent the amounts due to the Company for tips and instant transfer fees related to the Instacash earned wage access product. Subscription receivables represent the amounts billed to customers for subscription services.

The credit quality and future repayment of consumer receivables are dependent upon the customer's ability to perform under the terms of the agreement. Factors such as unemployment rates and housing values, among others, may impact the customer's ability to perform under the loan or Instacash advance terms though no direct correlation between charge-off rates and these factors has been identified in the Company's analysis. When assessing provision for losses on consumer receivables, the Company takes into account the composition and delinquency status of the outstanding consumer receivables and the related forecasted principal loss rates based on recent historical experience. Recent historical loss rates are updated on a quarterly basis. Charge-offs of consumer receivable balances occur after becoming 90 days past contractually due unless specific circumstances are identified on an individual or group of receivables that indicate charge-off is not appropriate. The level of exceptions to charge-offs occurring once 90 days past due is not material. Consumer receivable charge-off rates and aging by product for the three and six months ended June 30, 2024 and 2023.

Consumer receivables consisted of the following:

	J	June 30, 2024			
Loan receivables	\$	75,085	\$	66,815	
Instacash receivables		139,992		120,336	
Finance receivables		215,077		187,151	
Fees receivable		16,664		16,137	
Subscription receivables		4,170		3,491	
Deferred loan origination costs		69		86	
Accrued interest receivable		1,378		1,302	
Consumer receivables, before allowance for credit losses	\$	237,358	\$	208,167	

Changes in the allowance for losses on loan receivables were as follows:

	Three Months Ended June 30,			Six Months Ended June 30,				
		2024		2023		2024		2023
Beginning balance	\$	4,608	\$	5,791	\$	5,761	\$	5,784
Provision for credit losses on receivables		1,032		2,083		1,400		3,603
Loan receivables charged off		(1,572)		(2,568)		(3,722)		(6,757)
Recoveries		453		943		1,082		3,619
Ending balance	\$	4,521	\$	6,249	\$	4,521	\$	6,249

Changes in the allowance for losses on Instacash receivables were as follows:

	Three Months Ended June 30,			Six Months Ended June 30,				
		2024		2023		2024		2023
Beginning balance	\$	26,645	\$	19,686	\$	25,992	\$	23,240
Provision for credit losses on receivables		26,741		18,421		44,337		28,502
Instacash receivables charged off		(26,214)		(19,770)		(49,250)		(39,598)
Recoveries		4,325		3,974		10,418		10,167
Ending balance	\$	31,497	\$	22,311	\$	31,497	\$	22,311

Changes in the allowance for losses on fees receivable were as follows:

	Three Months Ended June 30,			Six Months Ended June 30,				
		2024		2023		2024		2023
Beginning balance	\$	1,999	\$	1,018	\$	2,552	\$	908
Provision for credit losses on receivables		3,931		3,869		5,526		8,043
Fees receivable charged off		(3,400)		(3,300)		(6,333)		(8,125)
Recoveries		581		513		1,366		1,274
Ending balance	\$	3,111	\$	2,100	\$	3,111	\$	2,100

Changes in the allowance for losses on subscription receivables were as follows:

	Th	Three Months Ended June 30,			Six Months Ended June 30,			ne 30,
		2024		2023		2024		2023
Beginning balance	\$	1,051	\$	978	\$	1,024	\$	1,292
Provision for credit losses on receivables		1,727		1,189		2,398		1,925
Subscription receivables charged off		(1,915)		(1,365)		(3,077)		(2,721)
Recoveries		531		293		1,049		599
Ending balance	\$	1,394	\$	1,095	\$	1,394	\$	1,095

The following is an assessment of the repayment performance of loan receivables as of June 30, 2024 and December 31, 2023 and presents the contractual delinquency of the loan receivables portfolio:

		June 30, 2024			r 31, 2023
	I	Amount	Percent	Amount	Percent
Current	\$	65,229	86.9%	\$ 58,980	88.2 %
Delinquency:					
31 to 60 days		5,357	7.1 %	4,451	6.7%
61 to 90 days		4,499	6.0%	3,384	5.1%
Total delinquency		9,856	13.1%	7,835	11.8%
Loan receivables before allowance for credit losses	\$	75,085	100.0%	\$ 66,815	100.0 %

Loan receivables that are 61 to 90 days contractually past due are placed on non-accrual status.

The following is an assessment of the repayment performance of Instacash receivables as of June 30, 2024 and December 31, 2023 and presents the contractual delinquency of the Instacash receivables portfolio:

	Jur	e 30, 2024	Decemb	er 31, 2023
	Amount	Percent	Amount	Percent
Current	\$ 117,112	83.7%	\$ 104,541	86.9%
Delinquency:				
31 to 60 days	13,033	9.3%	8,829	7.3 %
61 to 90 days	9,84	7.0%	6,966	5.8%
Total delinquency	22,880) 16.3 %	15,795	13.1 %
Instacash receivables before allowance for credit losses	\$ 139,992	2 100.0%	\$ 120,336	100.0 %

The following is an assessment of the repayment performance of fees receivable as of June 30, 2024 and December 31, 2023 and presents the contractual delinquency of the fees receivable portfolio:

		June 30, 2024			December 31, 2023		
	A	Amount	Percent	A	mount	Percent	
Current		13,644	81.9%	\$	13,971	86.6%	
Delinquency:							
31 to 60 days		1,700	10.2 %		1,197	7.4 %	
61 to 90 days		1,320	7.9%		969	6.0%	
Total delinquency		3,020	18.1%		2,166	13.4%	
Fees receivable before allowance for credit losses	\$	16,664	100.0 %	\$	16,137	100.0 %	

The following is an assessment of the repayment performance of subscription receivables as of June 30, 2024 and December 31, 2023 and presents the contractual delinquency of the subscription receivables portfolio:

	June 30, 2024				December 3	31, 2023
	A	mount	Percent	Am	iount	Percent
Current	\$	3,136	75.2%	\$	2,786	79.8%
Delinquency:						
31 to 60 days		693	16.6%		407	11.7%
61 to 90 days		341	8.2%		298	8.5 %
Total delinquency		1,034	24.8%		705	20.2 %
Subscription receivables before allowance for credit losses	\$	4,170	100.0 %	\$	3,491	100.0 %

4. SALE OF CONSUMER RECEIVABLES

On June 30, 2024 (the "Closing Date"), ML Plus LLC, an indirect, wholly-owned subsidiary of the Company (the "Seller"), entered into a Master Receivables Purchase Agreement (the "Purchase Agreement") with Sound Point Capital Management LP, as purchaser agent ("Sound Point"), and SP Main Street Funding I LLC and each additional purchaser from time to time party thereto, as purchasers (the "Purchasers"). The Purchase Agreement provides for the purchase, subject to certain conditions precedent, by the Purchasers from time to time during the term of the Purchase Agreement, on a committed basis, of a majority of the Company's eligible Instacash receivables, subject to certain concentration limits, up to an aggregate facility limit of \$175,000,000, which amount may be increased by up to \$75,000,000. The obligations of the Seller under the Purchase Agreement will be guaranteed by MoneyLion Technologies Inc., a direct, wholly-owned subsidiary of the Company. The Purchase Agreement terminates on the two-year anniversary of the Closing Date and may be extended for an additional one-year period upon mutual agreement of the Seller and Sound Point.

The initial price at which the Purchasers will purchase the eligible Instacash receivables with respect to each monthly cohort is based on the average loss rate at 360 days past the repayment date of the three most recent matured monthly cohorts and will be subject to adjustment for future monthly cohorts based on the performance of monthly cohorts at specified intervals past the repayment date compared to the expected loss rates of the reference matured monthly cohorts established at sale and a fixed discount percentage.

The Seller will pay the Purchasers a non-refundable fee, payable monthly, equal to 2.00% times the daily average available facility limit, subject to a maximum aggregate amount of \$1,750,000.

The Purchase Agreement contains customary representations and warranties; repurchase rights and obligations of the Seller upon the occurrence of certain events (subject to specified limitations); affirmative and negative covenants, including, among other things, financial reporting and notice requirements and restrictions on the ability of the Seller to incur liens on the purchased receivables; and events of default (subject to specified cure provisions), the occurrence of which will give Sound Point the right to terminate the Purchase Agreement.

In connection with the Seller's entry into the Purchase Agreement, on the Closing Date, MoneyLion Technologies Inc. (in such capacity, the "Servicer") entered into a Servicing Agreement (the "Servicing Agreement") with Sound Point and the Purchasers pursuant to which the Purchasers appointed the Servicer to, among other things, service the purchased Instacash receivables in accordance with agreed upon servicing guidelines, collect and remit collections therefrom and provide certain reporting and other information relating to its servicing duties. The Seller will receive a fixed percentage of net collections.

No Instacash receivables were sold under the Purchase Agreement during the three and six months ended June 30, 2024.

5. PROPERTY AND EQUIPMENT

Property and equipment consisted of the following:

	June 30, 2024	De	cember 31, 2023
Leasehold improvements	\$ 1,871	\$	1,932
Furniture and fixtures	260		361
Computers and equipment	2,773		2,551
	4,904		4,844
Less: accumulated depreciation	(2,943)		(2,980)
Property and equipment, net	\$ 1,961	\$	1,864

Total depreciation expense related to property and equipment was \$205 and \$283 for the three months ended June 30, 2024 and 2023, respectively, and \$391 and \$587 for the six months ended June 30, 2024 and 2023, respectively.

6. INTANGIBLE ASSETS

Intangible assets consisted of the following:

		June 30,	Dee	cember 31,
Useful Life	_	2024		2023
3 - 7 years	\$	46,969	\$	43,105
		1,706		1,695
10 - 15 years		160,500		160,500
9 - 15 years		15,960		15,960
		(56,833)		(44,719)
	\$	168,302	\$	176,541
	3 - 7 years 10 - 15 years	Useful Life 3 - 7 years \$ 10 - 15 years	Useful Life 2024 3 - 7 years \$ 46,969 1,706 1 10 - 15 years 160,500 9 - 15 years 15,960 (56,833) (56,833)	Useful Life 2024 3 - 7 years \$ 46,969 \$ 1,706 1,706 \$ 10 - 15 years 160,500 \$ 9 - 15 years 15,960 \$ (56,833)

The Company capitalizes certain internal-use software development costs, consisting primarily of contractor costs and employee salaries and benefits allocated to the software. Capitalization of costs incurred in connection with internally developed software commences when both the preliminary project stage is completed and management has authorized further funding for the project, based on a determination that it is probable the project will be completed and used to perform the function intended. Costs incurred for enhancements that are expected to result in additional functionalities are capitalized in a similar manner. Capitalization of costs ceases no later than the point at which the project is substantially complete and ready for its intended use, at which point amortization of capitalized costs begins. All other costs are expensed as incurred. Costs capitalized in connection with internal-use software were \$2,052 and \$1,689 for the three months ended June 30, 2024 and 2023, respectively, and were \$3,912 and \$2,851 during the six months ended June 30, 2024 and 2023, respectively.

For the three months ended June 30, 2024 and 2023, total amortization expense was \$6,126 and \$5,830, respectively. For the six months ended June 30, 2024 and 2023, total amortization expense was \$12,152 and \$11,710, respectively.

The following table summarizes estimated future amortization expense of intangible assets placed in service at June 30, 2024 for the years ending:

Remainder of 2024	\$ 12,318
2025	24,636
2026	24,636
2027	24,064
2028	21,755
Thereafter	59,187
	\$ 166,596

7. OTHER ASSETS

Other assets consisted of the following:

	June 30,		Dec	ember 31,
	2024		2023	
Receivable from payment processors	\$	29,502	\$	37,362
Prepaid expenses		5,241		5,987
Operating lease right-of-use assets		13,521		6,159
Other		4,226		4,051
Total other assets	\$	52,490	\$	53,559

8. DEBT

The Company's debt as of June 30, 2024 and December 31, 2023 is presented below:

	d	June 30, 2024				
Monroe Term Loans	\$	65,000	\$	65,000		
Unamortized discounts and debt issuance costs		(518)		(666)		
Total secured loans, net	\$	64,482	\$	64,334		
ROAR 1 SPV Credit Facility	\$	66,500	\$	64,500		
ROAR 2 SPV Credit Facility		64,500		62,500		
Unamortized discounts and debt issuance costs		(1,068)		(1,581)		
Total other debt, net	\$	129,932	\$	125,419		

For more information regarding debt instruments outstanding as of December 31, 2023, see Note 7, "Debt" in Part II, Item 8 of the Company's Annual Report on Form 10-K for the year ended December 31, 2023.

Monroe Term Loans—The Monroe Term Loans (as defined below) are comprised of term loans with a principal balance of \$65.0 million (the "Term A-1 Loans") and term loans that were fully repaid during 2023 (the "Term A-2 Loans" and together with the Term A-1 Loans, the "Monroe Term Loans"). The interest rate as of June 30, 2024 on the Term A-1 Loans was 12.33%.

Other Debt—In September 2021, ROAR 1 SPV Finance LLC, an indirect wholly owned subsidiary of the Company (the "ROAR 1 SPV Borrower"), entered into a \$100,000 credit agreement, which, during the first quarter of 2024, decreased to \$80,000 (the "ROAR 1 SPV Credit Facility"), with a lender for the funding of finance receivables, which secure the ROAR 1 SPV Credit Facility. The ROAR 1 SPV Credit Facility allows for increases in maximum borrowings under the agreement of up to \$200,000, bears interest at a rate of 12.5% and matures in March 2025, unless it is extended to March 2026. Under the terms of the ROAR 1 SPV Credit Facility, the ROAR 1 SPV Borrower is subject to certain covenants including minimum asset requirements to be held by ROAR 1 SPV Borrower.

9. LEASES

The Company is party to operating leases for all of its offices. Many leases contain options to renew and extend lease terms and options to terminate leases early. Reflected in the right-of-use asset and lease liability on the consolidated balance sheets are the periods provided by renewal and extension options that the Company is reasonably certain to exercise, as well as the periods provided by termination options that the Company is reasonably certain not to exercise. All long-term leases identified by the Company are classified as operating leases. Lease expenses related to long-term leases were \$1,283 and \$785 for the three months ended June 30, 2024 and 2023, respectively, and \$2,387 and \$1,581 for the six months ended June 30, 2024 and 2023, respectively. Short-term lease expense and variable lease expense were not material for the three and six months ended June 30, 2024 and 2023. Net rental income from subleases of \$166 and \$332 was recorded in other income for the three and six months ended June 30, 2024 and was not material for the three and six months ended June 30, 2023.

Maturities of the Company's long-term operating lease liabilities, which are included in other liabilities on the consolidated balance sheet, were as follows:

	Jun	e 30, 2024
Remainder of 2024	\$	2,295
2025		4,706
2026		3,486
2027		3,334
2028		3,271
Thereafter		2,939
Total lease payments		20,031
Less: imputed interest		5,333
Lease liabilities	\$	14,698
Weighted-average remaining lease term (years)		4.7
Weighted-average discount rate		13.3 %

10. INCOME TAXES

In calculating the provision for income taxes, the Company uses an estimate of the annual effective tax rate based upon the facts and circumstances known at the interim period. The effective tax rate for the six months ended June 30, 2024 was -27.7% and 0.8% for the six months ended June 30, 2023. The decrease in the effective tax rate for the six months ended June 30, 2024 compared to the six months ended June 30, 2023 was primarily attributable to the change in the valuation allowance and certain discrete items related to stock-based compensation.

11. COMMON AND PREFERRED STOCK

Class A Common Stock—Each holder of the shares of Class A Common Stock is entitled to one vote for each share of Class A Common Stock held of record by such holder on all matters on which stockholders generally are entitled to vote, as provide by the Company's Certificate of Incorporation (as amended from time to time). The holders of the shares of Class A Common Stock do not have cumulative voting rights in the election of directors. Generally, all matters to be voted on by the holders of Class A Common Stock must be approved by a majority (or, in the case of election of directors, by a plurality) of the votes entitled to be cast present in person or represented by proxy, unless otherwise specified by law, the Company's Certificate of Incorporation or the Company's Amended and Restated Bylaws (as amended from time to time).

Subject to preferences that may be applicable to any outstanding preferred stock, the holders of shares of Class A Common Stock are entitled to receive ratably such dividends, if any, as may be declared from time to time by MoneyLion's Board of Directors out of funds legally available therefor.

In the event of any voluntary or involuntary liquidation, dissolution or winding up of MoneyLion's affairs, the holders of the shares of Class A Common Stock are entitled to share ratably in all assets remaining after payment of MoneyLion's debts and other liabilities, subject to prior distribution rights of preferred stock or any class or series of stock having a preference over the shares of Class A Common Stock, then outstanding, if any.

The holders of shares of Class A Common Stock have no preemptive or conversion rights or other subscription rights. There are no redemption or sinking fund provisions applicable to the shares of Class A Common Stock. The rights, preferences and privileges of holders of shares of Class A Common Stock will be subject to those of the holders of any shares of the preferred stock MoneyLion may issue in the future.

Series A Preferred Stock—Prior to the Automatic Conversion Event (as described below), the Company had shares of Series A Preferred Stock outstanding. Holders of the shares of Series A Preferred Stock (other than certain regulated holders subject to the Bank Holding Company Act of 1956, as amended) were entitled to vote as a single class with the holders of the Class A Common Stock and the holders of any other class or series of capital stock of MoneyLion then entitled to vote.

Holders of the Series A Preferred Stock were entitled to a 30 cent cumulative annual dividend per share, payable at the Company's election in either cash or Class A Common Stock (or a combination thereof), with any dividends on the Class A Common Stock valued based on the per share volume-weighted average price of the shares of Class A Common Stock on the NYSE for the 20 trading days ending on the trading day immediately prior to the dividend payment date.

Holders of the Series A Preferred Stock were entitled to a liquidation preference in the event of the Company's liquidation equal to the greater of \$10.00 per share or the amount per share that such holder would have received had the Series A Preferred Stock been converted into Class A Common Stock immediately prior to the liquidation.

Shares of Series A Preferred Stock were convertible into shares of Class A Common Stock on a one-for-thirty basis, subject to customary anti-dilution adjustments. The Series A Preferred Stock was convertible (i) at any time upon the holder's election and (ii) automatically in the event that the per share volume-weighted average price of the shares of Class A Common Stock on the NYSE equaled or exceeded \$10.00 on any 20 trading days (consecutive or nonconsecutive) within any consecutive 30 trading day period ending no later than the last day of the lockup period applicable to such shares of Series A Preferred Stock.

As of the close of trading on the NYSE on May 26, 2023, the per share volume-weighted average price of the shares of Class A Common Stock on the NYSE equaled or exceeded \$10.00 for the twentieth trading day within a consecutive thirty trading day period ending no earlier than the last day of the lockup period applicable to such shares of Series A Preferred Stock (the "Automatic Conversion Event"). Accordingly, as a result of the Automatic Conversion Event, following the close of trading on the NYSE on May 26, 2023, all 30,049,053 shares of Series A Preferred Stock issued and outstanding automatically converted into 1,012,293 shares of newly issued Class A Common Stock based on the conversion rate provided in the Certificate of Designations of the Series A Preferred Stock (the "Certificate of Designations"). In lieu of any fractional shares otherwise issuable to any holder of the Series A Preferred Stock, the Company issued cash in accordance with the terms of the Certificate of Designations.

On June 30, 2023, the Company paid the accrued annual dividend on the previously outstanding shares of Series A Preferred Stock for the dividend payment period ending December 31, 2022 to all holders of record as of the applicable dividend record date (the "2022 Annual Dividend"). The 2022 Annual Dividend was paid in a mixture of Class A Common Stock and cash through the issuance of 229,605 shares of Class A Common Stock and payment of approximately \$3.0 million of cash.

12. STOCK-BASED COMPENSATION

Omnibus Incentive Plan

At the Company's 2022 Annual Meeting of Stockholders (the "2022 Annual Meeting"), Company stockholders approved the Company's Amended and Restated Omnibus Incentive Plan (as may be amended or restated from time to time, the "Incentive Plan"), as further described in the Company's Definitive Proxy Statement for the 2022 Annual Meeting, filed with the SEC on April 29, 2022.

Stock-based compensation of \$7,531 and \$5,250 was recognized during the three months ended June 30, 2024 and 2023, respectively, and stock-based compensation of \$14,028 and \$10,955 was recognized during the six months ended June 30, 2024 and 2023, respectively.

The number of units awarded under the Incentive Plan are generally based on a weighted average of the Class A Common Stock in the days leading up to the grant. Fair values for restricted stock units ("RSUs") and performance stock units ("PSUs") based on the Company's operating performance are valued based on the price of the Class A Common Stock at the time of grant. Fair values for options are calculated using a Black-Scholes option pricing model and PSUs with market conditions are fair valued using a Monte Carlo simulation model. The following table represents activity within the Incentive Plan for the six months ended June 30, 2024:

			Weighted Average Grant Date							
Туре	Vesting Conditions	Units Granted	Units Granted Fair Value							
Restricted Stock Unit	Service-based	374,776	\$	52.75	n/a					
Performance Stock Unit	Service and performance-based	85,090	\$	52.55	n/a					

The following table represents outstanding equity awards as of June 30, 2024:

Туре	Vesting Conditions	Units Outstanding	Weighted Average Strike Price		
Restricted Stock Unit	Service-based	923,396	\$ 39.10	n/a	
Performance Stock Unit	Service and performance-based	230,159	\$ 33.08	n/a	
Performance Stock Unit	Service and market-based	248,894	\$ 13.99	n/a	
Options	Service-based	534,857	\$ 20.81	\$ 31.73	

13. STOCK WARRANTS

Public Warrants and Private Placement Warrants

As a result of the Business Combination, MoneyLion acquired from Fusion Acquisition Corp., as of September 22, 2021, public warrants outstanding to purchase an aggregate of 583,333 shares of the Class A Common Stock (the "Public Warrants") and private placement warrants outstanding to purchase an aggregate of 270,000 shares of the Class A Common Stock (the "Private Placement Warrants") that expire on September 22, 2026.

The Public Warrants meet the conditions for equity classification in accordance with ASC 815-40. The Private Placement Warrants are accounted for as liabilities in accordance with ASC 815-40 and are presented within warrant liability on the consolidated balance sheets. The warrant liability is measured at fair value at inception and on a recurring basis, with changes in fair value presented within change in fair value of warrants liability in the consolidated statement of operations.

The Private Placement Warrants are valued based on the per warrant price of the Public Warrants, subject to adjustments to account for differences in contractual terms between the Private Placement Warrants and the Public Warrants. The per warrant price of the Public Warrants as of June 30, 2024 was \$0.10.

The following table presents the changes in the liability related to the Private Placement Warrants:

	Private Plac	ement
	Warran	its
Warrants payable balance, December 31, 2023	\$	810
Mark-to-market adjustment		-
Warrants payable balance, June 30, 2024	\$	810

For more information regarding the Public Warrants and Private Placement Warrants, see Note 12, "Stock Warrants" in Part II, Item 8 of the Company's Annual Report on Form 10-K for the year ended December 31, 2023.

14. NET INCOME (LOSS) PER SHARE

The following table sets forth the computation of net income (loss) per share of Class A Common Stock for the three and six months ended June 30, 2024 and 2023:

	Three Months	Ended	June 30,		Six Months Ended June 30,			
	2024		2023	2024			2023	
Numerator:								
Net income (loss)	\$ 3,137	\$	(27,723)	\$	10,212	\$	(36,940)	
Reversal of previously accrued dividends on preferred stock	 		2,667		_		690	
Net income (loss) attributable to common shareholders	\$ 3,137	7 \$ (25,056		\$ 10,212		\$	(36,250)	
Denominator:								
Weighted-average common shares outstanding - basic	10,862,853		9,234,238		10,694,635		8,944,836	
Plus: dilutive effect of common stock equivalents	 1,228,568		_		1,260,340		_	
Weighted-average common shares outstanding - diluted	12,091,421		9,234,238		11,954,975		8,944,836	
Net income (loss) per share attributable to common stockholders - basic	\$ 0.29	\$	(2.71)	\$	0.95	\$	(4.05)	
Net income (loss) per share attributable to common stockholders - diluted	\$ 0.26	\$	(2.71)	\$	0.85	\$	(4.05)	

For the three and six months ended June 30, 2024, 168,942 and 216,498, respectively, options to purchase Class A Common Stock and other rights to acquire Class A Common Stock were outstanding and anti-dilutive and, therefore, are excluded from the computation of diluted net income per share attributable to common stockholders. In addition, 85,090 PSUs are excluded from the computation of diluted net income per share attributable to common stockholders as the contingency has not yet been satisfied for the three and six months ended June 30, 2024. All Public Warrants and Private Placement Warrants to purchase Class A Common Stock and rights to receive Earnout Shares (as defined below) are excluded from the computation of diluted net income per share attributable to common stockholders as the relevant purchase price and milestones, respectively, were above the average price of the Class A Common Stock during the three and six months ended June 30, 2024.

For the three and six months ended June 30, 2023, the Company's potentially dilutive securities, which include stock options, RSUs, PSUs, preferred stock, the rights to receive Earnout Shares (as defined below) and warrants to purchase shares of common stock, have been excluded from the computation of diluted net loss per share as the effect would be antidilutive. Therefore, the weighted-average number of common shares outstanding used to calculate both basic and diluted net loss per share is the same for the three and six months ended June 30, 2023.

The following potential shares of Class A Common Stock have been excluded from the computation of diluted net income (loss) per share for the three and six months ended June 30, 2024 and 2023:

	Three Months En	ded June 30,	Six Months Ended June 30,		
	2024	2023	2024	2023	
Warrants to purchase common stock	853,330	853,330	853,330	853,330	
PSUs, RSUs and options to purchase common stock	254,032	2,448,661	301,588	2,448,661	
Right to receive Earnout Shares	583,333	583,333	583,333	583,333	
Total common stock equivalents	1,690,695	3,885,324	1,738,251	3,885,324	

In connection with the Business Combination, rights to receive Class A Common Stock (the "Earnout Shares") were issued, with the right to receive Class A Common Stock contingent upon the Class A Common Stock reaching certain price milestones. 250,000 and 333,333 shares of Class A Common Stock will be issued if the Class A Common Stock share price equals or is greater than \$375 and \$495, respectively, for twenty out of any thirty consecutive trading days. The right to receive the Earnout Shares will expire on September 22, 2026.

15. COMMITMENTS AND CONTINGENCIES

Legal Matters—From time to time, the Company is subject to various claims and legal proceedings in the ordinary course of business, including lawsuits, arbitrations, class actions and other litigation. The Company is also the subject of various actions, inquiries, investigations and proceedings by regulatory and other governmental agencies. The outcome of any such legal and regulatory matters, including those discussed in this Note 15, is inherently uncertain, and some of these matters may result in adverse judgments or awards, including penalties, injunctions or other relief, which could materially and adversely impact the Company's business, financial condition, operating results and cash flows. See Part I, Item 1A "Risk Factors — Risks Relating to Legal and Accounting Matters — Unfavorable outcomes in legal proceedings may harm our business, financial condition, results of operations and cash flows" in the Company's Annual Report on Form 10-K for the year ended December 31, 2023.

The Company has determined, based on its current knowledge, that the aggregate amount or range of losses that are estimable with respect to its legal proceedings, including the matters described below, would not have a material adverse effect on its business, financial position, results of operations or cash flows. As of June 30, 2024, amounts accrued were not material. Notwithstanding the foregoing, the ultimate outcome of legal proceedings involves judgments, estimates and inherent uncertainties, and cannot be predicted with certainty. It is possible that an adverse outcome of any matter could be material to the Company's business, financial position, results of operations or cash flows as a whole for any particular reporting period of occurrence. In addition, it is possible that a matter may prompt litigation or additional investigations or proceedings by other government agencies or private litigants.

The Company holds a number of state licenses in connection with its business activities, and must also comply with other applicable compliance and regulatory requirements in the states where it operates. In most states where the Company operates, one or more regulatory agencies have authority with respect to regulation and enforcement of the Company's business activities under applicable state laws, and the Company may also be subject to the supervisory and examination authority of such state regulatory agencies. Examinations by state regulators have and may continue to result in findings or recommendations that require the Company, among other potential consequences, to provide refunds to customers or to modify its internal controls and/or business practices.

In the ordinary course of its business, the Company is and has been from time to time subject to, and may in the future be subject to, governmental and regulatory examinations, information requests, investigations and proceedings (both formal and informal) in connection with various aspects of its activities by state agencies, certain of which could result in adverse judgments, settlements, fines, penalties, restitution, disgorgement, injunctions or other relief. The Company has responded to and cooperated with the relevant state agencies and will continue to do so in the future, as appropriate.

On September 29, 2022, the Consumer Financial Protection Bureau (the "CFPB") initiated a civil action in the United States District Court for the Southern District of New York ("SDNY") against MoneyLion Technologies Inc., ML Plus LLC and the Company's 38 state lending subsidiaries, alleging violations of the Military Lending Act and the Consumer Financial Protection Act. The CFPB is seeking injunctive relief, redress for allegedly affected consumers and civil monetary penalties. On January 10, 2023, the Company moved to dismiss the lawsuit, asserting various constitutional and merits-based arguments. On June 13, 2023, the CFPB filed its first amended complaint, alleging substantially similar claims as those asserted in its initial complaint. On July 11, 2023, the Company moved to dismiss the lawsuit, again asserting various constitutional and merit-based arguments. On October 9, 2023, the Company moved for a stay of the action pending a decision from the United States Supreme Court in CFPB v. Community Financial Services Association of America, Ltd., No. 22-448 (U.S. argued Oct. 3, 2023) ("CFSA"). On December 1, 2023, the Court issued an order granting the Company's motion to dismiss is now pending with the SDNY. The Company continues to maintain that the CFPB's claims are meritless and is vigorously defending against the lawsuit. Nevertheless, at this time, the Company cannot predict or determine the timing or final outcome of this matter or the effect that any adverse determinations in the lawsuit may have on its business, financial condition, results of operations or cash flows.

On July 21, 2023, Jeffrey Frommer, Lyusen Krubich, Daniel Fried and Pat Capra, the former equity owners of MALKA (collectively, the "Seller Members"), brought a civil action in the SDNY against MoneyLion Technologies Inc. alleging, among other things, breaches of the Membership Interest Purchase Agreement (the "MIPA") governing the acquisition of MALKA (the "MALKA Acquisition"). Among other claims, the Seller Members allege that they are entitled to payment of \$25.0 million of Class A Common Stock pursuant to the earnout provisions set forth in the MIPA, based on the Seller Members' assertion that MALKA achieved certain financial targets for the year ended December 31, 2022 (such payment, the "2022 Earnout Payment"). The Company believes that the Seller Members are not entitled to any portion of the 2022 Earnout Payment" under the terms of the MIPA and filed counterclaims against the Seller Members, alleging, among other things, fraud, negligent misrepresentation, conversion, breach of fiduciary duties and breach of contract and seeking compensatory damages and other remedies as a result of wrongdoing by the Seller Members. On October 17, 2023, the SDNY denied, in full, the Seller Members' motion to dismiss with respect to the Company's counterclaims and third-party complaint. On May 14, 2024, the SDNY denied the Seller Members' motion to dismiss with respect to the Company's counterclaims relating to declaratory judgment, unjust enrichment and conversion and breach of contract claims. The SDNY dismissed certain of the Company's counterclaims relating to declaratory judgment, unjust enrichment and conversion as duplicative of the fraud and misrepresentation conterclaims, as well as certain of ther Company believes are meritles. However, at this time, the Company cannot predict or determine the timing or final outcome of this matter or the effect that any adverse determinations in the lawsuit may have on its business, financial condition, results of operations or cash flows.

On July 27, 2023, MassMutual Ventures US II LLC, Canaan X L.P., Canaan XI L.P., F-Prime Capital Partners Tech Fund LP and GreatPoint Ventures Innovation Fund II, L.P., each of which are former holders of the Company's Series A Preferred Stock (collectively, the "Former Preferred Stockholders"), brought a civil action in the SDNY against MoneyLion Inc., the Company's Board of Directors and certain officers asserting claims under Section 14(a) relating to the Definitive Proxy Statement we filed with the SEC on March 31, 2023 in connection with the Special Meeting of Stockholders relating to the 1-for-30 Reverse Stock Split of the Class A Common Stock effected on April 24, 2023 and related state law claims. On November 6, 2023, the Company filed a motion to dismiss the lawsuit in its entirety. On May 15, 2024, the SDNY granted the Company's motion to dismiss the Former Preferred Stockholders' claim under Section 14(a) of the Securities Exchange Act of 1934 for failure to state a claim and declined to exercise supplemental jurisdiction over the remaining state law claims, dismissing such state law claims without prejudice. On June 14, 2024, Canaan X L.P., Canaan XI L.P., GreatPoint Ventures Innovation Fund II, L.P. filed a notice of appeal with the United States Court of Appeals for the Second Circuit. The Company continues to believe that the Former Preferred Stockholders' claims are meritless and intends to vigorously defend against the appeal if any Former Preferred Stockholders decide to pursue their appeal. Nevertheless, at this time, the Company cannot predict or determine the timing or final outcome of this matter or the effect that any adverse determinations in the lawsuit may have on its business, financial condition, results of operations or cash flows.

16. MERGERS AND ACQUISITIONS

Engine—On February 17, 2022, the Company completed the acquisition of all voting interest in Even Financial Inc., which was subsequently renamed to Engine. Engine powers the leading embedded finance marketplace solutions MoneyLion offers to its Enterprise Partners through which consumers are connected and matched with real-time, personalized financial product and service recommendations.

At the closing of the Engine Acquisition, the equityholders and advisors of Even Financial Inc. were entitled to receive a payment from the Company of up to an aggregate of 8,000,000 shares of Series A Preferred Stock, based on the attributed revenue of Engine's business during the 13-month period commencing January 1, 2022 (the "Earnout"), and certain recipients of options to acquire shares of the Company's Class A common stock were entitled to receive dividend equivalents in lieu of receiving Series A Preferred Stock, subject to certain conditions (the "Preferred Stock Equivalents").

The \$6,367 and \$6,433 decline in fair value of the Earnout and the Preferred Stock Equivalents for the three and six months ended June 30, 2023, respectively, was included on the consolidated statement of operations as a component of the change in fair value of contingent consideration from mergers and acquisitions.

In May 2023, the Earnout was settled through the issuance of 4,354,092 shares of Series A Preferred Stock, with cash paid in lieu of any fractional shares of Series A Preferred Stock. Cash payments relating to the settlement of the Earnout were \$459. In June 2023, the Preferred Stock Equivalents were settled through the issuance of 23,453 shares of Class A Common Stock, with cash paid in lieu of any fractional shares of Class A Common Stock. Cash payments relating to the settlement of the Preferred Stock Equivalents were \$307. Upon the Automatic Conversion Event, the MoneyLion Inc. Preferred Share Dividend Replacement Program governing the Preferred Stock Equivalents immediately and automatically terminated in accordance with its terms, following which all Preferred Stock Equivalents were forfeited.

MALKA—On November 15, 2021, MoneyLion completed the MALKA Acquisition. MALKA is a creator network and content platform that provides digital media and content production services to us and to its own clients in entertainment, sports, gaming, live streaming and other sectors.

The unsettled restricted shares payable relating to the MALKA Acquisition earnout and the related make-whole were settled during the first quarter of 2023. The \$180 decline in fair value for the six months ended June 30, 2023 was included on the consolidated statement of operations as a component of the change in fair value of contingent consideration from mergers and acquisitions.

17. SUBSEQUENT EVENTS

The Company has evaluated subsequent events through August 6, 2024, the date on which these consolidated financial statements were available to be issued, and concluded that the following subsequent event was required to be disclosed:

On August 2, 2024, ML Plus LLC, a wholly-owned indirect subsidiary of MoneyLion Inc., entered into the Third Amendment to Account Servicing Agreement ("Third Amendment") with Pathward, N.A. ("Pathward") to amend that certain Account Servicing Agreement, dated as of January 14, 2020 (as amended from time to time, the "Agreement"), pursuant to which ML Plus LLC provides its RoarMoney demand deposit accounts and debit cards with Pathward, as partner issuing bank (the "Program Services"), in order to enable the Company to offer overdraft protection in connection with the Program Services. The Third Amendment also extends the term of the Agreement to January 2029 (with automatic renewal for successive two-year periods unless either party provides written notice of non-renewal, which may be without cause, at least 180 days prior to the end of any such term).

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

The following Management's Discussion and Analysis of Financial Condition and Results of Operations summarizes the significant factors affecting the consolidated operating results, financial condition, liquidity and capital resources of MoneyLion and is intended to help the reader understand MoneyLion, our operations and our present business environment. This discussion should be read in conjunction with MoneyLion's unaudited consolidated financial statements and notes to those financial statements included in Part I, Item 1 "Financial Statements" within this Quarterly Report on Form 10-Q. References to "we," "us," "our," "Company" or "MoneyLion" refer to MoneyLion Inc. and, as context requires, its wholly-owned subsidiaries.

Overview

MoneyLion is a leader in financial technology, powering the next generation of personalized products and financial content for American consumers. MoneyLion was founded in 2013 with a vision to rewire the financial system. Our mission is to give everyone the power to make their best financial decisions. We believe that the financial wellness gap in America can be addressed by bridging the financial literacy and the financial access gaps, shortening the distance between education and action.

We design and offer modern personal finance products, tools and features and curate money-related content that delivers actionable insights and guidance to our users. We also operate and distribute embedded finance marketplace solutions that match consumers with personalized third-party offers from our partners, providing convenient access to an expansive breadth of financial solutions that enable consumers to borrow, spend, save and achieve better financial outcomes. Our leading marketplace solutions provide valuable distribution, acquisition, growth and monetization channels for our partners. In addition, we provide creative media and brand content services to clients across industries through our media division and leverage our adaptive, in-house content studio to produce and deliver engaging and dynamic content in support of our product and service offerings.

We have purposefully built our platform to help consumers navigate all of their financial inflection points, combining our deep first-party product expertise, engaging content, marketplaces, innovative technology, data and AI capabilities to create the ultimate marketplace solution. As of June 30, 2024, we had 17.0 million Total Customers who used 27.7 million Total Products and over 1,200 Enterprise Partners in our network. We strategically employ comprehensive, data-driven analytics and cutting-edge technology to enhance our platform, creating personalized experiences for our users based on our rich datasets. Utilizing innovative approaches to financial guidance that engage and educate our users within a peer community, we seek to empower consumers to take control of their financial lives.

In our Consumer business, we primarily earn revenue as follows:

- <u>RoarMoney Banking</u>: We earn revenue from interchange fees from payment networks based on customer expenditures on the debit card, as well as transaction volume-based incentive payments from the payment network. We also earn revenue from cardholder fees charged to our customers, such as the monthly administrative fee, an out-of-network ATM fee, a foreign transaction fee and instant transfer fees. Interchange fees, payment network payments and cardholder fees are reflected in service and subscription fees.
- Instacash: We earn revenue from optional tips and instant transfer fees, both reflected in service and subscription fees.
- <u>Membership Programs</u>: We earn revenue from the monthly subscription fee paid by our customers, which is reflected in service and subscription fees. Membership programs also provide customers access to Credit Builder Loans from which we earn revenue from interest income, which is reflected in net interest income on finance receivables.
- <u>MoneyLion Investing</u>: We earn revenue from the monthly administration fee paid by our customers, which is reflected in service and subscription fees.

• <u>MoneyLion Crypto</u>: We earn revenue from Zero Hash, which is reflected in service and subscription fees. Zero Hash pays us a share of the fees that they earn from our customers in exchange for us enabling Zero Hash to effect digital currency-related transactions for our customers.

In our Enterprise business, we primarily earn revenue, reflected in enterprise service revenues, as follows:

- <u>Consumer Marketplace</u>: We earn revenue from fees from our Product Partners based on a range of criteria depending on each Product Partner relationship, including, but not limited to, customers' clicks, impressions, completed transactions or a share of revenue generated for the Product Partner.
- <u>Enterprise Marketplace</u>: We earn revenue from fees from our Enterprise Partners based on a range of criteria depending on each Enterprise Partner relationship, including, but not limited to, customers' clicks, completed transactions or a share of revenue generated for the Product Partner. We also earn various SaaS and platform fees from our Enterprise Partners.
- <u>Media Services</u>: We earn revenue from our clients based on performance obligations within our contracts with them.

Recent Developments

On August 2, 2024, ML Plus LLC, a wholly-owned indirect subsidiary of MoneyLion Inc., entered into the Third Amendment to Account Servicing Agreement with Pathward, N.A. to amend that certain Account Servicing Agreement, dated as of January 14, 2020, pursuant to which ML Plus LLC provides its RoarMoney demand deposit accounts and debit cards with Pathward, as partner issuing bank, in order to enable the Company to offer overdraft protection in connection with the Program Services. The Third Amendment also extends the term of the Agreement to January 2029 (with automatic renewal for successive two-year periods unless either party provides written notice of non-renewal, which may be without cause, at least 180 days prior to the end of any such term).

Factors Affecting Our Performance

We are subject to a number of risks including, but not limited to, the need for successful development of products, services and functionality; the need for additional capital (or financing) to fund operating losses; competition with substitute products and services from larger companies; protection of proprietary technology and information; dependence on key individuals; and risks associated with changes in information technology. For additional information, see the "Risk Factors" section in our Annual Report on Form 10-K for the year ended December 31, 2023.

New Customer and Client Growth and Increasing Usage Across Existing Customers and Clients

Our ability to effectively acquire new customers and clients through our acquisition and marketing efforts and drive usage of our products and services across our existing customers and clients is key to our growth, particularly as a significant portion of the revenue we generate in our business is derived from transaction-based fees. We believe our customers' experience is enhanced by using our full suite of first-party financial products and services, complemented by the full spectrum of offers available in our marketplace, as we can better tailor the insights and recommendations we provide to them. In order to grow our business, we must engage and retain customers and continue to expand their use of our platform by cross-selling additional functionality, products and services to them. In our Enterprise business, we are dependent in part on our relationships with our Enterprise Partners, and any failure to effectively match consumers leads from our Channel Partners with product and service offerings from our Product Partners on our Enterprise platform, could adversely affect our business and results of operations.

Expansion and Innovation of Products, Services and Functionality

We will continue to invest in expanding and enhancing the products, services and functionality available through our platform for our customers and clients. Our ability to expand, enhance and sell additional functionality, products and services to our existing customers and clients may require more sophisticated and costly development, sales or engagement efforts. Any factors that impair our ability to do so may negatively impact our efforts towards retaining and attracting customers and clients.

General Economic and Market Conditions

Our performance is impacted by the relative strength of the overall economy, market volatility, consumer spending behavior and consumer demand for financial products and services. For example, with respect to our Consumer business, the willingness of our customers to spend, invest or borrow may fluctuate with their level of disposable income. Other factors such as interest rate fluctuations or monetary policies may also impact our customers' behavior and our own ability to fund Instacash advances and loan volume. In addition, in our Enterprise business, adverse macroeconomic conditions, such as significant tightening of credit markets, may cause our Product Partners to reduce their marketing spend or advertising on our platform or may cause a reduction in client spending in our Media Services division, which could adversely affect our business and results of operations.

Seasonality

We may experience seasonal fluctuations in our revenue. During the fourth quarter, revenue in our Consumer business may benefit from increased consumer spending during the holiday season, which may increase demand for our advance product as consumers seek additional liquidity. During the first quarter, we may see stronger collections on Instacash receivables resulting in a relatively lower provision for credit losses on consumer receivables as a result of the impact of tax refunds, as well as stronger demand for our banking and investment products and services. Seasonal trends may be superseded by market or macroeconomic events, which can have a significant impact on our business, as described above.

Competition

We compete across our business lines with a variety of competitors, including traditional banks and credit unions; new entrants obtaining banking licenses; non-bank digital providers offering banking-related services; specialty finance and other non-bank digital providers offering consumer lending-related or earned wage access products; digital wealth management platforms such as robo-advisors offering consumer investment services and other brokerage-related services; and digital financial platform, embedded finance and marketplace competitors, which aggregate and connect consumers to financial product and service offerings. In addition to competing for customers for our product and service offerings, we also compete to attract viewership of the content to which we connect customers, as there are other sources of financial-related content and news, many of which are more established and have a larger subscriber base. Furthermore, we compete with other advertising agencies and other service providers to attract marketing budget spending from our Enterprise clients. With respect to our Media Services division, we compete with others in the digital media and content creation industry, which range from large and established media companies, including social media companies, advertising agencies and production studios, to emerging start-ups. We expect our competition to continue to increase. The success of our business depends on our ability to compete effectively and attract new and retain existing customers and clients, which depends upon many factors both within and beyond our control.



Pricing of Our Products and Services

We derive a substantial portion of our revenue from fees earned from our products and services. The fees we earn are subject to a variety of external factors such as competition, interchange rates and other macroeconomic factors, such as interest rates and inflation, among others. We may provide discounts or other incentives and rewards that we pay to customers who utilize multiple products and services to expand usage of our platform. We may also lower pricing on our products and services to acquire new customers. As the market for our platform matures, or as new or existing competitors introduce new products, services or functionality that compete with ours, we may experience pricing pressure and be unable to retain current customers and clients and attract new customers and clients at prices that are consistent with our pricing model and operating budget. Our pricing strategy may prove to be unappealing to our customers and clients, and our competitors could choose to bundle certain products and services competitive with ours. If this were to occur, it is possible that we would have to change our pricing strategies or reduce our prices, which could adversely affect our business.

Product and Service Mix

We offer various products and services on our platform, including our core suite of first-party financial products and services, a broad range of financial and non-financial offers in our Consumer Marketplace, Enterprise Marketplace and Media Services in our Enterprise business. Each product and service has a different profitability profile. The relative usage of products and services with high or low profitability and their lifetime value could have an impact on our performance.

Access and Cost of Financing; Forward Flow Arrangement

Our credit products and Instacash product are financed by special purpose vehicle financings from third-party institutional lenders and, in the future, a forward flow financing arrangement pursuant to which we will sell a portion of our eligible Instacash receivables to third-party purchasers and receive a stable stream of servicing fee income, as described further under Part I, Item 1 "Financial Statements – Sale of Consumer Receivables." The loss of one or more of the financing sources we have for our credit products and Instacash product could have an adverse impact on our performance, and it could be costly to obtain new financing. In addition, the initial price at which we sell Instacash receivables under the forward flow financing arrangement will be based on the average loss rate at 360 days past the repayment date of the three most recent matured monthly cohorts and will be subject to adjustment for future monthly cohorts based on the performance of monthly cohorts at specified intervals past the repayment date compared to the expected loss rates of the reference matured monthly cohorts established at sale and a discount percentage. As a result, the loss on sale of the Instacash receivables were sold during the forward flow financing arrangement will be variable depending on the performance of the previously sold Instacash receivables. No Instacash receivables were sold during the three and six months ended June 30, 2024.

Key Performance Metrics

We regularly review several metrics, including the following key metrics, to evaluate our business, measure our performance, identify trends affecting our business, formulate financial projections and make strategic decisions.

Total Customers

We define Total Customers as the cumulative number of customers that have opened at least one account, including banking, membership subscription, secured personal loan, Instacash advance, managed investment account, cryptocurrency account and customers that are monetized through our marketplace and affiliate products. Total Customers also include customers that have submitted for, received or clicked on at least one marketplace credit offer. We consider Total Customers to be a key performance metric as it can be used to understand lifecycle efforts of our customers, as we look to cross-sell products to our customer base and grow our platform. Total Customers were 17.0 million and 9.9 million as of June 30, 2024 and 2023, respectively.

Total Products

We define Total Products as the total number of products that our Total Customers have opened, including banking, membership subscription, secured personal loan, Instacash advance, managed investment account, cryptocurrency account and monetized marketplace and affiliate products, as well as customers who signed up for our financial tracking services (with either credit tracking enabled or external linked accounts), whether or not the customer is still registered for the product. Total Products also include marketplace credit offers that our Total Customers have submitted for, received or clicked on through our marketplace. If a customer has funded multiple secured personal loans or Instacash advances or opened multiple products through our marketplace, it is only counted once for each product type. We consider Total Products to be a key performance metric as it can be used to understand the usage of our products across our customer base. Total Products were 27.7 million and 17.3 million as of June 30, 2024 and 2023, respectively.

Enterprise Partners

Enterprise Partners is comprised of Product Partners and Channel Partners. We define Product Partners as providers of the financial and non-financial products and services that we offer in our marketplaces, including financial institutions, financial services providers and other affiliate partners. We define Channel Partners as organizations that allow us to reach a wide base of consumers, including but not limited to news sites, content publishers, product comparison sites and financial institutions. Enterprise Partners were 1,254, comprising 613 Product Partners and 641 Channel Partners, and 1,110, comprising 505 Product Partners and 605 Channel Partners, as of June 30, 2024 and 2023, respectively.

Total Originations

We define Total Originations as the dollar volume of the secured personal loans originated and Instacash advances funded within the stated period. We consider Total Originations to be a key performance metric as it can be used to measure the usage and engagement of the customers across our secured personal lending product and Instacash earned wage access product and is a significant driver of net interest income on finance receivables and service and subscription fees. Total Originations were \$770 million and \$550 million for the three months ended June 30, 2024 and 2023, respectively, and \$1,487 million and \$1,056 million for the six months ended June 30, 2024 and 2023, respectively. All originations were originated directly by MoneyLion.

Adjusted EBITDA (Non-GAAP Measure)

Management believes Adjusted EBITDA, a non-U.S. GAAP measure, provides relevant and useful information to investors regarding the performance of the company. Refer to the "--- Non-GAAP Measures" section below for further discussion of Adjusted EBITDA.

Results of Operations for the Three and Six Months Ended June 30, 2024 and 2023

Revenues

The following table is reference for the discussion that follows.

	TI	hree Month 3	s Enc 0,	led June	Cha	nge	Si	x Months E	nded	June 30		Char	σe
		2024	•••	2023	 \$	%		2024	nucu	2023		\$ \$	%
					 (In thousands, except for percentages)								
Consumer revenues													
Service and subscription fees	\$	89,444	\$	66,268	\$ 23,176	35.0%	\$	174,653	\$	128,706	\$	45,947	35.7%
Net interest income on finance receivables		2,956		3,304	(348)	-10.5%		5,889		6,232		(343)	-5.5%
Total consumer revenues		92,400		69,572	 22,828	32.8%		180,542		134,938		45,604	33.8%
Enterprise service revenues		38,446		36,969	 1,477	4.0%		71,310		65,272		6,038	9.3%
Total revenue, net	\$	130,846	\$	106,541	\$ 24,305	22.8%	\$	251,852	\$	200,210	\$	51,642	25.8%

We generate revenue primarily from various product-related fees, providing membership subscriptions, performing enterprise services and originating loans.

Service and subscription fees

Service and subscription fees increased by \$23.2 million, or 35.0%, to \$89.4 million for the three months ended June 30, 2024, as compared to \$66.3 million for the same period in 2023. The increase in service and subscription fees was driven by increases in fee income related to instant transfer fees and tips from Instacash advances of \$22.0 million due to the growth of Instacash advances across both existing and new customers, an increase in subscription fees of \$0.6 million due to an increased number of customers using our membership programs and an increase of \$0.6 million in revenue from a transaction volume-based incentive payment program from a third-party payment network compared to the three months ended June 30, 2023.

Service and subscription fees increased by \$45.9 million, or 35.7%, to \$174.7 million for the six months ended June 30, 2024, as compared to \$128.7 million for the same period in 2023. The increase in service and subscription fees was driven by increases in fee income related to instant transfer fees and tips from Instacash advances of \$43.4 million due to the growth of Instacash advances across both existing and new customers, an increase in cardholder and interchange fees from RoarMoney accounts of \$0.4 million due to an increased number of customers using RoarMoney, an increase in membership revenues of \$1.5 million and an increase of \$0.8 million in revenue from a transaction volume-based incentive payment program from a third-party payment network compared to the six months ended June 30, 2023.

Net interest income on finance receivables

Net interest income on finance receivables is generated by interest earned on Credit Builder Loans, which is partially offset by the amortization of loan origination costs.

Net interest income on finance receivables decreased by \$0.3 million, or 10.5%, to \$3.0 million for the three months ended June 30, 2024, as compared to \$3.3 million for the same period in 2023. The decrease in net interest income on finance receivables of \$0.1 million was driven by the slightly lower receivables outstanding on our Credit Builder Plus loan program across both existing and new customers and higher provision on interest fees of \$0.2 million.

Net interest income on finance receivables decreased by \$0.3 million, or 5.5%, to \$5.9 million for the six months ended June 30, 2024, as compared to \$6.2 million for the same period in 2023. The decrease in net interest income on



finance receivables of \$0.3 million was driven by the slightly lower receivables outstanding on our Credit Builder Plus loan program across both existing and new customers and higher provision on interest fees of \$0.3 million, which was slightly offset by lower amortization of loan origination costs of \$0.2 million.

Enterprise service revenues

Enterprise service revenues increased by \$1.5 million, or 4.0%, to \$38.4 million for the three months ended June 30, 2024, as compared to \$37.0 million for the same period in 2023. This increase was primarily driven by stronger performance within our Enterprise Marketplace, which was partially offset by lower performance within our Consumer Marketplace and Media Services division.

Enterprise service revenues increased by \$6.0 million, or 9.3%, to \$71.3 million for the six months ended June 30, 2024, as compared to \$65.3 million for the same period in 2023. This increase was primarily driven by stronger performance within our Enterprise Marketplace, which was partially offset by lower performance in our Consumer Marketplace and Media Services division.

Operating Expenses

The following table is reference for the discussion that follows:

	Three Mon June		Cha	inge	5	Six Months I 30		ed June	Cha	inge	
	 2024	2023	 \$	%		2024	2023		 \$	%)
	 	 	 (In	thousands, exc	ept f	or percenta	ges))	 		
Operating expenses											
Provision for credit losses on consumer											
receivables	\$ 33,431	\$ 25,562	\$ 7,869	30.8%	5 \$	53,661	\$	42,073	\$ 11,588		27.5%
Compensation and benefits	24,852	22,572	2,280	10.1%	, D	49,638		46,980	2,658		5.7%
Marketing	10,530	6,549	3,981	60.8%	b	21,396		12,941	8,455		65.3%
Direct costs	34,449	32,230	2,219	6.9%	, D	65,838		62,032	3,806		6.1%
Professional services	11,007	4,518	6,489	143.6%	, D	16,773		9,517	7,256		76.2%
Technology-related costs	6,512	5,611	901	16.1%	, D	13,098		11,649	1,449		12.4%
Other operating expenses	4,338	11,219	(6,881)	-61.3%	,)	14,658		20,214	(5,556)		-27.5%
Total operating expenses	\$ 125,119	\$ 108,261	\$ 16,858	15.6%	5 \$	235,062	\$	205,406	\$ 29,656		14.4%
Other (expense) income											
Interest expense	\$ (6,714)	\$ (7,330)	\$ 616	-8.4%	5 \$	(13,531)	\$	(14,841)	\$ 1,310		-8.8%
Change in fair value of warrant liability	(81)	162	(243)	nm		_		13	(13)	-	100.0%
Change in fair value of contingent											
consideration from mergers and acquisitions		6,367	(6,367)	-100.0%	Ď	—		6,613	(6,613)	-	100.0%
Goodwill impairment loss	_	(26,721)	26,721	-100.0%	, D	_		(26,721)	26,721	-	100.0%
Other income	2,381	1,257	1,124	89.4%	, D	4,740		2,906	1,834		63.1%
Total other expense	\$ (4,414)	\$ (26,265)	\$ 21,851	-83.2%	5 \$	(8,791)	\$	(32,030)	\$ 23,239		-72.6%
Income tax benefit	\$ (1,824)	\$ (262)	\$ (1,562)	596.2%	5 \$	(2,213)	\$	(286)	\$ (1,927)		<u>673.8</u> %

Our operating expenses consist of the following:

Provision for credit losses on consumer receivables

Provision for credit losses on consumer receivables consists of amounts charged during the period to maintain an allowance for credit losses. The allowance represents management's estimate of the credit losses in our consumer receivable portfolio and is based on management's assessment of many factors, including changes in the nature, volume and risk characteristics of the consumer receivables portfolio, including trends in delinquency and charge-offs and current economic conditions that may affect the customer's ability to pay.

Provision for credit losses on consumer receivables increased by \$7.9 million, or 30.8%, to \$33.4 million for the three months ended June 30, 2024, as compared to \$25.6 million for the same period in 2023. This resulted primarily from an increase to provision related to Instacash advance receivables of \$8.3 million and an increase to provision related to subscription fees of \$0.5 million, which was partially offset by a decrease to provision related to Credit Builder Plus loan receivables of \$1.1 million.

Provision for credit losses on consumer receivables increased by \$11.6 million, or 27.5%, to \$53.7 million for the six months ended June 30, 2024, as compared to \$42.1 million for the same period in 2023. This resulted primarily from an increase to provision related to Instacash advance receivables of \$15.8 million and an increase to provision related to subscription fees of \$0.5 million, which was partially offset by a decrease to provision related to Credit Builder Plus loan receivables of \$2.2 million and a decrease to provision for Instacash instant transfer fees and tips of \$2.5 million.

Compensation and benefits

Compensation and benefits increased by \$2.3 million, or 10.1%, to \$24.9 million for the three months ended June 30, 2024, as compared to \$22.6 million for the same period in 2023. This increase was primarily driven by an increase in stock-based compensation of \$2.3 million, an increase in employee salary and benefits expenses of \$0.2 million and an increase in incentive compensation of \$0.1 million due to company performance, which was partially offset by higher capitalized salaries of \$0.4 million.

Compensation and benefits increased by \$2.7 million, or 5.7%, to \$49.6 million for the six months ended June 30, 2024, as compared to \$47.0 million for the same period in 2023. This increase was primarily driven by an increase in stock-based compensation of \$3.1 million, an increase in incentive compensation of \$0.6 million due to company performance and an increase in severance costs of \$0.3 million. This was partially offset by higher capitalized salaries of \$1.1 million and a \$0.2 million decrease in employee salary and benefits expenses.

Marketing

Marketing increased by \$4.0 million, or 60.8%, to \$10.5 million for the three months ended June 30, 2024, as compared to \$6.5 million for the same period in 2023. This increase resulted primarily from higher spend related to advertising through digital platforms and sponsorships.

Marketing increased by \$8.5 million, or 65.3%, to \$21.4 million for the six months ended June 30, 2024, as compared to \$12.9 million for the same period in 2023. This increase resulted primarily from higher spend related to advertising through digital platforms and sponsorships.

Direct costs

Direct costs increased by \$2.2 million, or 6.9%, to \$34.4 million for the three months ended June 30, 2024, as compared to \$32.2 million for the same period in 2023. The increase was primarily driven by an increase in payment processing fees of \$2.1 million and an increase in underwriting expenses of \$0.6 million, driven by growth in Total Originations and Total Customers, which was partially offset by a \$0.3 million decrease in costs related to our MoneyLion Investing and RoarMoney Banking offering and a decrease of \$0.2 million in direct costs of Enterprise revenues.

Direct costs increased by \$3.8 million, or 6.1%, to \$65.8 million for the six months ended June 30, 2024, as compared to \$62.0 million for the same period in 2023. The increase was primarily driven by \$0.7 million of direct costs related to the growth of Enterprise revenues, an increase in payment processing fees of \$2.6 million and an increase in underwriting expenses of \$1.1 million, driven by growth in Total Originations and Total Customers, which was partially offset by a \$0.5 million decrease in costs related to our RoarMoney Banking and MoneyLion Investing offering.

Professional services

Professional services increased by \$6.5 million, or 143.6%, to \$11.0 million for the three months ended June 30, 2024, as compared to \$4.5 million for the same period in 2023. This increase resulted primarily from an increase in outside legal expenses of \$4.8 million, an increase in accounting and auditing fees of \$0.5 million driven by regulatory compliance requirements, an increase in outside consulting expenses of \$0.4 million and an increase in recruiting fees of \$0.3 million.

Professional services increased by \$7.3 million, or 76.2%, to \$16.8 million for the six months ended June 30, 2024, as compared to \$9.5 million for the same period in 2023. This increase resulted primarily from an increase in outside legal expenses of \$5.5 million, an increase in accounting and auditing fees of \$0.4 million driven by regulatory compliance requirements, an increase in outside consulting expenses of \$0.7 million and an increase in recruiting fees of \$0.6 million.

Technology-related costs

Technology-related costs increased by \$0.9 million, or 16.1%, to \$6.5 million for the three months ended June 30, 2024, as compared to \$5.6 million for the same period in 2023. This increase resulted primarily from an increase in expenses for software licenses and subscriptions of \$0.6 million and depreciation and amortization related to equipment and software of \$0.3 million.

Technology-related costs increased by \$1.4 million, or 12.4%, to \$13.1 million for the six months ended June 30, 2024, as compared to \$11.6 million for the same period in 2023. This increase resulted primarily from an increase in depreciation and amortization related to equipment and software of \$0.4 million and an increase in expenses for software licenses and subscriptions of \$1.0 million.

Other operating expenses

Other operating expenses decreased by \$6.9 million, or 61.3%, to \$4.3 million for the three months ended June 30, 2024, as compared to \$11.2 million for the same period in 2023. The decrease was primarily driven by lower costs related to legal matters of \$5.4 million and a decrease in the provision for bad debts and increased recoveries of receivables in our Enterprise business of \$1.1 million.

Other operating expenses decreased by \$5.6 million, or 27.5%, to \$14.7 million for the six months ended June 30, 2024, as compared to \$20.2 million for the same period in 2023. The decrease was primarily driven by lower costs related to legal matters of \$5.2 million, a decrease in the provision for bad debts and increased recoveries of receivables in our Enterprise business of \$2.3 million and a decrease in insurance expenses of \$1.1 million, which was partially offset by a \$3.6 million increase in expenses related to processing transactions in our Consumer business.

Our other (expense) income consists of the following:

Interest expense

Interest expense decreased by \$0.6 million, or 8.4%, to \$6.7 million for the three months ended June 30, 2024, as compared to \$7.3 million for the same period in 2023. This decrease was primarily driven by a decrease in interest expense on secured debt of \$0.9 million due to a decrease in average outstanding principal of secured debt and a decrease in the variable interest rate, which was partially offset by an increase of \$0.3 million in interest expense on other debt due to an increase in the average outstanding principal of other debt. See Part I, Item 1 "Financial Statements — Debt" for more information.

Interest expense decreased by \$1.3 million, or 8.8%, to \$13.5 million for the six months ended June 30, 2024, as compared to \$14.8 million for the same period in 2023. This decrease was primarily driven by a decrease in interest expense on secured debt of \$1.6 million due to a decrease in average outstanding principal of secured debt and a decrease in the variable interest rate, which was partially offset by an increase of \$0.4 million in interest expense on other debt due to an increase in the average outstanding principal of other debt. See Part I, Item 1 "Financial Statements — Debt" for more information.

Change in fair value of warrant liability

Change in fair value of warrant liability was an expense of \$0.1 million for the three months ended June 30, 2024, as compared to a benefit of \$0.2 million for the same period in 2023. The change in fair value of warrant liability was due to changes in inputs that drive the warrant liability fair value calculations.

There was no significant change in fair value of warrant liability for the six months ended June 30, 2024, as compared to the same period in 2023. There were minimal changes in inputs that drove the warrant liability fair value calculations from the beginning to the end of both periods.

Change in fair value of contingent consideration from mergers and acquisitions

There was no change in fair value of contingent consideration from mergers and acquisitions for the three or six months ended June 30, 2024 since there was no unsettled contingent consideration outstanding during the three or six months ended June 30, 2024.

Goodwill impairment loss

There was no goodwill impairment loss for the three or six months ended June 30, 2024 since all goodwill had been written off by the end of the fiscal year ended December 31, 2023.

Other income

Other income increased by \$1.1 million to \$2.4 million for the three months ended June 30, 2024, as compared to other income of \$1.3 million for the same period in 2023. The increase was primarily driven by lower losses on debt extinguishments of \$0.5 million, an increase in interest income earned on interest bearing deposits of \$0.4 million and an increase in rental income of \$0.2 million.

Other income increased by \$1.8 million to \$4.7 million for the six months ended June 30, 2024, as compared to other income of \$2.9 million for the same period in 2023. The increase was primarily driven by an increase in interest income earned on interest bearing deposits of \$1.0 million, lower losses on debt extinguishments of \$0.5 million and an increase in rental income of \$0.3 million.

Income tax benefit

See Part I, Item 1 "Financial Statements --- Income Taxes" for an explanation of the tax activity recorded during the six months ended June 30, 2024.

Non-GAAP Measures

In addition to net income (loss), which is a measure presented in accordance with U.S. GAAP, management believes that Adjusted EBITDA provides relevant and useful information which is widely used by analysts, investors and competitors in our industry in assessing performance. Adjusted EBITDA is a supplemental measure of our performance that is neither required by nor presented in accordance with U.S. GAAP. Adjusted EBITDA should not be considered as a substitute for U.S. GAAP metrics such as net income (loss) or any other performance measures derived in accordance with U.S. GAAP and may not be comparable to similar measures used by other companies.

We define Adjusted EBITDA as net income (loss) plus interest expense related to corporate debt, income tax expense (benefit), depreciation and amortization expense, change in fair value of warrant liability, change in fair value of contingent consideration from mergers and acquisitions, goodwill impairment loss, stock-based compensation expense and certain other expenses that management does not consider in measuring performance. We believe that Adjusted EBITDA provides a meaningful understanding of an aspect of profitability based on our current product portfolio. In addition, Adjusted EBITDA is useful to an investor in evaluating our performance because it:

- is a measure widely used by investors, analysts and competitors to measure a company's operating performance;
- is a metric used by rating agencies, lenders and other parties to evaluate our credit worthiness; and
- is used by our management for various purposes, including as a measure of performance and as a basis for strategic planning and forecasting.

The reconciliation of net income (loss) to Adjusted EBITDA for the three and six months ended June 30, 2024 and 2023 is as follows:

	Three Months Ended June 30,					Six Months E	1ded June 30,		
		2024		2023		2024		2023	
	(unauc	dited)			(una	udited)			
Net income (loss)	\$	3,137	\$	(27,723)	\$	10,212	\$	(36,940)	
Add back:									
Interest related to corporate debt ⁽¹⁾		2,576		3,475		5,371		7,035	
Income tax benefit		(1,824)		(262)		(2,213)		(286)	
Depreciation and amortization expense		6,331		6,113		12,543		12,297	
Changes in fair value of warrant liability		81		(162)		-		(13)	
Change in fair value of contingent consideration from mergers and acquisitions		-		(6,367)		-		(6,613)	
Goodwill impairment loss		-		26,721		-		26,721	
Stock-based compensation expense		7,531		5,250		14,028		10,955	
Other expenses ⁽²⁾		686		2,188		2,062		3,373	
Adjusted EBITDA	\$	18,518	\$	9,233	\$	42,003	\$	16,529	

(1) We add back the interest expense related to all outstanding corporate debt, excluding outstanding principal balances related to the ROAR 1 SPV Credit Facility and the ROAR 2 SPV Credit Facility. For U.S. GAAP reporting purposes, interest expense related to corporate debt is included within interest expense in the consolidated statement of operations.

(2) We add back other expenses, including those related to transactions, including mergers and acquisitions and financings, that occurred, litigation-related expenses and certain costs or gains that management does not consider in measuring performance. Generally, these expenses are included within other expenses or professional fees in the consolidated statement of operations.

Changes in Financial Condition to June 30, 2024 from December 31, 2023

	June 30,		Dee	cember 31,	Change			
		2024		2023	\$		%	
Assets								
Cash and restricted cash	\$	103,110	\$	94,479	\$	8,631	9.1%	
Consumer receivables		237,358		208,167		29,191	14.0%	
Allowance for credit losses on consumer receivables		(40,523)		(35,329)		(5,194)	14.7%	
Consumer receivables, net		196,835		172,838		23,997	13.9%	
Enterprise receivables, net		23,045		15,978		7,067	44.2%	
Property and equipment, net		1,961		1,864		97	5.2 %	
Intangible assets, net		168,302		176,541		(8,239)	-4.7%	
Other assets		52,490		53,559		(1,069)	-2.0%	
Total assets	\$	545,743	\$	515,259	\$	30,484	5.9%	
Liabilities and Stockholders' Equity								
Liabilities:								
Debt agreements	\$	194,414	\$	189,753	\$	4,661	2.5 %	
Accounts payable and accrued liabilities		48,984		52,396		(3,412)	-6.5%	
Warrant liability		810		810		-	0.0%	
Other liabilities		23,079		15,077		8,002	53.1%	
Total liabilities		267,287		258,036		9,251	3.6%	
Stockholders' equity:								
Common Stock		1		1		-	0.0%	
Additional paid-in capital		980,662		969,641		11,021	1.1 %	
Accumulated deficit		(692,507)		(702,719)		10,212	-1.5 %	
Treasury stock		(9,700)		(9,700)		-	0.0%	
Total stockholders' equity		278,456		257,223		21,233	8.3 %	
Total liabilities and stockholders' equity	\$	545,743	\$	515,259	\$	30,484	5.9%	

Assets

Cash and restricted cash

Cash and restricted cash increased by \$8.6 million, or 9.1%, to \$103.1 million as of June 30, 2024, as compared to \$94.5 million as of December 31, 2023. Refer to the "— Cash Flows" section below for further discussion on the net change in cash and restricted cash from operating activities, investing activities and financing activities during the period.

Consumer receivables, net

Consumer receivables, net increased by \$24.0 million, or 13.9%, to \$196.8 million as of June 30, 2024, as compared to \$172.8 million as of December 31, 2023. The increase was primarily attributable to an increase in Instacash receivables, net of allowance for credit losses, of \$14.2 million and an increase in loan receivables, net of allowance for credit losses, of \$9.2 million. Refer to Part I, Item 1 "Financial Statements — Consumer Receivables" for additional information.

Enterprise receivables, net

Enterprise receivables, net increased by \$7.1 million, or 44.2%, to \$23.0 million as of June 30, 2024, as compared to \$16.0 million as of December 31, 2023. This increase was primarily attributable to growth in Enterprise Marketplace receivables of \$5.5 million and an increase in Media Services receivables of \$1.3 million.

Intangible assets, net

Intangible assets, net decreased by \$8.2 million, or 4.7%, to \$168.3 million as of June 30, 2024, as compared to \$176.5 million as of December 31, 2023. This decrease was primarily attributable to the amortization of intangible assets of \$12.1 million, which was partially offset by an increase in capitalized software of \$3.9 million.

Other assets

Other assets decreased by \$1.1 million, or 2.0%, to \$52.5 million as of June 30, 2024, as compared to \$53.6 million as of December 31, 2023. This decrease was primarily attributable to a decrease in the receivable from payment processors and a decrease in prepaid expenses, which was partially offset by an increase in operating lease right-of-use assets due to a lease of the Company's new corporate headquarters entered into during the six months ended June 30, 2024. Refer to Part I, Item 1 "Financial Statements — Other Assets" for additional information.

Liabilities

Debt agreements

Debt agreements increased by \$4.7 million, or 2.5%, to \$194.4 million as of June 30, 2024, as compared to \$189.8 million as of December 31, 2023. Refer to Part I, Item 1 "Financial Statements — Debt" for further discussion of financing transactions.

Accounts payable and accrued expenses

Accounts payable and accrued expenses decreased by \$3.4 million, or 6.5%, to \$49.0 million as of June 30, 2024, as compared to \$52.4 million as of December 31, 2023. The decrease was primarily attributable to a reduction in litigation accruals, annual bonuses and state taxes paid during the six months ending June 30, 2024, which was partially offset by increases in accounts payable for professional fees and payables owed to Channel Partners.

Warrant liability

Warrant liability activity between June 30, 2024 and December 31, 2023 was not significant. Refer to the "— Results of Operations for the Three and Six Months Ended June 30, 2024 and 2023" section above for further discussion on the change in fair value of warrant liability.

Other liabilities

Other liabilities increased by \$8.0 million, or 53.1%, to \$23.1 million as of June 30, 2024, as compared to \$15.1 million as of December 31, 2023. The increase was primarily driven by an increase in operating lease liabilities due to a lease of the Company's new corporate headquarters entered into during the six months ended June 30, 2024.

Liquidity and Capital Resources

We believe our existing cash and cash equivalents and cash flows from operating activities will be sufficient to meet our operating working capital needs for at least the next twelve months. Our future financing requirements will depend on several factors, including our growth, the timing and level of spending to support continued development of our platform, the expansion of marketing activities and merger and acquisition activity. In addition, growth of our finance receivables increases our liquidity needs, and any failure to meet those liquidity needs could adversely affect our business. Additional funds may not be available on terms favorable to us or at all. If the Company is unable to generate positive operating cash flows, additional debt and equity financings or refinancing of existing debt financings may be necessary to sustain future operations.

Receivables originated on our platform, including Credit Builder Loans and Instacash advances, were primarily financed through special purpose vehicle financings from thirdparty institutional lenders. As of June 30, 2024, there was an outstanding principal balance of \$66.5 million under the ROAR 1 SPV Credit Facility and an outstanding principal balance of \$64.5 million under the ROAR 2 SPV Credit Facility. For more information, see Note 7, "Debt" and Note 2, "Summary of Significant Accounting Policies" of the Company's Notes to Consolidated Financial Statements included in the Company's Annual Report on Form 10-K for the year ended December 31, 2023 for discussion of the ROAR 1 SPV Credit Facility and the ROAR 2 SPV Credit Facility and VIE considerations related to the ROAR 1 SPV Credit Facility and the ROAR 2 SPV Credit Facility, respectively.

In the future, the majority of our receivables for our Instacash product will be financed through a forward flow financing arrangement pursuant to which we will sell a portion of our eligible Instacash receivables at a discount to third-party purchasers and receive a stable stream of servicing fee income based on net collections. For more information, see Part I, Item 1 "Financial Statements – Sale of Consumer Receivables."

The following table presents the Company's cash, restricted cash and receivable from payment processor as of June 30, 2024 and December 31, 2023:

	J	June 30, 2024	D	December 31, 2023
Cash	\$	98,361	\$	92,195
Restricted cash		4,749		2,284
Receivable from payment processor	\$	29,502	\$	37,362

Cash Flows

The following table presents net change in cash and restricted cash from operating, investing and financing activities during the three and six months ended June 30, 2024 and 2023:

	1	hree Months	Ended	June 30,	Six Months E	nded J	une 30,
		2024		2023	 2024		2023
Net cash provided by operating activities	\$	47,798	\$	35,363	\$ 81,437	\$	38,043
Net cash used in investing activities		(43,920)		(35,589)	(73,799)		(56,623)
Net cash (used in) provided by financing activities		(2,670)		(8,356)	993		(32,955)
Net change in cash and restricted cash	\$	1,208	\$	(8,582)	\$ 8,631	\$	(51,535)

Operating Activities

Net cash provided by operating activities was \$47.8 million for the three months ended June 30, 2024 compared to net cash provided by operating activities of \$35.4 million for the three months ended June 30, 2023. This increase in net cash provided by operating activities was primarily driven by an increase in profitability, after adjusting for non-cash activity included in our net income (loss), of approximately \$21.5 million, which was partially offset by a decrease of \$9.1 million related to changes in working capital.

Net cash provided by operating activities was \$81.4 million for the six months ended June 30, 2024 compared to net cash provided by operating activities of \$38.0 million for the six months ended June 30, 2023. This increase in net cash provided by operating activities was primarily driven by an increase in profitability, after adjusting for non-cash activity included in our net income (loss), of approximately \$42.3 million and an increase of \$1.1 million related to changes in working capital.

Investing Activities

Net cash used in investing activities was \$43.9 million for the three months ended June 30, 2024 compared to net cash used in investing activities of \$35.6 million for the three months ended June 30, 2023. The increase in net cash used in investing activities was primarily related to increases in net finance receivable originations during the three months ended June 30, 2024 compared to the three months ended June 30, 2023.

Net cash used in investing activities was \$73.8 million for the six months ended June 30, 2024 compared to net cash used in investing activities of \$56.6 million for the six months ended June 30, 2023. The increase in net cash used in investing activities was primarily related to increases in net finance receivable originations during the six months ended June 30, 2024 compared to the six months ended June 30, 2023.

Financing Activities

Net cash used in financing activities was \$2.7 million for the three months ended June 30, 2024 compared to net cash used in financing activities of \$8.4 million for the three months ended June 30, 2023. The decrease in net cash used for financing activities was primarily attributable to a decrease in payments of debt principal of \$5.0 million and a decrease in preferred stock settlement payments of \$3.0 million, which was partially offset by an increase of \$2.5 million in cash used for tax payments owed on the vesting of stock compensation during the three months ended June 30, 2024 compared to the three months ended June 30, 2023.

Net cash provided by financing activities was \$1.0 million for the six months ended June 30, 2024 compared to net cash used in financing activities of \$33.0 million for the six months ended June 30, 2023. The increase in net cash provided by financing activities was primarily attributable to a decrease in payments of debt principal of \$29.0 million, an increase in proceeds from debt of \$4.0 million and a decrease in preferred stock settlement payments of \$3.0 million, which was partially offset by an increase of \$2.2 million in cash used for tax payments owed on the vesting of stock compensation during the six months ended June 30, 2024 compared to the six months ended June 30, 2023.

Financing Arrangements

Refer to Part I, Item 1 "Financial Statements -- Debt" for further discussion on financing transactions during the period.

Contractual Obligations

The table below summarizes debt, lease and other long-term minimum cash obligations outstanding as of June 30, 2024:

	Total	Rei	nainder of 2024	20	25 - 2026	202	7 – 2028	Th	ereafter
Monroe Term Loans	\$ 65,000	\$	_	\$	65,000	\$	_	\$	_
ROAR 1 SPV Credit Facility	66,500				66,500		—		—
ROAR 2 SPV Credit Facility	64,500				64,500		—		—
Operating lease obligations	20,031		2,295		8,192		6,605		2,939
Vendor unconditional purchase obligations	23,296				14,796		8,500		—
Total	\$ 239,327	\$	2,295	\$	218,988	\$	15,105	\$	2,939

Secured Loans and Other Debt

For more information regarding our secured loans and other debt, see Part I, Item 1 "Financial Statements - Debt" in this Quarterly Report on Form 10-Q.

Off-Balance Sheet Arrangements

At June 30, 2024, the Company did not have any material off-balance sheet arrangements.



Critical Accounting Policies and Estimates

See Part I, Item 1 "Financial Statements - Summary of Significant Accounting Policies" for a description of critical accounting policies and estimates.

Recently Issued and Adopted Accounting Pronouncements

See Part I, Item 1 "Financial Statements — Summary of Significant Accounting Policies" for a description of recently issued accounting pronouncements that may potentially impact our results of operations, financial condition or cash flows.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

We are exposed to market risks in the ordinary course of our business. Market risk represents the risk of loss that may impact our financial position due to adverse changes in financial market prices and rates. Our market risk exposure is primarily the result of fluctuations in interest rates.

Interest Rates Risk

Interest rates may adversely impact our customers' level of engagement on our platform and ability and willingness to pay outstanding amounts owed to us. While we do not charge interest on a lot of our products, higher interest rates could deter customers from utilizing our credit products and other loans. Moreover, higher interest rates may lead to increased delinquencies, charge-offs and allowances for loans and interest receivable, which could have an adverse effect on our operating results.

The Monroe Term Loans and future funding arrangements may bear a variable interest rate. The ROAR 1 SPV Credit Facility and ROAR 2 SPV Credit Facility have fixed interest rates. Given the fixed interest rates charged on many of our loans, a rising variable interest rate would reduce our interest margin earned in these funding arrangements. Dramatic increases in interest rates may make these forms of funding nonviable. A one percent change in the interest rate on our variable interest rate debt, based on principal balances as of June 30, 2024, would result in an approximately \$0.7 million impact to annual interest expense.

Item 4. Controls and Procedures

Evaluation of Disclosure Controls and Procedures

Disclosure controls are procedures that are designed with the objective of ensuring that information required to be disclosed in our reports filed under the Exchange Act, such as this Quarterly Report on Form 10-Q, is recorded, processed, summarized and reported within the time period specified in the SEC's rules and forms. Disclosure controls are also designed with the objective of ensuring that such information is accumulated and communicated to our management, including the Chief Executive Officer (Principal Executive Officer) and Chief Financial Officer (Principal Financial Officer), as appropriate to allow timely decisions regarding required disclosure. Our management evaluated, with the participation of our current Chief Executive Officer and Chief Financial Officer, the effectiveness of our disclosure controls and procedures as of June 30, 2024, pursuant to Rule 13a-15(b) under the Exchange Act. Based upon that evaluation, our Chief Executive Officer and Chief Financial Officer concluded that, as of June 30, 2024, our disclosure controls and procedures were effective in providing reasonable assurance that the information required for disclosure in reports filed or submitted under the Exchange Act are recorded, processed, summarized and reported within the time periods specified in SEC rules and forms, and that such information is accumulated and communicated to Company management, including our Chief Executive Officer and Chief Financial Officer, as appropriate, to allow timely decisions regarding required disclosures.

We do not expect that our disclosure controls and procedures will prevent all errors and all instances of fraud. Disclosure controls and procedures, no matter how well conceived and operated, can provide only reasonable, not absolute, assurance that the objectives of the disclosure controls and procedures are met. Further, the design of disclosure controls and procedures must reflect the fact that there are resource constraints, and the benefits must be considered relative to their costs. Because of the inherent limitations in all disclosure controls and procedures, no evaluation of disclosure controls and procedures can provide absolute assurance that we have detected all our control deficiencies and instances of fraud, if any. The design of disclosure controls and procedures also is based partly on certain assumptions about the likelihood of future events, and there can be no assurance that any design will succeed in achieving its stated goals under all potential future conditions.

Changes in Internal Control over Financial Reporting

There has been no change in our internal control over financial reporting during the quarter ended June 30, 2024 that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting.

Part II - OTHER INFORMATION

Item 1. Legal Proceedings

From time to time, we are subject to various claims and legal proceedings in the ordinary course of business, including lawsuits, arbitrations, class actions and other litigation. We are also the subject of various actions, inquiries, investigations and proceedings by regulatory and other governmental agencies. The outcome of any such legal and regulatory matters, including those discussed in this section, is inherently uncertain, and some of these matters may result in adverse judgments or awards, including penalties, injunctions or other relief, which could materially and adversely impact our business, financial condition, operating results and cash flows. See Part I, Item 1A "Risk Factors — Risks Relating to Legal and Accounting Matters — Unfavorable outcomes in legal proceedings may harm our business, financial condition, results of operations and cash flows" in the Company's Annual Report on Form 10-K for the year ended December 31, 2023.

We have determined, based on our current knowledge, that the aggregate amount or range of losses that are estimable with respect to our legal proceedings, including the matters described below, would not have a material adverse effect on our business, financial position, results of operations or cash flows. As of June 30, 2024, amounts accrued were not material. Notwithstanding the foregoing, the ultimate outcome of legal proceedings involves judgments, estimates and inherent uncertainties, and cannot be predicted with certainty. It is possible that an adverse outcome of any matter could be material to our business, financial position, results of operations or cash flows as a whole for any particular reporting period of occurrence. In addition, it is possible that a matter may prompt litigation or additional investigations or proceedings by other government agencies or private litigants.

State Regulatory Examinations and Investigations

We hold a number of state licenses in connection with our business activities, and must also comply with other applicable compliance and regulatory requirements in the states where we operate. In most states where we operate, one or more regulatory agencies have authority with respect to regulation and enforcement of our business activities under applicable state laws, and we may also be subject to the supervisory and examination authority of such state regulatory agencies. Examinations by state regulators have and may continue to result in findings or recommendations that require us, among other potential consequences, to provide refunds to customers or to modify our internal controls and/or business practices.

In the ordinary course of our business, we are and have been from time to time subject to, and may in the future be subject to, governmental and regulatory examinations, information requests, investigations and proceedings (both formal and informal) in connection with various aspects of our activities by state agencies, certain of which could result in adverse judgments, settlements, fines, penalties, restitution, disgorgement, injunctions or other relief. We have responded to and cooperated with the relevant state agencies and will continue to do so in the future, as appropriate.

CFPB Litigation

On September 29, 2022, the Consumer Financial Protection Bureau (the "CFPB") initiated a civil action in the United States District Court for the Southern District of New York ("SDNY") against MoneyLion Technologies Inc., ML Plus LLC and our 38 state lending subsidiaries, alleging violations of the Military Lending Act and the Consumer Financial Protection Act. The CFPB is seeking injunctive relief, redress for allegedly affected consumers and civil monetary penalties. On January 10, 2023, we moved to dismiss the lawsuit, asserting various constitutional and merits-based arguments. On June 13, 2023, the CFPB filed its first amended complaint, alleging substantially similar claims as those asserted in its initial complaint. On July 11, 2023, we moved to dismiss the lawsuit, again asserting various constitutional and merit-based arguments. On October 9, 2023, we moved for a stay of the action pending a decision from the United States Supreme Court in CFPB v. Community Financial Services Association of America, Ltd., No. 22-448 (U.S. argued Oct. 3, 2023) ("*CFSA*"). On December 1, 2023, the Court issued an order granting our motion and staying the action pending the United State Supreme Court's decision in *CFSA*. On May 16, 2024, the Supreme Court decided *CFSA*. Accordingly, our motion to dismiss is now pending with the SDNY. We continue to maintain that the CFPB's claims are meritless and are vigorously defending against the lawsuit. Nevertheless, at this time, we cannot predict or determine the timing or final outcome of this matter or the effect that any adverse determinations in the lawsuit may have on our business, financial condition, results of operations or cash flows.

MALKA Seller Members Litigation

On July 21, 2023, Jeffrey Frommer, Lyusen Krubich, Daniel Fried and Pat Capra, the former equity owners of MALKA (collectively, the "Seller Members"), brought a civil action in the SDNY against MoneyLion Technologies Inc. alleging, among other things, breaches of the Membership Interest Purchase Agreement (the "MIPA") governing our acquisition of MALKA. Among other claims, the Seller Members allege that they are entitled to payment of \$25.0 million of Class A Common Stock pursuant to the earnout provisions set forth in the MIPA, based on the Seller Members' assertion that MALKA achieved certain financial targets for the year ended December 31, 2022 (such payment, the "2022 Earnout Payment"). We believe that the Seller Members are not entitled to any portion of the 2022 Earnout Payment under the terms of the MIPA and filed counterclaims against the Seller Members, alleging, among other things, fraud, negligent misrepresentation, conversion, breach of fiduciary duties and breach of contract and seeking compensatory damages and other remedies as a result of wrongdoing by the Seller Members. On October 17, 2023, the SDNY denied, in full, the Seller Members' motion for a preliminary injunction to remove the restrictive legends on certain shares of Class A Common Stock previously issued to the Seller Members. Separately, on November 3, 2023, the Seller Members moved to dismiss our amended counterclaims and third-party complaint. On May 14, 2024, the SDNY denied the Seller Members' motion to dismiss with respect to our counterclaims alleging fraud, negligent misrepresentation, breach of fuduciary duty and certain conversion and breach of contract claims. The SDNY dismissed certain of our counterclaims relating to declaratory judgment, unjust enrichment and conversion as duplicative of our fraud and misrepresentation counterclaims, as well as certain other breach of contract counterclaims. We continue to vigorously pursue our remaining counterclaims and defend against the Seller Members' claims, which we believe are me

Former Series A Preferred Stockholders Litigation

On July 27, 2023, MassMutual Ventures US II LLC, Canaan X L.P., Canaan XI L.P., F-Prime Capital Partners Tech Fund LP and GreatPoint Ventures Innovation Fund II, L.P., each of which are former holders of the Company's Series A Preferred Stock (collectively, the "Former Preferred Stockholders"), brought a civil action in the SDNY against MoneyLion Inc., our Board of Directors and certain officers asserting claims under Section 14(a) relating to the Definitive Proxy Statement we filed with the SEC on March 31, 2023 in connection with the Special Meeting of Stockholders relating to the 1-for-30 Reverse Stock Split of the Class A Common Stock effected on April 24, 2023 and related state law claims. On November 6, 2023, we filed a motion to dismiss the lawsuit in its entirety.

On May 15, 2024, the SDNY granted our motion to dismiss the Former Preferred Stockholders' complaint in its entirety. The SDNY dismissed the Former Preferred Stockholders' claim under Section 14(a) of the Securities Exchange Act of 1934 for failure to state a claim and declined to exercise supplemental jurisdiction over the remaining state law claims, dismissing such state law claims without prejudice. On June 14, 2024, Canaan X L.P., Canaan XI L.P., GreatPoint Ventures Innovation Fund II, L.P. filed a notice of appeal with the United States Court of Appeals for the Second Circuit. We continue to believe that the Former Preferred Stockholders' claims are meritless and intend to vigorously defend against the appeal if any Former Preferred Stockholders decide to pursue their appeal. Nevertheless, at this time, we cannot predict or determine the timing or final outcome of this matter or the effect that any adverse determinations in the lawsuit may have on our business, financial condition, results of operations or cash flows.

Item 1A. Risk Factors

As of the date of this Quarterly Report on Form 10-Q, there have been no material changes to the risk factors disclosed in the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 2023, except as otherwise set forth below. We may disclose additional changes to such factors or disclose additional factors from time to time in our future filings with the SEC.

The legal and regulatory regimes governing certain of our products and services are uncertain and evolving. Changing or new laws, regulations, interpretations or regulatory enforcement priorities may have a material and adverse effect on our business, financial condition and results of operations.

We are required to comply with constantly changing federal, state and local laws, regulations and rules that regulate various aspects of the products and services that we offer. Federal and state regulators of consumer financial products and services are also enforcing existing laws, regulations and rules more aggressively and enhancing their supervisory expectations regarding the management of legal and regulatory compliance risks. Such laws, regulations and rules are complex and require us to incur significant expenses and devote significant management attention to ensure compliance. For example, the CFPB may adopt new model disclosures and regulations governing consumer financial services, including regulations defining unfair, deceptive or abusive acts or practices, and they could also issue advisory opinions or other similar soft tools to complement its rulemaking authority which could subject us to new or differing legal and regulatory requirements, which could materially and adversely affect our business. The CFPB's authority to change or interpret regulations adopted in the past by other regulators, or to rescind or alter past regulatory guidance, could increase our compliance costs and litigation exposure. If the CFPB or other similar regulatory bodies adopt, or customer advocacy groups are able to generate widespread support for, positions that are detrimental to our business, then our business, financial condition and results of operations could be harmed.

Changes in the laws, regulations and enforcement priorities applicable to our business, including reexamination of current enforcement practices, could have a material and adverse impact on our business, financial condition, results of operations and cash flows. We may not be able to respond quickly or effectively to regulatory, legislative and other developments. In particular, the regulatory landscape regarding earned wage access products (including our Instacash advance service) is uncertain and evolving given rapid growth in the use of earned wage access products in recent years. State and federal regulators, including the California Department of Financial Protection and Innovation, may in the future launch inquiries, reviews or similar investigations or issue new, or change or interpret, regulations or rules relating to earned wage access products, which could result in additional compliance requirements and other risks relating to our current and past business activities as described herein. For instance, on July 18, 2024, the CFPB issued a proposed interpretive rule, which, if finalized in its proposed form, would characterize certain earned wage access products as consumer loans subject to the Truth in Lending Act. It is possible that we could incur additional operational costs relating to our Instacash service and that other risks described herein could be heightened if the interpretive rule becomes final. State and federal regulators could also launch inquiries, reviews or similar investigations into our Instacash product.

⁴³

Proposals to change the statutes affecting financial services companies are frequently introduced in Congress and state legislatures that, if enacted, could affect our operating environment in substantial and unpredictable ways. We cannot determine with any degree of certainty whether any such legislative or regulatory proposals will be enacted and, if enacted, the ultimate impact that any such potential legislation or implemented regulations, or any such potential regulatory actions by federal or state regulators, would have upon our business or our operating environment. As a result, we could be forced, as we have in the past, to temporarily pause, limit or permanently cease to offer our earned wage access product in certain states depending on state regulatory regimes. In addition, numerous federal and state regulators have the authority to promulgate or change regulations that could have a similar effect on our third-party partners and service providers and restrict their business practices, such as the recent rulemaking under the TCPA by the Federal Communications Commission to require one-to-one consumer consents for telemarketing. These changes and uncertainties make our business planning more difficult and could result in changes to our business model, impair our ability to offer our existing or planned features, products and services or increase our cost of doing business.

Our failure to comply (or to ensure that our third-party partners, service providers or other agents comply) with these changing laws, regulations or rules may result in litigation, enforcement actions and penalties, including revocation of licenses and registrations; fines and other monetary penalties; civil and criminal liability; substantially reduced payments by our customers; modification of the original terms of loans and other products, permanent forgiveness of debt or inability to collect on amounts owed by our borrowers; and indemnification claims. Such consequences could, among other things, require changes to our business practices and scope of operations or harm our reputation, which in turn, could have a material adverse effect on our business, financial condition and results of operations. If our practices are not consistent or viewed as not consistent with legal and regulatory requirements, we may become subject to audits, inquiries, whistleblower complaints, adverse media coverage, investigations, litigation or criminal or civil sanctions, all of which may have an adverse effect on our reputation, business, results of operations and financial condition.

New laws, regulations, rules, guidance and policies could require us to incur significant expenses to ensure compliance, adversely impact our profitability, limit our ability to continue existing or pursue new business activities, require us to change certain of our business practices or alter our relationships with customers, affect retention of key personnel or expose us to additional costs (including increased compliance costs and/or customer remediation). These changes also may require us to invest significant resources or devote significant management attention in order to make any necessary changes. For example, the regulatory frameworks for an open banking paradigm and AI and machine learning technology are evolving and remain uncertain. It is possible that new laws and regulations will be adopted in the U.S., or existing laws and regulations may be interpreted in new ways, that would affect the operation of our platform and the way in which we use consumer data, AI and machine learning technology, including with respect to fair lending laws.

We rely on a variety of funding sources to support our business model. If our existing funding arrangements are not renewed or replaced or our existing funding sources are unwilling or unable to provide funding to us on terms acceptable to us, or at all, it could have a material adverse effect on our business, financial condition, results of operations and cash flows.

To support the origination of loans, cash advances and other receivables on our platform and the growth of our business, we must maintain a variety of funding arrangements. We cannot guarantee that we will be able to extend or replace our existing funding arrangements at maturity on reasonable terms or at all. For example, disruptions in the credit markets or other factors, such as the high inflation and interest rate environment in 2023 and to date in 2024, could adversely affect the availability, diversity, cost and terms of our funding arrangements. In addition, our funding sources may reassess their exposure to our industry or our business, including as a result of any significant underperformance of the consumer receivables facilitated through our platform or regulatory developments, in particular regarding earned wage access products, that impose significant requirements on, or increase potential risks and liabilities related to, the consumer receivables facilitated through our platform arrangements are not renewed or replaced or our existing funding sources are unwilling or unable to provide funding on terms acceptable to us, or at all, we would need to secure additional sources of funding or reduce our operations significantly, which could have a material adverse effect on our business, financial condition, results of operations and cash flows.

Further, as the volume of consumer receivables facilitated through our platform increases and in order to support future business growth, we may require the expansion of our funding capacity under our existing funding arrangements or the addition of new sources of capital. We may also change our funding strategy over time depending on the attractiveness and availability of alternative funding structures. For example, we entered into a new forward flow financing arrangement on June 30, 2024 to finance the origination of Instacash advances, as described further under Part I, Item 1 "Financial Statements – Sale of Consumer Receivables." The availability and diversity of new funding arrangements depends on various factors and are subject to numerous risks, many of which are outside of our control. In the event of a sudden or unexpected shortage of funds in the financial system, we may not be able to maintain necessary levels of funding without incurring high funding costs or a reduction in the term or size of funding instruments. In such a case, if we are unable to arrange new or alternative methods of financing on favorable terms, we would have to reduce our transaction volume or otherwise inhibit our business growth, which could have a material adverse effect on our business, financial condition, results of operations and cash flows.

The agreements governing our funding arrangements require us to comply with certain covenants. A breach of such covenants or other events of default under our funding agreements could result in the reduction or termination of our access to such funding, could increase our cost of such funding or, in some cases, could give our lenders the right to require repayment of the loans prior to their scheduled maturity. Certain of these covenants and restrictions limit our and our subsidiaries' ability to, among other things: incur additional indebtedness; create liens on certain assets; pay dividends on or make distributions in respect of their capital stock or make other restricted payments; consolidate, merge, sell, or otherwise dispose of all or substantially all of their assets; purchase or otherwise acquire assets or equity interests; modify organizational documents; enter into certain transactions with their affiliates; enter into restrictive agreements; engage in other business activities; and make investments. The Monroe Credit Agreement also contains certain financial covenants with respect to minimum adjusted revenue, EBITDA, liquidity and unrestricted cash (each as defined therein).

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

None.

Item 3. Default Upon Senior Securities

None.

Item 4. Mine Safety Disclosures

Not applicable.

Item 5. Other Information

Rule 10b5-1 Trading Arrangements

During the three months ended June 20, 2024, the officer set forth below adopted a "Rule 10b5-1 trading arrangement" (as defined in Item 408 of Regulation S-K) that is intended to satisfy the affirmative defense of Rule 10b5-1(c) under the Exchange Act. The 10b5-1 trading arrangement was entered into in writing during an open trading window under our Insider Trading Policy, has a trading period of at least one year, and is subject to a mandatory cooling off period as required by Rule 10b5-1. Shares in the Rule 10b5-1 trading arrangement underlying restricted stock units ("RSUs") and performance share units ("PSUs") may only be sold following satisfaction of the applicable time-based and performance-based vesting requirements. In addition, due to limit price requirements and timing conditions in certain of the Rule 10b5-1 trading arrangement prior to its expiration date, as indicated below.

On May 24, 2024, Adam VanWagner, our Chief Legal Officer, adopted a Rule 10b5-1 trading arrangement for the sale of up to:

(i) 5,702 shares of Class A Common Stock; and

(ii) the net shares (not yet determinable) after shares are withheld to satisfy tax obligations upon the vesting of 27,853 RSUs and, subject to the achievement of the applicable performance goals within a range of 80% to 120% of the awarded PSUs with respect to PSUs which have not yet been earned, 21,954 PSUs.

This Rule 10b5-1 trading arrangement is scheduled to expire on September 30, 2025.

Item 6. Exhibits

The agreements and other documents filed as exhibits to this report are not intended to provide factual information or other disclosure other than with respect to the terms of the agreements or other documents themselves, and you should not rely on them for that purpose. In particular, the representations, warranties, covenants and agreements and may be subject to limitations agreed upon by the contracting parties. The representations and warranties may have been made for the purposes of allocating contractual risk between the parties to such agreements instead of establishing these matters as facts and may be subject to standards of materiality applicable to the contracting parties that differ from those applicable to investors. Unless otherwise explicitly stated therein, investors and agreements or any descriptions thereof as characterizations of the actual state of facts or condition of the Company or any of its affiliates or businesses. Moreover, the assertions embodied in the representations and warranties may change after the respective dates of such agreements, which subsequent information may or may not be fully reflected in the Company's public disclosures.

Exhibit No.	Description
3.1	Fourth Amended and Restated Certificate of Incorporation of MoneyLion Inc. (incorporated by reference to Exhibit 3.1 to MoneyLion Inc.'s Registration Statement on Form S-1 (File 333-260254), filed with the SEC on October 14, 2021).
3.1.1	Certificate of Amendment to the MoneyLion Inc. Fourth Amended and Restated Certificate of Incorporation (incorporated by reference to Exhibit 3.1 to MoneyLion Inc.'s Current Report on Form 8-K (File 001-39346), filed with the SEC on April 24, 2023).
3.2	Amended and Restated Bylaws of MoneyLion Inc., effective as of March 15, 2023 (incorporated by reference to Exhibit 3.2 of MoneyLion Inc.'s Annual Report on Form 10-K (File No. 001-39346), filed with the SEC on March 16, 2023).
10.1*†	Amendment No. 3 to Credit Agreement, dated as of June 30, 2024, by and among MoneyLion Technologies Inc., as borrower, the various financial institutions party thereto, as lenders, and Monroe Capital Management Advisors, LLC, as administrative agent for the lenders.
10.2*†	Master Receivables Purchase Agreement, dated as of June 30, 2024, by and among ML Plus LLC, as seller, Sound Point Capital Management LP, as purchaser agent, and SP Main Street Funding I LLC and each additional purchaser from time to time party thereto, as purchasers.
10.3*†	Servicing Agreement, dated as of June 30, 2024, by and among MoneyLion Technologies Inc., as servicer, Sound Point Capital Management LP, as purchaser agent, and SP Main Street Funding I LLC and each additional purchaser from time to time party thereto, as purchasers.
31.1*	Certification of the Chief Executive Officer pursuant to Exchange Act Rules Rule 13a-14(a), as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
31.2*	Certification of the Chief Financial Officer pursuant to Exchange Act Rules Rule 13a-14(a), as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
32.1**	Certification of the Chief Executive Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
32.2**	Certification of the Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
101.INS	Inline XBRL Instance Document.
101.SCH	Inline XBRL Taxonomy Extension Schema Document.
101.CAL	Inline XBRL Taxonomy Extension Calculation Linkbase Document.
101.DEF	Inline XBRL Taxonomy Extension Definition Linkbase Document.
101.LAB	Inline XBRL Taxonomy Extension Label Linkbase Document.
101.PRE	Inline XBRL Taxonomy Extension Presentation Linkbase Document.
104	Cover Page Interactive Data File (formatted as inline XBRL with applicable taxonomy extension information contained in Exhibits 101).

* Filed herewith.

** The certifications furnished in Exhibits 32.1 and 32.2 hereto are deemed to accompany this Quarterly Report on Form 10-Q and will not be deemed "filed" for purposes of Section 18 of the Exchange Act, except to the extent that the registrant specifically incorporates it by reference. † Certain schedules and exhibits to this exhibit have been omitted pursuant to Regulation S-K Item 601(a)(5), or certain portions of this exhibit have been redacted pursuant to

Regulation S-K Item 601(b)(10)(iv).

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Date: August 6, 2024

Date: August 6, 2024

MONEYLION INC.

- By: /s/ Richard Correia Richard Correia President, Chief Financial Officer and Treasurer (Principal Financial Officer)
- By: /s/ Mark Torossian Mark Torossian Chief Accounting Officer (Principal Accounting Officer)

AMENDMENT NO. 3 TO CREDIT AGREEMENT

AMENDMENT NO. 3 TO CREDIT AGREEMENT, dated as of June 30, 2024 (this "<u>Amendment</u>"), to the Credit Agreement, dated as of March 24, 2022 (as amended, restated, supplemented or otherwise modified from time to time, the "<u>Credit Agreement</u>"), among MONEYLION TECHNOLOGIES INC., a Delaware corporation (the "<u>Company</u>" or "<u>Borrower</u>"); the financial institutions that are or may from time to time become parties thereto (together with their respective successors and permitted assigns, the "<u>Lenders</u>"); and MONROE CAPITAL MANAGEMENT ADVISORS, LLC, a Delaware limited liability company ("<u>Monroe Capital</u>"), as administrative agent for the Lenders (the "<u>Agent</u>").

WHEREAS, the Borrower has requested that the Agent and the Lenders amend certain terms and conditions of the Credit Agreement; and

WHEREAS, the Agent and the Lenders are willing to amend such terms and conditions of the Credit Agreement on the terms and conditions

set forth herein.

NOW THEREFORE, in consideration of the premises and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto hereby agree as follows:

1. <u>Definitions</u>. All terms used herein that are defined in the Credit Agreement and not otherwise defined herein shall have the meanings assigned to them in the Credit Agreement.

2. <u>Amendments</u>. The Credit Agreement is hereby amended to (i) delete the red or green stricken text (indicated textually in the same manner as the following examples: stricken text and stricken text); and (ii) add the blue or green double-underlined text (indicated textually in the same manner as the following examples: double-underlined text and double-underlined text), in each case, as set forth in the marked copy of the Credit Agreement attached as Annex A hereto and made a part hereof for all purposes.

3. <u>Representations and Warranties</u>. The Borrower hereby represents and warrants to the Agent and the Lenders as follows:

(a) <u>Representations and Warranties; No Event of Default</u>. The representations and warranties herein, in Section 9 of the Credit Agreement and in each other Loan Document, certificate or other writing delivered by or on behalf of the Loan Parties to the Agent or any Lender pursuant to the Credit Agreement or any other Loan Document on or immediately prior to the Amendment No. 3 Effective Date are true and correct in all material respects (unless any such representation or warranty is by its terms qualified by concepts of materiality, in which case that representation or warranty is true and correct in all respects after giving effect to any such materiality qualifier) with the same effect as if then made (except to the extent stated to relate to a specific earlier date, in which case that representation or warranty is true and correct in all material respects or in all respects, as applicable, as of that earlier date); and no Default or Event of Default has occurred and is continuing as of the Amendment No. 3 Effective Date or would result from this Amendment becoming effective in accordance with its terms.

(b) <u>Authorization, Etc.</u> The Borrower is duly authorized to execute and deliver this Amendment.

(c) <u>Enforceability of Loan Documents</u>. This Amendment is the legal, valid, and binding obligation of the Borrower, enforceable against the Borrower in accordance with its terms, subject to bankruptcy, insolvency, and similar laws affecting the enforceability of creditors' rights generally and to general principles of equity.

4. <u>Conditions to Effectiveness</u>. This Amendment shall become effective only upon satisfaction in full, in a manner satisfactory to the Agent, of the following conditions precedent (the first date upon which all such conditions shall have been satisfied being hereinafter referred to as the "<u>Amendment No. 3</u><u>Effective Date</u>"):

(a) <u>Payment of Fees, Etc.</u> The Borrower shall have paid on or before the Amendment No. 3 Effective Date all fees, costs and expenses then payable, if any, pursuant to the Credit Agreement and any other Loan Document.

(b) <u>Representations and Warranties</u>. The representations and warranties contained in this Amendment and in Section 9 of the Credit Agreement and in each other Loan Document shall be true and correct in all material respects (unless any such representation or warranty is by its terms qualified by concepts of materiality, in which case that representation or warranty is true and correct in all respects after giving effect to any such materiality qualifier) with the same effect as if then made (except to the extent stated to relate to a specific earlier date, in which case that representation or warranty shall be true and correct in all material respects or in all respects, as applicable, as of that earlier date).

(c) <u>No Default; Event of Default</u>. No Default or Event of Default shall have occurred and be continuing on the Amendment No. 3 Effective Date or result from this Amendment becoming effective in accordance with its terms.

(d) <u>Delivery of Documents</u>. The Agent shall have received on or before the Amendment No. 3 Effective Date, the following, each in form and substance satisfactory to the Agent:

- (i) this Amendment, duly executed by the Borrower, the Agent and the Lenders;
- (ii) the SP Intercreditor Agreement; and

(iii) a certificate of an Authorized Officer of the Borrower certifying (A) the attached copies of the SP Forward Flow Documents as in effect on the Amendment No. 3 Effective Date are true, complete and correct copies thereof and such agreements remain in full force and effect and (B) as to the matters set forth in subsections (b) and (c) of this Section 4.

5. <u>Release of Liens on InstaCash Receivables</u>. Pursuant to Section 14.11 of the Credit Agreement, (x) the Lenders irrevocably authorize Administrative Agent to, inter alia, release any Lien granted to or held by Administrative Agent under any Collateral Document upon property sold or to be sold or disposed of as part of or in connection with any disposition permitted under the Credit Agreement and (y) upon request by Administrative Agent at any time, the Lenders are to confirm in writing Administrative Agent's authority to release. In furtherance of the foregoing, (x) the Lenders hereby confirm Administrative

Agent's authority to release Liens on InstaCash Receivables disposed of in accordance with Section 11.4(xiv) and (y) the Administrative Agent hereby releases any and all Liens on such InstaCash Receivables so disposed of upon the disposition of such InstaCash Receivables in accordance with Section 11.4; provided that, for the avoidance of doubt, in the event any Loan Party repurchases any InstaCash Receivables previously disposed of, such repurchased InstaCash Receivables shall be automatically subject to the Lien in favor of Administrative Agent pursuant to and as provided in the Collateral Documents upon the consummation of such repurchase.

6. Continued Effectiveness of the Credit Agreement and Other Loan Documents. The Borrower hereby (a) acknowledges and consents to this Amendment for itself and each other Loan Party, (b) confirms and agrees that the Credit Agreement and each other Loan Document to which it is a party is, and shall continue to be, in full force and effect and is hereby ratified and confirmed in all respects, except that on and after the Amendment No. 3 Effective Date, all references in any such Loan Document to "the Credit Agreement", the "Agreement", "thereto", "thereof", "thereunder" or words of like import referring to the Credit Agreement shall mean the Credit Agreement as amended by this Amendment, and (c) confirms and agrees that, to the extent that any such Loan Document purports to assign or pledge to the Agent, for the benefit of the Agent and the Lenders, or to grant to the Agent, for the benefit of the Agreement (as amended hereby) and the other Loan Documents, such pledge, assignment and/or grant of the security interest or Lien is hereby ratified and confirmed in all respects. This Amendment does not and shall not affect any of the obligations of the Loan Parties, other than as expressly provided herein, including, without limitation, the Loan Parties' obligations to repay the Loans in accordance with the terms of Credit Agreement or the obligations of the Loan Parties under any Loan Document to which they are a party, all of which obligations shall remain in full force and effect. Except as expressly provided herein, the execution, delivery and effectiveness of this Amendment shall not operate as a waiver of any right, power or remedy of the Agent or any Lender under the Credit Agreement or any other Loan Document.

7. <u>No Novation</u>. Nothing herein contained shall be construed as a substitution or novation of the Obligations outstanding under the Credit Agreement or instruments securing the same, which shall remain in full force and effect, except as modified hereby.

8. <u>No Representations by Agent or Lenders</u>. The Borrower for itself and each other Loan Party hereby acknowledges that it has not relied on any representation, written or oral, express or implied, by the Agent or any Lender, other than those expressly contained herein, in entering into this Amendment.

9. <u>Release</u>. The Borrower for itself and each other Loan Party hereby acknowledges and agrees that: (a) neither it nor any of its Subsidiaries has any claim or cause of action against the Agent or any Lender (or any of the directors, officers, employees, agents, attorneys or consultants of any of the foregoing), and (b) the Agent and the Lenders have heretofore properly performed and satisfied in a timely manner all of their obligations to the Loan Parties, and all of their Subsidiaries and Affiliates. Notwithstanding the foregoing, the Agent and the Lenders wish (and the Borrower for itself and each other Loan Party agrees) to eliminate any possibility that any past conditions, acts, omissions, events or circumstances would impair or otherwise adversely affect any of their rights, interests, security and/or remedies. Accordingly, for and in consideration of the agreements contained in this Amendment and other

good and valuable consideration, the Borrower (for itself and its Subsidiaries and Affiliates and the successors, assigns, heirs and representatives of each of the foregoing) (collectively, the "<u>Releasors</u>") does hereby fully, finally, unconditionally and irrevocably release, waive and forever discharge the Agent and the Lenders, together with their respective Affiliates and Related Funds, and each of the directors, officers, employees, agents, attorneys and consultants of each of the foregoing (collectively, the "<u>Released Parties</u>"), from any and all debts, claims, allegations, obligations, damages, costs, attorneys' fees, suits, demands, liabilities, actions, proceedings and causes of action, in each case, whether known or unknown, contingent or fixed, direct or indirect, and of whatever nature or description, and whether in law or in equity, under contract, tort, statute or otherwise, which any Releasor has heretofore had or now or hereafter can, shall or may have against any Released Party by reason of any act, omission or thing whatsoever done or omitted to be done, in each case, on or prior to the Amendment No. 3 Effective Date directly arising out of, connected with or related to this Amendment, the Credit Agreement or any other Loan Document, or any act, event or transaction related or attendant thereto, or the agreements of the Agent or any Lender contained therein, or the possession, use, operation or control of any of the assets of any Loan Party, or the making of any Loans or other advances, or the management of such Loans or other advances or the Collateral. The Borrower for itself and each other Loan Party represents and warrants that it has no knowledge of any claim by any Releasor against any Released Party which would not be released hereby.

10. <u>Further Assurances</u>. The Borrower shall, and shall cause each other Loan Party to, execute any and all further documents, agreements and instruments, and take all further actions, as may be required under Applicable Law or as the Agent may reasonably request, in order to effect the purposes of this Amendment.

11. Miscellaneous.

(a) This Amendment may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which shall be deemed to be an original but all of which taken together shall constitute one and the same agreement. Delivery of an executed counterpart of this Amendment by facsimile or electronic mail shall be equally effective as delivery of an original executed counterpart of this Amendment.

(b) Section and paragraph headings herein are included for convenience of reference only and shall not constitute a part of this Amendment for any other purpose.

(c) This Amendment shall be governed by, and construed in accordance with, the laws of the State of New York.

(d) The Borrower for itself and each other Loan Party hereby acknowledges and agrees that this Amendment constitutes a "Loan Document" under the Credit Agreement.

(e) Any provision of this Amendment that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining portions hereof or affecting the validity or enforceability of such provision in any other jurisdiction.

[Remainder of page intentionally left blank.]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed and delivered as of the date set forth on the first page hereof.

> MONEYLION TECHNOLOGIES INC. as Borrower

By: /s/ Richard Correia Name: Richard Correia

Title: Chief Financial Officer

MONROE CAPITAL MANAGEMENT ADVISORS, LLC, as Administrative Agent

By:

/s/ Joseph Valickus Name: Joseph Valickus Title: Managing Director

LENDERS:

MONROE CAPITAL CORPORATION
in its capacity as Lender

/s/ Joseph Valickus

Name: Joseph Valickus Title: Managing Director

MC INCOME PLUS FINANCING SPV LLC, in its capacity as Lender

By: Monroe Capital Income Plus Corporation, as Collateral Manager

By: /s/ Joseph Valickus

1	
Name:	Joseph Valickus
Title:	Managing Director

MONROE CAPITAL FUND MARSUPIAL (LUX) FINANCING SPV LP, in its capacity as Lender

By: Monroe Capital Fund Marsupial (Lux) Financing GP LLC, its General Partner

By: Monroe Capital Management Advisors, LLC, as Designated Manager

By:	/s/ Joseph Valio	ckus
	Name:	Joseph Valickus
	Title:	Managing Director

MONROE CAPITAL OPPORTUNISTIC PRIVATE CREDIT FUND FINANCING SPV SCSP, in its capacity as Lender

- By: Monroe Capital Management Advisors, LLC, as Investment Manager
- By: /s/ Joseph Valickus

Name:	Joseph Valickus
Title:	Managing Director

MONROE CAPITAL PRIVATE CREDIT FUND 559 FINANCING SPV LLC,
in its capacity as a Lender

- By: Monroe Capital Private Credit Fund 559 LP, as its Designated Manager
- By: Monroe Capital Private Credit Fund 559 GP, LLC, its general partner
- By: /s/ Joseph Valickus Name: Joseph Valickus Title: Managing Director

MONROE CAPITAL PRIVATE CREDIT MASTER FUND IV SCSP, in its capacity as Lender

By: MONROE CAPITAL MANAGEMENT ADVISORS, LLC, as Investment Manager

By: /s/ Joseph Valickus

Name:	Joseph Valickus
Title:	Managing Director

MONROE CAPITAL PRIVATE CREDIT MASTER FUND IV (UNLEVERAGED) SCSP, in its capacity as Lender

- By: MONROE CAPITAL MANAGEMENT ADVISORS, LLC, as Investment Manager
- By: /s/ Joseph Valickus

 Name:
 Joseph Valickus

 Title:
 Managing Director

ANNEX A

Amended Financing Agreement

(see attached)

Agreed Form

CREDIT AGREEMENT

dated as of March 24, 2022

among

MONEYLION TECHNOLOGIES INC., as Borrower,

THE VARIOUS FINANCIAL INSTITUTIONS PARTY HERETO, as Lenders,

MONROE CAPITAL MANAGEMENT ADVISORS, LLC, as Administrative Agent and Lead Arranger

TABLE OF CONTENTS

i

SECTION 1 1.1 1.2 1.3 1.4 1.5	DEFINITIONS. Definitions Certain Interpretive Provisions. Accounting and Other Terms. Treatment of LLC Division Reclassifiable Items	1 34 35 36 36 <u>37</u>
SECTION 2		OWING AND CONVERSION PROCEDURES.
2.1	3637 Commitments 2.1.1 Term Loan Commitments. 2.1.2 Incremental Facilities.	36<u>37</u> 36<u>37</u> 3738
2.2	2.1.2Incremental Factures.Loan Procedures.2.2.1Various Types of Loans2.2.2Borrowing Procedures.2.2.3Conversion Procedures.	39<u>40</u> 39<u>40</u> <u>3940</u> 40
2.3	Commitments Several	41
2.4	Certain Conditions	41
SECTION 3	EVIDENCING OF LOANS.	41
3.1	Notes	41
3.2	Recordkeeping	<u>4142</u>
SECTION 4	INTEREST.	41 42
4.1	Interest Rates	41 42
1.1	4.1.1 Term Loans.	41 42
	4.1.2 Default Rate	42
	4.1.3 Interest Payment Dates	$\frac{42}{4243}$
4.2	Setting and Notice of Adjusted Term SOFR Rat	
4.2		
	Computation of Interest	42 <u>43</u> 42.42
4.4	Term SOFR Conforming Changes	<u>4243</u>
SECTION 5	FEES.	43
5.1	Administrative Agent's Fees	43
5.2	Applicable Premium	43
0.2	. pp. eucle : remain	
SECTION 6	PREPAYMENTS; REPAYMENTS.	44 45
6.1	Prepayments.	44 45
	6.1.1 Voluntary Prepayments	<u> </u>
	6.1.2 Mandatory Prepayments	<u>4445</u>
6.2	Manner of Prepayments.	45 46
	6.2.1 All Prepayments	4 <u>5</u> 46
6.3	Repayments	46
6.4	Option to Decline	4 <u>6</u> 47
0.1	-r	
SECTION 7	MAKING AND PRORATION OF PAYMENTS	S; SETOFF; TAXES. 4647
7.1	Making of Payments.	46 <u>47</u>

Page

7.2	Application of Certain Payments	46 47	,
7.3	Due Date Extension	47 48	
7.4	Setoff	48	
7.5	Proration of Payments	48	
7.6	Taxes.	48	
SECTION 8	INCREASED COSTS; SPECIAL PROVISION		R LOANS. 51
8.1	Increased Costs.	51	
8.2	Basis for Determining Interest Rate Inadequate		51<u>52</u>
8.3	Changes in Law Rendering SOFR Loans Unlaw		52 <u>53</u>
8.4	Right of Lenders to Fund through Other Offices		52<u>53</u>
8.5	Mitigation of Circumstances; Replacement of L		53
8.6	Conclusiveness of Statements; Survival of Prov		53<u>54</u>
8.7	Benchmark Replacement Setting.	53<u>54</u>	
SECTION 9	REPRESENTATIONS AND WARRANTIES.		55
9.1	Organization	55	
9.2	Authorization; No Conflict.	55	
9.3	Validity and Binding Nature	55 56	
9.4	Financial Condition	55 56	
9.5	No Material Adverse Change	55 56	
9.6	Litigation and Contingent Liabilities		55 56
9.7	Ownership of Properties; Liens		56
9.8	Equity Ownership	56	
9.9	Pension Plans.	56	
9.10		56 57	
9.11	Compliance with Laws Consents; Certain Actio		56<u>57</u>
9.12		57<u>58</u>	
9.13		57<u>58</u>	
9.14	Solvency, etc	57<u>58</u>	
9.15	Environmental Matters	58	
9.16	Insurance	58<u>59</u>	
9.17	Real Property	58<u>59</u>	
9.18	Information	58<u>59</u>	
9.19	Location of Bank Accounts	59	
9.20	[Reserved]	59	
9.21	Intellectual Property	59	
9.22	[Reserved].	59<u>60</u>	
9.23	Employee and Labor Matters	59 60	
9.24		59<u>60</u>	
9.25	Name; Jurisdiction of Organization; Organization 5960	onal ID Num	ber; Chief Place of Business; Chief Executive Office; FEIN
9.26	Locations of Collateral	60	
9.27	Security Interests	60	
9.28	No Default	60<u>61</u>	
9.29	Hedging Agreements	60<u>61</u>	
9.30	OFAC	60<u>61</u>	
9.31	Patriot Act	60<u>61</u>	

SECTION 10	AFFIRMATIVE COVENANTS.	61	
10.1	Reports, Certificates and Other Information	61	
	10.1.1 Annual Reports	61	
	10.1.2 Quarterly Reports	<u>6162</u>	
	10.1.3 Monthly Reports	<u>6162</u>	
	10.1.4 Compliance Certificates	62	
	10.1.5 Reports to the SEC and to Shar		
	10.1.6 Notice of Default, Litigation, a		
	10.1.7 Real Estate	63 64	
	10.1.7Real Estate10.1.8Management Reports	63<u>64</u>	
	10.1.9 Projections	63 64	
	10.1.10 Other Information	64	
	10.1.11 Instacash	64	
10.2	Books, Records, and Inspections	<u>6465</u>	
10.2	Maintenance of Property; Insurance.	64 65	
10.3			
	Compliance with Laws; Payment of Taxes an		
10.5	Maintenance of Existence, etc	<u>6667</u>	
10.6	Use of Proceeds	<u>6667</u>	
10.7	Employee Benefit Plans.	66<u>67</u>	
10.8	Environmental Matters	<u>6667</u>	
10.9	Lender Meetings	<u>6667</u>	
10.10	Further Assurances	67<u>68</u>	
10.11	Deposit Accounts	67<u>68</u>	
10.12	Collateral Access Agreements	67<u>68</u>	
10.13	Guarantor Joinders	67<u>68</u>	
SECTION 11	NEGATIVE COVENANTS	68 69	
11.1	Debt	68 69	
11.2	Liens	70 71	
11.3	Restricted Payments	7273	
11.4	Mergers, Consolidations, Sales	74 75	
11.5	Modification of Certain Documents; Organiz		
11.6	Transactions with Affiliates	76 78	
11.7	Inconsistent Agreements	77<u>78</u>	
11.8	Business Activities	78 80	
11.0	Investments	78 80	
11.10	Reserved]Indemnification Payments un		
11.11	Fiscal Year	80 <u>81</u>	
11.12	Financial Covenants	80 81	
11.12	11.12.1 Minimum Revenue	80 82	
	11.12.2 Minimum EBITDA	<u>80</u> 82	
	11.12.3 Minimum Liquidity	<u>8183</u>	
11.13	Compliance with Laws	<u>8183</u>	
11.13	Permitted Activities of Parent	81 83	
SECTION 12	EFFECTIVENESS; CONDITIONS OF LEN	DING, ETC. 82 84	
12.1	Conditions to Effectiveness	82 84	
	12.1.1 Agreement, Notes, and other L	oan Documents 8284	

iii

Page

12.1.3 Consents, etc \$\frac{835}{85}\$ 12.1.4 Letter of Direction \$\frac{835}{85}\$ 12.1.5 Collateral and Dilgence Questionnaire \$\frac{835}{85}\$ 12.1.6 [Reserved] \$\frac{835}{85}\$ 12.1.7 [Reserved] \$\frac{835}{85}\$ 12.1.8 [Reserved] \$\frac{835}{85}\$ 12.1.9 Opinions of Counsel \$\frac{935}{85}\$ 12.1.10 Insurance \$\frac{835}{85}\$ 12.1.11 Payment of Fees \$\frac{845}{85}\$ 12.1.12 Debt to be Repaid \$\frac{845}{85}\$ 12.1.13 Solvency Certificate \$\frac{845}{845}\$ 12.1.14 Search Results; Lien Terminations \$\frac{845}{845}\$ 12.1.15 Filings, Registrations, and Recordings \$\frac{845}{845}\$ 12.1.16 Closing Certificate \$\frac{845}{845}\$ 12.1.17 Financial Statements \$\frac{845}{846}\$ 12.1.20 Financial Condition \$\frac{845}{845}\$ 12.1.21 Know-Your-Customer and Anti-Money Laundering \$\frac{858}{856}\$ 12.1.2 Conditions Precedent to all Loans \$\frac{858}{857}\$ SECTION 13		12.1.2	Authorization Documents		82 84	
12.1.4 Letter of Direction 8385 12.1.5 Collateral and Diligence Questionnaire 8385 12.1.6 [Reserved] 8385 12.1.7 [Reserved] 8385 12.1.8 [Reserved] 8385 12.1.9 Opinions of Counsel 8385 12.1.10 Insurance 8385 12.1.11 Payment of Fees 8385 12.1.12 Debt to be Repaid 8485 12.1.13 Solvency Certificate 8485 12.1.14 Search Results; Lien Terminations 8486 12.1.15 Filings, Registrations, and Recordings 8486 12.1.16 Closing Certificate 8486 12.1.17 Financial Statements 8486 12.1.18 No Material Adverse Effect 8486 12.1.20 Financial Condition 8587 12.2.1 Comditions Precedent to all Loans 6587 12.2.1 Comfirmatory Certificate 8587 12.2.2 Conditions Precedent to all Loans, etc 8682 13.1.1 Non-Payment of the Loans, etc 8682 13.1.2 <						
12.1.5 Collateral and Diligence Questionnaire 6935 12.1.6 [Reserved] 6935 12.1.7 [Reserved] 6935 12.1.8 [Reserved] 6935 12.1.9 Opinions of Counsel 6935 12.1.10 Insurance 6935 12.1.11 Payment of Fees 6935 12.1.12 Debt to be Repaid 6935 12.1.13 Solvency Certificate 69485 12.1.14 Search Results; Lien Terminations 69486 12.1.15 Filings, Registrations, and Recordings 69486 12.1.16 Closing Certificate 69486 12.1.17 Financial Statements 69486 12.1.19 Capital Structure 69486 12.1.20 Financial Condition 69486 12.1.21 Know-Your-Customer and Anti-Money Laundering 6587 12.2.1 Conditions Precedent to all Loans 6587 12.2.1 Conditions Precedent to all Loans 6587 12.2.2 Confirmatory Certificate 6682 13.1.1 Non-Payment of Other Debt 6682 13.1						
12.1.6 [Reserved] 8385 12.1.7 [Reserved] 8385 12.1.8 [Reserved] 8385 12.1.9 Opinions of Counsel 8385 12.1.10 Insurance 8385 12.1.11 Payment of Fees 8485 12.1.12 Debt to be Repaid 8385 12.1.13 Solvency Certificate 8485 12.1.14 Scarch Results: Lien Terminations 8486 12.1.15 Filings, Registrations, and Recordings 8486 12.1.16 Closing Certificate 8486 12.1.17 Financial Statements 8486 12.1.10 Capital Structure 8486 12.1.20 Financial Condition 8486 12.1.21 Know-Your-Customer and Anti-Money Laundering 8587 12.2 Conditions Precedent to all Loans 9587 12.2.1 Compliance with Warranties, No Default, etc 8587 12.2.2 Confirmatory Certificate 8587 13.1 Events of Default 8587 13.1.1 Non-Payment of the Loans, etc 9682 13.1.3						
12.1.7 [Reserved] 8385 12.1.8 [Reserved] 8385 12.1.9 Opinions of Counsel 8385 12.1.10 Insurance 8385 12.1.11 Payment of Fees 8385 12.1.12 Debt to be Repaid 8385 12.1.13 Solvency Certificate 8486 12.1.14 Search Results; Lien Terminations 8486 12.1.15 Filings, Registrations, and Recordings 8486 12.1.16 Closing Certificate 8486 12.1.17 Financial Statements 8486 12.1.18 No Material Adverse Effect 8486 12.1.19 Capital Structure 8486 12.1.20 Financial Condition 8486 12.1.21 Know-Your-Customer and Anti-Money Laundering 8587 12.22 Confitions Precedent to all Loans 8587 12.21 Compliance with Warranties, No Default, etc 8587 12.22 Confirmatory Certificate 8587 13.11 Non-Payment of Other Debt 8682 13.12 Non-Payment of Other Debt 8682 13			•			
12.1.8 $[Reserved]$ Θ_{255} 12.1.9Opinions of Counsel Θ_{255} 12.1.10Insurance Θ_{255} 12.1.11Payment of Fees Θ_{255} 12.1.12Debt to be Repaid Θ_{255} 12.1.13Solvency Certificate Θ_{485} 12.1.14Search Results; Lien Terminations Θ_{486} 12.1.15Filings, Registrations, and Recordings Θ_{486} 12.1.16Closing Certificate Θ_{486} 12.1.17Financial Statements Θ_{486} 12.1.18No Material Adverse Effect Θ_{486} 12.1.20Financial Condition Θ_{486} 12.1.21Know-Your-Customer and Anti-Money Laundering Θ_{587} 12.22Conditions Precedent to all Loans Θ_{587} 12.2.1Compliance with Waranties, No Default, etc Θ_{587} 12.2.2Confirmatory Certificate Θ_{688} 13.1Events of DEFAULT AND THEIR EFFECT. Θ_{688} 13.1.1Non-Payment of the Loans, etc Θ_{688} 13.1.3[Reserved], Repurchase of Defective Receivables Θ_{688} 13.1.4Bankruptey, Insolvency, etc Θ_{688} 13.1.5Non-Compliance with Loan Documents. Θ_{789} 13.1.6Representations; Warranties Θ_{789} 13.1.9Invalidity of Loan Documents, etc			L J			
12.1.9 Opinions of Counsel 8385 12.1.10 Insurance 8385 12.1.11 Payment of Fees 8385 12.1.12 Debt to be Repaid 8385 12.1.13 Solvency Certificate 8485 12.1.14 Search Results; Lien Terminations 8486 12.1.15 Filings, Registrations, and Recordings 8486 12.1.16 Closing Certificate 8486 12.1.17 Financial Statements 8486 12.1.18 No Material Adverse Effect 8486 12.1.19 Capital Structure 8486 12.1.20 Financial Condition 6486 12.1.21 Know-Your-Customer and Anti-Money Laundering 8587 12.2 Conditions Precedent to all Loans 8587 12.2.1 Compliance with Warranties, No Default, etc 8587 12.2.2 Confirmatory Certificate 8587 12.2.1 Conditions Precedent to the Loans, etc 8682 13.1 Events of Default 8587 13.1 Events of Default 8587 13.1.3 [Reserved] Repurchase of Defective Receivables						
12.1.10 Insurance \$385 12.1.11 Payment of Fees \$385 12.1.12 Debt to be Repaid \$385 12.1.13 Solvency Certificate \$485 12.1.14 Search Results; Lien Terminations \$4486 12.1.15 Filings, Registrations, and Recordings \$4486 12.1.16 Closing Certificate \$4486 12.1.17 Financial Statements \$4486 12.1.18 No Material Adverse Effect \$4486 12.1.20 Financial Condition \$4486 12.1.20 Financial Condition \$4486 12.1.21 Know-Your-Customer and Anti-Money Laundering \$587 12.2 Conditions Precedent to all Loans \$4587 12.2.1 Compliance with Warranties, No Default, etc \$587 12.2.2 Confirmatory Certificate \$6587 13.1 EVENTS OF DEFAULT AND THEIR EFFECT. \$6587 13.1.1 Non-Payment of Other Debt \$6688 13.1.2 Non-Compliance with Loan Documents. \$6688 13.1.4 Bakruptcy, Insolvency, etc \$6688 13.1.6 Represent				+		
12.1.11 Payment of Fees 8385 12.1.12 Debt to be Repaid 8485 12.1.13 Solvency Certificate 8485 12.1.14 Search Results; Lien Terminations 8486 12.1.15 Filings, Registrations, and Recordings 8486 12.1.16 Closing Certificate 8486 12.1.17 Financial Statements 8486 12.1.19 Capital Structure 8486 12.1.20 Financial Condition 8486 12.1.21 Know-Your-Customer and Anti-Money Laundering 8587 12.2 Conditions Precedent to all Loans 8587 12.2.1 Compliance with Warranties, No Default, etc 8587 12.2.2 Confirmatory Certificate 8587 SECTION 13 EVENTS OF DEFAULT AND THEIR EFFECT. 8587 13.1 Events of Default 8587 13.1 Events of Default 8587 13.1.3 Hearved/Repurchase of Defective Receivables 8688 13.1.4 Bankruptey, Insolvency, etc 8688 13.1.5 Non-Compliance with Loan Documents. 8688 13.1.6 Repres						
12.1.12 Debt to be Repaid \$#355 12.1.13 Solvency Certificate \$#485 12.1.14 Search Results; Lien Terminations \$#486 12.1.15 Filings, Registrations, and Recordings \$#486 12.1.16 Closing Certificate \$#486 12.1.17 Financial Adverse Effect \$#486 12.1.19 Capital Structure \$#486 12.1.20 Financial Condition \$#486 12.1.21 Know-Your-Customer and Anti-Money Laundering \$#586 12.1.20 Financial Condition \$#587 12.2.2 Confirmatory Certificate \$#587 12.2.1 Compliance with Warranties, No Default, etc \$#587 12.2.2 Confirmatory Certificate \$#587 13.1 EVENTS OF DEFAULT AND THEIR EFFECT. \$#587 13.1 Events of Default \$#587 13.1 Events of Default \$#587 13.1.3 IReserved]Repurchase of Defective Receivables \$#688 13.1.3 IReserved]Repurchase of Defective Receivables \$#688 13.1.4 Bakruptcy, Insolvency, etc \$#688 13.		12.1.10	Insurance	+	83<u>85</u>	
12.1.13 Solvency Certificate 8485 12.1.14 Search Results; Lien Terminations 8486 12.1.15 Filings, Registrations, and Recordings 8486 12.1.16 Closing Certificate 8486 12.1.17 Financial Statements 8486 12.1.18 No Material Adverse Effect 8486 12.1.19 Capital Structure 8486 12.1.20 Financial Condition 8486 12.1.21 Know-Your-Customer and Anti-Money Laundering 8587 12.2.2 Conditions Precedent to all Loans 8587 12.2.2 Conditions Precedent to all Loans 8587 12.2.2 Confirmatory Certificate 8587 12.2.2 Confirmatory Certificate 8587 13.1 EVENTS OF DEFAULT AND THEIR EFFECT. 8587 13.1 Events of Default 8587 13.1 Events of Default 8582 13.1.1 Non-Payment of the Loans, etc 8688 13.1.2 Non-Payment of Other Debt 8688 13.1.3 [Reserved]Repurchase of Defective Receivables 8668 13.1.3 [Reserved		12.1.11				
12.1.14 Search Results; Lien Terminations 8486 12.1.15 Filings, Registrations, and Recordings 8486 12.1.16 Closing Certificate 8486 12.1.17 Financial Statements 8486 12.1.18 No Material Adverse Effect 8486 12.1.19 Capital Structure 8486 12.1.20 Financial Condition 8486 12.1.21 Know-Your-Customer and Anti-Money Laundering 8586 12.2.2 Conditions Precedent to all Loans 8587 12.2.2 Confirmatory Certificate 8587 12.2.2 Confirmatory Certificate 8587 12.2.2 Confirmatory Certificate 8587 SECTION 13 EVENTS OF DEFAULT AND THEIR EFFECT. 8587 13.1 Non-Payment of the Loans, etc 8682 13.1.2 Non-Payment of Other Debt 8688 13.1.4 Bankruptcy, Insolvency, etc 8688 13.1.4 Bankruptcy, Insolvency, etc 8688 13.1.4 Bankruptcy, Insolvency, etc 8688 13.1.6 Representations; Warranties 8789 13.1.8 Ju		12.1.12		+	83<u>85</u>	
12.1.15 Filings, Registrations, and Recordings 8486 12.1.16 Closing Certificate 8486 12.1.17 Financial Statements 8486 12.1.18 No Material Adverse Effect 8486 12.1.19 Capital Structure 8486 12.1.20 Financial Condition 8486 12.1.20 Financial Condition 8486 12.1.21 Know-Your-Customer and Anti-Money Laundering 8587 12.2 Conditions Precedent to all Loans 8587 12.2.2 Confirmatory Certificate 8587 12.2.2 Confirmatory Certificate 8587 13.1 EVENTS OF DEFAULT AND THEIR EFFECT. 8587 13.1 Events of Default 8587 13.1 Non-Payment of the Loans, etc 8682 13.1.1 Non-Payment of Other Debt 8688 13.1.3 Heserved/Repurchase of Defective Receivables 8688 13.1.4 Bankruptcy, Insolvency, etc 8688 13.1.5 Non-Compliance with Loan Documents. 8789 13.1.6 Representations; Warrantics 8789 13.1.7 Pension		12.1.13	Solvency Certificate		84<u>85</u>	
12.1.16 Closing Certificate 8486 12.1.17 Financial Statements 8486 12.1.18 No Material Adverse Effect 8486 12.1.19 Capital Structure 8486 12.1.20 Financial Condition 8486 12.1.21 Know-Your-Customer and Anti-Money Laundering 8587 12.2 Conditions Precedent to all Loans 8587 12.2.1 Compliance with Warranties, No Default, etc 8587 12.2.2 Confirmatory Certificate 8587 12.2.2 Confirmatory Certificate 8587 13.1 EVENTS OF DEFAULT AND THEIR EFFECT. 8587 13.1 Events of Default 8587 13.1 Non-Payment of the Loans, etc 8688 13.1.2 Non-Payment of Other Debt 8688 13.1.3 [Reserved]Repurchase of Defective Receivables 8688 13.1.4 Bankruptcy, Insolvency, etc 8688 13.1.7 Pension Plans 8789 13.1.8 Judgments 8789 13.1.9 Invalidity of Loan Documents, etc 8789 13.1.10 Charge of Proceedings		12.1.14	Search Results; Lien Terminations	5	84<u>86</u>	
12.1.16 Closing Certificate 8486 12.1.17 Financial Statements 8486 12.1.18 No Material Adverse Effect 8486 12.1.19 Capital Structure 8486 12.1.20 Financial Condition 8486 12.1.21 Know-Your-Customer and Anti-Money Laundering 8587 12.2 Conditions Precedent to all Loans 8587 12.2.1 Compliance with Warranties, No Default, etc 8587 12.2.2 Confirmatory Certificate 8587 12.2.2 Confirmatory Certificate 8587 13.1 EVENTS OF DEFAULT AND THEIR EFFECT. 8587 13.1 Events of Default 8587 13.1 Non-Payment of the Loans, etc 8688 13.1.2 Non-Payment of Other Debt 8688 13.1.3 [Reserved]Repurchase of Defective Receivables 8688 13.1.4 Bankruptcy, Insolvency, etc 8688 13.1.7 Pension Plans 8789 13.1.8 Judgments 8789 13.1.9 Invalidity of Loan Documents, etc 8789 13.1.10 Charge of Proceedings		12.1.15	Filings, Registrations, and Record	ings	84	<u>86</u>
12.1.17 Financial Statements \$486 12.1.18 No Material Adverse Effect \$486 12.1.19 Capital Structure \$486 12.1.20 Financial Condition \$486 12.1.21 Know-Your-Customer and Anti-Money Laundering \$556 12.2 Conditions Precedent to all Loans \$557 12.2 Conditions Precedent to all Loans \$587 12.2.1 Compliance with Warranties, No Default, etc \$587 12.2.2 Confirmatory Certificate \$587 12.2.1 Comfirmatory Certificate \$587 13.1 Events of Default \$587 13.1 Non-Payment of Other Det \$688 13.1.1 Non-Payment of Other Det \$688 13.1.3 [Reserved]Repurchase of Defective Receivables \$668 13.1.4 Bankruptcy, Insolvency, etc \$6688 13.1.5 Non-Compliance with Loan Documents. \$668 13.1.7 Pension Plans \$789 13.1.8 Judgments \$789 13.1.9 Invalidity of Loan Documents, etc \$789 13.1.10 Charge of Proceedings		12.1.16			84 86	
12.1.18No Material Adverse Effect\$448612.1.19Capital Structure\$448612.1.20Financial Condition\$448612.1.21Know-Your-Customer and Anti-Money Laundering\$55612.2Conditions Precedent to all Loans\$658712.2.1Compliance with Warranties, No Default, etc\$658712.2.2Confirmatory Certificate\$658713.1EVENTS OF DEFAULT AND THEIR EFFECT.\$658713.1EVENTS OF DEFAULT AND THEIR EFFECT.\$668713.1.1Non-Payment of the Loans, etc\$668813.1.2Non-Payment of Other Debt\$668813.1.3IPReserved]Repurchase of Defective Receivables\$668813.1.4Bankruptcy, Insolvency, etc\$668813.1.5Non-Compliance with Loan Documents.\$668813.1.6Representations; Warranties\$78913.1.7Pension Plans\$78913.1.8Judgments\$78913.1.9Invalidity of Loan Documents, etc\$78913.1.1Charges or Proceedings\$892013.1.2Instacash9013.2Effect of Event of Default\$892013.3Credit Bidding\$892113.4Curative Equity.\$99113.2Effect of Event of Default\$892013.4Curative Equity.\$99113.2Effect of Event of Default\$892013.4Curative Equity.\$99113.4Curative Equity.\$99213.4Curative Equity. <td< td=""><td></td><td>12.1.17</td><td></td><td></td><td></td><td></td></td<>		12.1.17				
12.1.19Capital Structure848612.1.20Financial Condition848612.1.21Know-Your-Customer and Anti-Money Laundering858612.2Conditions Precedent to all Loans858712.2.1Compliance with Warranties, No Default, etc858712.2.2Confirmatory Certificate858712.2.2Confirmatory Certificate858713.1EVENTS OF DEFAULT AND THEIR EFFECT.858713.1Events of Default858713.1Events of Default858713.1.1Non-Payment of the Loans, etc868813.1.2Non-Payment of Other Debt868813.1.3IRcestrued]Repurchase of Defective Receivables868813.1.4Bankruptcy, Insolvency, etc868813.1.5Non-Compliance with Loan Documents.868813.1.6Representations; Warranties878913.1.7Pension Plans878913.1.8Judgments878913.1.10Charges or Proceedings889013.1.2Instacash9013.2Effect of Event of Default889013.3Credit Bidding889113.4Curative Equity.899113.4Curative Equity.899113.4Curative Equity.89921SECTION 14AGENCY.902214.1Appointment and Authorization909214.3Delegation of Duties9092						
12.1.20 Financial Condition 8486 12.1.21 Know-Your-Customer and Anti-Money Laundering 8585 12.2 Conditions Precedent to all Loans 8587 12.2.1 Compliance with Warranties, No Default, etc 8587 12.2.2 Confirmatory Certificate 8587 12.2.2 Confirmatory Certificate 8587 13.1 EVENTS OF DEFAULT AND THEIR EFFECT. 8587 13.1 Events of Default 8587 13.1 Events of Default 8587 13.1.1 Non-Payment of the Loans, etc 8688 13.1.2 Non-Payment of Other Debt 8688 13.1.4 Bankruptcy, Insolvency, etc 8688 13.1.5 Non-Compliance with Loan Documents. 8688 13.1.6 Representations; Warranties 8789 13.1.7 Pension Plans 8789 13.1.10 Charges or Proceedings 8890 13.1.11 Change of Control 8890 13.1.2 Instacash 90 13.2 Effect of Event of Default 8890 13.3 Creatit Bidding 8491				-		
12.1.21 Know-Your-Customer and Anti-Money Laundering 8586 12.2 Conditions Precedent to all Loans 8587 12.2.1 Compliance with Warranties, No Default, etc 8587 12.2.2 Confirmatory Certificate 8587 SECTION 13 EVENTS OF DEFAULT AND THEIR EFFECT. 8587 13.1 Events of Default 8587 13.1.1 Non-Payment of the Loans, etc 8687 13.1.2 Non-Payment of Other Debt 8688 13.1.3 Heserved/Repurchase of Defective Receivables 8688 13.1.4 Bankruptcy, Insolvency, etc 8688 13.1.5 Non-Compliance with Loan Documents. 8688 13.1.6 Representations; Warranties 8789 13.1.7 Pension Plans 8789 13.1.9 Invalidity of Loan Documents, etc 8789 13.1.10 Charges or Proceedings 8890 13.1.1 Charges of Control 8890 13.1.10 Charges of Control 8890 13.1.11 Charges of Proceedings 8890 13.1.2 Instacash 90 13.2 Effect of Ev			1		8486	
12.2 Conditions Precedent to all Loans \$587 12.2.1 Compliance with Warranties, No Default, etc \$587 12.2.2 Confirmatory Certificate \$587 SECTION 13 EVENTS OF DEFAULT AND THEIR EFFECT. \$587 13.1 Events of Default \$587 13.1 Events of Default \$587 13.1 Non-Payment of the Loans, etc \$687 13.1.2 Non-Payment of Other Debt \$688 13.1.3 [Reserved]Repurchase of Defective Receivables \$668 13.1.4 Bankruptcy, Insolvency, etc \$688 13.1.5 Non-Compliance with Loan Documents. \$688 13.1.6 Representations; Warranties \$789 13.1.7 Pension Plans \$789 13.1.8 Judgments \$789 13.1.9 Invalidity of Loan Documents, etc \$789 13.1.10 Charges or Proceedings \$8901 13.2 Effect of Event of Default \$8901 13.3 Credit Bidding \$8991 13.4 Curative Equity. \$9921 SECTION 14 AGENCY. \$9022 <td></td> <td></td> <td></td> <td>Ioney I aun</td> <td></td> <td>0586</td>				Ioney I aun		0586
12.2.1 Compliance with Warranties, No Default, etc 8587 12.2.2 Confirmatory Certificate 8587 SECTION 13 EVENTS OF DEFAULT AND THEIR EFFECT. 8587 13.1 Events of Default 8587 13.1 Events of Default 8587 13.1 Non-Payment of the Loans, etc 8687 13.1.1 Non-Payment of Other Debt 8688 13.1.3 fReserved]Repurchase of Defective Receivables 8688 13.1.4 Bankruptcy, Insolvency, etc 8688 13.1.5 Non-Compliance with Loan Documents. 8688 13.1.6 Representations; Warranties 8789 13.1.8 Judgments 8789 13.1.9 Invalidity of Loan Documents, etc 8789 13.1.10 Charges or Proceedings 8890 13.1.2 Instacash 90 13.2 Effect of Event of Default 8890 13.3 Credit Bidding 8891 13.4 Curative Equity. 8991 13.4 Curative Equity. 8991 13.4 Curative Equity. 89921 SE	12.2					85 <u>80</u>
12.2.2Confirmatory Certificate8587SECTION 13EVENTS OF DEFAULT AND THEIR EFFECT.858713.1Events of Default858713.1.1Non-Payment of the Loans, etc868213.1.2Non-Payment of Other Debt868813.1.3[Reserved]Repurchase of Defective Receivables868813.1.4Bankruptcy, Insolvency, etc868813.1.5Non-Compliance with Loan Documents.868813.1.6Representations; Warranties878913.1.7Pension Plans878913.1.8Judgments878913.1.9Invalidity of Loan Documents, etc878913.1.10Charges or Proceedings889013.1.2Instacash9013.2Effect of Event of Default889013.3Credit Bidding889113.4Curative Equity.8991SECTION 14AGENCY.909214.1Appointment and Authorization909214.2[Reserved]909214.3Delegation of Duties9092	12.2					0507
SECTION 13 EVENTS OF DEFAULT AND THEIR EFFECT. \$587 13.1 Events of Default \$587 13.1 Events of Default \$687 13.1.1 Non-Payment of the Loans, etc \$688 13.1.2 Non-Payment of Other Debt \$688 13.1.3 [Reserved]Repurchase of Defective Receivables \$6688 13.1.3 [Reserved]Repurchase of Defective Receivables \$6688 13.1.4 Bankruptcy, Insolvency, etc \$6688 13.1.5 Non-Compliance with Loan Documents. \$6688 13.1.6 Representations; Warranties \$789 13.1.7 Pension Plans \$789 13.1.9 Invalidity of Loan Documents, etc \$789 13.1.10 Charges or Proceedings \$890 13.1.11 Change of Control \$890 13.1.2 Instacash \$90 13.2 Effect of Event of Default \$8890 13.3 Credit Bidding \$8891 13.4 Curative Equity. \$9791 SECTION 14 AGENCY. \$9022 14.1 Appointment and Authorization \$9092				Jelault, etc		03<u>07</u>
13.1Events of Default 8587 13.1.1Non-Payment of the Loans, etc 8687 13.1.2Non-Payment of Other Debt 8688 13.1.3[Reserved]Repurchase of Defective Receivables 8688 13.1.4Bankruptcy, Insolvency, etc 8688 13.1.5Non-Compliance with Loan Documents. 8688 13.1.6Representations; Warranties 8789 13.1.7Pension Plans 8789 13.1.8Judgments 8789 13.1.9Invalidity of Loan Documents, etc 8789 13.1.10Charges or Proceedings 8890 13.1.11Charge of Control 8890 13.2Effect of Event of Default 8890 13.3Credit Bidding 8891 13.4Curative Equity. 8991 SECTION 14AGENCY. 90922 14.1Appointment and Authorization 90922 14.2[Reserved] 90922 14.3Delegation of Duties 90922		12.2.2	Commatory Certificate		65<u>67</u>	
13.1Events of Default 8587 13.1.1Non-Payment of the Loans, etc 8687 13.1.2Non-Payment of Other Debt 8688 13.1.3[Reserved]Repurchase of Defective Receivables 8688 13.1.4Bankruptcy, Insolvency, etc 8688 13.1.5Non-Compliance with Loan Documents. 8688 13.1.6Representations; Warranties 8789 13.1.7Pension Plans 8789 13.1.8Judgments 8789 13.1.9Invalidity of Loan Documents, etc 8789 13.1.10Charges or Proceedings 8890 13.1.11Charge of Control 8890 13.2Effect of Event of Default 8890 13.3Credit Bidding 8891 13.4Curative Equity. 8991 SECTION 14AGENCY. 90922 14.1Appointment and Authorization 90922 14.2[Reserved] 90922 14.3Delegation of Duties 90922	SECTION 13	EVENTS OF	DEFAULT AND THEIR FEFECT		<u>95</u> 97	
13.1.1Non-Payment of the Loans, etc 8687 13.1.2Non-Payment of Other Debt 8688 13.1.3[Reserved]Repurchase of Defective Receivables 8688 13.1.4Bankruptcy, Insolvency, etc 8688 13.1.5Non-Compliance with Loan Documents. 8688 13.1.6Representations; Warranties 8789 13.1.7Pension Plans 8789 13.1.8Judgments 8789 13.1.9Invalidity of Loan Documents, etc 8789 13.1.10Charges or Proceedings 8890 13.1.11Change of Control 8890 13.2Effect of Event of Default 8890 13.3Credit Bidding 8891 13.4Curative Equity. 8991 SECTION 14AGENCY. 90922 14.1Appointment and Authorization 90922 14.2[Reserved] 90922 14.3Delegation of Duties 90922					05 <u>07</u>	
13.1.2Non-Payment of Other Debt868813.1.3[Reserved]Repurchase of Defective Receivables868813.1.4Bankruptcy, Insolvency, etc868813.1.5Non-Compliance with Loan Documents.868813.1.6Representations; Warranties878913.1.7Pension Plans878913.1.8Judgments878913.1.9Invalidity of Loan Documents, etc878913.1.10Charges or Proceedings889013.1.11Change of Control889013.1.2Instacash9013.2Effect of Event of Default889113.4Curative Equity.8991SECTION 14AGENCY.902214.1Appointment and Authorization909214.2[Reserved]909214.3Delegation of Duties9092	15.1			85 <u>87</u>	2607	
13.1.3[Reserved]Repurchase of Defective Receivables868813.1.4Bankruptcy, Insolvency, etc868813.1.5Non-Compliance with Loan Documents.868813.1.6Representations; Warranties878913.1.7Pension Plans878913.1.8Judgments878913.1.9Invalidity of Loan Documents, etc878913.1.10Charges or Proceedings889013.1.11Charge of Control889013.2Effect of Event of Default889113.3Credit Bidding889113.4Curative Equity.8991SECTION 14AGENCY.909214.1Appointment and Authorization909214.2[Reserved]909214.3Delegation of Duties9092			2			
13.1.4Bankruptcy, Insolvency, etc 8688 13.1.5Non-Compliance with Loan Documents. 8688 13.1.6Representations; Warranties 8789 13.1.7Pension Plans 8789 13.1.8Judgments 8789 13.1.9Invalidity of Loan Documents, etc 8789 13.1.10Charges or Proceedings 8890 13.1.11Charge of Control 8890 13.1.2Instacash 90 13.2Effect of Event of Default 8891 13.3Credit Bidding 8891 13.4Curative Equity. 8991 SECTION 14AGENCY. 9092 14.1Appointment and Authorization 9092 14.2[Reserved] 9092 14.3Delegation of Duties 9092						00
13.1.5Non-Compliance with Loan Documents.868813.1.6Representations; Warranties878913.1.7Pension Plans878913.1.8Judgments878913.1.9Invalidity of Loan Documents, etc878913.1.10Charges or Proceedings889013.1.11Charge of Control889013.1.2Instacash9013.2Effect of Event of Default889113.3Credit Bidding889113.4Curative Equity.8991SECTION 14AGENCY.909214.1Appointment and Authorization909214.2[Reserved]909214.3Delegation of Duties9092				e Receivad		<u>00</u>
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$						0.0
13.1.7Pension Plans878913.1.8Judgments878913.1.9Invalidity of Loan Documents, etc878913.1.0Charges or Proceedings889013.1.10Charges or Proceedings889013.1.11Change of Control889013.1.2Instacash9013.2Effect of Event of Default889113.3Credit Bidding889113.4Curative Equity.8991SECTION 14AGENCY.909214.1Appointment and Authorization909214.2[Reserved]909214.3Delegation of Duties9092			1	ments.		<u>88</u>
13.1.8 Judgments 8789 13.1.9 Invalidity of Loan Documents, etc 8789 13.1.0 Charges or Proceedings 8890 13.1.10 Charges or Proceedings 8890 13.1.11 Change of Control 8890 13.2 Effect of Event of Default 8890 13.3 Credit Bidding 8891 13.4 Curative Equity. 8991 SECTION 14 AGENCY. 9092 14.1 Appointment and Authorization 9092 14.2 [Reserved] 9092 14.3 Delegation of Duties 9092			1 ,	0.500	87<u>89</u>	
13.1.9 Invalidity of Loan Documents, etc 8789 13.1.10 Charges or Proceedings 8890 13.1.11 Charges or Proceedings 8890 13.1.12 Instacash 90 13.2 Effect of Event of Default 8890 13.3 Credit Bidding 8891 13.4 Curative Equity. 8991 SECTION 14 AGENCY. 9092 14.1 Appointment and Authorization 9092 14.2 [Reserved] 9092 14.3 Delegation of Duties 9092						
13.1.10Charges or Proceedings889013.1.11Charges or Proceedings889013.1.11Charge of Control889013.1.12Instacash9013.2Effect of Event of Default889013.3Credit Bidding889113.4Curative Equity.8991SECTION 14AGENCY.909214.1Appointment and Authorization909214.2[Reserved]909214.3Delegation of Duties9092						
13.1.11Change of Control889013.1.12Instacash9013.2Effect of Event of Default889013.3Credit Bidding889113.4Curative Equity.8991SECTION 14AGENCY.909214.1Appointment and Authorization909214.2[Reserved]909214.3Delegation of Duties9092						
13.1.12Instacash9013.2Effect of Event of Default889013.3Credit Bidding889113.4Curative Equity.8991SECTION 14AGENCY.909214.1Appointment and Authorization909214.2[Reserved]909214.3Delegation of Duties9092						
13.2Effect of Event of Default889013.3Credit Bidding889113.4Curative Equity.8991SECTION 14AGENCY.909214.1Appointment and Authorization909214.2[Reserved]909214.3Delegation of Duties9092		13.1.11	Change of Control	+	88<u>90</u>	
13.3Credit Bidding889113.4Curative Equity.8991SECTION 14AGENCY.909214.1Appointment and Authorization909214.2[Reserved]909214.3Delegation of Duties9092		<u>13.1.12</u>	Instacash	<u>90</u>		
13.4Curative Equity.8991SECTION 14AGENCY.909214.1Appointment and Authorization909214.2[Reserved]909214.3Delegation of Duties9092	13.2	Effect of Eve	ent of Default	88 90	<u>)</u>	
SECTION 14AGENCY.909214.1Appointment and Authorization909214.2[Reserved]909214.3Delegation of Duties9092	13.3	Credit Biddin	ng	88 91	_	
14.1Appointment and Authorization909214.2[Reserved]909214.3Delegation of Duties9092	13.4	Curative Equ	uity.	89<u>91</u>		
14.1Appointment and Authorization909214.2[Reserved]909214.3Delegation of Duties9092	SECTION 14	AGENCY		90 92		
14.2 [Reserved] 9092 14.3 Delegation of Duties 9092					0000	
14.3 Delegation of Duties 9092	14 1	Appointment	t and Authorization		90 92	
			t and Authorization		90<u>92</u>	
14.4 Exculnation 9002	14.2	[Reserved]		90<u>92</u>	_	
•	14.2 14.3	[Reserved] Delegation o		90<u>92</u> 90<u>92</u>	_	
	14.2 14.3 14.4	[Reserved] Delegation o Exculpation	f Duties	90 <u>92</u> 90 <u>92</u> 90 <u>92</u>	_	
	14.2 14.3	[Reserved] Delegation o	f Duties	90 <u>92</u> 90 <u>92</u> 90 <u>92</u>	_	

iv

14.6	Notice of Default	91 93
14.7	Credit Decision	91 93
14.8	Indemnification	91 94
14.9	Administrative Agent in Individual Capacitie	es 92 94
14.10	Successor Administrative Agent	92 94
14.11	Collateral Matters	92 95
14.12	Restriction on Actions by Lenders	93 95
14.13	Administrative Agent May File Proofs of	
14.14	Other Agents; Arrangers and Managers	94 96
14.15	Protective Advances	94 96
SECTION 15	GENERAL.	94<u>97</u>
15.1	Waiver; Amendments.	94<u>97</u>
15.2	Confirmations	96 98
15.3		96<u>98</u>
	15.3.1 Generally	96 98
	15.3.2 Electronic Communications	96 98
15.4	Computations	96 99
15.5	Costs and Expenses	97<u>99</u>
15.6	Assignments; Participations.	97<u>99</u>
	15.6.1 Assignments.	97<u>99</u>
	15.6.2 Participations	98<u>100</u>
15.7	Register	99<u>101</u>
15.8	GOVERNING LAW	99<u>102</u>
15.9	Confidentiality	99<u>102</u>
15.10	Severability	100 102
15.11	Nature of Remedies	100 103
15.12	Entire Agreement	<u>100103</u>
15.13	Counterparts	101<u>103</u>
15.14	Successors and Assigns	101 <u>103</u>
15.15	Captions	101<u>103</u>
15.16	Customer Identification—USA Patriot A	Act Notice <u>101103</u>
15.17	INDEMNIFICATION BY LOAN PARTIES	101 <u>103</u>
15.18	Non-Liability of Lenders.	102 104
15.19	Forum Selection and Consent to Juri	SDICTION 102104
15.20	WAIVER OF JURY TRIAL	102<u>105</u>
15.21	Acknowledgement and Consent to Bail-	In of EEA Financial Institutions 103105
	-	

Page

v

ANNEXES

Lenders and Pro Rata Shares Addresses for Notices

SCHEDULES

SCHEDULE 1.1	Enterprise Value
SCHEDULE 9.6	Litigation and Contingent Liabilities
SCHEDULE 9.8	Equity Ownership
SCHEDULE 9.11(d)	Compliance with Laws; Consents; Certain Actions
SCHEDULE 9.16	Insurance
SCHEDULE 9.17	Real Property
SCHEDULE 9.19	Deposit and Securities Accounts
SCHEDULE 9.21	Intellectual Property
SCHEDULE 9.25	Loan Party Information
SCHEDULE 9.26	Locations of Collateral
SCHEDULE 11.1	Existing Debt
SCHEDULE 11.2	Existing Liens
SCHEDULE 11.6	Transactions with Affiliates
SCHEDULE 11.7	Inconsistent Agreements
SCHEDULE 11.9	Investments
SCHEDULE 12.1	Debt to be Repaid
	EXHIBITS
EXHIBIT A	Form of Note (Section 3.1)
EXHIBIT B	Form of Compliance Certificate (Section 10.1.4)
EXHIBIT C	Form of Assignment Agreement (Section 15.6.1)
EXHIBIT D	Form of Notice of Borrowing (Section 2.2.2)
EXHIBIT E	Form of Notice of Conversion (Section 2.2.3)
EXHIBIT F-1	Form of U.S. Tax Compliance Certificate)
EXHIBIT F-2	Form of U.S. Tax Compliance Certificate)
EXHIBIT F-3	Form of U.S. Tax Compliance Certificate)
EXHIBIT F-4	Form of U.S. Tax Compliance Certificate)
EXHIBIT G	Form of Intercompany Subordination Agreement

ANNEX A

ANNEX B

vi

CREDIT AGREEMENT

THIS CREDIT AGREEMENT dated as of March 24, 2022 (this "<u>Agreement</u>") is entered into among MONEYLION TECHNOLOGIES INC., a Delaware corporation (the "<u>Company</u>" or "<u>Borrower</u>"); the financial institutions that are or may from time to time become parties hereto (together with their respective successors and permitted assigns, the "<u>Lenders</u>"); and MONROE CAPITAL MANAGEMENT ADVISORS, LLC, a Delaware limited liability company ("<u>Monroe Capital</u>"), as administrative agent for the Lenders.

The Company has requested that the Lenders make Loans (a) to provide the funds required to repay the Debt to be Repaid, (b) to provide for the ongoing general corporate purposes and working capital needs of Borrower and its Subsidiaries, (c) to consummate any other transaction not prohibited by this Agreement (including repayment of Term A-2 Loans with Term B Loans or Incremental Loans) and (d) to fund fees and expenses associated with the foregoing, as further provided in this Agreement, in an aggregate principal amount of \$110,000,000 in the form of (a) Term A-1 Loans to Borrower on the Closing Date in an aggregate principal amount of \$70,000,000, (b) Term A-2 Loans to Borrower on the Closing Date in an aggregate principal amount of \$70,000,000 and (c) Term B Loans to Borrower during the Term B Loan Availability Period in an aggregate principal amount of \$20,000,000, and the Lenders are willing to do so on the terms and conditions set forth in this Agreement.

To secure the Loans and other Obligations, Borrower and the other Loan Parties are granting to Administrative Agent, for the benefit of Administrative Agent and Lenders, a security interest in and lien upon substantially all of Borrower's and the other Loan Parties' real and personal property (subject to the limitations and exclusions set forth in the Loan Documents).

In consideration of the mutual agreements contained in this Agreement, the parties hereby agree as follows:

SECTION 1 DEFINITIONS.

1.1 Definitions. When used in this Agreement the following terms have the following meanings:

"Account Debtor" is used as defined in the Guaranty and Collateral Agreement.

"Acquisition" means any transaction or series of related transactions for the purpose of or resulting, directly or indirectly, in (a) the acquisition of all or substantially all of the assets of a Person, or of all or substantially all of any business or division of a Person, (b) the acquisition of all or a majority of the Equity Interests of any Person, or otherwise causing any Person to become a Subsidiary, or (c) a merger or consolidation or any other combination with another Person (other than a Person that is already a Subsidiary).

"Adjusted Revenue" means, with respect to the Company and its Subsidiaries for any period, the "Total Revenues, net" of the Company and its Subsidiaries for that period determined in accordance with GAAP, as adjusted by amortization of loan origination costs, provisions for loss on receivables related to membership and fee receivables, and revenue derived from products that have been phased out and non-operating income, in each case, determined in a manner consistent with the Company's financial reporting practices in effect as of the Closing Date.

"Adjusted Term SOFR" means, for purposes of any calculation, the rate per annum equal to the sum of (a) Term SOFR for such calculation plus (b) the Term SOFR Adjustment; provided that if Adjusted

Term SOFR as so determined shall ever be less than the Floor, then Adjusted Term SOFR shall be deemed to be the Floor.

"<u>Administrative Agent</u>" means Monroe Capital in its capacity as administrative agent for the Lenders under this Agreement and any successor thereto in that capacity appointed in accordance with <u>Section 14.10</u>.

"Affected Loan" is defined in Section 8.3.

"Affiliate" of any Person means (a) any other Person which, directly or indirectly, controls or is controlled by or is under common control with that Person, (b) solely for purposes of Sections 9.16, 9.30, 11.3, and 11.6 of this Agreement, any officer or director of that Person and (c) with respect to any Lender, any entity administered or managed by that Lender or an Affiliate or investment advisor thereof and which is engaged in making, purchasing, holding or otherwise investing in commercial loans. A Person will be deemed to be "controlled by" any other Person if that other Person possesses, directly or indirectly, power to vote 10% or more of the securities (on a fully diluted basis) having ordinary voting power for the election of directors or managers or power to direct or cause the direction of the management and policies of that Person whether by contract or otherwise. Unless expressly stated otherwise in this Agreement, none of the following Persons will be deemed an Affiliate of any Loan Party: (i) Administrative Agent; or (ii) any Lender.

"Agent Fee Letter" means the fee letter dated as of the date of this Agreement between Borrower and Administrative Agent, as amended and restated as of April 28, 2023.

"Agreement" is defined in the introductory clause of this Agreement.

"Amendment No. 2" means the Amendment No. 2 to Credit Agreement, dated as of April 28, 2023, among the Borrower, the Lenders and the Administrative Agent.

"Amendment No. 2 Effective Date" has the meaning specified therefor in Section 4 of Amendment No. 2, which date is April 28, 2023.

"Amendment No. 3" means the Amendment No. 3 to Credit Agreement, dated as of June 30, 2024, among the Borrower, the Lenders and the Administrative Agent.

"Amendment No. 3 Effective Date" has the meaning specified therefor in Section 4 of Amendment No. 3, which date is June 30, 2024.

"<u>Applicable Margin</u>" means, as of any date of determination, the applicable rate per annum set forth in the following table that corresponds to the "Enterprise Value" calculation as set forth in (or otherwise furnished or delivered contemporaneously with) the most recent Compliance Certificate delivered to Administrative Agent pursuant to Section 10.1.4. At any time prior to the EBITDA Trigger Date, the rate per annum in the row styled "Level IV" shall not apply.

Level	Enterprise Value	Base Rate Loans	SOFR Loans
Ι	Enterprise Value less than or equal to \$300,000,000	8.25%	9.25%
Π	Enterprise Value greater than \$300,000,000 and less than or equal to \$800,000,000	7.50%	8.50%

<u>2</u>

Level	Enterprise Value	Base Rate Loans	SOFR Loans
III	Enterprise Value greater than \$800,000,000	6.75%	7.75%
IV	Enterprise Value greater than \$800,000,000 and, from and after the EBITDA Trigger Date, Total Debt to EBITDA Ratio less than or equal to 4.00:1.00	6.00%	7.00%

The Enterprise Value shall be calculated as of the end of each Fiscal Quarter of the Company and its Subsidiaries (the "EV Calculation"), and such EV Calculation shall be set forth in or otherwise furnished to the Administrative Agent contemporaneously with the furnishing of the Compliance Certificate pursuant to Section 10.1.4. The Applicable Margin will be based upon the most recent EV Calculation furnished as set forth above and will take effect on the first day of the month following the date such EV Calculation is furnished and will be re-determined quarterly. If Company fails to furnish an EV Calculation when due as set forth above (or fails to furnish a Compliance Certificate when due), then the Applicable Margin will be the rate per annum in the row styled "Level I" as of the first day of the month following the date on which the EV Calculation (or Compliance Certificate) was required to be delivered until the date on which that EV Calculation or Compliance Certificate is delivered, on which date (but not retroactively), without constituting a waiver of any Default or Event of Default occasioned by the failure to timely deliver that EV Calculation or Compliance Certificate, the Applicable Margin will be set at the rate per annum based upon the calculations disclosed by that EV Calculation or Compliance Certificate. If any information contained in any EV Calculation or Compliance Certificate delivered pursuant to the foregoing or Section 10.1.4 is shown to be inaccurate, and that inaccuracy, if corrected, would have led to the application of a higher Applicable Margin for any period than the Applicable Margin actually applied for that period, then (i) Borrower shall promptly upon a responsible officer becoming aware thereof deliver or cause to be delivered to Administrative Agent and each Lender a correct EV Calculation and/or Compliance Certificate for that period; (ii) the Applicable Margin will be determined as if the correct Applicable Margin (as set forth in the table above) were applicable for that period (irrespective of whether a correct EV Calculation and/or Compliance Certificate is delivered); and (iii) Borrower shall promptly (but in any event within five Business Days after delivery of that corrected EV Calculation and/or Compliance Certificate or after demand by Administrative Agent if such corrected EV Calculation and/or Compliance Certificate is not timely delivered) deliver to Administrative Agent full payment in respect of the accrued additional interest as a result of the increased Applicable Margin for that period, which payment Administrative Agent shall promptly apply to the affected Obligations.

"Applicable Premium" is defined in Section 5.2.

"Applicable Premium Trigger Event" means:

(a) any prepayment by any Loan Party of all, or any part, of the principal balance of any Term Loan for any reason (including, but not limited to, any optional prepayment or mandatory prepayment, and distribution in respect thereof, and any refinancing thereof (but for the avoidance of doubt, not including the \$5,000,000 amortization payment in respect of Term A-1 Loan to be made pursuant to Section 6.3(a)(i) or the \$5,000,000 and \$10,000,000 amortization payments in respect of Term A-2 Loan to be made on May 1, 2023 and July 15, 2023, respectively, pursuant to Section 6.3(b)), whether in whole or in part, and whether before or after (i) the occurrence of an Event of Default, or (ii) the commencement of any Insolvency Proceeding, and notwithstanding any acceleration (for any reason) of the Obligations; provided that any payments required to be made pursuant to Section 6.1.2(a)(ii) and Section 6.1.2(a)(v) of this Agreement shall not constitute an Applicable Premium Trigger Event;

<u>3</u>

(b) the acceleration of the Obligations for any reason, including, but not limited to, acceleration in accordance with Section 13.2 of the Credit Agreement, including as a result of the commencement of an Insolvency Proceeding;

(c) the satisfaction, release, payment, restructuring, reorganization, replacement, reinstatement, defeasance or compromise of any of the Obligations in any Insolvency Proceeding or any foreclosure (whether by power of judicial proceeding or otherwise) or deed in lieu of foreclosure or the making of a distribution of any kind in any Insolvency Proceeding to Administrative Agent, for the account of the Lenders in full or partial satisfaction of the Obligations; or

(d) the termination of the Credit Agreement for any reason.

Solely for purposes of calculating the term Applicable Premium, if an Applicable Premium Trigger Event occurs under clause (b), (c) or (d), the entire then outstanding principal amount of the Term Loan shall be deemed to have been prepaid on the date on which such Applicable Premium Trigger Event occurs.

"<u>Approved Fund</u>" means (a) any Person (other than a natural person) engaged in making, purchasing, holding, or investing in commercial loans and similar extensions of credit and that is advised, administered, or managed by a Lender, an Affiliate of a Lender (or an entity or an Affiliate of an entity that administers, advises or manages a Lender); (b) with respect to any Lender that is an investment fund, any other investment fund that invests in loans and that is advised, administered or managed by the same investment advisor as that Lender or by an Affiliate of that investment advisor; and (c) any third party that provides "warehouse financing" to a Person described in <u>clause (a) or (b)</u> (and any Person described in <u>clause (a) or (b)</u> will also be deemed an Approved Fund with respect to any such third party providing warehouse financing); <u>provided</u> that in no event shall a Person that constitutes a Disqualified Institution constitute an Approved Fund.

"<u>Asset Disposition</u>" means the sale, lease, assignment, disposition, conveyance or other transfer for value by any Loan Party to any Person of any asset of that Loan Party (including, the loss, destruction or damage of any thereof or any actual or threatened (in writing to any Loan Party) condemnation, confiscation, requisition, seizure or taking thereof). For the avoidance of doubt, none of (x) the sale of any Permitted Convertible Debt by Parent, (y) the sale of any Permitted Warrant Transaction by Parent and (z) the performance by Parent of its obligations under any Permitted Convertible Debt or any Permitted Warrant Transaction shall constitute an Asset Disposition.

"Assignee" is defined in Section 15.6.1.

"Assignment Agreement" is defined in Section 15.6.1.

"Attorney Costs" means, with respect to any Person, all reasonable and documented out-of-pocket invoiced fees and charges of any counsel (excluding in -house counsel) to that Person and all court costs and similar legal expenses.

"Available Tenor" means, as of any date of determination and with respect to the then-current Benchmark, as applicable, (x) if such Benchmark is a term rate, any tenor for such Benchmark (or component thereof) that is or may be used for determining the length of an interest period pursuant to this Agreement or (y) otherwise, any payment period for interest calculated with reference to such Benchmark (or component thereof) that is or may be used for determining any frequency of making payments of interest calculated with reference to such Benchmark pursuant to this Agreement, in each case, as of such date and not including, for the avoidance of doubt, any tenor for such Benchmark that is then-removed from the definition of "Interest Period" pursuant to <u>Section 8.7.4</u>.

<u>4</u>

"Bail-In Action" means the exercise of any Write-Down and Conversion Powers by the applicable EEA Resolution Authority in respect of any liability of an EEA Financial Institution.

"Bail-In Legislation" means, with respect to any EEA Member Country implementing Article 55 of Directive 2014/59/EU of the European Parliament and of the Council of the European Union, the implementing law for such EEA Member Country from time to time which is described in the EU Bail-In Legislation Schedule.

"Bankruptcy Code" means the Federal Bankruptcy Reform Act of 1978 (11 U.S.C. § 101 et seq.) and the regulations issued from time to time thereunder.

"Baseline Average Weekly Volume" means, with respect to any Baseline Test Period, the average Instacash Weekly Purchased Amount during such Baseline Test Period.

"Baseline Test Period" means, with reference to the occurrence of any Regulatory Trigger Event or Event of Termination under the SP Forward Flow Agreement, the period of six full calendar weeks most recently ended immediately prior to the occurrence of such Regulatory Trigger Event or Event of Termination.

"Base Rate" means at any time a fluctuating rate per annum equal to the greatest of (a) the Federal Funds Rate plus 0.50%, (b) the Prime Rate, (c) 2.00%, and (d) the Adjusted Term SOFR for a one-month tenor in effect on such day plus 1.00 %.

"Base Rate Loan" means any Loan which bears interest at or by reference to the Base Rate.

"Base Rate Term SOFR Determination Day" has the meaning specified in the definition of "Term SOFR".

"Benchmark" means, initially, the Term SOFR Reference Rate; provided that if a Benchmark Transition Event has occurred with respect to the Term SOFR Reference Rate or the then-current Benchmark, then "Benchmark" means the applicable Benchmark Replacement to the extent that such Benchmark Replacement has replaced such prior benchmark rate pursuant to Section 8.7.1.

"Benchmark Replacement" means, with respect to any Benchmark Transition Event, the sum of: (a) the alternate benchmark rate that has been selected by Administrative Agent and Borrower giving due consideration to (i) any selection or recommendation of a replacement benchmark rate or the mechanism for determining such a rate by the Relevant Governmental Body or (ii) any evolving or then-prevailing market convention for determining a benchmark rate as a replacement to the then-current Benchmark for Dollar-denominated syndicated credit facilities at such time and (b) the related Benchmark Replacement Adjustment; provided, that if the Benchmark Replacement as determined pursuant to clause (a) or (b) above would be less than the Floor, the Benchmark Replacement will be deemed to be a rate per annum equal to the Floor for the purposes of this Agreement and the other Loan Documents.

"Benchmark Replacement Adjustment" means, with respect to any replacement of the then-current Benchmark with an Unadjusted Benchmark Replacement, the spread adjustment, or method for calculating or determining such spread adjustment, (which may be a positive or negative value or zero) that has been selected by Administrative Agent and Borrower Representative giving due consideration to (a) any selection or recommendation of a spread adjustment, or method for calculating or determining such spread adjustment of such Benchmark with the applicable Unadjusted Benchmark Replacement by the Relevant Governmental Body or (b) any evolving or then-prevailing market convention for determining a spread adjustment, or method for calculating or determining such spread adjustment, for

<u>5</u>

the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement for Dollar-denominated syndicated credit facilities at such time.

"Benchmark Replacement Date" means the earlier to occur of the following events with respect to the then-current Benchmark:

(a) in the case of clause (a) or (b) of the definition of "Benchmark Transition Event", the later of (i) the date of the public statement or publication of information referenced therein and (ii) the date on which the administrator of such Benchmark (or the published component used in the calculation thereof) permanently or indefinitely ceases to provide all Available Tenors of such Benchmark (or such component thereof); or

(b) in the case of clause (c) of the definition of "Benchmark Transition Event", the first date on which such Benchmark (or the published component used in the calculation thereof) has been determined and announced by the regulatory supervisor for the administrator of such Benchmark (or such component thereof) to be non-representative; provided that such non-representativeness, non-compliance or non-alignment will be determined by reference to the most recent statement or publication referenced in such clause (c) and even if any Available Tenor of such Benchmark (or such component thereof) continues to be provided on such date.

For the avoidance of doubt, the "Benchmark Replacement Date" will be deemed to have occurred in the case of clause (a) or (b) with respect to any Benchmark upon the occurrence of the applicable event or events set forth therein with respect to all then-current Available Tenors of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Transition Event" means the occurrence of one or more of the following events with respect to the then-current Benchmark:

(a) a public statement or publication of information by or on behalf of the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that such administrator has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof), permanently or indefinitely; provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof);

(b) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof), the Federal Reserve Board, the Federal Reserve Bank of New York, an insolvency official with jurisdiction over the administrator for such Benchmark (or such component), a resolution authority with jurisdiction over the administrator for such Benchmark (or such component) or a court or an entity with similar insolvency or resolution authority over the administrator for such Benchmark (or such component) has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof) permanently or indefinitely; provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof); or

(c) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that all Available Tenors of such Benchmark (or such component thereof) are not, or as of a specified future date will not be, representative.

<u>6</u>

For the avoidance of doubt, a "Benchmark Transition Event" will be deemed to have occurred with respect to any Benchmark if a public statement or publication of information set forth above has occurred with respect to each then-current Available Tenor of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Transition Start Date" means, in the case of a Benchmark Transition Event, the earlier of (a) the applicable Benchmark Replacement Date and (b) if such Benchmark Transition Event is a public statement or publication of information of a prospective event, the 90th day prior to the expected date of such event as of such public statement or publication (or if the expected date of such prospective event is fewer than 90 days after such statement or publication, the date of such statement or publication).

"Benchmark Unavailability Period" means, the period (if any) (a) beginning at the time that a Benchmark Replacement Date has occurred if, at such time, no Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Loan Document in accordance with Section 8.7 and (b) ending at the time that a Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Loan Document in accordance with Section 8.7.

"Borrower" is defined in the introductory clause of this Agreement.

"Borrowing" means a borrowing consisting of simultaneous Loans of the same Type and, in the case of a SOFR Borrowing, having the same Interest Period made by the Lenders.

"Business Day" means any day other than a Saturday, Sunday or other day on which commercial banks are authorized to close under the laws of, or are in fact closed in, New York

"<u>Capital Expenditures</u>" means all expenditures that, in accordance with GAAP, would be required to be capitalized and shown on the consolidated balance sheet of the Company and its Subsidiaries, including expenditures in respect of Capital Leases, but excluding any such expenditures made in connection with the replacement, substitution, or restoration of assets to the extent financed (i) from insurance proceeds (or other similar recoveries) paid on account of the loss of or damage to the assets being replaced or restored, (ii) with awards of compensation arising from the taking by eminent domain or condemnation of the assets being replaced, (iii) with assets traded or exchanged for that replacement, substitution, or restoration of assets, or (iv) with Net Cash Proceeds from a sale, lease, assignment, disposition, or other transfer for value of the type specifically described in clause (a) of the definition of "Permitted Asset Disposition."

"<u>Capital Lease</u>" means, with respect to any Person, any lease of (or other agreement conveying the right to use) any real or personal property by that Person that, in conformity with GAAP, is accounted for as a capital lease on the balance sheet of that Person.

"<u>Cash Collateralize</u>" means, with respect to any inchoate, contingent, or other Obligations, the delivery of cash to Administrative Agent, as security for the payment of those Obligations, in an amount equal to Administrative Agent's good faith estimate of the amount due or to become due, including all fees and other amounts relating to those Obligations. "Cash Collateralization" has a correlative meaning.

"Cash Equivalent Investment" means, at any time, (a) any evidence of Debt, maturing not more than one year after that time, issued or guaranteed by the United States Government or any agency thereof; (b) commercial paper, maturing not more than one year from the date of issue, or corporate demand notes, in each case (unless issued by a Lender or its holding company) rated at least A-l by Standard & Poor's Ratings Services, a division of The McGraw-Hill Companies, Inc. or P-l by Moody's Investors Service,

Inc.; (c) any certificate of deposit, time deposit, or banker's acceptance, maturing not more than one year after that time, or any overnight Federal Funds transaction that is issued or sold by any Lender or its holding company (or by a commercial banking institution that is a member of the Federal Reserve System and has a combined capital and surplus and undivided profits of not less than \$500,000,000); (d) any repurchase agreement entered into with any Lender (or commercial banking institution of the nature referred to in clause (c)) which (i) is secured by a fully perfected security interest in any obligation of the type described in any of clauses (a) through (c) above and (ii) has a market value at the time that repurchase agreement is entered into of not less than 100% of the repurchase obligation of that Lender (or other commercial banking institution) thereunder; (e) money market accounts or mutual funds which invest exclusively in assets satisfying the foregoing requirements; and (f) other short-term liquid investments approved in writing by Administrative Agent (such approval not to be unreasonably withheld, delayed or conditioned).

"<u>Cash-in-Transit</u>" means, as of any date of determination, the aggregate amount of receivables of the Company and its Subsidiaries from payment processors (for debit card collections and other sources).

"<u>Change in Law</u>" means the adoption or phase-in of, or any change in, in each case after the date of this Agreement, any applicable law, rule, or regulation, or any change in the interpretation or administration of any applicable law, rule, or regulation by any governmental authority, central bank, or comparable agency charged with the interpretation or administration thereof, or compliance by any Lender with any request or directive (whether or not having the force of law) of any such authority, central bank, or comparable agency. For purposes of this Agreement, the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines, or directives thereunder or issued in connection therewith, and all requests, rules, guidelines, or directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States regulatory authorities, in each case pursuant to Basel III, will, in each case, be deemed to have been adopted and gone into effect after the date of this Agreement.

"<u>Change of Control</u>" means the occurrence of any of the following events: (a) the acquisition, directly or indirectly, by any person or group (within the meaning of Section 13(d)(3) of the Exchange Act) of beneficial ownership of more than 30% of the aggregate outstanding voting or economic power of the Equity Interests of Parent; or (b) except to the extent otherwise expressly permitted by the terms of this Agreement (including without limitation, pursuant to Section 11.4), Parent ceases to, directly or indirectly, own and control 100% of the outstanding Equity Interests of Borrower.

"Closing Date" is defined in Section 12.1.

"Code" means the Internal Revenue Code of 1986.

"Collateral" means "Collateral" (as defined in the Guaranty and Collateral Agreement) and any and all other property now or hereafter securing Obligations.

"<u>Collateral Access Agreement</u>" means an agreement in form and substance reasonably satisfactory to Administrative Agent pursuant to which a lessor of real property on which collateral or books or records are stored or otherwise located, or a warehouseman, processor, or other bailee of inventory or other property owned by any Loan Party, acknowledges the Liens of Administrative Agent, waives or subordinates any Liens held by that Person on Collateral held at that property, and, in the case of any such agreement with a lessor, permits Administrative Agent reasonable access to and use of the applicable real property following the occurrence and during the continuance of an Event of Default to assemble, complete and remove any Collateral stored or otherwise located on that real property.

<u>8</u>

"Collateral and Diligence Questionnaire" means a collateral and diligence questionnaire executed and delivered to Administrative Agent by a Loan Party.

"<u>Collateral Documents</u>" means, collectively, the Guaranty and Collateral Agreement, each Mortgage, each Mortgage-Related Document, each Collateral Access Agreement, each Control Agreement, each Intellectual Property Security Agreement, and any other agreement or instrument pursuant to which Borrower, any Subsidiary or any other Person grants or purports to grant Collateral to Administrative Agent for the benefit of Administrative Agent and the Lenders or otherwise relates to any such Collateral.

"<u>Commitment</u>" means, as to any Lender, that Lender's commitment to make Loans under this Agreement. The initial amount of each Lender's Commitment is set forth on Annex A.

"Competitor" means any company or business that operates a personal finance platform including mobile applications designed to help users simplify personal financial management.

"Compliance Certificate" means a Compliance Certificate in substantially the form of Exhibit B.

"Computation Period" means each period of four consecutive Fiscal Quarters ending on the last day of a Fiscal Quarter.

"<u>Conforming Changes</u>" means, with respect to either the use or administration of Term SOFR or the use, administration, adoption or implementation of any Benchmark Replacement, any technical, administrative or operational changes (including changes to the definition of "Base Rate," the definition of "Business Day," the definition of "U.S. Government Securities Business Day," the definition of "Interest Period" or any similar or analogous definition (or the addition of a concept of "interest period"), timing and frequency of determining rates and making payments of interest, timing of borrowing requests or prepayment, conversion or continuation notices, the applicability and length of lookback periods and other technical, administrative or operational matters) that Administrative Agent decides may be appropriate to reflect the adoption and implementation of any such rate or to permit the use and administration thereof by Administrative Agent in a manner substantially consistent with market practice (or, if Administrative Agent decides that adoption of any portion of such market practice is not administratively feasible or if Administrative Agent determines that no market practice for the administration of any such rate exists, in such other manner of administration as Administrative Agent decides is reasonably necessary in connection with the administration of this Agreement and the other Loan Documents).

"<u>Connection Income Taxes</u>" means Other Connection Taxes that are imposed on or measured by net income (however denominated) or that are franchise Taxes or branch profits Taxes.

"<u>Consolidated Net Income</u>" means, with respect to the Company and its Subsidiaries for any period, in each case as determined in accordance with GAAP, the consolidated net income (or loss) of the Company and its Subsidiaries for that period, excluding (a) any gains from Asset Dispositions; (b) any extraordinary gains; (c) the income (or loss) of any Subsidiary during that period in which any other Person (other than the Company or a Subsidiary) has a joint interest and contractual provisions actually restrict the amount of dividends or other distributions payable in cash to the Company or any of its Subsidiaries during such period; (d) the income (or loss) of any Person during that period and accrued prior to the date it becomes a Subsidiary of the Company or is merged into or consolidated with a Loan Party or that Person's assets are acquired by a Loan Party, except to the extent that the declaration or payment of dividends or similar distributions by that Subsidiary to the Company or any of its Subsidiaries of that period is subsidiary to the Company or any of the Subsidiary dividends or similar distributions permitted or required by the terms of this Agreement; (e) the income of any Subsidiary to the extent that the declaration or payment of dividends or similar distributions by that Subsidiary to the Company or any of its Subsidiaries of that

<u>9</u>

income is not at the time permitted by operation of the terms of its organizational documents, its governing documents, or any agreement, instrument, judgment, decree, order, statute, rule; or governmental regulation applicable to that Subsidiary, except to the extent of the amount of cash dividends or other distributions actually paid in cash to the Company or any Subsidiary during such period; and (f) any gains (or losses) from discontinued operations.

"Contingent Liability" means, with respect to any Person, each obligation and liability of that Person and all such obligations and liabilities of that Person incurred pursuant to any agreement, undertaking or arrangement by which that Person: (a) guarantees, endorses or otherwise becomes or is contingently liable upon (by direct or indirect agreement, contingent or otherwise, to provide funds for payment, to supply funds to, or otherwise to invest in, a debtor, or otherwise to assure a creditor against loss) the indebtedness, dividend, obligation or other liability of any other Person in any manner (other than by endorsement of instruments in the course of collection), including any indebtedness, dividend or other obligation which may be issued or incurred at some future time; (b) guarantees the payment of dividends or other distributions upon the Equity Interests of any other Person; (c) undertakes or agrees (whether contingently or otherwise): (i) to purchase, repurchase, or otherwise acquire any indebtedness, obligation or liability of any other Person or any property or assets constituting security therefor, (ii) to advance or provide funds for the payment or discharge of any indebtedness, obligation or liability of any other Person (whether in the form of loans, advances, stock purchases, capital contributions, or otherwise), or to maintain solvency, assets, level of income, working capital, or other financial condition of any other Person, or (iii) to make payment to any other Person other than for value received; (d) agrees to lease property or to purchase securities, property, or services from any other Person with the purpose or intent of assuring the owner of that indebtedness or obligation of the ability of that other Person; or (f) undertakes or agrees obligation; (e) to induce the issuance of, or in connection with the issuance of, any letter of credit for the benefit of any other Person; or (f) undertakes or agrees otherwise to assure a creditor against loss. The amount of any Contingent Liabilit

"<u>Control Agreement</u>" means each deposit account control agreement or securities account control agreement, as applicable, entered into by a Loan Party, each depository institution or securities intermediary party thereto and Administrative Agent in form and substance reasonably satisfactory to Administrative Agent.

"<u>Controlled Group</u>" means all members of a controlled group of corporations, all members of a controlled group of trades or businesses (whether or not incorporated) under common control and all members of an affiliated service group which, together with Borrower or any Subsidiary of Borrower, are treated as a single employer under Section 414 of the Code or Section 4001(b) of ERISA.

"Credit Facilities" means the credit facilities provided under this Agreement and the other Loan Documents.

"<u>Curative Equity</u>" means the making of capital contributions to the Company or the issuance of Equity Interests by the Company (other than Disqualified Equity Interests), for the purposes of, and in accordance with, Section 13.4.

"Debt" of any Person at a particular time means, without duplication, (a) all indebtedness of that Person for borrowed money; (b) all indebtedness evidenced by bonds, debentures, notes or similar instruments; (c) all obligations of that Person as lessee under Capital Leases which have been or should be recorded as liabilities on a balance sheet (excluding the footnotes thereto) of that Person in accordance with GAAP; (d) all obligations of that Person to pay the deferred purchase price of property or services (excluding (x) trade accounts payable in the ordinary course of business and (y) any earn out obligation,

purchase price adjustment or other similar obligation until such obligation (A) becomes a liability on the balance sheet of such Person (excluding the footnotes thereto) in accordance with GAAP and (B) has not been paid within 60 days after becoming due and payable following expiration of any dispute resolution mechanics set forth in the applicable agreement governing the applicable transaction (it being understood and agreed that, in addition to the foregoing limitations, earn-out obligations shall only constitute Debt to the extent payable in cash)); (e) all indebtedness secured by a Lien on the property of that Person, whether or not that indebtedness has been assumed by that Person, but if that Person has not assumed or otherwise become liable for that indebtedness, then the amount of that indebtedness will be deemed to be the lesser of (i) the amount of such indebtedness and (ii) at the fair market value of the property securing that indebtedness at the time of determination; (f) all obligations, contingent or otherwise, with respect to the face amount of all letters of credit (whether or not drawn), bankers' acceptances, and similar obligations issued for the account of that Person; (g) all Hedging Obligations of that Person; (h) all Contingent Liabilities of that Person; (i) all Debt of any partnership of which that Person is a general partner unless such Debt is expressly made non-recourse to such Person; (j) (x) all monetary obligations under any receivables factoring, receivables sales, or similar transactions (including any warehouse financing) and (y) all monetary obligations (to the extent such obligations become payable in accordance with GAAP) under any synthetic lease, tax ownership/operating lease, off-balance sheet financing, or similar financing; and (k) any Disqualified Equity Interests of that Person and any other equity interest or instrument of that Person, whether or not mandatorily redeemable, that under GAAP is characterized as debt, whether pursuant to financial accounting standards b

"Debt to be Repaid" means the Debt listed on Schedule 12.1.

"Default" means any event that, if it continues uncured, will, with lapse of applicable cure or grace periods or notice or both, constitute an Event of Default.

"Defaulting Lender" means any Lender that (a) has failed to fund any portion of the Loans or participations in Letters of Credit required to be funded by it under this Agreement within two (2) Business Days of the date required to be funded by it under this Agreement; (b) has otherwise failed to pay over to Administrative Agent or any other Lender any other amount required to be paid by it under this Agreement within two (2) Business Days of the date when due, unless the subject of a good faith dispute; (c) has or has a direct or indirect parent company that has (i) been deemed insolvent or become the subject of an Insolvency Proceeding or (ii) become the subject of a Bail-In Action; (d) has notified any Borrower, Administrative Agent or any Lender that it does not intend to comply with any of its funding obligations under this Agreement or has made a public statement to the effect that it does not intend to comply with its funding obligations under this Agreement or under other agreements in which it commits to extend credit or (e) has failed to confirm within three (3) Business Days of a request by Administrative Agent (including any request by Borrower that such request be made) that it will comply with the terms of this Agreement relating to its obligations to fund Loans and participations in Letters of Credit.

"Disqualified Equity Interest" means any Equity Interest that, by its terms (or by the terms of any security or other Equity Interest into which it is convertible or for which it is exchangeable), or upon the happening of any event or condition, in each case before the date that is one hundred eighty (180) days after the Termination Date, (a) matures or is mandatorily redeemable (other than solely for Qualified Equity Interests), pursuant to a sinking-fund obligation or otherwise (except as a result of a change of control or asset sale so long as any rights of the holders thereof upon the occurrence of a change of control or asset sale event are subject to the prior repayment in full of the Loans and all other Obligations that are accrued and payable and the termination of the Commitments); (b) is redeemable at the option of the holder thereof (other than solely for Qualified Equity Interests), in whole or in part; (c) provides for the scheduled payments of dividends in cash before the date that is one hundred eighty (180) days after the Termination

<u>11</u>

Date; or (d) is or becomes convertible into or exchangeable for Debt or any other Equity Interests that would constitute Disqualified Equity Interests.

"Disqualified Institution" means (i)(x) Persons that are Competitors of any of the Loan Parties and (y) those banks, financial institutions and other Persons to the extent identified by the Company to Administrative Agent and each Initial Lender by name in writing prior to the Closing Date, which written list of Disqualified Institutions under clause (y) may be updated from time to time after the Closing Date with the consent of Administrative Agent and each Lender (such consents not to be unreasonably withheld, conditioned or delayed) or (ii) any reasonably identifiable controlled Affiliate of such Persons (provided that neither Administrative Agent nor any Lender shall have any obligation to carry out due diligence in order to identify such controlled Affiliates but shall act in good faith), it being understood that any supplement thereof shall not apply retroactively to disqualify any Persons that have previously acquired an assignment or participation interest in the Loans and/or Commitments as permitted herein. Disqualified Institutions exclude any Person that the Company has designated in writing as no longer being a Disqualified Institution by written notice delivered to Administrative Agent and each Lender from time to time.

"Dollar" and the sign "\$" mean lawful money of the United States of America.

"EBITDA" means, for any period, the result of the following, in each case as determined in accordance with GAAP:

(a) Consolidated Net Income for that period;

<u>plus</u>

(b) to the extent deducted in determining that Consolidated Net Income (other than in respect of (b)(xiv)(i)), without duplication, the sum of the following during that period:

(i) Interest Expense;

(ii) any federal, state, local and foreign income tax expense for that period and, without duplication, Tax Distributions made in (or payable with respect to) that period;

(iii) depreciation and amortization for that period;

(iv) non-cash compensation expense, or other non-cash expenses or charges, for that period arising from the granting of stock options, stock appreciation rights or similar equity arrangements;

(v) non-cash extraordinary or non-cash non-recurring expenses or losses (including non-cash adjustments due to changes in accounting) and other non-cash charges (including changes in fair value of warrants issued by Parent) (in each case as determined in accordance with GAAP) incurred during that period;

(vi) reasonable and documented out-of-pocket fees, costs, and expenses paid during that period in connection with the negotiation, execution, and delivery of this Agreement and the other Loan Documents and the consummation of the transactions contemplated by this Agreement and the other Loan Documents, so long as (A) those fees, costs, and expenses were paid not later than 180 days after the Closing Date, and (B) the aggregate amount of all such fees, costs, and expenses, whenever incurred or paid, does not exceed \$1,800,000.

(vii) (A) (x) Malka Transaction Expenses, (y) Even Transaction Expenses and (z) at the election of Borrower, other transaction fees, costs and expenses incurred in connection with Permitted Acquisitions and Investments, Permitted Asset Dispositions and other transactions not prohibited by this Agreement or any other Loan Documents or Agent Fee Letter (whether or not any such transaction has been consummated) but only to the extent the aggregate amount thereof added back pursuant to this clause (z) does not exceed \$5,000,000 for any Computation Period or \$10,000,000 in the aggregate during the term of this Agreement and (B) other reasonable and customary transaction fees, costs and expenses incurred in connection with issuance of Permitted Convertible Debt and Equity Issuances, in each case, negotiated with non-Affiliates on arm's length customary market terms (as determined by Borrower in good faith);

(viii) (A) extraordinary charges, losses and expenses (determined in accordance with GAAP) and (B) unusual, one-time or non-recurring charges, losses and expenses, but only to the extent the aggregate amount thereof added back pursuant to this clause (viii) does not exceed \$3,500,000 for any Computation Period;

(ix)[reserved];

(x) reasonable and documented out-of-pocket fees, costs, and expenses of the Loan Parties incurred after the Closing Date and paid during that period in connection with the Loan Documents (including any amendment, modification or waiver of terms thereof) but only to the extent the aggregate amount thereof added back pursuant to this clause (x) does not exceed \$500,000 for any Computation Period;

(xi) restructuring and similar charges, including any charge attributable to the undertaking and/or implementation of cost savings initiatives, cost rationalization programs, operating expense reductions and/or synergies (excluding "revenue" synergies, but including, without limitation, in connection with any integration, and/or restructuring or transition), any business optimization or business organization charge, any restructuring charge (including any charge relating to any tax restructuring), and/or any charge relating to the closure or consolidation of any facility and/or discontinued operations (including but not limited to severance, rent termination costs, moving costs and legal costs), any systems implementation charge, any consulting charge, any severance charge and/or any other transaction costs or other costs associated with operational changes or improvements, but only to the extent the aggregate amount thereof pursuant to this clause (xi) does not exceed \$2,500,000 for any Computation Period;

(xii) non-cash charges relating to grants of stock appreciation rights, stock options, restricted stock units or restricted stock, non-cash impairment of goodwill and other long term intangible assets, unrealized non-cash losses (or minus unrealized non-cash gains) under Hedging Agreements or other derivative contracts, unrealized non-cash losses (or minus unrealized non-cash gains) in such period due solely to fluctuations in currency values, but excluding any non-cash loss or expense relating to a write-down, write-off or reserve with respect to accounts receivable or inventory in any Fiscal Year;

(xiii) reasonable and customary fees, costs, and expenses of board of directors (or similar governing body) (including compensation expenses) of the Company for that period incurred in the ordinary course of business;

(xiv) (i) business interruption insurance proceeds and (ii) other amounts paid during such period which have been indemnified or reimbursed in cash by third parties that are not a Loan

Party or a Subsidiary of a Loan Party (or an Affiliate of a Loan Party with respect to expenses related to operating expenses of such Loan Party or Subsidiary);

(xv) the amount of any earn-out payments paid in such period to the extent the earn-out payment is permitted by the terms of this Agreement and made pursuant to and in accordance with documentation that has been delivered to Administrative Agent;

(xvi) one-time costs associated with Public Company Compliance in an aggregate amount not to exceed \$1,000,000 during the term of this Agreement;

<u>plus</u>

(c) proceeds of any Curative Equity made in accordance with Section 13.4 and subject to the limitations therein;

<u>minus</u>

(d) to the extent included in determining Consolidated Net Income, without duplication: non-cash gains or profits during that period;

provided that (x) there shall be included in determining EBITDA for any period, without duplication and to the extent reasonably satisfactory to the Required Lenders, the EBITDA of any Person acquired by the Company or any of its Subsidiaries pursuant to a Permitted Acquisition during such period (but not the EBITDA of any related Person, property, business or assets to the extent not so acquired), to the extent not subsequently sold, transferred or otherwise disposed of by the Company or such Subsidiary during such period and (y) there shall be excluded in determining EBITDA for any period, without duplication and to the extent reasonably satisfactory to the Required Lenders, the EBITDA of any Person sold, transferred or otherwise disposed of by the Company or any of its Subsidiary pursuant to Section 11.4 during such period, in each case, as if such acquisition or disposition occurred on the first day of the relevant measurement period.

Notwithstanding anything to the contrary contained herein, for purposes of calculating EBITDA, for the monthly periods set forth below, EBITDA for such month shall be deemed the respective EBITDA amount set forth below (the "Deemed EBITDA"):

Period	EBITDA
February 2021	(\$1,958,987)
March 2021	\$689,217
April 2021	(\$2,227,366)
May 2021	(\$2,152,412)
June 2021	(\$6,969,332)
July 2021	(\$1,003,636)
August 2021	(\$10,914,497)
September 2021	(\$5,476,686)
October 2021	(\$6,289,667)
November 2021	(\$8,079,961)
December 2021	(\$14,789,862)
January 2022	(\$4,788,940)

<u>14</u>

For the avoidance of doubt, with respect to any addback or adjustment subject to a cap or other limitation, (x) such cap or limitation shall not apply with respect to any addbacks or adjustments included in the Deemed EBITDA and (y) addbacks or adjustments included in Deemed EBITDA shall not be counted as usage of such cap or limitation.

"EBITDA Trigger Date" the first date by which EBITDA for the most recently ended two consecutive Fiscal Quarters is greater than \$0 (whether or not sustained).

"ECF Percentage" means, with respect to the Excess Cash Flow for any Fiscal Year following the EBITDA Trigger Date, the following percentages, as applicable: (a) if the Total Debt to EBITDA Ratio as of the last day of the last Fiscal Quarter of that Fiscal Year is equal to or less than 1.50 to 1.00, 0%; (b) if the Total Debt to EBITDA Ratio as of the last Fiscal Quarter of that Fiscal Year is greater than 1.50 to 1.00 but equal to or less than 3.00 to 1.00, 25.0%; and (c) otherwise, 50.0%.

"EEA Financial Institution" means (a) any credit institution or investment firm established in any EEA Member Country which is subject to the supervision of an EEA Resolution Authority; (b) any entity established in an EEA Member Country which is a parent of an institution described in clause (a) of this definition; or (c) any financial institution established in an EEA Member Country which is a subsidiary of an institution described in clauses (a) or (b) of this definition and is subject to consolidated supervision with its parent;

"EEA Member Country" means any of the member states of the European Union, Iceland, Liechtenstein, and Norway.

"EEA Resolution Authority" means any public administrative authority or any person entrusted with public administrative authority of any EEA Member Country (including any delegee) having responsibility for the resolution of any EEA Financial Institution.

"Eligible Assignee" means any commercial bank, any finance company, any investment fund or other fund that invests in loans, or any Affiliate of any of the foregoing.

"Environmental Claims" means all claims, however asserted, by any governmental, regulatory or judicial authority or other Person alleging potential liability or responsibility for violation of any Environmental Law, or for release of Hazardous Substances or injury to the environment.

"Environmental Laws" means all present or future federal, state or local laws, statutes, common law duties, rules, regulations, ordinances and codes, together with all administrative or judicial orders, consent agreements, directed duties, requests, licenses, authorizations and permits of, and agreements with, any governmental authority, in each case relating to any matter arising out of or relating to public health and safety, or pollution or protection of the environment or workplace, including any of the foregoing relating to the presence, use, production, generation, handling, transport, treatment, storage, disposal, distribution, discharge, emission, release, threatened release, control or cleanup of any Hazardous Substance.

"Enterprise Value" means, as of any date of determination (or, with respect to restricted stock units and options, Funded Debt and cash and Cash Equivalent Investments, as of the last day of the Fiscal Quarter most recently ended prior to such date of determination) (a) Equity Value <u>plus</u> (b) all outstanding Funded Debt of Parent and its Subsidiaries <u>minus</u> (c) cash and Cash Equivalent Investments of Parent and its Subsidiaries consistent with past practice). For illustrative purposes, set forth on Schedule 1.1 are reasonably detailed calculations

<u>15</u>

demonstrating the Enterprise Value as of March 23, 2022, and its components set forth in clauses (a), (b) and (c) above.

"Equity Value" means, as of any date of determination (or, with respect to restricted stock units and options, as of the last day of the Fiscal Quarter most recently ended prior to such date of determination), the product of (a) the number of issued and outstanding Equity Interests of Parent (which for the purpose of this definition shall include all common shares, preferred shares, options and restricted stock units) on a fully diluted basis using the treasury stock method multiplied by (b) the price of a share of Parent's common stock, par value \$0.0001 per share as of the close of business of such date.

"Equity Interests" means, with respect to any Person, all shares, interests, participations or other equivalents (however designated, whether voting or nonvoting) of that Person's equity capital, whether now outstanding or issued or acquired after the Closing Date, including common shares, preferred shares, membership interests in a limited liability company, limited or general partnership interests in a partnership, interests in a trust, interests in other unincorporated organizations, or any other equivalent of any such ownership interest, but excluding, for the avoidance of doubt, any Debt that is convertible into or exchangeable for any of the foregoing.

"Equity Issuance" means either (a) the sale or issuance by any Loan Party or any of its Subsidiaries of any shares of its Equity Interests or (b) the receipt by Borrower of any cash capital contributions.

"ERISA" means the Employee Retirement Income Security Act of 1974.

"EU Bail-In Legislation Schedule" means the EU Bail-In Legislation Schedule published by the Loan Market Association (or any successor person), as in effect from time to time.

"EV Calculation" has the meaning specified in the definition of "Applicable Margin".

"Even Earn-Out" means the earn-out payment payable in either shares of common stock, par value \$0.0001 per share, of Parent or Series A Convertible Preferred Stock, par value \$0.0001 per share, of Parent pursuant to the terms of the Even Acquisition Agreement.

"Even Financial Acquisition" means the acquisition of Even Financial Inc. by MoneyLion Inc., pursuant to that certain Agreement and Plan of Merger dated December 15, 2021, by and among MoneyLion Inc., Epsilon Merger Sub Inc., Even Financial Inc., and Fortis Advisors LLC, solely in its capacity as representative of the equityholders of Even Financial Inc. (the "Even Acquisition Agreement").

"Even Financial Dividend" means dividends paid in cash by MoneyLion Inc. to (i) holders of the Series A Convertible Preferred Stock, par value \$0.0001, of MoneyLion Inc. pursuant to the Certificate of Designation filed immediately prior to, and issued upon, the consummation of the transactions contemplated by the Even Financial Acquisition, as such Certificate of Designation is in effect on the Closing Date and (ii) participants in the MoneyLion Inc. Preferred Share Dividend Replacement Program adopted concurrently with the consummation of the Even Financial Acquisition.

"Even Transaction Expenses" means all fees, costs and expenses incurred in connection with the Even Financial Acquisition.

"Event of Default" means any of the events described in Section 13.1.

"Excess Cash Flow" means, for any period, the result of:

<u>16</u>

(a) EBITDA for that period,

<u>minus</u>

(b) the sum, without duplication, of

(i) (A) scheduled payments of principal of the Term Loans and other Funded Debt (other than (x) payments of revolving Debt that do not include a dollar-for-dollar commitment reduction with respect to such Debt and (y) for the avoidance of doubt, any Excess Cash Flow payments made during such period pursuant to Section 6.1.2(a)(v)) permitted under this Agreement and made during that period and (B) the amount of Net Cash Proceeds from (x) any Asset Disposition pursuant to Section 11.4(ix) or (y) Extraordinary Receipts, in each case, to the extent such Net Cash Proceeds are required to be used to prepay the Term Loans or are reinvested (or anticipated to be reinvested) in the business pursuant to Section 6.1.2(a)(i) or (iv), respectively, in each case, solely to the extent the receipt of such Net Cash Proceeds resulted in an increase to Consolidated Net Income or EBITDA (and not in excess of the amount of such increase), plus

(ii) cash payments permitted under this Agreement and made during that period with respect to any Capital Expenditures, Permitted Acquisitions and any Investments made pursuant to Section 11.9(p) and/or (r), in each case, to the extent funded with Internally Generated Cash, <u>plus</u>

(iii) all income taxes (or taxes from capital gains or profits) paid (or payable) in cash by the Loan Parties during (or with respect to) that period net of refunds actually received in cash during that period, <u>plus</u>

(iv) Tax Distributions paid (or payable) in cash during (or with respect to) such period, plus

(v) cash Interest Expense of the Loan Parties during that period, plus

(vi) the positive difference, if any, of working capital as of the end of that period over working capital as of the beginning of that period (provided that cash and Cash Equivalent Investments shall be excluded in determining any such change in working capital for purposes of this clause (vi)), <u>plus</u>

(vii) the cash portion paid (or payable) by the Company and its Subsidiaries during (or with respect to) such period of all cash items added back to EBITDA during such period pursuant to clauses (b)(vi), (b)(vii), (b)(viii), (b)(xi), (b)(xiii), (b)(xv), and (b)(xvi) of the definition of EBITDA, in each case, to the extent funded with Internally Generated Cash, <u>plus</u>

(viii) other overhead and administrative fees, costs, and expenses of the Company incurred in the ordinary course of business and paid in cash during such period to the extent permitted under Section 11.3 to the extent added back to EBITDA for such period, <u>plus</u>

(ix) payments in respect of earn-out obligations permitted hereunder by the Loan Parties in connection with Permitted Acquisitions, plus

(x) net cash payments made in connection with Hedging Agreements during such period; plus

<u>17</u>

(xi) an amount equal to (A) all cash charges excluded in calculating Consolidated Net Income and (B) any non-cash item of gain or income included in calculating Consolidated Net Income or EBITDA; plus

(xii) the positive difference, if any, of the aggregate outstanding principal amount of originated loans funded with cash into haircut capital of Loan Parties (i.e., excluding amounts funded with proceeds of warehouse lines or other financing) as of the end of that period over the aggregate outstanding principal amount of originated loans funded with cash into haircut capital of Loan Parties (i.e., excluding amounts funded with proceeds of warehouse lines or other financing) as of the beginning of that period, assuming a consistent advance rate from such warehouse lines of 90% or higher (i.e., discount rate of 10% or lower);

<u>plus</u>

(c) the sum, without duplication, of:

(i) the positive difference, if any, of working capital as of the beginning of that period over working capital as of the end of that period (provided that cash and Cash Equivalent Investments shall be excluded in determining any such change in working capital for purposes of this clause (c)); <u>plus</u>

(ii) the positive difference, if any, of the aggregate outstanding principal amount of originated loans funded with cash into haircut capital of Loan Parties (i.e., excluding amounts funded with proceeds of warehouse lines or other financing) as of the beginning of that period over the aggregate outstanding principal amount of originated loans funded with cash into haircut capital of Loan Parties (i.e., excluding amounts funded with proceeds of warehouse lines or other financing) as of the beginning of that period over the aggregate outstanding principal amount of originated loans funded with cash into haircut capital of Loan Parties (i.e., excluding amounts funded with proceeds of warehouse lines or other financing) as of the end of that period, assuming a consistent advance rate from such warehouse lines of 90% or higher (i.e., discount rate of 10% or lower).

"Excluded Deposit Accounts" means (i) deposit accounts the balance of which consists exclusively of (A) withheld income taxes and federal, state or local employment taxes required to be paid to the Internal Revenue Service or state or local government agencies with respect to employees of any of the Loan Parties and their Subsidiaries and (B) amounts required to be paid over to an employee benefit plan pursuant to DOL Reg. Sec. 2510.3 102 on behalf of or for the benefit of employees of any of the Loan Parties and their Subsidiaries; (ii) all segregated deposit accounts constituting (and the balance of which consists solely of funds set aside in connection with) payroll accounts, trust accounts, fiduciary accounts and accounts dedicated to the payment of accrued employee benefits, medical, dental and employee benefits claims to employees of any of the Loan Parties and their Subsidiaries; (iii) other deposit accounts, excluding collection or primary operating accounts, where the average daily balance thereof, calculated on a monthly basis, does not exceed \$250,000, up to an aggregate amount for all such deposit accounts not to exceed \$750,000; (iv) any deposit account established as an escrow account (but only so long as such account remains an escrow account); (v) any deposit account constituting a zero balance account and; (vi) any deposit account with respect to which, in the reasonable judgment of Administrative Agent, the burden or cost of providing a Guaranty shall be excessive in view of the benefits to be obtained by the Lenders thereform and (vii) the Collection Account (as defined in the SP Forward Flow Agreement) so long as the related Servicing Agreement (as defined therein) remains in effect.

"Excluded Subsidiary" means any Subsidiary that is (a) not a wholly owned Subsidiary of a Loan Party, (b) prohibited or restricted by applicable Law or by contractual obligations existing on the Closing Date (or, in the case of any Subsidiary acquired after the Closing Date, in existence at the time of acquisition but not entered into in contemplation thereof) from guaranteeing the Obligations or if guaranteeing the Obligations (i) would require governmental (including regulatory) consent, approval, license or authorization in order to provide such guarantee or (ii) where the delivery of such guaranty or pledge, as applicable, would reasonably be expected to result in an adverse Tax consequence (including, without limitation, as a result of the operation of Sections 245A and 956 of the Internal Revenue Code, taking into account any proposed or final regulations issued thereunder, or any similar Law or regulation in any applicable jurisdiction (other than any de minimis consequences)) as reasonably determined by Administrative Agent in consultation with Borrower, (c) a Subsidiary with respect to which, in the reasonable judgment of Administrative Agent, the burden or cost of providing a Guaranty shall be excessive in view of the benefits to be obtained by the Lenders therefrom, (d) any SPV Financing Entity, (e) any Subsidiary that is a licensed broker-dealer, registered investment adviser or licensed insurance broker and (f) MoneyLion of Malaysia SDN DHB.

"Excluded Taxes" means any of the following Taxes imposed on or with respect to a Recipient or required to be withheld or deducted from a payment to a Recipient: (a) Taxes imposed on or measured by net income (however denominated), franchise Taxes, and branch profits Taxes, in each case imposed (i) in a jurisdiction in which the relevant recipient is organized, (ii) in a jurisdiction which the relevant recipient's principal office is located, (iii) in a jurisdiction in which the relevant recipient's lending office (or branch) in respect of which payments under this Agreement are made is located or (iv) that are Other Connection Taxes; (b) any withholding Taxes imposed under FATCA; (c) in the case of a Lender, U.S. federal withholding Taxes imposed on amounts payable to or for the account of such Lender with respect to an applicable interest in a Loan or Commitment pursuant to a law in effect on the date on which (i) such Lender changes its lending office, except in each case to the extent that, pursuant to Section 7.6, amounts with respect to such Taxes were payable either to such Lender's assignor immediately before such Lender became a party hereto or to such Lender immediately before it changed its lending office, and (d) Taxes attributable to such Recipient's failure to comply with Section 7.6.4.

"Extraordinary Receipts" means any cash in excess of \$750,000 (in the case of clauses (b) (other than in respect of business interruption insurance) and (d) below) and \$3,000,000 (in the case of clauses (a), (b) (to the extent relating to business interruption insurance), (c), (e) and (f) below), in each case, in any Fiscal Year received by or paid to or for the account of any Loan Party not in the ordinary course of business consisting of (a) pension plan reversions, (b) proceeds of insurance, (c) judgments, proceeds of settlements or other consideration of any kind in connection with any cause of action (other than with respect to reimbursement of third party claims), (d) condemnation awards (and payments in lieu thereof), (e) indemnity payments (other than with respect to reimbursement of third party claims), (f) any purchase price adjustment received in connection with any purchase, and (g) foreign, United States, state or local tax refunds to the extent not included in the calculation of EBITDA.

"FATCA" means Sections 1471 through 1474 of the Code, as of the date of the Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with), and any current or future regulations or official interpretations thereof and any agreement entered into pursuant to Section 1471(b)(1) of the Code and any fiscal or regulatory legislation, rules or practices adopted pursuant to any intergovernmental agreement treaty or convention among Governmental Authorities and implementing such Sections of the Code.

"Federal Funds Rate" means, for any day, a fluctuating interest rate equal for each day during the applicable period to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers, as published for that day (or, if that day is not a Business Day, for the next preceding Business Day) by the Federal Reserve Bank of New York, or, if that rate is not so published for any day which is a Business Day, the average of the quotations for that day on those transactions received by Administrative Agent from three Federal funds brokers of recognized

<u>19</u>

standing selected by Administrative Agent. Administrative Agent's determination of the Federal Funds Rate will be binding and conclusive absent manifest error.

"Financial Statements" means that certain audited consolidated balance sheet, statement of operations, stockholders equity and cash flows for the fiscal year ended December 31, 2020.

"Fiscal Quarter" means a fiscal quarter of a Fiscal Year, which period is the 3-month period ending on the last day of each of March, June, September, and December of each year.

"Fiscal Year" means the fiscal year of the Company and its Subsidiaries, which period will be the 12-month period ending on the last day of December of each year.

"Floor" means a rate of interest equal to 1.00%.

"FRB" means the Board of Governors of the Federal Reserve System or any successor thereto.

"Funded Debt" means, as to any Person, all Debt of the type set forth in clauses (a), (b), (c), (d), (e), (f) (the amount of such Debt will be limited to the extent of unreimbursed amounts thereunder), (i), (j) (excluding Debt of SPV Financing Entities of the type set forth in clause (j)(x)) (in each case, only to the extent such Debt is characterized as debt or considered a liability under GAAP) and (k) of such definition, and Contingent Liabilities with respect to each of the foregoing types of Debt. For the avoidance of doubt, in no event shall Debt of an SPV Financing Entity of the type set forth in clause (j)(x) of "Debt" (even if such Debt would also constitute Debt under another clause of the definition thereof) constitute Funded Debt.

"GAAP" means generally accepted accounting principles set forth from time to time in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board (or agencies with similar functions of comparable stature and authority within the U.S. accounting profession) and the Securities and Exchange Commission, which are applicable to the circumstances as of the date of determination.

"Guarantor" means each Person that guarantees the Obligations of Borrower.

"Guaranty" means each guaranty executed and delivered by any Guarantor, together with any joinders thereto and any other guaranty agreement executed by a Guarantor, in each case in form and substance reasonably satisfactory to Administrative Agent. The Guaranty and Collateral Agreement is a Guaranty.

"Guaranty and Collateral Agreement" means the Guaranty and Collateral Agreement dated as of the date of this Agreement executed and delivered by each Loan Party, together with any joinders thereto and any other guaranty and collateral agreement executed by a Loan Party, in each case in form and substance reasonably satisfactory to Administrative Agent.

"Hazardous Substances" means hazardous waste, hazardous substance, pollutant, contaminant, toxic substance, oil, hazardous material, chemical, or other substance regulated by any Environmental Law.

"Hedging Agreement" means any interest rate, currency or commodity swap agreement, cap agreement, collar agreement, spot foreign exchange, forward foreign exchange option (or series of options) and any other agreement or arrangement designed to protect a Person against fluctuations in interest rates, currency exchange rates or commodity prices.

<u>20</u>

"<u>Hedging Obligation</u>" means, with respect to any Person, any liability of that Person under any Hedging Agreement determined (a) for any date on or after the date that Hedging Agreement has been closed out and termination value determined in accordance therewith, using that termination value; and (b) for any date prior to the date referenced in clause (a), using the amount determined as the mark-to-market value for that Hedging Agreement, as determined based upon one or more mid-market or other readily available quotations provided by any recognized dealer in that Hedging Agreement (which may include a Lender or any Affiliate of a Lender).

"Incremental Effective Date" is defined in Section 2.1.2(a).

"Incremental Equivalent Debt" is defined in Section 2.1.2(f).

"Incremental Facility Amendment" is defined in Section 2.1.2(e).

"Incremental Facility Request" is defined in Section 2.1.2(a).

"Incremental Facility" is defined in Section 2.1.2(a).

"Incremental Loan" is defined in Section 2.1.2(a).

"Incremental Term Loan" is defined in Section 2.1.2(a).

"Incremental Term Loan Commitment" is defined in Section 2.1.2(a).

"Indemnified Liabilities" is defined in Section 15.17.

"Indemnified Taxes" means (a) Taxes other than Excluded Taxes, imposed on or with respect to any payment made by or on account of any obligation of any Loan Party under any Loan Document and (b) to the extent not otherwise described in (a), Other Taxes.

"Insolvency Proceeding" means any case or proceeding commenced by or against a Person under any state, federal or foreign law for, or any agreement of a Person to, (a) the entry of an order for relief under the Bankruptcy Code, or any other insolvency, debtor relief, or debt adjustment law; (b) the appointment of a receiver, trustee, liquidator, administrator, conservator or other custodian for that Person or any part of its Property; or (c) an assignment or trust mortgage for the benefit of creditors.

"Instacash Receivables" means each "Receivable," together with all "Related Rights" relating thereto, in each case as defined in the SP Forward Flow Agreement.

"Instacash Servicer Account" means that certain deposit account, maintained by the Company as servicer in respect of Instacash, into which the proceeds of Instacash Receivables are deposited, which account is subject to a Control Agreement and the SP Intercreditor Agreement.

"Instacash Weekly Purchased Amount" means, with respect to any calendar week, the amount (by principal balance) of Instacash Receivables purchased by the Purchasers under the SP Forward Flow Agreement during such calendar week.

"Intellectual Property" is defined in Section 9.21.

"Intellectual Property Security Agreement" is used as defined in the Guaranty and Collateral Agreement.

<u>21</u>

"Intercompany Subordination Agreement" means an Intercompany Subordination Agreement substantially in the form of Exhibit G.

"Interest Expense" means, for any period, as determined in accordance with GAAP, the consolidated interest expense of the Company and its Subsidiaries for that period (including all imputed interest on Capital Leases) in each case calculated net of interest income.

"Interest Payment Date" means as to any Loan, the last Business Day of each month and the Termination Date.

"Interest Period" means, as to any Borrowing, the period commencing on the date of such Loan or Borrowing and ending on the numerically corresponding day in the calendar month that is one or three months thereafter (in each case, subject to the availability thereof), as specified in the applicable Notice of Borrowing or Notice of Conversion; provided that (i) if any Interest Period would end on a day other than a Business Day, such Interest Period shall be extended to the next succeeding Business Day unless such next succeeding Business Day would fall in the next calendar month, in which case such Interest Period shall end on the next preceding Business Day, (ii) any Interest Period that commences on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the last calendar month of such Interest Period) shall end on the last Business Day of the last calendar month of such Interest Period, (iii) no Interest Period shall extend beyond the Termination Date and (iv) no tenor that has been removed from this definition pursuant to Section 8.7.4 shall be available for specification in such Notice of Borrowing or Notice of Conversion. For purposes hereof, the date of a Loan or Borrowing initially shall be the date on which such Loan or Borrowing is made and thereafter shall be the effective date of the most recent conversion or continuation of such Loan or Borrowing.

"Internally Generated Cash" means cash generated from the operations of the business of Borrower and its Subsidiaries; provided that, notwithstanding the foregoing, "Internally Generated Cash" shall not include (i) the proceeds of any Debt (ii) the proceeds of any Equity Issuance, (iii) the proceeds of any insurance, indemnification or other payments or (iv) the proceeds of any Extraordinary Receipts (including, without limitation, any non-ordinary course tax refunds or returns).

"Investment" means, with respect to any Person, any investment in another Person, whether by acquisition of any debt or Equity Interest, by making any loan or advance, by becoming obligated with respect to a Contingent Liability in respect of obligations of that other Person (other than travel and similar advances to employees in the ordinary course of business) or by making an Acquisition. The amount of any Investment shall be the original cost of such Investment, plus the cost of any addition thereto that otherwise constitutes an Investment, without any adjustments for increases or decreases in value, or write-ups, write-downs or write-offs with respect thereto, but giving effect to any repayments of principal in the case of any Investment in the form of a loan and any return of capital or return on Investment in the case of any equity Investment (whether as a distribution, dividend, redemption or sale).

"IRS" means the United States Internal Revenue Service.

"Lender" is defined in the introductory clause of this Agreement.

"Lender Party" is defined in Section 15.17.

"Lien" means, with respect to any Person, any interest granted by that Person in any real or personal property, asset, or other right owned or being purchased or acquired by that Person (including an interest in respect of a Capital Lease) that secures payment or performance of any obligation and includes any mortgage, lien, encumbrance, title retention lien, charge, or other security interest of any kind, whether

<u>22</u>

arising by contract, as a matter of law, by judicial process, or otherwise, provided that in no event shall an operating lease in and of itself be deemed to constitute a Lien.

"Liquidity" means, as of any date of determination, (I) the sum of (x) Unrestricted Cash plus (y) Cash-in-Transit, <u>minus</u> (II) the unpaid amount of any final non-appealable judgments or settlements (whether in respect of an action by a Governmental Authority, class action or substantially similar claim or otherwise; and whether from a court or arbitration) or final non-appealable penalties or fines imposed by a Governmental Authority (in each case, except to the extent covered by insurance (including D&O insurance) pursuant to which the insurer has been notified within seven calendar days of such judgment, settlement, penalty or fine (and which Borrower believes in good faith is covered by such insurance at the time of such notification) and has not denied coverage so long as that insurance is paid to the applicable Loan Party within sixty (60) days of the rendering of those judgments, entering into of those settlements or imposition of those penalties or fines).

"Loan" or "Loans" means, as the context may require, any of the Term Loans.

"Loan Documents" means this Agreement, the Notes, the Agent Fee Letter, each Collateral and Diligence Questionnaire, the Collateral Documents, and all documents, instruments, and agreements delivered in connection with the foregoing, as any of the foregoing are amended or modified in accordance with their respective terms.

"Loan Party" means Borrower and each Guarantor.

"Malka Acquisition" means that certain Membership Interest Purchase Agreement dated as of November 15, 2021, by and among, among others, MoneyLion Technologies Inc. and Malka Media Group LLC.

"Malka Earn-Out" means the earn-out payment payable in cash or shares of common stock, par value \$0.0001 per share, of Parent, in the sole discretion of Borrower, pursuant to the terms of the Malka Acquisition Agreement.

"Malka Make-Whole" means those certain make-whole payments payable pursuant to Section 2.06(e) of the Malka Acquisition Agreement.

"Malka Transaction Expenses" means all fees, costs and expenses incurred in connection with the Malka Acquisition Agreement.

"Margin Stock" means any "margin stock" as defined in Regulation U.

"<u>Material Adverse Effect</u>" means (a) a material adverse change in, or a material adverse effect upon, the financial condition, operations, assets, business or properties of the Loan Parties taken as a whole, (b) a material impairment of the ability of any Loan Party to perform any of the Obligations under any Loan Document, (c) a material adverse effect upon any substantial portion of the Collateral under the Collateral Documents or upon the legality, validity, binding effect or enforceability against any Loan Party of any Loan Document, or (d) a material impairment of the ability of Administrative Agent to enforce or collect any Obligations or to realize upon any material portion of the Collateral.

"Monroe Capital" is defined in the introductory clause of this Agreement.

"Mortgage" means a mortgage, deed of trust, leasehold mortgage or similar instrument granting Administrative Agent a Lien on real property of any Loan Party.

<u>23</u>

"Mortgage-Related Documents" means with respect to any real property subject to a Mortgage, the following, in form and substance reasonably satisfactory to Administrative Agent: (a) a mortgagee title policy (or binder therefor) covering Administrative Agent's interest under the Mortgage, in a form and amount and by an insurer reasonably acceptable to Administrative Agent, which must be fully paid on that effective date; (b) all assignments of leases, estoppel letters, attornment agreements, consents, waivers, and releases as Administrative Agent reasonably requires with respect to other Persons having an interest in such real estate; (c) a current, as-built survey of the real property, containing a metes-and-bounds property description and certified by a licensed surveyor reasonably acceptable to Administrative Agent; (d) a life-of-loan flood hazard determination and, if such real property is located in a flood plain, an acknowledged notice to the applicable Loan Party and flood insurance in an amount, with endorsements and by an insurer reasonably acceptable to Administrative Agent; (e) a current appraisal of such real property, prepared by an appraiser reasonably acceptable to Administrative Agent, and in form and substance reasonably satisfactory to Required Lenders; (f) copies of any existing Phase I or Phase II environmental site assessments in the possession of any Loan Party with respect to such real property.

"<u>Multiemployer Pension Plan</u>" means a multiemployer plan, as defined in Section 4001(a)(3) of ERISA, to which Borrower or any other member of the Controlled Group contributes or has any liability, whether direct or indirect or absolute or contingent.

"Net Cash Proceeds" means:

(a) with respect to any Asset Disposition, the aggregate cash proceeds (including cash proceeds received pursuant to policies of insurance or by way of deferred payment of principal pursuant to a note, installment receivable or otherwise, but only as and when received) received by any Loan Party pursuant to that Asset Disposition net of (i) the direct costs relating to that sale, transfer or other disposition (including sales commissions and legal, accounting and investment banking fees); (ii) taxes paid or reasonably estimated by Borrower to be payable as a result thereof (after taking into account any available tax credits or deductions and any tax sharing arrangements); (iii) amounts required to be applied to the repayment of any Debt secured by a Lien on the asset subject to that Asset Disposition (other than the Loans); (iv) all amounts that are set aside as a reserve (A) for adjustments in respect of the purchase price of the applicable assets, (B) amounts provided as a reserve against any liabilities under any indemnification obligation, (C) for any liabilities associated with such sale or casualty, to the extent such reserve is required by GAAP, and (D) for the payment of unassumed liabilities relating to the assets sold or otherwise disposed of at the time of, or within 30 days after, the date of such sale or other disposition; (v) in the case of any Asset Disposition by any non-Wholly-Owned Subsidiary, the pro rata portion of the Net Cash Proceeds thereof (calculated without regard to this clause (v)) attributable to any minority interest and not available for distribution to or for the account of the applicable Borrower or a Wholly-Owned Subsidiary as a result thereof; and (vi) in the case of a Subsidiary that is not a Loan Party, payment of the outstanding principal amount of, premium or penalty, if any, and interest and other amounts on any Debt (other than the Loans) permitted to be repaid hereunder that is required to be repaid or otherwise comes due or would be in default under the terms thereof

(b) with respect to any issuance of Equity Interests, the aggregate cash proceeds received by any Loan Party pursuant to that issuance, net of the direct costs relating to that issuance (including sales and underwriters' commissions and provisions for taxes); and

(c) with respect to any issuance of Debt, the aggregate cash proceeds received by any Loan Party pursuant to that issuance, net of the direct costs of that issuance (including up-front,

<u>24</u>

underwriters' and placement fees, any other fees and expenses incurred in connection therewith and provision for taxes in connection therewith).

"Non-Consenting Lender" is defined in Section 15.1(k).

"Non-U.S. Lender" is defined in Section 7.6.4.

"Note" means a promissory note substantially in the form of Exhibit A.

"Notice of Borrowing" is defined in Section 2.2.2(a).

"Notice of Conversion" is defined in Section 2.2.3(b).

"Obligations" means all obligations (monetary (including post-petition interest, allowed or not) or otherwise) of any Loan Party under this Agreement and any other Loan Document, including Attorney Costs, all in each case howsoever created, arising or evidenced, whether direct or indirect, absolute or contingent, now or hereafter existing, or due or to become due.

"OFAC" is defined in Section 9.30.

"Operating Lease" means any lease of (or other agreement conveying the right to use) any real or personal property by any Loan Party, as lessee, other than any Capital Lease.

"Other Connection Taxes" means, with respect to any to Recipient pursuant to the terms of this Agreement, Taxes imposed as a result of a present or former connection between such Administrative Agent, Lender or other Person, as the case may be, and the jurisdiction imposing such Tax (other than connections arising from such recipient having executed, delivered, become a party to, performed its obligations under, received payments under, received or perfected a security interest under, engaged in any other transaction pursuant to or enforced any Loan Document, or sold or assigned an interest in any Loan or Loan Document).

"Other Taxes" means any and all present or future stamp or documentary taxes or any other excise or property taxes, charges or similar levies, arising from any payment made under this Agreement or from the execution, delivery or enforcement of, or otherwise with respect to any Loan Document, except any such Taxes that are Other Connection Taxes imposed with respect to an assignment.

"Parent" means MoneyLion Inc., which directly owns 100% of the outstanding Equity Interests of Borrower.

"Participant" is defined in Section 15.6.2.

"Patriot Act" is defined in Section 15.16.

"Payment in Full" means (a) the payment in full in cash of all Loans and other Obligations, other than contingent indemnification obligations for which no claims have been asserted; (b) the termination of all Commitments; and (c) the release of any claims of the Loan Parties against Administrative Agent and Lenders arising on or before the payment date.

"PBGC" means the Pension Benefit Guaranty Corporation and any entity succeeding to any or all of its functions under ERISA.

<u>25</u>

"Pension Plan" means a "pension plan," as that term is defined in Section 3(2) of ERISA, which is subject to Title IV of ERISA or the minimum funding standards of Section 302 of ERISA (other than a Multiemployer Pension Plan), and as to which Borrower, any Subsidiary of Borrower or any member of Borrower's Controlled Group, has any liability, whether direct or indirect or absolute or contingent, including any liability by reason of having been a substantial employer within the meaning of Section 4063 of ERISA at any time during the preceding five years, or by reason of being deemed to be a contributing sponsor under Section 4069 of ERISA.

"Periodic Term SOFR Determination Day" has the meaning specified in the definition of "Term SOFR".

"<u>Permitted Acquisition</u>" means (x) the Even Financial Acquisition and (y) any other Acquisition by Borrower or any of its Subsidiaries that, in the case of this clause (y) satisfies the following terms:

- (d) the business, division or assets acquired are for use, or the Person acquired is engaged, in a line of business permitted by Section 11.8;
- (e) immediately before and after giving effect to that Acquisition, no Event of Default exists;

(f) the aggregate upfront consideration (cash and non-cash) to be paid by the Loan Parties or their Subsidiaries (including any Debt assumed or issued in connection therewith, the maximum amount payable in connection with any deferred purchase price obligation (including any earnout obligation) (as estimated by Borrower in good faith) and the value of any upfront Equity Interests of any Loan Party issued to the seller in connection with that Acquisition) in connection with that Acquisition is less than \$50,000,000 (unless otherwise waived in writing by Administrative Agent in its sole discretion) except to the extent funded with the Net Cash Proceeds of an Equity Issuance (other than issuance of Disqualified Equity Interests), in each case to the extent such Net Cash Proceeds do not constitute Curative Equity; <u>provided</u> that any earn-out obligations incurred by any Loan Party or its Subsidiaries in connection with a Permitted Acquisition shall be subordinated to the Obligations pursuant to terms reasonably satisfactory to Administrative Agent;

(g) immediately after giving effect to that Acquisition, Borrower is in pro forma compliance with all the financial covenants set forth in Section 11.12 (recomputed for the most recent Fiscal Quarter for which financial statements have been delivered);

(h) in the case of the Acquisition of any Person, the board of directors or similar governing body of that Person has approved that Acquisition;

(i) not less than ten (10) Business Days prior to that Acquisition (or any later date approved by Administrative Agent in its reasonable discretion), Administrative Agent has received an acquisition summary with respect to the Person and/or business, division or assets to be acquired, which summary must include a reasonably detailed description thereof (including financial information) and operating results (including financial statements for the most recent 12 month period for which they are available), the material terms and conditions, including economic terms, of the proposed Acquisition, and Borrower's calculation of pro forma EBITDA relating thereto; provided that Borrower may update, supplement or amend documents provided under this clause (f) (and any such update, supplement or amendment shall be deemed not to restart such ten (10) Business Day period);

<u>26</u>

(j) not less than five (5) Business Days prior to that Acquisition (or any later date approved by Administrative Agent in its reasonable discretion), Administrative Agent has received substantially complete drafts of each material document, instrument and agreement to be executed in connection with that Acquisition and prior to consummation of a Permitted Acquisition Administrative Agent has received executed versions of such material documents, instruments and agreements;

(k) Borrower's computation of pro forma EBITDA is satisfactory to Administrative Agent;

(1) unless otherwise waived in writing by Administrative Agent, the business, division, assets or Person acquired generated EBITDA (calculated in a manner reasonably acceptable to Administrative Agent) for the four consecutive Fiscal Quarters most recently ended prior to that Acquisition of no less than (\$15,000,000) on a stand-alone basis for any single Acquisition (and (\$40,000,000) in the aggregate for all such Acquisitions);

(<u>m</u>) the definitive documents for such Acquisition shall not prohibit the collateral assignment of rights and indemnities under the related acquisition documents or material contracts and agreements of the target in favor of Administrative Agent;

 (\underline{n}) to the extent required pursuant to the terms thereof, the provisions of Section 10.10 have been satisfied (or will be satisfied within 30 days after the date of consummation of such Acquisition (or such other date as agreed by Administrative Agent in its sole discretion)); and

(<u>o</u>) Borrower has provided Administrative Agent with all available business, financial, environmental, accounting, legal and other due diligence materials, reports and memoranda, including insurance reports and like third party professional reports, and any other information with respect to that Acquisition reasonably requested by Administrative Agent.

"Permitted Asset Disposition" means an Asset Disposition involving (a) the sale or lease of inventory in the ordinary course of business; (b) any sale, lease, assignment, disposition, or other transfer for value between or among Loan Parties or (c) liquidation of Cash Equivalent Investments into cash in the ordinary course of business.

"<u>Permitted Bond Hedge Transaction</u>" means any call or capped call option (or substantively equivalent derivative transaction) relating to Parent's common stock (or other securities or property following a merger event, reclassification or other change of the common stock of Parent) purchased by Parent in connection with the issuance of any Permitted Convertible Debt and settled in common stock of Parent (or such other securities or property), cash or a combination thereof (such amount of cash determined by reference to the price of Parent's common stock or such other securities or property) and cash in lieu of fractional shares.

"Permitted Convertible Debt" means senior, unsecured Debt of Parent that is convertible into shares of common stock of Parent (or other securities or property following a merger event, reclassification or other change of the common stock of Parent), cash or a combination thereof (such amount of cash determined by reference to the price of Parent's common stock or such other securities or property) and cash in lieu of any fractional shares, provided that such Permitted Convertible Debt has customary market terms for capital markets convertible indebtedness.

"Permitted Lien" means a Lien expressly permitted under this Agreement pursuant to Section 11.2.

"Permitted Parent Debt" means unsecured Debt of Parent (A) that is not subject to any guarantee by any Subsidiary of Parent, (B) that will not mature until after the latest maturity date in effect with respect to the Loans on the date of issuance or incurrence thereof, (C) that is not subject to mandatory redemption, mandatory amortization, repurchase, prepayment or sinking fund obligation (other than customary offers to repurchase and prepayment events upon a change of control, asset sale or event of loss and a customary acceleration right after an event of default) prior to ninety-one (91) days after such latest maturity date (it being understood that such Debt may have mandatory prepayment, repurchase or redemption provisions satisfying the requirements of clause (D) below prior to such date), (D) that does not require any payments in cash of interest or other amounts in respect of the principal thereof prior to a date that is ninety-one (91) days after the latest maturity date in effect with respect to the Loans on the date of such issuance or incurrence, (E) that has covenant, default and remedy provisions no more restrictive (taken as a whole) than those set forth in this Agreement (taken as a whole) (except in a manner customary for holding company debt securities, including senior discount notes) (in each case as determined by Borrower in good faith) and (F) Permitted Convertible Debt.

"Permitted Warrant Transaction" means any call option, warrant or right to purchase (or substantively equivalent derivative transaction) relating to Parent's common stock (or other securities or property following a merger event, reclassification or other change of the common stock of Parent) sold by Parent substantially concurrently with any purchase by Parent of a Permitted Bond Hedge Transaction and settled in common stock of Parent (or such other securities or property), cash or a combination thereof (such amount of cash determined by reference to the price of Parent's common stock or such other securities or property) and cash in lieu of fractional shares.

"Person" means any natural person, corporation, partnership, trust, limited liability company, association, governmental authority or unit, or any other entity, whether acting in an individual, fiduciary or other capacity.

"Prepayment Percentage" is defined in Section 5.2.

"Prime Rate" means, for any day, the rate of interest in effect for that day equal to the prime rate in the United States as reported from time to time by *Bloomberg L.P.* (or other authoritative source selected by Administrative Agent in its reasonable discretion), or as Prime Rate is otherwise determined by Administrative Agent in its sole and absolute discretion. Administrative Agent's determination of the Prime Rate will be conclusive, absent manifest error. Any change in the Prime Rate will take effect at the opening of business on the day of that change. In the event *Bloomberg L.P.* (or any other authoritative source) publishes a range of "prime rates," the Prime Rate will be the highest of the "prime rates."

"Proceeding" means any investigation, inquiry, litigation, review, hearing, suit, claim, audit, arbitration, proceeding or action (in each case, whether civil, criminal, administrative, investigative or informal) commenced, brought, conducted or heard by or before, or otherwise involving, any governmental authority or arbitrator.

"Pro Rata Share"

(a) with respect to a Lender's obligation to make a Term A-1 Loan and receive payments of interest, fees, and principal with respect thereto, (i) prior to the Term A-1 Loan Commitment being terminated or reduced to zero, the percentage obtained by dividing (A) that Lender's Term A-1 Loan Commitment plus the unpaid principal amount of that Lender's Term A-1 Loans, by (B) the aggregate Term A-1 Loan Commitment of all Lenders plus the unpaid principal amount of all Term A-1 Loans of all Lenders; and (ii) from and after the time the Term A-1 Loan Commitment has been terminated or reduced

to zero, the percentage obtained by dividing (A) the aggregate unpaid principal amount of that Lender's Term A-1 Loans by (B) the aggregate unpaid principal amount of all Term A-1 Loans;

(b) with respect to a Lender's obligation to make a Term A-2 Loan and receive payments of interest, fees, and principal with respect thereto, (i) prior to the Term A-2 Loan Commitment being terminated or reduced to zero, the percentage obtained by dividing (A) that Lender's Term A-2 Loan Commitment plus the unpaid principal amount of that Lender's Term A-2 Loans, by (B) the aggregate Term A-2 Loan Commitment of all Lenders plus the unpaid principal amount of all Term A-2 Loans of all Lenders; and (ii) from and after the time the Term A-2 Loan Commitment has been terminated or reduced to zero, the percentage obtained by dividing (A) the aggregate unpaid principal amount of that Lender's Term A-2 Loans by (B) the aggregate unpaid principal amount of all Term A-2 Loans by (B) the aggregate unpaid principal amount of all Term A-2 Loans;

(c) with respect to a Lender's obligation to make Term B Loans and receive payments of interest, fees, and principal with respect thereto, (i) prior to the Term B Loan Commitment being terminated or reduced to zero, the percentage obtained by dividing (A) that Lender's Term B Loan Commitment plus the unpaid principal amount of that Lender's Term B Loans, by (B) the aggregate Term B Loan Commitment of all Lenders plus the unpaid principal amount of all Term B Loans of all Lenders; and (ii) from and after the time the Term B Loan Commitment has been terminated or reduced to zero, the percentage obtained by dividing (A) the aggregate unpaid principal amount of that Lender's Term B Loans; and

(d) with respect to all other matters as to a particular Lender, (i) prior to the time that the Commitments have been terminated or reduced to zero, the percentage obtained by dividing (A) that Lender's Term A-1 Loan Commitment plus the aggregate unpaid principal amount of that Lender's A-1 Loans plus that Lender's A-2 Loan Commitment plus the aggregate unpaid principal amount of that Lender's Term B Loans, by (B) the Term A-1 Loan Commitment of all Lenders plus the aggregate unpaid principal amount all Term A-1 Loans of all Lenders plus the Term A-2 Loan Commitment of all Lenders plus the aggregate unpaid principal amount all Term A-2 Loans of all Lenders plus the Term A-2 Loan Commitment of all Lenders plus the aggregate Term B Loan Commitment of all Lenders plus the aggregate Term B Loan Commitment of all Lenders plus the aggregate or reduced to zero, the percentage obtained by dividing (A) the aggregate unpaid principal amount of that Lender's Term A-1 Loans plus the aggregate unpaid principal amount of that Lender's Term A-1 Loans plus the aggregate unpaid principal amount of that Lender's Term A-1 Loans of all Lenders plus the Term A-2 Loan Commitment of all Lenders plus the aggregate unpaid principal amount all Term B Loans of all Lenders; and (ii) if the Commitments have been terminated or reduced to zero, the percentage obtained by dividing (A) the aggregate unpaid principal amount of that Lender's Term A-1 Loans plus the aggregate unpaid principal amount of that Lender's Term A-1 Loans plus the aggregate unpaid principal amount of that Lender's Term B Loans of all Lenders plus the aggregate unpaid principal amount of all Term A-1 Loans of all Lenders plus the aggregate unpaid principal amount of all Term A-1 Loans of all Lenders plus the aggregate unpaid principal amount of that Lender's Term B Loans of all Lenders plus the aggregate unpaid principal amount of all Term A-1 Loans of all Lenders plus the aggregate unpaid principal amount of all Term A-1 Loans of all Lenders.

"Protective Advances" is defined in Section 14.15.

"Recipient" means (a) the Administrative Agent and (b) any Lender, as applicable.

"<u>Regulation D</u>" means Regulation D of the FRB.

"<u>Regulation U</u>" means Regulation U of the FRB.

"<u>Relevant Governmental Body</u>" means the Federal Reserve Board or the Federal Reserve Bank of New York, or a committee officially endorsed or convened by the Federal Reserve Board or the Federal Reserve Bank of New York, or any successor thereto.

"Reportable Event" means a reportable event as defined in Section 4043 of ERISA and the regulations issued thereunder as to which the PBGC has not waived the notification requirement of Section

<u>29</u>

4043(a), or the failure of a Pension Plan to meet the minimum funding standards of Section 412 of the Code or under Section 302 of ERISA.

"Required Lenders" means, at any time, Lenders whose Pro Rata Shares exceed 50% as determined pursuant to clause (d) of the definition of "Pro Rata Share"; provided that the Pro Rata Shares held or deemed held by, any Defaulting Lender will be excluded for purposes of making a determination of Required Lenders.

"SEC" means the Securities and Exchange Commission or any other governmental authority succeeding to any of the principal functions thereof.

"Second Lien Loan" means the "Term Loans" under and as defined in the Second Lien Loan and Security Agreement.

"Second Lien Loan and Security Agreement" means that certain Loan and Security Agreement, dated as of April 17, 2020, entered into between MoneyLion Technologies Inc. (f/k/a MoneyLion Inc.), ML Plus LLC, as co-borrower, each lender thereto, and Monroe Capital, as collateral agent and as administrative agent.

"Senior Officer" means, with respect to any Loan Party, any of the president, chief executive officer or the chief financial officer of that Loan Party.

"SOFR" means a rate equal to the secured overnight financing rate as administered by the SOFR Administrator.

"SOFR Administrator" means the Federal Reserve Bank of New York (or a successor administrator of the secured overnight financing rate).

"SOFR Borrowing" means, as to any Borrowing, the SOFR Loans comprising such Borrowing.

"SOFR Loan" means a Loan that bears interest at a rate based on Adjusted Term SOFR, other than pursuant to clause (c) of the definition of "Base Rate".

"Sound Point" means Sound Point Capital Management, LP.

"SP Forward Flow Agreement" means that certain Master Receivables Purchase Agreement made and entered into as of June 30, 2024, by and among Sound Point, as purchaser agent, SP Main Street Funding I LLC, each additional Purchaser that may from time to time become party thereto and ML Plus LLC, as seller.

"SP Forward Flow Documents" means the SP Forward Flow Agreement, the related Servicing Agreement (as defined therein) and each other document, agreement or instrument related thereto or executed in connection therewith.

"SP Intercreditor Agreement" means that certain Account Intercreditor Agreement, dated as of June 30, 2024, by and among Sound Point, Monroe Capital and the Company.

"<u>SPV Financing Entity</u>" means a Subsidiary of Borrower which (a) is a "special purpose vehicle" or similar entity which is or was formed or acquired for purposes of (i) acquiring consumer loans, cash advance receivables and/or membership fees or similar periodic membership receivables, and/or (ii) obtaining related financing or equity investments from lenders or equity investors, as applicable; (b) issues

<u>30</u>

promissory notes or similar evidences of indebtedness to lenders or investors in connection with consumer loans, cash advance receivables and/or membership receivables acquired by such Subsidiary or an Affiliate of such Subsidiary; and/or (c) acquires consumer loans, cash advance receivables and/or membership receivables and transfers all or a portion of such consumer loans, cash advance receivables and/or membership receivables to another Subsidiary that is special purpose vehicle and a trust.

"Subordination Agreement" means any subordination, intercreditor, or other similar agreement in form and substance satisfactory to Administrative Agent.

"Subordinated Debt" means Debt subject to a Subordination Agreement.

"Subsidiary" means, with respect to any Person, a corporation, partnership, limited liability company, or other entity of which that Person owns, directly or indirectly, outstanding Equity Interests having more than 50% of the ordinary voting power for the election of directors or other managers of that corporation, partnership, limited liability company, or other entity. Unless the context otherwise requires, each reference to Subsidiaries in this Agreement refers to Subsidiaries of Parent.

"<u>Tax Distributions</u>" means for any taxable period in which Borrower is a member of a consolidated, combined, unitary or similar income tax group of which a direct or indirect parent of Borrower is Parent ("<u>Tax Group</u>"), distributions by Borrower or any Subsidiary of Borrower in amounts necessary to enable Parent to pay federal, foreign, state, territorial and local income Taxes and franchise Taxes imposed in lieu of income Taxes of such Tax Group that are attributable to the taxable income of each Borrower and/or any Subsidiary, as applicable, in respect of consolidated, combined, unitary or similar returns for the relevant jurisdiction of Parent that include such Borrower and/or any Subsidiaries attributable to Borrower and/or any Subsidiaries, if such Borrower and/or any Subsidiaries the income tax liability of Borrower and/or its applicable Subsidiaries, if such Borrower and such Subsidiaries had been a stand-alone corporate taxpayer or a consolidated, combined, unitary or affiliated group that consists solely of Borrower and such Subsidiaries for all relevant taxable periods.

"Taxes" means any and all present and future taxes, duties, levies, imposts, deductions, assessments, charges, similar fees or withholdings (including backup withholdings) imposed under applicable law and/or by any governmental authority that are in the nature of a tax, and any and all liabilities (including interest and penalties and other additions to taxes) with respect to any of the foregoing.

"<u>Term A-1 Loan Commitment</u>" means, as to any Lender, that Lender's commitment to make Term A-1 Loans under this Agreement. The amount of each Lender's Term A-1 Loan Commitment is set forth on Annex A. The initial aggregate amount of the Term A-1 Loan Commitments of all Lenders is \$70,000,000.

"Term A-1 Loan" is defined in Section 2.1.2(a).

"<u>Term A-2 Interest Rate</u>" means, for any given day, the greater of (a) 12%, and (b) a fluctuating rate of interest per annum equal to the Prime Rate plus 5.75%, but shall not exceed 15%. Any change in the Term Loan Interest Rate due to a change in the Prime Rate shall be effective from and including the effective day of such change in the Prime Rate.

"Term A-2 Lender" means any Lender with a Term A-2 Loan Commitment or a Term A-2 Loan.

"Term A-2 Loan Commitment" means, as to any Lender, that Lender's commitment to make Term A-2 Loans under this Agreement. The amount of each Lender's Term A-2 Loan Commitment is set forth

<u>31</u>

on Annex A. The initial aggregate amount of the Term A-2 Loan Commitments of all Lenders is \$20,000,000.

"Term A-2 Loan" is defined in Section 2.1.2(a).

"Term A-2 Termination Date" means October 15, 2023.

"Term B Loan Availability Period" means the period (i) beginning on and including the day after the Closing Date and (ii) ending on the Amendment No. 2 Effective Date.

"<u>Term B Loan Commitment</u>" means, as to any Lender, that Lender's commitment to make Term B Loans under this Agreement. The amount of each Lender's Term B Loan Commitment is set forth on Annex A. The initial aggregate amount of the Term B Loan Commitments of all Lenders is \$20,000,000.

"Term B Loan" is defined in Section 2.1.2(b).

"Term Loan" means, as the context may require, any Term A-1 Loans, Term A-2 Loans and/or any Term B Loan and, if applicable, any Incremental Term

Loan.

"Term SOFR" means,

(a) for any calculation with respect to a SOFR Loan, the Term SOFR Reference Rate for a tenor comparable to the applicable Interest Period on the day (such day, the "Periodic Term SOFR Determination Day") that is two (2) U.S. Government Securities Business Days prior to the first day of such Interest Period, as such rate is published by the Term SOFR Administrator; provided, however, that if as of 4:00 p.m. (Chicago time) on any Periodic Term SOFR Determination Day the Term SOFR Reference Rate for the applicable tenor has not been published by the Term SOFR Administrator and a Benchmark Replacement Date with respect to the Term SOFR Reference Rate has not occurred, then Term SOFR will be the Term SOFR Reference Rate for such tenor as published by the Term SOFR Administrator on the first preceding U.S. Government Securities Business Day for which such Term SOFR Reference Rate for such tenor was published by the Term SOFR Administrator so long as such first preceding U.S. Government Securities Business Day is not more than three (3) U.S. Government Securities Business Days prior to such Periodic Term SOFR Determination Day, and

(b) for any calculation with respect to an Base Rate Loan on any day, the Term SOFR Reference Rate for a tenor of one month on the day (such day, the "<u>Base Rate Term SOFR Determination Day</u>") that is two (2) U.S. Government Securities Business Days prior to such day, as such rate is published by the Term SOFR Administrator; <u>provided, however</u>, that if as of 4:00 p.m. (Chicago time) on any Base Rate Term SOFR Determination Day the Term SOFR Reference Rate for the applicable tenor has not been published by the Term SOFR Administrator and a Benchmark Replacement Date with respect to the Term SOFR Reference Rate has not occurred, then Term SOFR will be the Term SOFR Reference Rate for such tenor as published by the Term SOFR Administrator on the first preceding U.S. Government Securities Business Day for which such Term SOFR Reference Rate for such tenor was published by the Term SOFR Administrator so long as such first preceding U.S. Government Securities Business Day is not more than three (3) U.S. Government Securities Business Days prior to such Base Rate SOFR Determination Day.

"<u>Term SOFR Adjustment</u>" means, for any calculation with respect to a Base Rate Loan or a SOFR Loan, a percentage per annum equal to the percentage set forth below for the applicable Type of such Loan and (if applicable) Interest Period therefor:

Base Rate Loans: 0.11448%

<u>32</u>

SOFR Loans:

Interest Period	<u>Percentage</u>
One month	0.11448 %
Three months	0.26161%

"Term SOFR Administrator" means CME Group Benchmark Administration Limited (CBA) (or a successor administrator of the Term SOFR Reference Rate selected by Administrative Agent in its reasonable discretion).

"Term SOFR Reference Rate" means the forward-looking term rate based on SOFR.

"Termination Date" means the earliest to occur of (a) March 24, 2026 or (b) any other date on which the Loans are required to be paid in full pursuant to Section 6 or Section 13.

"Termination Event" means, with respect to a Pension Plan that is subject to Title IV of ERISA, the following: (a) a Reportable Event; (b) the withdrawal of Borrower or any other member of the Controlled Group from that Pension Plan during a plan year in which that Borrower or other member of the Controlled Group was a "substantial employer" as defined in Section 4001(a)(2) of ERISA; (c) the termination of that Pension Plan, the filing of a notice of intent to terminate the Pension Plan or the treatment of an amendment of that Pension Plan as a termination under Section 4041 of ERISA; (d) the institution by the PBGC of proceedings to terminate that Pension Plan; or (e) any event or condition that could reasonably constitute grounds under Section 4042 of ERISA for the termination of, or appointment of a trustee to administer, that Pension Plan.

"Threshold Amount" means \$30,000,000.

"Threshold Percentage" means 10.0%.

"Total Debt" means, as of any date of determination, all Funded Debt of the Company and its Subsidiaries as of such date, determined on a consolidated basis in accordance with GAAP.

"Total Debt to EBITDA Ratio" means, as of the last day of any Fiscal Quarter, the ratio of (a) Total Debt as of that day minus the lesser of (i) Unrestricted Cash and (ii) \$5,000,000 to (b) EBITDA for the Computation Period ending on that day.

"Total Plan Liability" means, at any time, the present value of all vested and unvested accrued benefits under all Pension Plans, determined as of the then most recent valuation date for each Pension Plan, using PBGC actuarial assumptions for single-employer plan terminations.

"Treasury Rate" means, as of any prepayment date, the yield to maturity as of such prepayment date of United States Treasury securities with a constant maturity (as compiled and published in the most recent Federal Reserve Statistical Release H.15 (519) that has become publicly available at least two Business Days prior to the prepayment date (or, if such Statistical Release is no longer published, any publicly available source of similar market data)) most nearly equal to the period from the prepayment date to the date that is 12 months after the Closing Date; provided, however, that if the period from the prepayment date to the date that is 12 months after the Closing Date; provided, however, that if the period from the prepayment date to the date that be constant maturity of

<u>33</u>

a United States Treasury security for which a weekly average yield is given, the Treasury Rate shall be the weekly average yield on actually traded United States Treasury securities adjusted to a constant maturity of one year shall be used.

"Type" is defined in Section 2.2.1.

"UCC" means the Uniform Commercial Code as in effect from time to time in the State of New York.

"Unadjusted Benchmark Replacement" means the applicable Benchmark Replacement excluding the related Benchmark Replacement Adjustment.

"Unfunded Liability" means the amount (if any) by which the present value of all vested and unvested accrued benefits under all Pension Plans exceeds the fair market value of all assets allocable to those benefits, all determined as of the then most recent valuation date for each Pension Plan, using PBGC actuarial assumptions for single employer plan terminations.

"Unrestricted Cash" means, as of any date of determination, the aggregate amount of unrestricted cash on-hand and Cash Equivalent Investments of the Company and its Subsidiaries, maintained in deposit accounts in the United States in the name of a Loan Party, which deposit accounts are subject to Control Agreements.

"U.S. Government Securities Business Day" means any day except for (a) a Saturday, (b) a Sunday or (c) a day on which the Securities Industry and Financial Markets Association recommends that the fixed income departments of its members be closed for the entire day for purposes of trading in United States government securities

"Weighted Average Life to Maturity" means, when applied to any Debt at any date, the number of years obtained by dividing: (a) the sum of the products obtained by multiplying (i) the amount of each then remaining installment, sinking fund, serial maturity or other required scheduled payments of principal, including payment at final maturity, in respect thereof by (ii) the number of years (calculated to the nearest one-twelfth) that will elapse between such date and the making of such payment; by (b) the then outstanding principal amount of such Debt; provided that the effect of (x) any prepayment made in respect of such Debt shall be disregarded in making such calculation and (y) any "AHYDO catch-up" payment that may be required to be made in respect of such Debt shall be disregarded in making such calculation.

"<u>Wholly-Owned Subsidiary</u>" means, as to any Person, a Subsidiary all of the Equity Interests of which (except directors' qualifying Equity Interests) are at the time directly or indirectly owned by that Person and/or another Wholly-Owned Subsidiary of that Person. Unless the context otherwise requires, each reference to Wholly-Owned Subsidiaries refers to Wholly-Owned Subsidiaries of the Company.

"Withholding Certificate" is defined in Section 7.6.4.

"<u>Write-Down and Conversion Powers</u>" means, with respect to any EEA Resolution Authority, the write-down and conversion powers of such EEA Resolution Authority from time to time under the Bail-In Legislation for the applicable EEA Member Country, which write-down and conversion powers are described in the EU Bail-In Legislation Schedule.

1.2 Certain Interpretive Provisions.

<u>34</u>

(a) The meanings of defined terms are equally applicable to the singular and plural forms of the defined terms.

(b) Section, Annex, Schedule and Exhibit references are to this Agreement unless otherwise specified.

(c) The term "including" is not limiting and means "including without limitation."

(d) In the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including"; the words "to" and "until" each mean "to but excluding", and the word "through" means "to and including."

(e) Unless otherwise expressly provided in this Agreement, (i) references to agreements (including this Agreement and the other Loan Documents) and other contractual instruments include all subsequent amendments, restatements, supplements, and other modifications thereto, but only to the extent that those amendments, restatements, supplements, and other modifications are not prohibited by the terms of any Loan Document, and (ii) references to any statute or regulation are to be construed as including all statutory and regulatory provisions amending, replacing, supplementing, or interpreting that statute or regulation.

(f) This Agreement and the other Loan Documents may use several different limitations, tests, or measurements to regulate the same or similar matters. All such limitations, tests, and measurements are cumulative and each is to be performed in accordance with its terms.

(g) This Agreement and the other Loan Documents are the result of negotiations among and have been reviewed by counsel to Administrative Agent, Borrower, the Lenders, and the other parties thereto and are the products of all parties. Accordingly, they are not to be construed against Administrative Agent or the Lenders merely because of Administrative Agent's or Lenders' involvement in their preparation.

(h) If any delivery due date specified in Section 10.1 for the delivery of reports, certificates, and other information required to be delivered pursuant to Section 10.1 falls on a day which is not a Business Day, then that due date will be extended to the immediately following Business Day.

(i) A Default or Event of Default will be deemed to exist at all times during the period commencing on the date that Default or Event of Default occurs to the date on which that Default or Event of Default is waived in writing pursuant to this Agreement or, in the case of a Default, is cured within any period of cure expressly provided for in this Agreement, and an Event of Default will "continue" or be "continuing" until that Event of Default has been waived in writing by the Required Lenders.

prohibited".

(j) Unless the context requires otherwise, the words "permitted" shall be construed to also refer to actions or undertakings that are "not

(k) For purposes of determining compliance with the financial covenants set forth in Section 11.12, (i) all revenues of Malka Media Group LLC and its Subsidiaries prior to November 15, 2021 and (ii) all revenues of Even Financial Inc. and its Subsidiaries prior to February 17, 2022, shall be excluded.

1.3 Accounting and Other Terms.

(a) Unless otherwise expressly provided in this Agreement, each accounting term used in this Agreement has the meaning given it under GAAP applied on a basis consistent with those used in

<u>35</u>

preparing the Financial Statements and using the same inventory valuation method as used in the Financial Statements, except for any change required or permitted by GAAP if Borrower's certified public accountants concur in that change, the change is disclosed to Administrative Agent, and <u>Section 11.12</u> is amended in a manner satisfactory to Administrative Agent to take into account the effects of the change.

(b) If at any time any change in GAAP would affect the computation of any financial ratio or requirement set forth in any Loan Document, and either Borrower or the Required Lenders shall so request, Administrative Agent, the Lenders and Borrower shall negotiate in good faith to amend such ratio or requirement to preserve the original intent thereof in light of such change in GAAP (subject to the approval of the Required Lenders); provided that, until so amended, (A) such ratio or requirement shall continue to be computed in accordance with GAAP prior to such change therein and (B) Borrower shall provide to Administrative Agent and the Lenders financial statements and other documents required under this Agreement or as reasonably requested hereunder setting forth a reconciliation between calculations of such ratio or requirement made before and after giving effect to such change in GAAP. Notwithstanding the foregoing, any lease treated as an Operating Lease on the date it is entered into shall continue to be treated as an operating lease during the term of this Agreement notwithstanding a change in the treatment thereof to a Capital Lease in accordance with any change in GAAP.

(c) All terms used in this Agreement which are defined in Article 8 or Article 9 of the UCC and which are not otherwise defined in this Agreement have the same meanings in this Agreement as set forth therein, except that terms used in this Agreement which are defined in the UCC as in effect in the State of New York on the date of this Agreement will continue to have the same meaning notwithstanding any replacement or amendment of that statute except as Administrative Agent may otherwise determine.

(d) The Administrative Agent does not warrant or accept responsibility for, and shall not have any liability with respect to (a) the continuation of, administration of, submission of, calculation of or any other matter related to Base Rate, the Term SOFR Reference Rate, Adjusted Term SOFR or Term SOFR, or any component definition thereof or rates referred to in the definition thereof, or any alternative, successor or replacement rate thereto (including any Benchmark Replacement), including whether the composition or characteristics of any such alternative, successor or replacement rate (including any Benchmark Replacement) will be similar to, or produce the same value or economic equivalence of, or have the same volume or liquidity as, Base Rate, the Term SOFR Reference Rate, Adjusted Term SOFR, Term SOFR or any other Benchmark prior to its discontinuance or unavailability, or (b) the effect, implementation or composition of any Conforming Changes. The Administrative Agent and its affiliates or other related entities may engage in transactions that affect the calculation of Base Rate, the Term SOFR Reference Rate, Term SOFR, Adjusted Term SOFR, any alternative, successor or replacement rate (including any Benchmark Replacement) or any relevant adjustments thereto, in each case, in a manner adverse to Borrowers; provided, that Administrative Agent will not engage in such transactions with the primary purpose of negatively impacting Borrowers. The Administrative Agent may select information sources or services in its reasonable discretion to ascertain Base Rate, the Term SOFR Reference Rate, Term SOFR, Adjusted Term SOFR, Adjusted Term SOFR or any other Benchmark, in each case pursuant to the terms of this Agreement, and shall have no liability to Borrower, any Lender or any other person or entity for damages of any kind, including direct or indirect, special, punitive, incidental or consequential damages, costs, losses or expenses (whether in tort, contract or otherwise and whether at law or in equity), for any

1.4 Treatment of LLC Division. Any restriction, condition or prohibition applicable to a merger, transfer, consolidation, amalgamation, consolidation, assignment, sale or transfer, or similar term set forth in the Loan Documents shall be deemed to apply to a division of or by a limited liability company, or an allocation of assets to a series of a limited liability companies, including any "Division" or other process or action permitted under Section 18-217 of Title 6 of the Delaware Code, as if it were a merger,

<u>36</u>

transfer, consolidation, amalgamation, consolidation, assignment, sale or transfer, or similar term, as applicable. Any reference in any Loan Document to a merger, transfer, consolidation, amalgamation, consolidation, assignment, sale, disposition or transfer, or similar term, shall be deemed to apply to a division of or by a limited liability company, or an allocation of assets to a series of a limited liability companies (or the unwinding of such a division or allocation), as if it were a merger, transfer, consolidation, amalgamation, consolidation, assignment, sale or transfer, or similar term, as applicable, to, of or with a separate Person. Any division of a limited liability company shall constitute a separate Person under the Loan Documents (and each division of any limited liability company that is a Subsidiary, joint venture or any other like term shall also constitute such a Person or entity). For all purposes under the Loan Documents, in connection with any division or plan of division under Delaware law (or any comparable event under a different jurisdiction's laws): (a) if any asset, right, obligation or liability of a different Person, then it shall be deemed to have been transferred from the original Person to the subsequent Person, and (b) if any new Person comes into existence, such new Person shall be deemed to have been organized on the first date of its existence by the holders of its Equity Interests at such time.

1.5 Reclassifiable Items. For purposes of determining compliance with this Agreement, in the event that any Debt, Lien, Restricted Payment, Investment or Asset Disposition or portion thereof, as applicable, at any time meets the criteria of more than one of the categories of transactions or items permitted pursuant hereto (each of the foregoing, a "Reclassifiable Item"), Borrower, in its sole discretion, may, from time to time, divide, classify or reclassify such Reclassifiable Item (or portion thereof) under one or more clauses of each such Section and will only be required to include such Reclassifiable Item (or portion thereof) in any one category. It is understood and agreed that any Debt, Liens, Restricted Payment, Investment, Asset Disposition and/or Affiliate transaction need not be permitted solely by reference to one category of permitted Debt, Lien, Restricted Payment, Investment, Asset Disposition and/or hereunder, but may instead be permitted in part under any combination thereof or under any other available exception.

SECTION 2 COMMITMENTS OF THE LENDERS; BORROWING AND CONVERSION PROCEDURES.

2.1 Commitments. On and subject to the terms and conditions of this Agreement, each of the Lenders, severally and for itself alone, agrees to make loans to Borrower as follows:

2.1.1 Term Loan Commitments.

(a) Each Lender with a Term A-1 Loan Commitment agrees to make a loan to Borrower (each such loan, a "<u>Term A-1 Loan</u>") on the Closing Date in that Lender's Pro Rata Share of the aggregate Term A-1 Loan Commitments of all Lenders. The Commitments of the Lenders to make Term A-1 Loans will expire concurrently with the making of Term A-1 Loans on the Closing Date.

(b) Each Lender with a Term A-2 Loan Commitment agrees to make a loan to Borrower (each such loan, a "<u>Term A-2 Loan</u>") on the Closing Date in that Lender's Pro Rata Share of the aggregate Term A-2 Loan Commitments of all Lenders. The Commitments of the Lenders to make Term A-2 Loans will expire concurrently with the making of Term A-2 Loans on the Closing Date.

Notwithstanding the foregoing or any other provision of this Agreement, each Term A-2 Lender shall not make its Term A-2 Loan in cash on the Closing Date but shall be deemed to have made its Term A-2 Loan under this Agreement by exchanging (i.e. rolling over) (the "Second Lien Rollover") its Second Lien Loans for the same aggregate principal amount as such Term A-2 Lender's Term A-2 Loan Commitment and Term A-2 Loan. The parties hereto acknowledge and agree that, as of the Closing Date,

after giving effect to the Second Lien Rollover, the Obligations in respect of the Second Lien Loans shall be deemed satisfied and paid in full.

(c) Each Lender with a Term B Loan Commitment agrees to make one or more loans (each such loan, a "<u>Term B Loan</u>") during the Term B Loan Availability Period in that Lender's Pro Rata Share of the aggregate amounts that Borrower requests from all Lenders. The aggregate amount of all Term B Loans made will not exceed the aggregate Term B Loan Commitments of all Lenders. The Commitments of the Lenders to make Term B Loans will decrease concurrently with the making of Term B Loans on each applicable borrowing date by an amount equal to the aggregate amount of the Term B Loans made on that borrowing date. In no event shall more than four (4) borrowings of Term B Loans be advanced to Borrower. The Commitments of the Lenders to make Term B Loans will expire at the end of the Term B Loan Availability Period.

2.1.2 Incremental Facilities.

(a) <u>Requests</u>. Borrower may, from time to time by written notice to Administrative Agent, no later than fifteen (15) days prior to the effectiveness of any such facility (each, an "<u>Incremental Facility Request</u>"), request one or more increases in the Term A-1 Commitment or Term B Commitment or one or more tranches of new term loans denominated in dollars (the "<u>Incremental Term Loan Commitment</u>" and the term loans thereunder, each an "<u>Incremental Loans</u>"; the Incremental Term Loan Commitments are sometimes referred to herein individually as an "<u>Incremental Loan</u>" and, collectively, as the "<u>Incremental Loans</u>"; the Incremental Term Loan Commitments are sometimes referred to herein individually as an "<u>Incremental Facility</u>" and collectively as the "<u>Incremental Facilities</u>") in an aggregate principal amount not to exceed \$60,000,000; provided that no commitment of any Lender shall be increased without the consent of such Lender. Such Incremental Facility Request shall set forth (A) the amount of the Incremental Term Loan Commitment being requested (which shall be in a minimum amount of \$5,000,000 and multiples of \$100,000 in excess thereof), (B) the date (an "<u>Incremental Effective Date</u>") on which such Incremental Facility is requested to become effective (which shall not be (a) unless otherwise agreed by Administrative Agent, less than fifteen (15) days after the date of such notice and (b) more than thirty (30) days after the date of such notice (or such longer period as Administrative Agent may agree), and (C) whether the related Incremental Loan is initially to be a SOFR Loan or a Base Rate Loan (and, if a SOFR Loan, the Interest Period therefor).

(b) Allocations. Upon delivery of the Incremental Facility Request, (x) such Incremental Facility shall be offered to each Lender according its Pro Rata Share, provided that none of the Lenders will be required to provide all or any portion of such Incremental Facility, and any decision whether or not to so provide all or any portion of such Incremental Facility by any Lender shall be made at the sole discretion of such Lender and (y) any Lender that fails to commit to provide all or any portion of its Pro Rata Share, as applicable, in which case such portion may instead be provided in the form of Incremental Equivalent Debt in accordance with clause (b), (i) to the extent Borrower does not have written commitments from one or more existing Lenders to provide such Incremental Facility in its entirety within ten (10) Business Days after Borrower has delivered such Incremental Facility Request, all such existing Lenders shall be deemed to have declined to provide any portion of such Incremental Facility, in which case such portion may instead be provided in the form of Incremental Facility Request, all such existing Lenders shall be deemed to have declined to provide any portion of such Incremental Facility, in which case such portion may instead be provided in the form of Incremental Facility Request, all such existing Lenders shall be deemed to have declined to provide any portion of such Incremental Facility, in which case such portion may instead be provided in the form of Incremental Facility in accordance with clause (f) below.

(c) <u>Conditions</u>. No Incremental Facility shall become effective under this Section 2.1.2 unless (A) immediately before and immediately after giving effect to such Incremental Facility, the loans to be made thereunder and the application of the proceeds therefrom, (I) no Event of Default shall

have occurred or be continuing and (II) the representations and warranties made in this Agreement or the other Loan Documents are true and correct in all material respects (without duplication of any materiality qualifier contained therein), (B) Administrative Agent shall have received a certificate of an Senior Officer of Borrower certifying as to the foregoing and attaching reasonably detailed supporting calculations related thereto, in form reasonably satisfactory to Administrative Agent and (C) Administrative Agent shall have received such other information and documentation as it may reasonably and promptly request in connection with such Incremental Facility and the use of proceeds thereof.

(d) Terms. Each Incremental Term Loan in the form of an increase to existing Term A-1 Loans or Term B Loans shall be on the same terms (including the maturity date) as, and pursuant to documentation (other than an Incremental Facility Amendment, if required) applicable to, the Term A-1 Loan or Term B Loan, as applicable. Each Incremental Term Loan in the form of one or more tranches of new term loans shall be on the same terms (except as provided below) as, and pursuant to documentation (other than an Incremental Facility Amendment, if required) applicable to, the Term A-1 Loan or Term B Loan, as applicable; provided that, to the extent agreed by Borrower, Administrative Agent and the lenders providing such Incremental Facility, the Incremental Loans may have call protection, applicable fees and amortization schedules that differ from those applicable to the Term A-1 Loan or Term B Loans. Each Incremental Term Loan shall have the same Guaranty as and be secured on a pari passu basis by the same Collateral securing the existing Term A-1 Loans and Term B Loans and any other Incremental Facilities.

(e) Required Amendments. Each of the parties hereto hereby agrees that, upon the effectiveness of any Incremental Facility, this Agreement shall be amended to the extent (but only to the extent) necessary to reflect the existence of such Incremental Facility and the Loans evidenced thereby (an "Incremental Facility Amendment"), and any such Incremental Facility Amendment may without the consent of the other Lenders effect such amendments to this Agreement and the other Loan Documents as may be necessary or appropriate, in the reasonable opinion of Administrative Agent and Borrower, to effectuate the provisions of this Section 2.1.2, and, for the avoidance of doubt, this Section 2.1.2 shall supersede any provisions in Section 15.1. From and after each Incremental Effective Date, the Loans and Commitments established pursuant to this Section 2.1.2 shall constitute Loans and Commitments under, and shall be entitled to all the benefits afforded by, this Agreement and the other Loan Documents. The Loan Parties shall take any actions reasonably required by Administrative Agent to ensure and/or demonstrate that the Liens and security interests granted by the applicable Loan Documents continue to be perfected under the UCC or otherwise after giving effect to the establishment of any such new Loans and Commitments, including compliance with Section 10.9.

(f) Incremental Equivalent Debt. Borrower may issue or incur Incremental Equivalent Debt in lieu of any Incremental Facility upon the delivery to the Administrative Agent of notice prior to the proposed effective date thereof, so long as the aggregate principal amount of Incremental Equivalent Debt incurred and Incremental Commitments provided pursuant to Section 2.1.2 does not exceed the \$60,000,000. As used herein, "Incremental Equivalent Debt" means Debt consisting of one or more series of notes or loans, a bridge facility in lieu of the foregoing, or debt securities; provided that:

(i) each applicable condition set forth in Section 2.1.2(c) shall be satisfied (or waived in accordance with the terms hereof) with respect to the incurrence of such Debt;

(ii) such Debt shall rank junior in right of payment with the Term Loans and shall rank junior in right of security with the Term Loans or may be unsecured, and a representative acting on behalf of the lenders or investors providing such Debt shall have entered into a Subordination Agreement (and the Lenders hereby authorize and direct the Administrative Agent to enter into any such Subordination Agreement);

<u>39</u>

(iii) such Debt shall not at any time be incurred or guaranteed by any Person other than a Loan Party, and such Debt shall not be secured by property other than Collateral;

(iv) the final scheduled maturity date applicable to such Debt shall not be earlier than the date that is one year after the then final scheduled maturity date of the Term Loans with the latest maturity date then in effect and (B) the Weighted Average Life to Maturity of such Series shall not be shorter than the remaining Weighted Average Life to Maturity of the Term Loans with the latest maturity date then in effect (without giving effect to any prepayments thereof); and

 (\underline{v}) except as otherwise expressly set forth herein, the pricing (including interest, fees and premiums), optional prepayment and redemption terms with respect to such Debt shall be determined by Borrower and the lenders or investors providing such Debt.

2.2 Loan Procedures.

2.2.1 <u>Various Types of Loans</u>. Each Loan (other than a Term A-2 Loan) may be divided into tranches which are, either a Base Rate Loan or a SOFR Loan (each, a "Type" of Loan), as Borrower specifies in the related Notice of Borrowing pursuant to <u>Section 2.2.2</u> or Notice of Conversion pursuant to <u>Section 2.2.3</u>. Subject to the other terms and conditions of this Agreement, Base Rate Loans and SOFR Loans may be outstanding at the same time. All borrowings, conversions, and repayments of Loans will be effected so that each Lender will have a ratable share (according to its Pro Rata Share) of all Types of Loans.

2.2.2 Borrowing Procedures.

(a) Borrower shall give written notice (each such written notice, a "<u>Notice of Borrowing</u>") substantially in the form of <u>Exhibit D</u> to Administrative Agent and each Lender with an applicable Commitment of each proposed borrowing not later than 10:00 a.m. (Chicago time) three Business Days prior to the proposed date of that borrowing. Each such notice will be effective upon receipt by Administrative Agent, will be irrevocable, and must specify the date, amount, Interest Period and Type of borrowing; <u>provided</u> that any borrowing of Term B Loans must be at least \$5,000,000 and an integral multiple of \$100,000.

(b) On the requested borrowing date, each Lender with an applicable Commitment shall provide Administrative Agent with immediately available funds, to Administrative Agent's Account, covering that Lender's Pro Rata Share of that borrowing so long as the applicable Lender has not received written notice that the conditions precedent set forth in <u>Section 12</u> with respect to that borrowing have not been satisfied. After Administrative Agent's receipt of the proceeds of the applicable Loans from Lenders with applicable Commitments, Administrative Agent shall make the proceeds of those Loans available to Borrower on the applicable borrowing date by transferring to Borrower immediately available funds equal to the proceeds received by Administrative Agent. Each Base Rate borrowing must be on a Business Day. Each Lender shall, upon request of Administrative Agent, deliver to Administrative Agent a list of all Loans made by that Lender, together with all information related thereto as Administrative Agent reasonably requests. Notwithstanding any provision of this Agreement to the contrary, Borrower may not request, and Lenders will not be required to fund, any borrowing of any Loan that is not a SOFR borrowing unless, subject to and as more particularly described in <u>Section 8</u>, SOFR is unavailable or unlawful.

2.2.3 Conversion Procedures.

(a) Subject to Section 2.2.1 and to the other terms and conditions of this Agreement, Borrower may, upon irrevocable written notice to Administrative Agent in accordance with Section 2.2.3(b), elect, as of any Business Day, to convert any Loans (other than Term A-2 Loans) (or any part thereof in an aggregate amount not less than \$1,000,000 or a higher integral multiple of \$100,000) into Loans of the other Type. After giving effect to any prepayment or conversion, the aggregate principal amount of SOFR Loans must be at least \$5,000,000 and an integral multiple of \$100,000. Notwithstanding any provision of this Agreement to the contrary, Borrower may not request the conversion of, and Administrative Agent and the Lenders will not be required to convert, any SOFR Loan into a Base Rate Loan unless, subject to and as more particularly described in Section 8, SOFR is unavailable or unlawful.

(b) Borrower shall give written notice (each such written notice, a "Notice of Conversion") substantially in the form of Exhibit \underline{E} or telephonic notice (followed immediately by a Notice of Conversion) to Administrative Agent of each proposed conversion not later than 1:00 p.m. (Chicago time) on the proposed date of that conversion, specifying in each case:

- (i) the proposed date of conversion;
- (ii) the aggregate amount of Loans to be converted;
- (iii) the Type of Loans resulting from the proposed conversion; and
- (iv) if the resulting Borrowing is a SOFR Borrowing, the Interest Period therefor after giving effect to such conversion.
- (c) Administrative Agent will promptly notify each Lender of its receipt of a Notice of Conversion pursuant to this Section 2.2.3.

(d) Failure to Submit a Conversion Notice; Events of Default. If Borrower fails to deliver a timely and complete Conversion Notice with respect to a SOFR Borrowing prior to the end of the Interest Period therefor, then, unless such SOFR Borrowing is repaid as provided herein, Borrower shall be deemed to have selected that such SOFR Borrowing shall automatically be continued as a SOFR Borrowing with an Interest Period of one month at the end of such Interest Period. Notwithstanding any contrary provision hereof, if an Event of Default exists and Administrative Agent, at the request of the Required Lenders, so notifies Borrower, then, so long as such Event of Default exists (i) no outstanding Borrowing may be converted to or continued as a SOFR Borrowing and (ii) unless repaid as provided herein, each SOFR Borrowing shall automatically be converted to a Base Rate Borrowing at the end of the Interest Period therefor.

2.3 Commitments Several. The failure of any Lender to make a requested Loan on any date will not relieve any other Lender of its obligation (if any) to make a Loan on that date, but no Lender will be responsible for the failure of any other Lender to make any Loan to be made by that other Lender.

2.4 Certain Conditions. Except as otherwise provided in Section 2.2.3, no Lender will have an obligation to make any Loan or to permit any conversion into any SOFR Loan, if an Event of Default or Default exists.

SECTION 3 EVIDENCING OF LOANS.

3.1 Notes. At a Lender's request, (a) the Term A-1 Loans of that Lender may be evidenced by a Note, with appropriate insertions, payable to the order of that Lender in a face principal amount equal to the principal amount of that Lender's Term A-1 Loans; (b) the Term A-2 Loans of that Lender may be evidenced by a Note, with appropriate insertions, payable to the order of that Lender in a face principal amount of that Lender in a face principal amount of that Lender is Term A-2 Loans; and (c) the Term B Loans of that Lender may be evidenced by a Note, with appropriate insertions, payable to the order of that Lender is Term B Loans of that Lender is Term B Loans plus the principal amount of that Lender's Term B Loan Commitment.

3.2 Recordkeeping. Administrative Agent, on behalf of each Lender, shall record in its records, the date and amount of each Loan made by each Lender and each repayment or conversion thereof. The aggregate unpaid principal amount so recorded will be rebuttably presumptive evidence of the principal amount of the Loans owing and unpaid. The failure to so record any such amount or any error in so recording any such amount will not, however, limit or otherwise affect the Obligations of Borrower under this Agreement or under any Note to repay the principal amount of the Loans under this Agreement, together with all interest accruing thereon.

SECTION 4 INTEREST.

4.1 Interest Rates. Borrower agree to pay interest on the unpaid principal amount of each Loan for the period commencing on the date that Loan is made until that Loan is paid in full as follows:

4.1.1 Term Loans.

(a) Term A-1 Loans.

(i) at all times while a Term A-1 Loan is a Base Rate Loan, at a rate per annum equal to the sum of the Base Rate from time to time in effect plus the Applicable Margin; and

(ii) at all times while a Term A-1 Loan is a SOFR Loan, at a rate per annum equal to the sum of the Adjusted Term SOFR from time to time in effect plus the Applicable Margin.

(b) Term A-2 Loans.

(i) at a rate per annum equal to the Term A-2 Interest Rate.

(c) Term B Loans.

(i) at all times while a Term B Loan is a Base Rate Loan, at a rate per annum equal to the sum of the Base Rate from time to time in effect plus the Applicable Margin; and

(ii) at all times while a Term B Loan is a SOFR Loan, at a rate per annum equal to the sum of the Adjusted Term SOFR from time to time in effect plus the Applicable Margin.

<u>42</u>

<u>4.1.2</u> <u>Default Rate</u>. Notwithstanding the foregoing, at any time an Event of Default exists, the interest rate applicable to each Loan, at the election of Administrative Agent or the Required Lenders (upon notice to Borrower), will be increased by 3.00% from the date such Event of Default occurred and otherwise during the existence of such Event of Default (and, in the case of Obligations not bearing interest, those Obligations will, during the existence of an Event of Default (to the extent then due and not yet paid) bear interest at the highest interest rate applicable to the Term Loans <u>plus</u> 3.00%), but any such increase may be rescinded by Required Lenders, notwithstanding <u>Section 15.1</u>. Notwithstanding the foregoing, upon the occurrence of an Event of Default under <u>Sections 13.1.1</u> or <u>13.1.4</u>, the increase provided for in this <u>Section 4.1.2</u> will occur automatically. In no event will interest payable by Borrower to any Lender under this Agreement exceed the maximum rate permitted under applicable law, and if any such provision of this Agreement is in contravention of any such law, then that provision will be deemed modified to limit that interest to the maximum rate permitted under that law.

4.1.3 Interest Payment Dates. Accrued interest on each Base Rate Loan is payable in arrears on the first Business Day of each month, upon a prepayment of that Loan, and at maturity. Accrued interest on each SOFR Loan is payable on the first Business Day of each month, upon a prepayment of that Loan, and at maturity. After maturity, and at any time an Event of Default exists, accrued interest on all Loans will be payable on demand.

4.2 Setting and Notice of Adjusted Term SOFR Rates. The Adjusted Term SOFR will be determined by Administrative Agent. Each determination of the applicable Adjusted Term SOFR by Administrative Agent will be conclusive and binding upon the parties to this Agreement, absent manifest error.

4.3 Computation of Interest. Interest will be computed for the actual number of days elapsed on the basis of a year of (a) 360 days for interest calculated at the Adjusted Term SOFR and (b) 365/366 days for interest calculated at the Base Rate. The applicable interest rate for each Base Rate Loan will change simultaneously with each change in the Base Rate and the applicable interest rate for each SOFR Loan will change simultaneously with each change in the Adjusted Term SOFR.

4.4 Term SOFR Conforming Changes. In connection with the use or administration of Term SOFR, Administrative Agent will have the right, in consultation with Borrower, to make Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in any other Loan Document, any amendments implementing such Conforming Changes will become effective without any further action or consent of any other party to this Agreement or any other Loan Document. The Administrative Agent will promptly notify Borrower and the Lenders of the effectiveness of any Conforming Changes in connection with the use or administration of Term SOFR.

SECTION 5 FEES.

5.1 Administrative Agent's Fees. Borrower shall pay to Administrative Agent all agent's fees as are mutually agreed to from time to time by Borrower and Administrative Agent, including the fees set forth in the Agent Fee Letter.

5.2 Applicable Premium. Upon the occurrence of an Applicable Premium Trigger Event, Borrower shall pay to Administrative Agent, for the sole and separate account of Administrative Agent, the applicable premium (each such applicable premium, an "Applicable Premium"), calculated in accordance with this Section 5.2.

<u>43</u>

(a) Without limiting the generality of Sections 6.2 and 6.3, and notwithstanding anything to the contrary in this Agreement or any other Loan Document, it is understood and agreed that if the Obligations are accelerated as a result of the occurrence and continuance of any Event of Default (including by operation of law or otherwise), the Applicable Premium, if any, determined as of the date of acceleration, will also be due and payable and will be treated and deemed as though the Term Loans were repaid or prepaid as of such date and shall constitute part of the Obligations for all purposes herein. Any Applicable Premium payable in accordance with this Section 5.2 shall be presumed to be equal to the liquidated damages sustained by the Lenders as the result of the occurrence of the Applicable Premium Trigger Event, and Borrower and the other Loan Parties agree that it is reasonable under the circumstances currently existing. The Applicable Premium, if any, shall also be payable in the event the Obligations (and/or this Agreement) are satisfied or released by foreclosure (whether by power of judicial proceeding), deed in lieu of foreclosure or by any other similar remedial actions taken by Administrative Agent or any Lender or by any other means. BORROWER EXPRESSLY WAIVES THE PROVISIONS OF ANY PRESENT OR FUTURE STATUTE OR LAW THAT PROHIBITS OR MAY PROHIBIT THE COLLECTION OF THE FOREGOING APPLICABLE PREMIUM IN CONNECTION WITH ANY SUCH ACCELERATION.

(b) Borrower expressly agrees that (a) the Applicable Premium is reasonable and is the product of an arm's length transaction between sophisticated business people, ably represented by counsel, (b) the Applicable Premium shall be payable notwithstanding the then prevailing market rates at the time payment is made, (c) there has been a course of conduct between Lenders and the Loan Parties giving specific consideration in this transaction for such agreement to pay the Applicable Premium, (d) the Loan Parties shall be estopped hereafter from claiming differently than as agreed to in this Section 5.2, (E) their agreement to pay the Applicable Premium is a material inducement to the Lenders to provide the Commitments and make the Term Loans, and (f) the Applicable Premium represents a good faith, reasonable estimate and calculation of the lost profits or damages of the Lenders and that it would be impractical and extremely difficult to ascertain the actual amount of damages to the Lenders or profits lost by the Lenders as a result of such Applicable Premium Trigger Event.

(c) Each Applicable Premium will be equal to:

(I) in the case of any Term Loan (other than Term A-2 Loans),

- (A) if the repayment of the applicable Term Loan occurs within 12 months after the Closing Date, the sum of (i) the principal amount of such repayment multiplied by 2.00% and (ii) the present value of all interest that would have otherwise been payable on the amount of such principal prepayment from the date of such prepayment to and including the date that is twelve months after the Closing Date, computed using a discount rate equal to the Treasury Rate plus 50 basis points, and
- (B) thereafter, an amount determined by multiplying the principal amount of the Term Loans repaid by the following applicable percentage amount: (i) 2.00% if that repayment occurs more than 12 months after the Closing Date but within 24 months after the Closing Date; (iii) 1.00% if that repayment occurs more than 24 months after the Closing Date but within 36 months after the Closing Date; and (iv) 0% if that repayment occurs more than 36 months after the Closing Date.

(II) in the case of any Term A-2 Loans, a payment equal to the principal amount of the Term A-2 Loans being prepaid multiplied by the applicable Prepayment Percentage.

For purposes of this Section 5.2(II), "Prepayment Percentage" means (A) three percent (3%) with respect to Term A-2 Loans prepaid after February 27, 2022 and prior to April 17, 2022 and one percent (1%) with respect to Term A-2 Loans prepaid after April 17, 2022 but prior to the Term A-2 Termination Date.

(d) All fees payable pursuant to this Section 5.4 will be deemed fully earned and non-refundable as of the Closing Date.

SECTION 6 PREPAYMENTS; REPAYMENTS.

6.1 Prepayments.

6.1.1 Voluntary Prepayments. Borrower may from time to time prepay the Term Loans in whole or in part. Borrower shall give Administrative Agent (which shall promptly advise each applicable Lender) notice of any such prepayment not later than 10:00 a.m. (Chicago time) on the day of that prepayment (which must be a Business Day), specifying the Loans to be prepaid and the date and amount of prepayment. Any such partial prepayment must be in an amount equal to \$1,000,000 or a higher integral multiple of \$100,000.

6.1.2 Mandatory Prepayments.

(a) <u>Term Loans</u>. Borrower shall make a prepayment of the Term Loans until paid in full upon the occurrence of any of the following at the following times and in the following amounts:

(i) within five (5) Business Days of the receipt by any Loan Party of any Net Cash Proceeds from any Asset Disposition pursuant to Section 11.4(ix) which exceeds in any Fiscal Year \$500,000, in an amount equal to 100% of those Net Cash Proceeds which are in excess of \$500,000 in such Fiscal Year; provided that, at the option of Borrower, and so long as no Default or Event of Default shall have occurred and be continuing at the time of receipt of such Net Cash Proceeds, Borrower may reinvest up to \$5,000,000 of such Net Cash Proceeds in any Fiscal Year in assets useful in their business so long as such reinvestment is made within 270 days or, in the case in which any portion of such Net Cash Proceeds (as certified by Borrower in writing to Administrative Agent); provided further, that any Net Cash Proceeds not so reinvested prior to the end of such applicable period shall be promptly applied to the prepayment of the Term Loans as set forth in this Section 6.1.2(a)(i) upon the expiration of such applicable period;

(ii) within two (2) Business Days of the receipt by any Loan Party of any Net Cash Proceeds from any issuance of Equity Interests of any Loan Party, whether in connection with the issuance of any Curative Equity or otherwise (excluding any issuance of Equity Interests (a) issued by Parent or (b) comprised of Curative Equity for the purpose of curing breach of the financial covenant set forth in Section 11.12.3), in an amount equal to 100% of those Net Cash Proceeds to the extent in excess of \$500,000 during the term of this Agreement.

(iii) concurrently with the receipt by any Loan Party of any Net Cash Proceeds from any issuance of any Debt of any Loan Party (excluding Debt permitted by Section 11.1), in an amount equal to 100% of those Net Cash Proceeds;

(iv) within five (5) Business Days of the receipt by any Loan Party of any Net Cash Proceeds in respect of Extraordinary Receipts, in an amount equal to 100% of those Extraordinary Receipts; provided that, at the option of Borrower (as elected by Borrower in writing to Administrative Agent on or prior to the fifth Business Day after the date of receipt of such Extraordinary Receipts), and so long as no Default or Event of Default shall have occurred and be continuing at the time of receipt of such Net Cash Proceeds, Borrower may reinvest up to \$7,500,000 in the aggregate over the term of the Agreement of such Extraordinary Receipts in assets useful in their business so long as such reinvestment is made within 270 days or, in the case in which any portion of such Extraordinary Receipts (as certified by Borrower in writing to Administrative Agent); provided further, that any Extraordinary Receipts not so reinvested prior to the end of such applicable period shall be promptly applied to the prepayment of the Term Loans as set forth in this Section 6.1.2(a)(iv) upon the expiration of such applicable period; and

(v) within five (5) Business Days after the earlier of (A) the date that Fiscal Year-end financial statements are required to be delivered pursuant to Section 10.1.1, and (B) the date of Borrower's actual delivery of Fiscal Year-end financial statements delivered pursuant to Section 10.1.1 (commencing with the first Fiscal Year period ending after the EBITDA Trigger Date), in an amount equal to the result of (A) the ECF Percentage of Excess Cash Flow for that Fiscal Year <u>minus</u> (B) the aggregate amount of all voluntary prepayments in respect of the outstanding principal balance of the Term Loan made by Borrower during such Fiscal Year; <u>provided</u>, that any Excess Cash Flow payment made pursuant to this Section 6.1.2(b)(v) shall exclude the portion of Excess Cash Flow that is attributable to the target of a Permitted Acquisition or similar permitted Investment and that accrued prior to the closing date of such Permitted Acquisition or similar permitted Investment.

6.2 Manner of Prepayments.

6.2.1 <u>All Prepayments</u>. Any prepayment of Term Loans is subject to <u>Section 5.2</u>. All prepayments of the Term Loans will be applied ratably to the Term Loans in the inverse order of maturity to the remaining installments thereof (including, without limitation, the final installment thereof). Except as otherwise provided by this Agreement, all principal payments in respect of the Loans will be applied first to repay outstanding Base Rate Loans and then to repay outstanding SOFR Loans in direct order of Interest Period maturities.

6.3 Repayments.

(a) <u>Term A-1 Loans</u>. Unless sooner paid in full, (i) \$5,000,000 of the outstanding principal balance of the Term A-1 Loans must be paid on October 15, 2023 and (ii) the remaining outstanding principal balance of the Term A-1 Loans must be paid in full on the Termination Date.

(b) <u>Term A-2 Loans</u>. Unless sooner paid in full, the outstanding principal amount of the Term A-2 Loans must be paid on the following dates and in the following amounts set forth opposite such dates:

Date	Amount
May 1, 2023	\$5,000,000
July 15, 2023	\$10,000,000

<u>46</u>

Date	Amount
Term A-2 Termination Date (i.e., October 15, 2023)	Remaining outstanding principal balance of the Term A-2 Loans to be paid in full

(c) <u>Term B Loans</u>. Unless sooner paid in full, the outstanding principal balance of the Term B Loans must be paid in full on the Termination Date.

(d) Incremental Loans. Terms of the repayments of any Incremental Facilities shall be set forth in the applicable Incremental Amendment.

6.4 Option to Decline. Any mandatory prepayment required to be made pursuant to Section 6.2.2(b) (other than with respect to clauses (iii) and (vi) therein) may be declined in whole or in part by any Lender without prejudice to such Lender's rights hereunder to accept or decline any future payments in respect of any mandatory prepayments, by providing notice to Administrative Agent no later than 2:00 p.m. (Chicago time) one (1) Business Day (or such other date acceptable to Administrative Agent) prior to the date of such prepayment. If a Lender chooses not to accept payment in respect of a mandatory prepayment in whole or in part, such declined proceeds shall be retained by Borrower.

SECTION 7 MAKING AND PRORATION OF PAYMENTS; SETOFF; TAXES.

7.1 Making of Payments.

7.1.1 Borrower shall make all payments of principal or interest on the Loans, and of all fees, to Administrative Agent in immediately available funds to Administrative Agent's Account not later than noon (Chicago time) on the date due, and funds received after that time will be deemed to have been received by Administrative Agent on the following Business Day. Borrower shall make all payments to Administrative Agent and the Lenders without set-off, counterclaim, recoupment, deduction, or other defense. Administrative Agent shall promptly remit to each Lender its share of all such payments received in collected funds by Administrative Agent for the account of that Lender. Notwithstanding the foregoing, Borrower shall make all payments under Section 8.1 directly to the Lender entitled thereto.

7.2 Application of Certain Payments.

7.2.1 So long as no Default or Event of Default has occurred and is continuing, (a) payments matching specific scheduled payments then due will be applied to those scheduled payments and (b) voluntary and mandatory prepayments will be applied as set forth in <u>Sections 6.1</u> and <u>6.2</u>.

7.2.2 Subject to any written agreement among Administrative Agent and the Lenders:

(a) All payments of principal and interest in respect of outstanding Loans, all payments of fees, and all other payments in respect of any other Obligations, will be allocated by Administrative Agent among Administrative Agent and the Lenders, as applicable, in proportion to their respective Pro Rata Shares or otherwise as provided in this Agreement or, in respect of payments not made on account of Loans, as designated by the Person making payment when the payment is made.

(b) After the occurrence and during the continuance of an Event of Default, Administrative Agent may, and upon the direction of the Required Lenders shall, apply all

<u>47</u>

payments in respect of any Obligations and all proceeds of the Collateral, subject to the provisions of this Agreement, as follows: (i) <u>first</u>, ratably to pay the Obligations in respect of any fees, expense reimbursements, indemnities and other amounts then due and payable to Administrative Agent until paid in full; (ii) <u>second</u>, ratably to pay the Obligations in respect of any fees (other than any Applicable Premium) and indemnities then due and payable to the Lenders until paid in full; (iii) <u>third</u>, ratably (to Administrative Agent in accordance with Administrative Agent's outstanding Protective Advances) to pay interest then due and payable in respect of Protective Advances until paid in full; (iv) <u>fourth</u>, ratably (to Administrative Agent accordance with Administrative Agent's outstanding Protective Advances) to pay principal of the Protective Advances until paid in full; (v) <u>fifth</u>, ratably, to pay interest then due and payable in respect of the Term Loans until paid in full; (vi) <u>sixth</u>, ratably to pay principal of the Term Loans until paid in full; (vi) <u>seventh</u>, ratably to pay the Obligations in respect of any Applicable Premium then due and payable until paid in full; and (viii) <u>eighth</u>, to the ratable payment of all other Obligations then due and payable.

(c) For purposes of Section 7.2.2(b), "paid in full" means payment in cash of all amounts owing under the Loan Documents (or, to the extent those Obligations are contingent, to provide Cash Collateral in respect of those Obligations) according to the terms thereof, including loan fees, service fees, professional fees, interest (and specifically including interest accrued after, or that would have accrued but for, the commencement of any Insolvency Proceeding), default interest, interest on interest, and expense reimbursements, whether or not same would be or is allowed or disallowed in whole or in part in any Insolvency Proceeding.

(d) In the event of a direct conflict between the priority provisions of this <u>Section 7.2.2</u> and other provisions contained in any other Loan Document, it is the intention of the parties to this Agreement that all such priority provisions be read together and construed, to the fullest extent possible, to be in concert with each other. In the event of any actual, irreconcilable conflict that cannot be resolved as aforesaid, the terms and provisions of this <u>Section 7.2.2</u> will control and govern.

7.3 Due Date Extension. If any payment of principal or interest with respect to any of the Loans, or of any fees, falls due on a day which is not a Business Day, then that due date will be extended to the immediately following Business Day (unless, in the case of a SOFR Loan, that immediately preceding Business Day is the first Business Day of a month, in which case that due date will be the immediately preceding Business Day), and, in the case of principal, additional interest will accrue and be payable for the period of any such extension.

7.4 Setoff. Borrower, for itself and each other Loan Party, agrees that Administrative Agent and each Lender have all rights of set-off and bankers' lien provided by applicable law, and in addition thereto, Borrower, for itself and each other Loan Party, agrees that at any time any Event of Default exists, Administrative Agent and each Lender may apply to the payment of any Obligations of Borrower and each other Loan Party under this Agreement, whether or not then due, any and all balances, credits, deposits, accounts, or moneys of Borrower and each other Loan Party then or thereafter with Administrative Agent or that Lender.

7.5 Proration of Payments. If any Lender obtains any payment or other recovery (whether voluntary, involuntary, by application of offset, or otherwise), on account of principal of or interest on any Loan (but excluding (i) any payment pursuant to Section 8 or 15.6 and (ii) payments of interest on any Affected Loan) in excess of its applicable Pro Rata Share of payments and other recoveries obtained by all Lenders on account of principal of and interest on the Loans, then held by them, then that Lender shall purchase from the other Lenders such participations in the Loans held by them as are necessary to cause that purchasing Lender to share the excess payment or other recovery ratably with each of them, but if all

<u>48</u>

or any portion of the excess payment or other recovery is thereafter recovered from that purchasing Lender, then that purchase will be rescinded and the purchase price restored to the extent of that recovery.

7.6 Taxes.

7.6.1 Borrower shall make all payments under this Agreement or under any Loan Documents without setoff, counterclaim, or other defense. Unless required by applicable law (as determined in the good faith discretion of the applicable withholding agent), all payments under this Agreement or under the Loan Documents (including any payment of principal, interest, or fees) to, or for the benefit, of any Recipient will be made by Borrower free and clear of and without deduction or withholding for, or account of, any Taxes now or hereafter imposed by any taxing authority.

<u>7.6.2</u> If Borrower makes any payments to a Recipient under this Agreement or under any other Loan Document in respect of which Borrower is required by applicable law to deduct or withhold any Indemnified Taxes, then Borrower shall increase such payment under this Agreement or under any other Loan Document such that after the reduction for the amount of Indemnified Taxes withheld (and any Indemnified Taxes withheld or imposed with respect to the additional payments required under this <u>Section 7.6.2</u>), the amount paid equals the amount that was payable under this Agreement or under any other Loan Document without regard to this <u>Section 7.6.2</u>. To the extent Borrower withholds any Taxes on payments under this Agreement or under any other Loan Document, Borrower shall pay the full amount deducted to the relevant taxing authority within the time allowed for payment under applicable law and shall deliver to Administrative Agent within 30 days after Borrower have made payment to that taxing authority the original or certified copy of a receipt issued by that taxing authority (or other evidence satisfactory to Administrative Agent) evidencing the payment of all amounts so required to be deducted or withheld from that payment. Borrower shall timely pay to the relevant Governmental Authority in accordance with applicable law, or at the option of Administrative Agent timely reimburse it for the payment of, any Other Taxes.

<u>7.6.3</u> If any Recipient is required by law to make any payments of any Indemnified Taxes on or in relation to any amounts received or receivable under this Agreement or under any other Loan Document, or any Indemnified Tax is assessed against Recipient with respect to amounts received or receivable under this Agreement or under any other Loan Document, Borrower will indemnify such Recipient within 30 days after demand therefor against (i) that Indemnified Tax and (ii) any Taxes imposed as a result of the receipt of the payment under this <u>Section 7.6.3</u>. A certificate prepared in good faith as to the amount of any such payment by that Lender or Administrative Agent or other recipient will, absent manifest error, be final, conclusive, and binding on all parties.

7.6.4 (a) To the extent permitted by applicable law, each Lender that is not a United States person within the meaning of Code Section 7701(a)(30) (a "Non-U.S. Lender") shall deliver to Borrower and Administrative Agent on or prior to the Closing Date (or in the case of a Lender that is an Assignee, on the date of the assignment to that Lender), whichever of the following is applicable:

(i) in the case of a Non-U.S. Lender claiming the benefits of an income tax treaty to which the United States is a party (x) with respect to payments of interest under any Loan Document, two properly completed and duly executed copies of IRS Form W-8BEN establishing that Lender's entitlement to a complete exemption from, or a reduced rate in, United States withholding Tax on interest payments to be made under this

<u>49</u>

Agreement or with respect to any Loan pursuant to such tax treaty and (y) with respect to any other applicable payments under any Loan Document, IRS Form W-8BEN establishing an exemption from, or reduction of, U.S. federal withholding Tax pursuant to the such tax treaty;

(ii) two properly completed and duly executed copies of IRS Form W-8ECI;

(iii) in the case of a Non-U.S. Lender claiming the benefits of the exemption for portfolio interest under Section 881(c) of the Code, (x) a certificate substantially in the form of Exhibit F-1 to the effect that such Non-U.S. Lender is not a "bank" within the meaning of Section 881(c)(3)(A) of the Code, a "10 percent shareholder" of Borrower within the meaning of Section 881(c)(3)(B) of the Code, or a "controlled foreign corporation" described in Section 881(c)(3)(C) of the Code (a "U.S. Tax Compliance Certificate") and (y) two properly completed and duly executed copies of IRS Form W-8BEN; or

(iv) to the extent a Non-U.S. Lender is not the beneficial owner, two properly completed and duly executed copies of IRS Form W-8IMY, accompanied by IRS Form W-8ECI, IRS Form W-8BEN, a U.S. Tax Compliance Certificate substantially in the form of Exhibit F-2 or Exhibit F-3, IRS Form W-9, and/or other certification documents from each beneficial owner, as applicable; provided that if the Non-U.S. Lender is a partnership and one or more direct or indirect partners of such Non-U.S. Lender are claiming the portfolio interest exemption, such Non-U.S. Lender may provide a U.S. Tax Compliance Certificate substantially in the form of Exhibit F-4 on behalf of each such direct and indirect partner;

(b) any Non-U.S. Lender shall, to the extent it is legally entitled to do so, deliver to Borrower and Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Non-U.S. Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of Borrower or Administrative Agent), executed originals of any other form prescribed by applicable law as a basis for claiming exemption from or a reduction in U.S. federal withholding Tax, duly completed, together with such supplementary documentation as may be prescribed by applicable law to permit Borrower or Administrative Agent to determine the withholding or deduction required to be made; and

(c) Each Lender that is not a Non-U.S. Lender (other than any such Lender that is taxed as corporation for U.S. federal income tax purposes) shall provide two properly completed and duly executed copies of IRS Form W-9 (or any successor or other applicable form) to Borrower and Administrative Agent certifying that that Lender is exempt from United States federal backup withholding Tax. To the extent that a form provided pursuant to this Section 7.6.4(c) is rendered obsolete or inaccurate in any material respect as result of change in circumstances with respect to the status of a Lender or Administrative Agent, then that Lender or Administrative Agent shall, to the extent permitted by applicable law, deliver to Borrower and, as applicable, Administrative Agent revised forms necessary to confirm or establish the entitlement to that Lender's exemption from United States federal backup withholding Tax.

(d) No Borrower will be required to pay additional amounts to any Lender, or indemnify any Lender, under this <u>Section 7.6</u> to the extent that those obligations would not have arisen but for the failure of that Lender to comply with this <u>Section 7.6.4</u>.

(e) Each Lender shall indemnify Administrative Agent and hold Administrative Agent harmless for the full amount of any and all present or future Taxes and related liabilities (including penalties, interest, additions to Tax and expenses, and any Taxes imposed by any jurisdiction on amounts payable to Administrative Agent under this <u>Section 7.6</u>) which are imposed on or with respect to principal, interest, or fees payable to that Lender under this Agreement and which are not paid by Borrower pursuant to this <u>Section 7.6</u>, whether or not those Taxes or related liabilities were correctly or legally asserted. This indemnification must be made within 30 days from the date Administrative Agent makes written demand therefor.

7.6.5 If Administrative Agent or a Lender determines, in its sole discretion, that it has received a refund of any Indemnified Taxes as to which it has been indemnified by Borrower or with respect to which Borrower has paid additional amounts pursuant to this Section 7.6, then Administrative Agent or that Lender, as applicable, shall pay over that refund to Borrower (but only to the extent of indemnity payments made, or additional amounts paid, by Borrower under this Section 7.6 with respect to the Indemnified Taxes giving rise to that refund), net of any Taxes imposed by reason of receipt of that refund and all out-of-pocket expenses of Administrative Agent or that Lender, as applicable, and without interest (other than any interest paid by the relevant governmental authority with respect to that refund, which interest must be paid to Borrower). Upon the request of Administrative Agent or any such Lender, Borrower shall repay any amount paid to Borrower (plus any penalties, interest, or other charges imposed by the relevant governmental authority. Nothing in this Section 7.6.5 is to be construed to require Administrative Agent or any Lender to make available its tax returns (or any other information which it deems confidential) to Borrower or any other Person.

<u>7.6.6</u> If a payment made to a Lender under any Loan Document would be subject to U.S. federal income withholding Tax imposed by FATCA if that Lender were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the Code, as applicable), then that Lender shall deliver to Administrative Agent (or, in the case of a Participant, to the Lender granting the participation only) at the time or times prescribed by law and at any other time or times reasonably requested by Administrative Agent (or, in the case of a Participant, the Lender granting the participation) all documentation prescribed by applicable law (including as prescribed by Section 1471(b)(3)(C)(i) of the Code) and all additional documentation reasonably requested by Administrative Agent (or, in the case of a Participant, the Lender granting the participation) as is necessary for Administrative Agent or Borrower to comply with their obligations under FATCA and to determine that that Lender has complied with that Lender's obligations under FATCA or to determine the amount to deduct and withhold from that payment. Solely for purposes of this Section 7.6.6, "FATCA" is deemed to include any amendments made to FATCA after the date of this Agreement.

7.6.7 Each party's obligations under this Section shall survive the resignation or replacement of Administrative Agent or any assignment of rights by, or the replacement of, a Lender, the termination of the Commitments and the repayment, satisfaction or discharge of all obligations under any Loan Document.

SECTION 8 INCREASED COSTS; SPECIAL PROVISIONS FOR SOFR LOANS.

8.1 Increased Costs.

(a) If any Change in Law (i) imposes, modifies, or deems applicable any reserve (including any reserve imposed by the FRB, but excluding any reserve included in the determination of the

Adjusted Term SOFR pursuant to <u>Section 4</u>), special deposit, or similar requirement against assets of, deposits with, or for the account of, or credit extended by, any Lender; or (ii) subject any Lender to any Taxes (other than (A) Indemnified Taxes and (B) Connection Income Taxes) on its loans, loan principal, letters of credit, commitments, or other obligations, or its deposits, reserves, other liabilities or capital attributable thereto; or (iii) imposes on any Lender any other condition affecting its SOFR Loans, its Note(s), or its obligation to make SOFR Loans, and the result of anything described in <u>clauses (i)</u> and (<u>ii)</u> above is to increase the cost to (or to impose a cost on) that Lender (or any SOFR Office of that Lender) of making or maintaining any Loan, or to reduce the amount of any sum received or receivable by that Lender (or its SOFR Office) under this Agreement or under its Note(s) with respect thereto, then upon demand by that Lender (which demand must be accompanied by a statement setting forth the basis for that demand and a calculation of the amount thereof in reasonable detail, a copy of which must be furnished to Administrative Agent), Borrower shall pay directly to that Lender such additional amount as will compensate that Lender for that increased cost or that reduction, so long as the applicable amounts have accrued on or after the day that is 180 days prior to the date on which that Lender first made demand therefor.

(b) If any Lender reasonably determines that any change in, or the adoption or phase-in of, any applicable law, rule, or regulation regarding capital adequacy, or any change in the interpretation or administration thereof by any governmental authority, central bank, or comparable agency charged with the interpretation or administration thereof, or the compliance by any Lender or any Person controlling any Lender with any request or directive regarding capital adequacy (whether or not having the force of law) of any such authority, central bank, or comparable agency, has or would have the effect of reducing the rate of return on that Lender's or that controlling Person's capital as a consequence of that Lender's obligations under this Agreement to a level below that which that Lender or that controlling Person could have achieved but for that change, adoption, phase-in, or compliance (taking into consideration that Lender's or that controlling Person's policies with respect to capital adequacy) by an amount deemed by that Lender or that controlling Person to be material, then from time to time, upon demand by that Lender (which demand must be accompanied by a statement setting forth the basis for that demand and a calculation of the amount thereof in reasonable detail, a copy of which must be furnished to Administrative Agent), Borrower shall pay to that Lender such additional amount as will compensate that Lender first made demand therefor.

8.2 Basis for Determining Interest Rate Inadequate or Unfair.

(a) Administrative Agent shall promptly notify the other parties of the following:

(i) Administrative Agent reasonably determines (which determination will be binding and conclusive on Borrower) that by reason of circumstances affecting the interbank SOFR market adequate and reasonable means do not exist for ascertaining the applicable Adjusted Term SOFR; or

(ii) the Required Lenders advise Administrative Agent that the Adjusted Term SOFR as determined by Administrative Agent will not adequately and fairly reflect the cost to those Lenders of maintaining or funding SOFR Loans (taking into account any amount to which those Lenders may be entitled under <u>Section 8.1</u>) or that the making or funding of SOFR Loans has become impracticable as a result of an event occurring after the date of this Agreement which in the opinion of those Lenders materially affects those Loans.



(b) So long as any circumstances described in a notice delivered pursuant to Section 8.2(a) continue, (i) no Lender will be required to make or convert any Base Rate Loans into SOFR Loans, and (ii) each such Loan will, unless then repaid in full, automatically convert to a Base Rate Loan.

If Administrative Agent reasonably determines (in consultation with Borrower) after the Closing Date that adequate and reasonable means do not exist for ascertaining the Adjusted Term SOFR, then (i) a comparable or successor floating rate that is, at such time, broadly accepted by the market for loans denominated in Dollars in lieu of the London interbank offered rate as determined by Administrative Agent (in consultation with Borrower) or (ii) if no such broadly accepted comparable successor rate exists at such time, Administrative Agent and Borrower shall endeavor to establish an alternate rate of interest to the Adjusted Term SOFR that gives due consideration to the then prevailing market convention for determining a rate of interest for syndicated loans in the United States at such time, and shall enter into an amendment to this Agreement to reflect such alternate rate of interest and such other related changes to this Agreement as may be applicable. Such selection of a successor rate within five Business Days of notice thereof. Notwithstanding anything to the contrary in this Agreement, such amendment shall become effective without any further action or consent of any other party to this Agreement.

<u>8.3</u> Changes in Law Rendering SOFR Loans Unlawful. If, after the date of this Agreement, any change in, or the adoption of any new, law or regulation, or any change in the interpretation of any applicable law or regulation by any governmental authority or other regulatory body charged with the administration thereof, makes it (or in the good faith judgment of any Lender causes a substantial question as to whether it is) unlawful for any Lender to make, maintain, or fund SOFR Loans, then that Lender shall promptly notify each of the other parties to this Agreement and, so long as those circumstances continue, (a) that Lender will not be required to make or convert any Base Rate Loan into a SOFR Loan (but that Lender shall, subject to the other terms of this Agreement, make Base Rate Loans concurrently with the making of or conversion of Base Rate Loans into SOFR Loans by the Lenders which are not so affected, in each case in an amount equal to the amount of SOFR Loans which would be made or converted into by that Lender at that time in the absence of those circumstances), and (b) each such SOFR Loan will, unless then repaid in full, automatically convert to a Base Rate Loan. Each Base Rate Loan made by a Lender which, but for the circumstances described in the foregoing sentence, would be a SOFR Loan (an "<u>Affected Loan</u>") will remain outstanding for the period corresponding to the SOFR Loans of which that Affected Loan would be a part absent those circumstances.

8.4 Right of Lenders to Fund through Other Offices. Each Lender may, if it so elects, fulfill its commitment as to any SOFR Loan by causing a foreign branch or Affiliate of that Lender to make that Loan, but each such Loan will be deemed to have been made by that Lender and the obligation of Borrower to repay that Loan will be to that Lender and will be deemed held by the Lender, to the extent of that Loan, for the account of that branch or Affiliate.

8.5 Mitigation of Circumstances; Replacement of Lenders.

(a) Each Lender shall promptly notify Borrower and Administrative Agent of any event of which it has knowledge that will result in, and will use reasonable commercial efforts available to it (and not, in that Lender's sole judgment, otherwise disadvantageous to that Lender) to mitigate or avoid, (i) any obligation by Borrower to pay any amount pursuant to Sections 7.6 or 8.1 or (ii) the occurrence of any circumstances described in Sections 8.2 or 8.3 (and, if any Lender has given notice of any such event described in clause (i) or (ii) and thereafter that event ceases to exist, that Lender shall promptly so notify Borrower and Administrative Agent). Without limiting the foregoing, each Lender shall designate a different funding office if that designation will avoid (or reduce the cost to Borrower of) any event described



in clause (i) or (ii) and that designation will not, in that Lender's sole judgment, be otherwise disadvantageous to that Lender.

(b) If Borrower become obligated to pay additional amounts to any Lender pursuant to <u>Sections 7.6</u> or <u>8.1</u>, or any Lender gives notice of the occurrence of any circumstances described in <u>Sections 8.2</u> or <u>8.3</u>, or any Lender becomes a Defaulting Lender, then Borrower may designate another financial institution that is acceptable to Administrative Agent in its reasonable discretion (a "<u>Replacement Lender</u>") to purchase the Loans of that Lender and that Lender's rights under this Agreement, without recourse to or warranty by, or expense to, that Lender, for a purchase price equal to the outstanding principal amount of the Loans payable to that Lender <u>plus</u> any accrued but unpaid interest on those Loans and all accrued but unpaid fees owed to that Lender and any other amounts owed to that Lender under this Agreement and any other Loan Document, and to assume all the obligations of that Lender under this Agreement. Upon any such purchase and assumption (pursuant to an Assignment Agreement), the applicable Lender will no longer be a party to this Agreement or have any rights under this Agreement (other than rights with respect to indemnities and similar rights applicable to that Lender prior to the date of that purchase and assumption) and will be relieved from all obligations to Borrower under this Agreement, and the Replacement Lender will succeed to the rights and obligations of that Lender under this Agreement.

<u>8.6</u> <u>Conclusiveness of Statements; Survival of Provisions</u>. Determinations and statements of any Lender pursuant to <u>Sections 8.1, 8.2</u>, or <u>8.3</u> will be conclusive absent demonstrable error. Lenders may use reasonable averaging and attribution methods in determining compensation under <u>Section 8.1</u>, and the provisions of <u>Section 8.1</u> will survive repayment of the Obligations, cancellation of any Note(s), and termination of this Agreement.

8.7 Benchmark Replacement Setting.

8.7.1 Benchmark Replacement. Notwithstanding anything to the contrary herein or in any other Loan Document, upon the occurrence of a Benchmark Transition Event, Administrative Agent and Borrower may amend this Agreement to replace the then-current Benchmark with a Benchmark Replacement. Any such amendment with respect to a Benchmark Transition Event will become effective at 5:00 p.m. (New York City time) on the fifth (5th) Business Day after Administrative Agent has posted such proposed amendment to all affected Lenders and Borrower so long as Administrative Agent has not received, by such time, written notice of objection to such amendment from Lenders comprising the Required Lenders. No replacement of a Benchmark with a Benchmark Replacement pursuant to this Section 8.7.1 will occur prior to the applicable Benchmark Transition Start Date.

<u>8.7.2</u> <u>Benchmark Replacement Conforming Changes</u>. In connection with the use, administration, adoption or implementation of a Benchmark Replacement, Administrative Agent will have the right, in consultation with Borrower, to make Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in any other Loan Document, any amendments implementing such Conforming Changes will become effective without any further action or consent of any other party to this Agreement or any other Loan Document.

<u>8.7.3</u> <u>Notices; Standards for Decisions and Determinations</u>. The Administrative Agent will promptly notify Borrower Representative and the Lenders of (i) the implementation of any Benchmark Replacement and (ii) the effectiveness of any Conforming Changes in connection with the use, administration, adoption or implementation of a Benchmark Replacement. The Administrative Agent will notify Borrower Representative of (x) the removal or reinstatement of any tenor of a Benchmark pursuant to Section 8.7.4 and (y) the commencement of any Benchmark

Unavailability Period. Any determination, decision or election that may be made by Administrative Agent or, if applicable, any Lender (or group of Lenders) pursuant to this Section 8.7, including any determination with respect to a tenor, rate or adjustment or of the occurrence or non-occurrence of an event, circumstance or date and any decision to take or refrain from taking any action or any selection, will be conclusive and binding absent manifest error and may be made in its or their sole discretion and without consent from any other party to this Agreement or any other Loan Document, except, in each case, as expressly required pursuant to this Section 8.7.

<u>8.7.4</u> Unavailability of Tenor of Benchmark. Notwithstanding anything to the contrary herein or in any other Loan Document, at any time (including in connection with the implementation of a Benchmark Replacement), (i) if the then-current Benchmark is a term rate (including the Term SOFR Reference Rate) and either (A) any tenor for such Benchmark is not displayed on a screen or other information service that publishes such rate from time to time as selected by Administrative Agent in its reasonable discretion or (B) the regulatory supervisor for the administrator of such Benchmark has provided a public statement or publication of information announcing that any tenor for such Benchmark is not or will not be representative, then Administrative Agent may modify the definition of "Interest Period" (or any similar or analogous definition) for any Benchmark settings at or after such time to remove such unavailable or non-representative tenor and (ii) if a tenor that was removed pursuant to clause (i) above either (A) is subsequently displayed on a screen or information service for a Benchmark (including a Benchmark Replacement) or (B) is not, or is no longer, subject to an announcement that it is not or will not be representative for a Benchmark (including a Benchmark Replacement), then Administrative Agent may modify the definition of "Interest Period" (or any similar or analogous definition) for all Benchmark settings at or after such time to reinstate such previously removed tenor.

8.7.5 Benchmark Unavailability Period. Upon Borrower Representative's receipt of notice of the commencement of a Benchmark Unavailability Period, Borrower Representative may revoke any pending request for a SOFR Borrowing of, conversion to or continuation of SOFR Loans to be made, converted or continued during any Benchmark Unavailability Period and, failing that, Borrowers will be deemed to have converted any such request into a request for a Borrowing of or conversion to Base Rate Loans. During a Benchmark Unavailability Period or at any time that a tenor for the then-current Benchmark is not an Available Tenor, the component of Base Rate based upon the then-current Benchmark or such tenor for such Benchmark, as applicable, will not be used in any determination of Base Rate.

SECTION 9 REPRESENTATIONS AND WARRANTIES.

To induce Administrative Agent and the Lenders to enter into this Agreement and to induce the Lenders to make Loans under this Agreement, Borrower represents and warrants to Administrative Agent and the Lenders on the Closing Date, on the date of any borrowing hereunder and on each other date on which such representations and warranties are expressly deemed made or expressly required to be made under the terms of the Loan Documents, that:

9.1 Organization. Each of the Loan Parties and their Subsidiaries is validly existing and in good standing under the laws of its jurisdiction of organization, and each of the Loan Parties and their Subsidiaries is duly qualified to do business in each jurisdiction where, because of the nature of its activities or properties, that qualification is required, except for any jurisdiction where the failure to so qualify could not reasonably be expected to have a Material Adverse Effect.

9.2 Authorization; No Conflict.

<u>55</u>

(a) Each Loan Party is duly authorized to execute and deliver each Loan Document to which it is a party, Borrower is duly authorized to borrow monies under this Agreement, and each Loan Party is duly authorized to perform its Obligations under each Loan Document to which it is a party.

(b) The execution, delivery, and performance by each Loan Party of each Loan Document to which it is a party, and the borrowings by Borrower under this Agreement, do not and will not (i) require any consent or approval of any governmental agency or authority (other than any consent or approval that has been obtained and is in full force and effect); (ii) conflict with (A) any provision of applicable law in any respect that would reasonably be expected to have a Material Adverse Effect, (B) the organizational documents or governing documents of any Loan Party, or (C) any agreement, indenture, instrument, or other document, or any judgment, order, or decree, that is binding upon any Loan Party or any of their respective properties that would reasonably be expected to have a Material Adverse Effect; or (iii) require, or result in, the creation or imposition of any Lien on any asset of any Loan Party (other than Liens in favor of Administrative Agent created pursuant to the Collateral Documents or otherwise permitted under the terms of this Agreement).

<u>9.3 Validity and Binding Nature</u>. Each of this Agreement and each other Loan Document to which any Loan Party is a party is the legal, valid, and binding obligation of that Person, enforceable against that Person in accordance with its terms, subject to bankruptcy, insolvency, and similar laws affecting the enforceability of creditors' rights generally and to general principles of equity.

9.4 Financial Condition. The Financial Statements, copies of each of which have been delivered to each Lender, were prepared in accordance with GAAP (except as expressly stated therein and subject, in the case of any such unaudited statements, to the absence of footnotes and to normal year-end adjustments) and present fairly in all material respects the consolidated financial condition of the Company and its Subsidiaries as at the dates covered in the Financial Statements and the results of their operations for the periods then ended.

9.5 No Material Adverse Change. Since December 31, 2020, there has been no material adverse change in the financial condition, operations, assets, business or properties of the Loan Parties and their Subsidiaries, taken as a whole.

9.6 Litigation and Contingent Liabilities. No litigation (including derivative actions), arbitration proceeding or governmental investigation or proceeding is pending or, to the Loan Parties' knowledge, threatened in writing against any of the Loan Parties and their Subsidiaries that could reasonably be expected to have a Material Adverse Effect, except as set forth in <u>Schedule 9.6</u>. Other than liability incident to any such litigation or proceedings, none of the Loan Parties and their Subsidiaries has any material contingent liabilities that are not listed in <u>Schedule 9.6</u> or permitted by <u>Section 11.1</u>.

9.7 Ownership of Properties; Liens. Each of the Loan Parties and their Subsidiaries owns good title to and, in the case of owned real property, marketable title to, and in the case of leased real property, a valid leasehold interest in, all of its properties and assets, real and personal, tangible and intangible, of any nature whatsoever (including Intellectual Property), free and clear of all Liens, charges, and claims, except as permitted by Section 11.2.

<u>9.8 Equity Ownership</u>. All issued and outstanding Equity Interests of each of the Loan Parties and their Subsidiaries are, to the extent applicable, duly authorized and validly issued, fully paid, non-assessable, and free and clear of all Liens other than those in favor of Administrative Agent, and all such Equity Interests were issued in compliance with all applicable state and federal laws concerning the issuance of securities. <u>Schedule 9.8</u> sets forth the authorized Equity Interests of each of the Loan Parties and their Subsidiaries as of the Closing Date. All of the issued and outstanding Equity Interests of each

<u>56</u>

Wholly-Owned Subsidiary is, directly or indirectly, owned by the Company. As of the Closing Date, except as set forth on <u>Schedule 9.8</u>, there are no pre-emptive or other outstanding rights, options, warrants, conversion rights, or other similar agreements or understandings for the purchase or acquisition of any Equity Interests of any of the Loan Parties and their Subsidiaries.

9.9 Pension Plans.

(a) The Unfunded Liability of all Pension Plans does not in the aggregate exceed 20% of the Total Plan Liability for all such Pension Plans. Except as could not reasonably be expected to result, either individually or in the aggregate, in a Material Adverse Effect, (i) each Pension Plan complies with all applicable requirements of law and regulations; (ii) no contribution failure under Section 430 of the Code, Section 303 of ERISA, or the terms of any Pension Plan has occurred with respect to any Pension Plan sufficient to give rise to a Lien under Section 303(k) of ERISA; (iii) there are no pending or, to the knowledge of Borrower, threatened claims, actions, investigations, or lawsuits against any Pension Plan, any fiduciary of any Pension Plan, or Borrower or any other member of the Controlled Group with respect to a Pension Plan or Multiemployer Plan; (iv) neither Borrower nor any other member of the Controlled Group has engaged in any prohibited transaction (as defined in Section 4975 of the Code or Section 406 of ERISA) with respect to any Pension Plan; and (vi) no Termination Event has occurred or is reasonably expected to occur with respect to any Pension Plan.

(b) To the knowledge of Borrower, all contributions (if any) have been made to any Multiemployer Pension Plan that are required to be made by Borrower or any other member of the Controlled Group under the terms of the plan or of any collective bargaining agreement or by applicable law. Except as could not reasonably be expected to result, either individually or in the aggregate, in a Material Adverse Effect, (i)neither Borrower nor any other member of the Controlled Group has withdrawn or partially withdrawn from any Multiemployer Pension Plan, incurred any withdrawal liability with respect to any such plan, or received notice of any claim or demand for withdrawal liability or partial withdrawal liability from any such plan; and (ii) neither Borrower nor any other member of the Controlled Group has received any notice that any Multiemployer Pension Plan is or may become insolvent.

9.10Investment Company Act. No Loan Party is required to register as an "investment company" under the Investment Company Act of 1940.

9.11Compliance with Laws Consents; Certain Actions.

(a) Each of the Loan Parties and their Subsidiaries is in compliance in all respects with the requirements of all laws and all orders, writs, injunctions, and decrees applicable to it or to its properties, except where (a) that requirement of law or order, writ, injunction, or decree is being contested in good faith by appropriate proceedings diligently conducted, or (b) the failure to comply therewith, either individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect.

(b) Each of the Loan Parties and their Subsidiaries have obtained all consents, approvals and authorizations of, made all declarations or filings with, and given all notices to, all Government Authorities that are necessary as of the Effective Date to continue their respective businesses as currently conducted, in each case except where the failure to do so could not reasonably be expected to cause a Material Adverse Effect.

(c) As of the Closing Date, no Governmental Authority has taken action to suspend, revoke or terminate any material permit or material license of any Loan Party (excluding, for avoidance of doubt, any voluntary termination or expiration of any such license or permit initiated by a Loan Party prior and unrelated to any such action by Governmental Authority).



(d) As of the Closing Date, (i) no class action claims (in court or in arbitration) have been asserted or threatened in writing and no more than 1,000 substantially similar claims (in court or in arbitration) have been asserted or threatened in writing, such that one or more such claims or actions individually or in the aggregate involve potential monetary relief against the Loan Parties exceeding the Threshold Amount or challenge the legality of revenues generated by the Loan Parties exceeding the Threshold Amount, as measured over the most recently ended calendar year, except where such claim(s) could not reasonably be expected to result in a Material Adverse Effect, and (ii) no actions have been initiated or threatened in writing by any Governmental Authority (or initiated defensively by any Loan Party against any Governmental Authority), such that one or more such claims or actions individually or in the aggregate involve potential monetary relief against the Loan Parties exceeding the Threshold Amount or challenge the legality of revenues generated by the Loan Parties exceeding the Threshold Amount or challenge the legality of revenues generated by the Loan Parties exceeding the Threshold Amount or challenge the legality of revenues generated by the Loan Parties exceeding the Threshold Amount, as measured over the most recently ended calendar year, provided that the actions identified on Schedule 9.11(d) shall not by themselves be deemed a violation of this representation and warranty.

9.12Regulation T, U, X. Borrower is not engaged principally, or as one of its important activities, in the business of extending credit for the purpose of purchasing or carrying Margin Stock (under Regulations T, U or X of the Federal Reserve Board of Governors).

9.13Taxes. Each of the Loan Parties and their Subsidiaries has timely filed all material tax returns and reports required by law to have been filed by it and has paid all material Taxes due and payable with respect to each such return, except any such Taxes or charges that (a) are not delinquent, (b) remain payable without penalty or interest, or (c) are being diligently contested in good faith by appropriate proceedings and for which adequate reserves in accordance with GAAP have been set aside on that Loan Party's or that Subsidiary's books. The Loan Parties and their Subsidiaries have made adequate reserves on their books and records in accordance with GAAP for all Taxes that have accrued but which are not yet due and payable.

9.14Solvency, etc. On the Closing Date, and immediately prior to and after giving effect to each borrowing under this Agreement and the use of the proceeds thereof, with respect to Borrower, individually, and the Loan Parties taken as a whole, (a) the fair value of its or their assets is greater than the amount of its or their liabilities (including disputed, contingent and unliquidated liabilities) as that value is established and liabilities evaluated in accordance with GAAP; (b) the present fair saleable value of its or their assets is not less than the amount that will be required to pay the probable liability on its or their debts as they become absolute and matured; (c) it is, and they are, able to realize upon its or their assets and pay its or their debts and other liabilities (including disputed, contingent and unliquidated liabilities) as they mature in the normal course of business; (d) it does not, and they do not, intend to, and it does not, and they do not, believe that it or they will, incur debts or liabilities beyond its or their ability to pay as those debts and liabilities mature; and (e) it is not, and they are not, engaged in or about to engage in business or a transaction for which its or their property would constitute unreasonably small capital.

<u>9.15Environmental Matters</u>. The on-going operations of each of the Loan Parties and their Subsidiaries comply in all respects with all Environmental Laws, except for non-compliance that could not (if enforced in accordance with applicable law) reasonably be expected to result, either individually or in the aggregate, in a Material Adverse Effect. Each of the Loan Parties and their Subsidiaries has obtained, and maintained in good standing, all licenses, permits, authorizations, registrations, and other approvals required under any Environmental Law and required for their respective ordinary course operations and each of the Loan Parties and their Subsidiaries is in compliance with all terms and conditions thereof, except where the failure to do so could not reasonably be expected to result in material liability to any of the Loan Parties and their Subsidiaries, and none of the properties or operations of the Loan Parties and their Subsidiaries, is subject to, and none of the Loan Parties

<u>58</u>

and their Subsidiaries reasonably anticipates the issuance of, (a) any written order from or agreement with any federal, state, or local governmental authority, or (b) any judicial or docketed administrative or other proceeding respecting any Environmental Law, Environmental Claim, or Hazardous Substance that could reasonably be expected to result, either individually or in the aggregate, in a Material Adverse Effect. No Hazardous Substances have been released or are present at any property, arising from operations prior to the Closing Date, or relating to any waste disposal of any Loan Party or any Subsidiary thereof that could reasonably be expected to result, either individually or in the aggregate, in a Material Adverse Effect.

<u>9.16Insurance</u>. Set forth on <u>Schedule 9.16</u> is a complete and accurate summary of the property and casualty insurance program of the Loan Parties and their Subsidiaries as of the Closing Date (including the names of all insurers, policy numbers, expiration dates, amounts and types of coverage, annual premiums, exclusions, deductibles, self-insured retention, and a description in reasonable detail of any self-insurance program, retrospective rating plan, fronting arrangement, or other risk assumption arrangement involving any of the Loan Parties and their Subsidiaries). Each of the Loan Parties and their Subsidiaries and their respective properties are insured, in all material respects, with what are reasonably believed by Borrower to be financially sound and reputable insurance companies that are not Affiliates of the Loan Parties, in such amounts, with such deductibles, and covering such risks as are customarily carried by companies of similar size, engaged in similar businesses, and owning similar properties in localities where the Loan Parties and their Subsidiaries operate.

9.17Real Property. Set forth on Schedule 9.17 is a complete and accurate list, as of the Closing Date, of the address of all real property owned or leased by any of the Loan Parties and their Subsidiaries.

<u>9.18Information</u>. All information (other than the projections and forecasts referred to below and information of a general economic or general industry nature) heretofore or contemporaneously with this Agreement furnished in writing by any of the Loan Parties and their Subsidiaries to Administrative Agent or any Lender for purposes of or in connection with this Agreement and the transactions contemplated by this Agreement is, and all written information (other than the projections and forecasts referred to below and information of a general economic or general industry nature) hereafter furnished by or on behalf of any of the Loan Parties and their Subsidiaries to Administrative Agent or any Lender pursuant to or in connection with this Agreement will be, when furnished, taken as a whole, true and accurate in all material respects as of the date on which such information is delivered or certified and does not or will not, when furnished, contain any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements contained therein not materially misleading in light of the circumstances under which such statements are made (after giving effect to all supplements and updates thereto that are made in accordance with the terms of the Loan Documents). All projections and forecasts provided by Borrower are based on good faith estimates and assumptions believed by Borrower to be reasonable as of the date of the applicable projections, assumptions or forecasts; provided that actual results during the period or periods covered by any such projections and forecasted results.

<u>9.19Location of Bank Accounts</u>. <u>Schedule 9.19</u> sets forth a complete and accurate list as of the Closing Date of all deposit, checking, and other bank accounts, all securities and other accounts maintained with any broker dealer or other securities intermediary, and all other similar accounts maintained by each Loan Party, together with a description thereof (including the bank, broker dealer, or securities intermediary at which each such account is maintained and the account number and the purpose thereof).

9.20[Reserved].

9.21Intellectual Property. Except as set forth on Schedule 9.21, each of the Loan Parties and their Subsidiaries owns or licenses or otherwise has the right to use all patents, patent applications,

trademarks, trademark applications, service marks, trade names, copyrights, copyright applications, and other intellectual property rights ("Intellectual Property") that are reasonably necessary for the operation of its business as currently conducted, without infringement upon or conflict with the Intellectual Property rights of any other Person with respect thereto, except for any infringements and conflicts that, individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect. Set forth on <u>Schedule 9.21</u> is a complete and accurate list as of the Closing Date of all such material Intellectual Property of each of the Loan Parties and their Subsidiaries. The operation of the business of the Loan Parties, as currently conducted, does not infringe upon or conflict with any Intellectual Property rights of any other Person, and no claim or litigation regarding any Intellectual Property rights of any other Person is pending or threatened against the Loan Parties or any of their Subsidiaries, except for any infringements and conflicts that could not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

9.22[Reserved].

<u>9.23Employee and Labor Matters</u>. Except as could not reasonably be expected to result, individually or in the aggregate, in Material Adverse Effect, there is (a) no unfair labor practice complaint pending or, to the knowledge of any Borrower, threatened against any Loan Party or any Subsidiary thereof before any governmental authority and no grievance or arbitration proceeding pending or threatened against any of the Loan Parties and their Subsidiaries that arises out of or under any collective bargaining agreement; and (b) no strike, labor dispute, slowdown, stoppage or similar action or grievance pending or, to the knowledge of Borrower, threatened against any of the Loan Parties and their Subsidiaries.

9.24[Reserved].

9.25Name; Jurisdiction of Organization; Organizational ID Number; Chief Place of Business; Chief Executive Office; FEIN. Schedule 9.25 sets forth a complete and accurate list as of the Closing Date of (a) the exact legal name of each of the Loan Parties and their Subsidiaries; (b) the jurisdiction of organization of each of the Loan Parties and their Subsidiaries; (c) the organizational identification number of each Loan Party (or indicates that that Loan Party has no organizational identification number); (d) each place of business of each of the Loan Parties and their Subsidiaries; (e) the chief executive office of each of the Loan Parties and their Subsidiaries; and (f) the federal employer identification number of each Loan Party.

<u>9.26Locations of Collateral</u>. There is no location at which any Loan Party has any material portion of Collateral (except for inventory in transit in the ordinary course of business) other than those locations listed on <u>Schedule 9.26</u>. <u>Schedule 9.26</u> contains a true, correct, and complete list, as of the Closing Date, of the names and addresses of each warehouse at which Collateral of each Loan Party is stored. None of the receipts received by any Loan Party from any warehouse states that the goods covered thereby are to be delivered to bearer or to the order of a named Person or to a named Person and that named Person's assigns.

<u>9.27Security Interests</u>. The Guaranty and Collateral Agreement creates in favor of Administrative Agent, for the benefit of Administrative Agent and the Lenders, a legal, valid, and enforceable security interest in the Collateral. Upon the filing of the UCC-1 financing statements described in <u>Section 12.1.15</u> and the recording of the collateral assignments referred to in the Guaranty and Collateral Agreement in the United States Patent and Trademark Office and the United States Copyright Office, as applicable, the security interests in and Liens on the Collateral granted under the Guaranty and Collateral Agreement (to the extent such Collateral can be perfected by the filing of UCC-1 financing statements or recording such collateral assignments and the taking of such other action required by this Agreement and the Collateral Documents) will be perfected, first-priority security interests, and no further recordings or

filings are or will be required in connection with the creation, perfection, or enforcement of those security interests and Liens, other than (a) the filing of continuation statements in accordance with applicable law; and (b) the recording of the collateral assignments referred to in the Guaranty and Collateral Agreement in the United States Patent and Trademark Office and the United States Copyright Office, as applicable, with respect to after-acquired U.S. patent and trademark applications and registrations and U.S. copyright registrations.

9.28No Default. No Default or Event of Default exists or would result from the incurrence by any Loan Party of any Debt under this Agreement or under any other Loan Document.

<u>9.29Hedging Agreements</u>. None of the Loan Parties and their Subsidiaries is a party to, nor will it be a party to, any Hedging Agreement other than a bona fide (not speculative) Hedging Agreement.

<u>9.300FAC</u>. Each of Borrower and its Subsidiaries is and will remain in compliance in all material respects with all U.S. economic sanctions laws, Executive Orders and implementing regulations as promulgated by the U.S. Treasury Department's Office of Foreign Assets Control ("<u>OFAC</u>"), and all applicable anti-money laundering and counter-terrorism financing provisions of the Bank Secrecy Act and all regulations issued pursuant to it. None of Borrower, its Subsidiaries or its or their respective Affiliates is (a) a Person designated by the U.S. government on the list of the Specially Designated Nationals and Blocked Persons (the "<u>SDN List</u>") with which a U.S. Person cannot deal with or otherwise engage in business transactions; (b) a Person who is otherwise the target of U.S. economic sanctions laws such that a U.S. Person cannot deal or otherwise engage in business transactions with that Person; or (c) controlled by (including, without limitation ,by virtue of that Person being a director or owning voting shares or interests), or acts, directly or indirectly, for or on behalf of, any Person on the SDN List or a foreign government that is the target of U.S. economic sanctions prohibitions such that the entry into, or performance under, this Agreement or any other Loan Document would be prohibited under U.S. law.

<u>9.31Patriot Act</u>. Each of Borrower and its Subsidiaries and its and their Affiliates are in compliance with (a) the Trading with the Enemy Act, and each of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B Chapter V) and any other enabling legislation or executive order relating thereto; (b) the Patriot Act; and (c) other federal or state laws relating to *"know your customer*" and anti-money laundering rules and regulations. No part of the proceeds of any Loan will be used directly or indirectly for any payments to any government official or employee, political party, official of a political party, candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977.

SECTION 10 AFFIRMATIVE COVENANTS.

Until Payment in Full, Borrower shall, unless at any time the Required Lenders otherwise expressly consent in writing, do the following:

10.1Reports, Certificates and Other Information. Furnish to Administrative Agent and each Lender:

<u>10.1.1</u> Annual Reports. On or prior to the date that is the earlier of (x) one hundred twenty (120) days after the end of each Fiscal Year (commencing with the Fiscal Year ending December 31, 2021) and (y) the date following the end of a Fiscal Year (commencing with the Fiscal Year ending December 31, 2021) by which Parent is required to file Form 10-K with the SEC, (i) the consolidated balance sheet of Borrower as at the end of such Fiscal Year and the related consolidated statements of operations, stockholders' equity and cash flows of Borrower for such

<u>61</u>

Fiscal Year and setting forth, in reasonable detail, in comparative form the corresponding figures for the previous Fiscal Year and (ii) with respect to such consolidated financial statements, a report thereon of an independent certified public accountant of recognized national standing or another accounting firm reasonably acceptable to Administrative Agent (which report shall not be subject to a "going concern" or scope of audit qualification (except for any such qualification pertaining to, or disclosure of an exception or qualification resulting from, (x) the maturity (or impending maturity) of any Debt or (y) any breach or anticipated breach of any financial covenant hereunder), and shall state that such consolidated financial statements fairly present, in all material respects, the consolidated financial position of Borrower as at the dates indicated and its income and cash flows for the periods indicated in conformity with GAAP in all material respects.

10.1.2 Quarterly Reports. Within sixty (60) days after the end of the first three Fiscal Quarters of each Fiscal Year, unaudited consolidated balance sheets of the Company and its Subsidiaries as of the end of that Fiscal Quarter, together with consolidated statements of earnings and a consolidated statement of cash flows for that Fiscal Quarter and for the period beginning with the first day of that Fiscal Year and ending on the last day of that Fiscal Quarter, together with a comparison with the corresponding period of the previous Fiscal Year and a comparison with the budget for that period of the current Fiscal Year, together with a management discussion and analysis, all certified by a Senior Officer of the Company.

10.1.3 <u>Monthly Reports</u>. Promptly when available and in any event within 30 days after the end of each month, (i) unaudited consolidated balance sheets of the Company and its Subsidiaries as of the end of that month, together with consolidated statements of earnings and, if prepared for internal use, a consolidated statement of cash flows for that month and for the period beginning with the first day of that Fiscal Year and ending on the last day of that month and (ii) a certificate setting forth (1) Adjusted Revenue for such month, (2) the amount of Liquidity and Unrestricted Cash as at the end of such month and (3) a snapshot and roll forward report with supporting documents reflecting credit performance of the Company's credit and financial products, including Credit Builder Plus and InstaCashInstacash and such other products as Administrative Agent may request from time to time.

Notwithstanding the foregoing, the obligations in Sections 10.1.1 and 10.1.2 may be satisfied with respect to financial information of the Company and its Subsidiaries by furnishing the Form 10-K or 10-Q of Parent filed with the SEC; provided that, (i) if (1) such financial statements relate to such other Person, such financial statements or the Form 10-K or Form 10-Q, as applicable, shall be accompanied by consolidating information (which need not be audited) that summarizes in reasonable detail the differences between the information relating to such Person, on the one hand, and the information relating to Borrower and its consolidated subsidiaries on a standalone basis, on the other hand, which consolidating information shall be certified by a Senior Officer of Borrower as having been fairly presented in all material respects and (ii) to the extent such statements are in lieu of statements required to be provided under Section 10.1.1, such statements shall be accompanied by an audit report that would satisfy the applicable requirements set forth in Section 10.1.1 as if the references to "Borrower" therein were references to such Person; provided further, the Company shall notify Administrative Agent upon any such filing.

<u>10.1.4</u> <u>Compliance Certificates</u>. Contemporaneously with the furnishing of a copy of each annual audit report pursuant to <u>Section 10.1.1</u> and each set of quarterly statements pursuant to <u>Section 10.1.2</u>, a duly certified calculation of Enterprise Value as of the last day of the most recently ended annual and/or quarterly period, as applicable, and a duly completed compliance certificate in the form of <u>Exhibit B</u>, with appropriate insertions, dated the date of that annual report or those statements and signed by a Senior Officer of the Company, containing (a) a computation of each of the financial ratios and restrictions set forth in <u>Section 11.12</u>; (b) a certification to the effect that

<u>62</u>

that Senior Officer has not become aware of any Default or Event of Default that has occurred and is continuing or, if there is any such event, describing it and the steps, if any, being taken to cure it; and (c) a written statement of the Company's management setting forth a discussion of the Company's and its Subsidiaries' financial condition, changes in financial condition, and results of operations.

10.1.5 <u>Reports to the SEC and to Shareholders</u>. Promptly upon the filing or sending thereof, copies of all regular, periodic, or special reports of any Loan Party filed with the SEC; copies of all registration statements of any Loan Party filed with the SEC (other than on Form S-8); and copies of all proxy statements or other communications made to security holders generally.

10.1.6 Notice of Default, Litigation, and ERISA Matters. Promptly upon becoming aware of any of the following, written notice describing the same and the steps being taken by the Loan Parties and their Subsidiaries affected thereby with respect thereto:

(a) the occurrence of a Default or an Event of Default;

(b) the commencement of, or any material adverse development in, any litigation or proceeding affecting any of the Loan Parties and their Subsidiaries or their respective property (i) in which the amount of damages claimed is \$2,000,000 (or its equivalent in another currency or currencies) or more in the aggregate for all such litigations or proceedings; (ii) in which the relief sought is an injunction or other stay of the performance of this Agreement or any other Loan Document; or (iii) which, if adversely determined, could reasonably be expected to have a Material Adverse Effect;

(c) (i) the institution of any steps by Borrower or any other member of the Controlled Group or the PBGC to terminate any Pension Plan under Section 4041(c) or Section 4042 of ERISA; (ii) the failure of any member of the Controlled Group to make a required contribution to any Pension Plan (if that failure is sufficient to give rise to a Lien under Section 303(k) of ERISA) or to any Multiemployer Pension Plan; (iii) the taking of any action with respect to a Pension Plan that would result in the requirement that any Loan Party furnish a bond or other security to the PBGC or that Pension Plan; (iv) the occurrence of any event with respect to any Pension Plan or Multiemployer Pension Plan that could result in the incurrence by any member of the Controlled Group of any material liability, fine, or penalty (including any claim or demand for withdrawal liability or partial withdrawal from any Multiemployer Pension Plan); (v) any notice received by Borrower or any other member of the Controlled Group that any Multiemployer Pension Plan is or is reasonably expected to become insolvent; or (vi) a Termination Event has occurred or is reasonably expected to occur with respect to any Pension Plan that could reasonably be expected to result in Obligations or other liability to any Loan Party in excess of \$3,000,000;

(d) (i) any action by any Governmental Authority to suspend, revoke or terminate any material permit or material license of any Loan Party (excluding, for avoidance of doubt, any voluntary termination or expiration of any such license or permit initiated by a Loan Party prior and unrelated to any such action by Governmental Authority), (ii) any assertion or threat in writing of any class action claims (in court or in arbitration) or 1,000 substantially similar claims (in court or in arbitration), such that one or more such claims or actions individually or in the aggregate involve potential monetary relief against the Loan Parties exceeding the Threshold Amount or challenge the legality of revenues generated by the Loan Parties exceeding the Threshold Amount, as measured over the most recently ended calendar year and (iii) any action initiated or threatened in writing by any Governmental Authority (or initiated defensively by any Loan Party against any Governmental Authority), such that one or more such claims or actions individually or in the

<u>63</u>

aggregate involve potential monetary relief against the Loan Parties exceeding the Threshold Amount or challenge the legality of revenues generated by the Loan Parties exceeding the Threshold Amount, as measured over the most recently ended calendar year;

- (e) any cancellation or material change in any material insurance maintained by any Loan Party;
- (f) any material violation of, or material non-compliance with, any material requirement of law by any Loan Party; or

(g) any other event (including (i) any violation of any Environmental Law or the assertion of any Environmental Claim, or (ii) the enactment or effectiveness of any law, rule, or regulation) in each case that could reasonably be expected to result in a Material Adverse Effect.

10.1.7 Real Estate. Promptly upon any of the Loan Parties and their Subsidiaries acquiring or leasing any real property after the Closing Date, an updated version of Schedule 9.17 showing information as of the date of delivery.

10.1.8 <u>Management Reports</u>. Promptly upon receipt thereof, copies of all detailed financial and management reports submitted to Borrower by independent auditors in connection with each annual or interim audit made by those auditors of the books of Borrower.

<u>10.1.9</u> Projections. Unless otherwise waived in the sole discretion of Administrative Agent, as soon as practicable, and in any event not later than 75 days after the first day of each Fiscal Year (and, solely to the extent otherwise prepared by Borrower, within 75 days after the consummation of any Permitted Acquisition; provided that such financial projections shall be required to be prepared and delivered by Borrower for any Permitted Acquisition or series of related Permitted Acquisitions involving consideration individually or in the aggregate in excess of \$50,000,000), financial projections for the Company and its Subsidiaries for such current Fiscal Year (including a business plan, monthly operating and cash flow budgets and a capital expenditures budget) prepared in a manner consistent with the projections delivered by Borrower to Administrative Agent prior to the Closing Date or otherwise in a manner reasonably satisfactory to Administrative Agent (which financial projections and forecasts shall be based on good faith estimates and assumptions believed by Borrower to be reasonable as of the date of delivery by Borrower to Administrative Agent (it being understood that actual results during the period or periods covered by any such projections may differ materially from projected or forecasted results)).

10.1.10 Other Information. Promptly from time to time, all other information (including, without limitation, business or financial data, reports, appraisals and projections) concerning any of the Loan Parties and their Subsidiaries or their respective properties or business as any Lender or Administrative Agent reasonably requests.

10.1.11 Instacash.

(a) (i) Promptly upon receipt or sending thereof (as applicable), copies of each Monthly Servicing Report delivered under the Servicing Agreement and (ii) promptly upon receipt or sending thereof (as applicable), all notices and reports received from or sent to Sound Point or its Affiliates relating to, or any Loan Party or its Subsidiaries otherwise becoming aware of (w) a Regulatory Trigger Event (as defined in the SP Forward Flow Agreement), (x) an Event of Termination (as defined in the SP Forward Flow Agreement) and any cure of an Event of

Termination (or any other event that causes a prospective Event of Termination to no longer be deemed an Event of Termination pursuant to the terms of the SP Forward Flow Agreement), (y) a Material Underperformance Event (as defined in the SP Forward Flow Agreement) or (z) any repurchase made by any Loan Party or its Subsidiaries of Instacash Receivables.

(b) Bi-weekly, by no later than Friday of each week beginning with the week ending July 5, 2024 and until the "Group Outstanding Purchased Amount" (as defined in the SP Forward Flow Agreement) first exceeds \$30,000,000, a report in form and substance mutually agreed and reasonably acceptable to the Administrative Agent, providing certain details of sales of Instacash Receivables under the SP Forward Flow Agreement since the date of the last such bi-weekly report.

10.2Books, Records, and Inspections. Keep, and cause each of the Loan Parties and their Subsidiaries to keep, its books and records in accordance with sound business practices sufficient to allow the preparation of financial statements in accordance with GAAP; permit, and cause each other Loan Party and each Subsidiary of each Loan Party to permit, any Lender or Administrative Agent or any representative, agent, or advisor thereof to inspect the properties and operations of the Loan Parties and their Subsidiaries; and permit, and cause each other Loan Party and each Subsidiary of each Loan Party to permit, at any reasonable time (on a Business Day) and with reasonable notice (or at any time without notice if an Event of Default exists), any Lender or Administrative Agent or any representative, agent, or advisor thereof to visit any or all of its offices, to discuss its financial matters with its officers and its independent auditors (and Borrower hereby authorizes all such independent auditors to discuss from) any of its books or other records; and permit, and cause each other Loan Party and each Subsidiaries, to perform appraisals of the equipment of the Loan Parties and their Subsidiaries, and each Subsidiaries, to perform appraisals of the equipment of the Loan Parties and their Subsidiaries, and to inspect, audit, check and make copies of and extracts from the books, records, computer data, computer programs, journals, orders, receipts, correspondence and other data relating to inventory, accounts, and any other Collateral. All such visits, inspections, appraisals or audits conducted by Administrative Agent in any Fiscal Year and (ii) Borrower will not be required to reimburse Administrative Agent in any Fiscal Year for more than \$100,000 for such visits, inspections, appraisals, or audits.

10.3Maintenance of Property; Insurance.

(a) Except as otherwise permitted hereunder, keep, and cause each of the Loan Parties and their Subsidiaries to keep, all material property useful and necessary in the business of the Loan Parties and their Subsidiaries in good working order and condition, ordinary wear and tear and casualty excepted.

(b) Maintain, and cause each of the Loan Parties and their Subsidiaries to maintain, with responsible insurance companies, all insurance coverage as may be required by any law or governmental regulation or court decree or order applicable to it, general liability insurance and business interruption insurance in such amounts and duration, and with such deductibles, is customarily carried by companies engaged in similar businesses and owning similar properties in localities where Borrower, or the applicable Subsidiary operates, and all other insurance, to such extent and against such hazards and liabilities, as is customarily maintained by companies similarly situated, but which must insure against all risks and liabilities of the type identified on <u>Schedule 9.16</u> and must have insured amounts no less than, and deductibles no higher than, those set forth on that schedule; and, upon request of Administrative Agent or any Lender, furnish to Administrative Agent or that Lender original or electronic copies of policies

<u>65</u>

evidencing that insurance and a certificate setting forth in reasonable detail the nature and extent of all insurance maintained by the Loan Parties and their Subsidiaries. Borrower shall cause each issuer of an insurance policy in respect of any Loan Party to provide Administrative Agent with an endorsement (i) showing Administrative Agent as lender's loss payee with respect to each policy of property or casualty insurance and naming Administrative Agent as an additional insured with respect to each policy of liability insurance; (ii) providing that thirty (30) days' notice will be given to Administrative Agent prior to any cancellation of, material reduction or change in coverage provided by or other material modification to that policy; and (iii) reasonably acceptable in all other respects to Administrative Agent. Each Loan Party shall execute and deliver to Administrative Agent a collateral assignment, in form and substance satisfactory to Administrative Agent, of each business interruption insurance policy maintained by that Loan Party.

(c) Unless Borrower provides Administrative Agent with evidence of the insurance coverage required by this Agreement, Administrative Agent may purchase insurance at Borrower's expense, after notice to Borrower, to protect Administrative Agent's and the Lenders' interests in the Collateral. This insurance may, but need not, protect any Loan Party's interests. The coverage that Administrative Agent purchases might not pay any claim that is made against any Loan Party in connection with the Collateral. Borrower may later cancel any insurance purchased by Administrative Agent, but only after providing Administrative Agent with evidence that Borrower has obtained insurance as required by this Agreement. If Administrative Agent purchases insurance for the Collateral, Borrower will be responsible for the costs of that insurance, including interest and any other charges that might be imposed with the placement of the insurance, until the effective date of the cancellation or expiration of the insurance. The costs of the insurance may be added to the principal amount of the Loans owing under this Agreement. The costs of the insurance may be more than the cost of the insurance the Loan Parties might be able to obtain on their own.

10.4Compliance with Laws; Payment of Taxes and Liabilities.

(a) Comply, and cause each of the Loan Parties and their Subsidiaries to comply, in all respects with all applicable laws, rules, regulations, decrees, orders, judgments, licenses, and permits, except where failure to comply could not reasonably be expected to have a Material Adverse Effect;

(b) Without limiting Section 10.4(a), ensure, and cause each of the Loan Parties and their Subsidiaries to ensure, that no Person who owns a controlling interest in or otherwise controls any of the Loan Parties and their Subsidiaries is (i) listed on the SDN List maintained by OFAC and/or any other similar lists maintained by OFAC pursuant to any authorizing statute, Executive Order, or regulation; or (ii) a Person designated under Section 1(b), (c), or (d) of Executive Order No. 13224 (September 23, 2001), any related enabling legislation, or any other similar Executive Orders.

(c) Without limiting Section 10.4(a), comply, and cause each of the Loan Parties and their Subsidiaries to comply, with all applicable Bank Secrecy Act and anti-money laundering laws and regulations.

(d) File all federal, state and other material tax returns and reports required to be filed, and cause each of the Loan Parties and their Subsidiaries to pay all federal, state and other taxes, assessment, fees and other governmental charges levied or imposed upon them or their properties, income or assets as the same shall become due and payable, but none of the Loan Parties and their Subsidiaries will be required under this Section 10.4(d) to pay any such tax or charge so long as the failure to do so could not reasonable be expected to have a Material Adverse Effect or that Loan Party or that Subsidiary is contesting the validity thereof in good faith by appropriate proceedings and has set aside on its books adequate reserves with respect thereto in accordance with GAAP, and, in the case of a claim that could become a Lien on any

<u>66</u>

Collateral, those contest proceedings stay the foreclosure of that Lien or the sale of any portion of the Collateral to satisfy that claim.

(e) Each of the Loan Parties and their Subsidiaries obtain and maintain all consents, approvals and authorizations of, make all declarations or filings with, and give all notices to, all Government Authorities that are necessary to continue their respective businesses as currently conducted, in each case except where the failure to do so would not reasonably be expected to cause a Material Adverse Effect.

<u>10.5Maintenance of Existence, etc.</u> Maintain and preserve, and (subject to <u>Section 11.4</u>) cause each of the Loan Parties and their Subsidiaries to maintain and preserve, (a) its existence and good standing in the jurisdiction of its organization and (b) its qualification to do business and good standing in each jurisdiction where the nature of its business makes that qualification necessary (other than any such jurisdictions in which the failure to be qualified or in good standing could not reasonably be expected to have a Material Adverse Effect).

10.6Use of Proceeds. Use the proceeds of the Loans solely to repay the Debt to be Repaid, for working capital purposes, repayment of Term A-2 Loans with Term B Loans or Incremental Loans and for other general business purposes and any other purpose not prohibited by this Agreement (including Permitted Acquisitions); and not use or permit any proceeds of any Loan to be used, either directly or indirectly, for the purpose, whether immediate, incidental or ultimate, of "purchasing or carrying" any Margin Stock.

10.7Employee Benefit Plans.

(a) Maintain, and cause each other member of the Controlled Group to maintain, each Pension Plan in compliance with all applicable requirements of law and regulations;

(b) Make, and cause each other member of the Controlled Group to make, on a timely basis, all required contributions to any Multiemployer Pension Plan; and

(c) Not, and not permit any other member of the Controlled Group to, (i) seek a waiver of the minimum funding standards of ERISA, (ii) terminate or withdraw from any Pension Plan or Multiemployer Pension Plan or (iii) take any other action with respect to any Pension Plan that could reasonably be expected to entitle the PBGC to terminate, impose liability in respect of, or cause a trustee to be appointed to administer, any Pension Plan;

unless the actions or events described in <u>clauses (a)</u>, (b), and/or (c) individually or in the aggregate could not reasonably be expected to have a Material Adverse Effect.

<u>10.8Environmental Matters</u>. If any release or threatened release or other disposal of Hazardous Substances occurs or has occurred on any real property or any other assets of any of the Loan Parties and their Subsidiaries, then Borrower shall, or shall cause the applicable Loan Party or the applicable Subsidiary of a Loan Party to, cause the prompt containment and removal of those Hazardous Substances and the remediation of that real property or other assets as necessary to materially comply with all applicable Environmental Laws and to preserve in all material respects the value of that real property or other assets. Without limiting the generality of the foregoing, Borrower shall, and shall cause the Loan Parties and their Subsidiaries to, comply in all material respects with any applicable federal or state judicial or administrative order requiring the performance at any real property of any of the Loan Parties and their Subsidiaries of activities in response to the release or threatened release of a Hazardous Substance. Borrower shall, and shall cause its Subsidiaries to, dispose of all Hazardous Substances in compliance in all material respects with Environmental Laws.

<u>67</u>

<u>10.9Lender Meetings</u>. Upon the request of Administrative Agent (which request, so long as no Event of Default shall have occurred and be continuing, shall not be made more than once in any Fiscal Quarter) and upon reasonable prior notice, participate in customary meetings with Administrative Agent and Lenders (at Borrower's corporate offices (or at such other location as may be agreed to by Borrower and Administrative Agent) or by teleconference at such time as may be reasonably agreed to by Borrower and Administrative Agent, to discuss business, financial, legal, regulatory, product offering and/or other customary matters relating to the Company and its Subsidiaries (including related participation and follow-up by counsels to the Company and Administrative Agent).

<u>10.10Further Assurances</u>. Take, and cause each other Loan Party to take, all actions as are necessary or as Administrative Agent or the Required Lenders reasonably request from time to time to ensure that the Obligations of each Loan Party under the Loan Documents are secured by a first-priority perfected Lien in favor of Administrative Agent (subject to Permitted Liens) on the Collateral and guaranteed by each Loan Party (including, within thirty (30) days of the acquisition or creation thereof (or any longer period Administrative Agent agrees to in its sole discretion), any Subsidiary acquired or created after the Closing Date, but excluding each Excluded Subsidiary) and as Administrative Agent reasonably determines, including (i) the execution and delivery of guaranties, security agreements, pledge agreements, mortgages (including, without limitation, leasehold mortgages), deeds of trust (including, without limitation, leasehold deeds of trust), financing statements, opinions of counsel, and other documents, in each case in form and substance reasonably satisfactory to Administrative Agent, and the filing or recording of any of the foregoing; (ii) the delivery of certificated securities and other Collateral with respect to which perfection is obtained by possession to the extent required by the terms of the Guaranty and Collateral Agreement; and (iii) with respect to any real property acquired by any Loan Party after the Closing Date in which such Loan Party has a fee interest, the delivery with respect to real property with a value in excess of \$500,000 (to the extent requested by Administrative Agent, in all right, title and interest of the applicable Loan Party in that real property, together with all Mortgage-Related Documents and a legal opinion of special counsel for the applicable Loan Party for the state in which that real property is located in form and substance reasonably acceptable to Administrative Agent.

10.11Deposit Accounts. Unless Administrative Agent otherwise consents in writing, from and after the date which is forty-five (45) days after the Closing Date (or such later date as Administrative Agent may agree), maintain, and cause each other Loan Party to maintain, all of their deposit accounts and securities accounts (including, for avoidance of doubt, the Instacash Servicer Account), other than Excluded Deposit Accounts, with an institution that has entered into one or more Control Agreements with Administrative Agent and the applicable Loan Party granting "control" (as defined in the UCC) of each applicable account to Administrative Agent.

10.12Collateral Access Agreements. If requested by Administrative Agent in writing with respect to any property leased by any Loan Party in which Collateral is located or books and records relating thereto are maintained, the Company shall use reasonable best efforts to promptly deliver a Collateral Access Agreement with respect to such premises. The parties agree that failure to deliver any such Collateral Access Agreement after application of such commercially reasonable efforts shall not be a Default or Event of Default hereunder.

10.13Guarantor Joinders. Within 45 days after the Closing Date (or such longer period as the Administrative Agent may agree), cause each Subsidiary of Parent as of the Closing Date that is not a Loan Party as of the Closing Date (other than any Excluded Subsidiary) to become a Guarantor under the Loan Documents and otherwise satisfy the requirements set forth in Section 10.10 hereof with respect thereto (it

<u>68</u>

being understood and agreed that no opinions of counsel shall be required in connection with the satisfaction of this Section 10.13 to the extent that the Subsidiaries becoming Guarantors pursuant to this Section 10.13 do not in the aggregate comprise more than 10% of assets or revenues of Parent and its Subsidiaries on a consolidated basis).

SECTION 11 NEGATIVE COVENANTS

Until Payment in Full, Borrower (and, for purposes of Section 11.4, Parent) shall, unless at any time the Required Lenders otherwise expressly consent in writing, do the following:

11.1Debt. Not, and not permit any of the Loan Parties and their Subsidiaries to, create, incur, assume, or suffer to exist any Debt, except the following:

(a) Obligations under this Agreement and the other Loan Documents;

(b) Debt of any of the Loan Parties and their Subsidiaries secured by Liens permitted by Section 11.2(d), so long as the aggregate amount of all such Debt at any time outstanding does not exceed \$2,000,000;

(c) (i) Debt of any Loan Party to any other Loan Party; (ii) Debt of any Subsidiary which is not a Loan Party owing to a Loan Party so long as the aggregate amount at any time outstanding is otherwise permitted as an Investment under Section 11.9; (iii) Debt of any Loan Party owing to any Subsidiary which is not a Loan Party, which is expressly subordinated to the Obligations pursuant to the Intercompany Subordination Agreement; and (iv) Debt of any Subsidiary which is not a Loan Party to any other Subsidiary which is not a Loan Party to an

(d) Debt arising in connection with endorsement of instruments for deposit in the ordinary course of business;

(e) Debt of any Loan Party to any employee, officer, or director or any such Person's spouse, estate, or estate-planning vehicle to repurchase Equity Interests from that Person upon the death, disability, or termination of employment of that employee, officer of director, so long as the aggregate amount of all such Debt at any time outstanding does not exceed \$1,000,000;

(f) Hedging Obligations incurred for bona fide hedging purposes and not for speculative purposes;

(g) Debt described on Schedule 11.1 and any extension, renewal, replacement or refinancing thereof so long as the principal amount thereof is not increased;

(h) the Debt to be Repaid (so long as that Debt is repaid on the Closing Date with the proceeds of the initial Loans under this Agreement);

(i) Contingent Liabilities arising with respect to (i) customary indemnification obligations by any of the Loan Parties and their Subsidiaries in favor of purchasers in connection with dispositions permitted under Section 11.4, and (ii) agreements providing for indemnification, workers' compensation claims, self-insurance obligations, adjustment of purchase price or similar obligations, or from guarantees or letters of credit, surety bonds, appeals bonds or performance bonds securing the performance of any Loan Party pursuant to such agreements, in connection with Permitted Acquisitions, in each case in each of the foregoing, not in the form of Funded Debt and (iii) the guaranty by a Loan Party or any Subsidiary thereof of a lease, sublease, license, or sublicense entered into in the ordinary course of

business or such other Debt permitted by the terms of this Agreement by another Loan Party or any Subsidiary thereof;

(j) Debt in respect of credit cards, credit card processing services, debit cards, stored value cards, purchasing cards, clearinghouse arrangements, netting services, overdraft protection, and other like services, in each case, incurred in the ordinary course of business;

- (k) Debt existing on the Closing Date and set forth on Schedule 11.1;
- (1) Subordinated Debt;

(m) Debt acquired in connection with a Permitted Acquisition, including Debt to pay indemnification, contingent purchase price payments, or other purchase price adjustments, but solely to the extent (i) the business, division or assets acquired generated positive EBITDA for the four consecutive fiscal quarters most recently ended prior to that Acquisition and (ii) the amount of Debt so assumed in connection with any such Permitted Acquisition is not more than 4.0x the amount of EBITDA generated by the business, division or assets acquired, in an aggregate outstanding amount not at any time exceeding \$25,000,000; provided that (i) such Debt was not incurred in contemplation of such Permitted Acquisition and (ii) such Debt is secured only by the assets acquired in the applicable Permitted Acquisition (including any acquired Equity Interests) and the proceeds, products, replacements and accessions thereof;

(n) Debt of any SPV Financing Entity with respect to any purchase of accounts receivable, loan financing, warehouse, or other similar agreement, which shall not include any obligation or Debt of Borrower or any other Subsidiary that is not an SPV Financing Entity, so long as such debt is non-recourse to the Loan Parties and their Subsidiaries (other than any SPV Financing Entity) other than customary limited recourse items;

(<u>o</u>) Debt consisting of the financing of insurance premiums;

(p) other unsecured Debt of the Loan Parties and their Subsidiaries in an aggregate outstanding amount not at any time exceeding (x) prior to the EBITDA Trigger Date, \$5,000,000 and (y) following the EBITDA Trigger Date, \$15,000,000;

(q) to the extent constituting Debt, earn-out obligations incurred in connection with Permitted Acquisitions, so long as (i) the aggregate amount of all such Debt at any time outstanding (determined by reference to the maximum potential amount of any such Debt if the earn-out obligations become fully-earned) does not exceed \$10,000,000 plus the Malka Earn-Out, the Malka Make-Whole and the Even Financial Earn-Out and (ii) any such earn-out obligations other than the Malka Earn-Out, the Malka Make-Whole and Even Earn-Out are subordinated in right of payment to the Obligations;

(<u>r</u>) Debt constituting taxes, assessments, municipal or other governmental charges to the extent non-payment of any such amounts does not otherwise constitute a Default or Event of Default;

(s) unsecured Debt to trade creditors and pursuant to credit cards incurred in the ordinary course of business;

- (t) Debt incurred as a result of endorsing negotiable instruments received in the ordinary course of business;
- (<u>u</u>) Debt secured by Liens permitted under Section 11.2(c);

(v) to the extent constituting Debt, Investments permitted pursuant to Section 11.9;

(w) Debt comprised of letters of credit and other credit support obligations in an aggregate principal amount not to exceed \$10,000,000 which, if secured, is secured by Liens permitted pursuant to Section 11.2(s).

 (\underline{x}) (x) (i) any obligations incurred under ERISA or under any employee consulting agreements, (ii) accrued expenses, trade accounts payable, accruals for payroll and other liabilities accrued in the ordinary course of business (including on an intercompany basis) and (iii) liabilities associated with customer prepayments and deposits;

- (y) any Permitted Convertible Debt;
- (z) any Incremental Equivalent Debt; and

(aa) extensions, refinancings, modifications, amendments and restatements of any items of Debt incurred pursuant to (a), (b), (c), (l), (m), (q), (t), (w), (y) and (z) above, provided that the principal amount thereof is not increased—; and

(bb) Debt in respect of repurchase obligations pursuant to the SP Forward Flow Documents, provided that (i) any "Defective Receivables" repurchased as a result of a breach of the Eligibility Criteria set forth in clause 17 of Annex A of the SP Forward Flow Agreement may not be selected with the intention of repurchasing lower quality Instacash Receivables, taken as a whole, than the similar Instacash Receivables that could have been, but were not selected to be, repurchased at the same time, and (ii) the aggregate amount of Repurchase Price (as defined in the SP Forward Flow Agreement) paid by the Borrower (as repurchaser) pursuant to the SP Forward Flow Documents shall not exceed with respect to repurchases of "Underperforming Receivables" or "Optional Repurchased Receivables" under the SP Forward Flow Agreement (x) so long as Unrestricted Cash, after giving pro forma effect to any such repurchase, is at least \$85,000,000, the amount set forth in the proviso to Section 10(b) of the SP Forward Flow Agreement or (y) otherwise an amount per calendar year that results in or is otherwise reasonably projected or expected to result in, losses in excess of \$10,000,000 (provided that in any case such repurchases of "Underperforming Receivables" or "Optional Repurchased Receivables" may not occur more frequently than is permitted by the proviso to Section 10(b) of the SP Forward Flow Agreement).

To the extent Debt which is permitted to be incurred hereunder is incurred directly by a Loan Party or other Person and guaranteed by another Loan Party or any other Person the guarantee of such Debt shall be permitted under this Section 11.1 and this Agreement, without duplication.

11.2Liens. Not, and not permit any of the Loan Parties and their Subsidiaries to, create or permit to exist any Lien on any of its real or personal properties, assets, or rights of whatsoever nature (whether now owned or hereafter acquired), except the following:

(a) Liens for taxes or other governmental charges not at the time delinquent or thereafter payable without penalty, or being diligently contested in good faith by appropriate proceedings (as determined by Borrower in good faith), and, in each case, for which it maintains adequate reserves in accordance with GAAP and the execution or other enforcement of which is effectively stayed;

(b) Liens arising in the ordinary course of business any of the Loan Parties and their Subsidiaries (such as (i) Liens of landlords, carriers, warehousemen, mechanics and materialmen and other similar Liens imposed by law and (ii) Liens in the form of deposits or pledges incurred in connection with worker's compensation, unemployment compensation and other types of social security (excluding Liens

<u>71</u>

arising under ERISA) or in connection with surety bonds, bids, performance bonds, and similar obligations) for sums not overdue or being diligently contested in good faith by appropriate proceedings and not involving any advances or borrowed money or the deferred purchase price of property or services and, in each case, for which it maintains adequate reserves in accordance with GAAP and the execution or other enforcement of which is effectively stayed;

(c) Liens described on <u>Schedule 11.2</u> as of the Closing Date and replacements, renewals and extensions thereof on the assets currently subject to those Liens;

(d) subject to the limitation set forth in <u>Section 11.1(b)</u>, the following: (i) Liens arising in connection with Capital Leases (and attaching only to the property being leased); (ii) Liens existing on property at the time of the acquisition thereof by any of the Loan Parties and their Subsidiaries (and not created in contemplation of that acquisition); and (iii) Liens that constitute purchase money security interests on any property securing debt incurred for the purpose of financing all or any part of the cost of acquiring that property, so long as any such Lien attaches to the applicable property within 20 days of the acquisition thereof and attaches solely to the property so acquired;

(e) easements, rights of way, restrictions (including zoning restrictions), covenants, encroachments, and other similar real property charges or encumbrances, minor defects or irregularities in title, and other similar real property Liens not interfering in any material respect with the ordinary conduct of the business of any Loan Party or any Subsidiary thereof;

(f) leases, subleases, licenses, or sublicenses of the assets or properties of any of the Loan Parties and their Subsidiaries, in each case entered into in the ordinary course of business and not interfering in any material respect with the business of any of the Loan Parties and their Subsidiaries;

(g) customary set-off rights against depository accounts permitted under this Agreement in favor of banks at which any of the Loan Parties and their Subsidiaries maintains any such depository accounts, so long as those set-off rights secure only the obligations of that Loan Party or that Subsidiary to pay ordinary course fees and bank charges;

(h) Liens consisting of precautionary filings of UCC financing statements filed with respect to Operating Leases permitted under this Agreement and any interest of title of a lessor under any Operating Lease permitted under this Agreement;

- (i) Liens arising under the Loan Documents;
- (j) Liens arising from judgments in circumstances not constituting an Event of Default;

(k) Liens arising out of the sale or transfer to an SPV Financing Entity of consumer loans, cash advance receivables or membership receivables made in compliance with the terms of this Agreement, including Liens perfected by the filing of a Uniform Commercial Code financing statement covering such consumer loans, cash advance receivables or membership receivables naming Borrower, as "debtor" or "seller," and any grant of a "back-up" security interest in favor of such SPV Financing Entity in connection with any such sale or transfer;

(1) Liens in the form of cash collateralization to secure reimbursement obligations with respect to automated clearing house treasury management arrangements, credit cards, credit card processing services, debit cards, stored value cards, and purchase cards obtained in the ordinary course of business and otherwise permitted hereunder;

<u>72</u>

(m) to the extent constituting a Lien, any interest of a lessor or licensor (including any sublessor or sublicensor) under any lease or license (including any sublease or sublicense) entered into by Borrower or any Subsidiary in the ordinary course of business and covering only the assets so leased or licensed (but excluding any Capital Lease or purchase-money transaction covered by clause (c) above);

(n) Liens arising from judgments, decrees or attachments in circumstances not constituting an Event of Default under Section 13.1.8;

(<u>o</u>) Liens existing on the Closing Date and set forth on Schedule 11.2;

(p) other Liens with respect to obligations that do not in the aggregate exceed \$3,000,000 at any time outstanding;

(q) Liens securing Debt permitted in accordance with Section 11.1(r); provided that such Liens were not granted in contemplation of such Permitted Acquisition;

(I) deposits to secure the performance of bids, trade contracts and leases (other than Debt), statutory obligations, surety and appeal bonds, performance bonds and other obligations of a like nature incurred in the ordinary course of business;

(s) Liens on cash, Cash Equivalent Investments or other marketable securities to secure letters of credit and other credit support obligations of the Company and its Subsidiaries in an amount up to 105% of the face amount thereof; and

(t) Liens on securing Incremental Equivalent Debt permitted pursuant to Section 2.1.2(f); provided that a representative acting on behalf of the lenders or investors providing such Debt shall have entered into a Subordination Agreement (and the Lenders hereby authorize and direct the Administrative Agent to enter into any such Subordination Agreement)—; and

(u) Liens on Instacash Receivables (and related deposit accounts) arising pursuant to the SP Forward Flow Documents and subject, as applicable, to the SP Intercreditor Agreement.

<u>11.3Restricted Payments</u>. Not, and not permit any of the Loan Parties and their Subsidiaries to, (i) make any dividend or distribution to any holders of its Equity Interests; (ii) purchase or redeem any of its Equity Interests; (iii) pay any management fees, transaction-based fees, or similar fees to any of its equity holders or any Affiliate thereof; (iv) make any payment on account of Debt that has been contractually subordinated in right of payment to the Obligations if that payment is not permitted at that time under the applicable subordination terms and conditions; (v) make any voluntary or mandatory prepayment of principal with respect to any unsecured Debt or any Debt secured by a Lien that is junior to the Liens securing the Obligations (any prepayments or payments of the type describe in foregoing clauses (d) and (e), a "<u>Junior Debt Payment</u>"; or (vi) set aside funds for any of the foregoing, except that any Subsidiary may pay dividends or make other distributions to Borrower or any Subsidiary of Borrower (each of the foregoing clauses (a) though (f), a "<u>Restricted Payment</u>"), except that:

(a) Borrower may make distributions to Parent to pay (x) overhead and administrative legal, accounting and similar fees, costs, and expenses of Parent and (y) customary expenses of directors of Parent and customary expenses for meeting of the board of directors (including compensation expenses), in each case, incurred in the ordinary course of business;

<u>73</u>

(b) Loan Parties and their Subsidiaries may make payments on account of earn-out obligations permitted pursuant to Section 11.1(q), so long as in the case of such earn-out obligations, (1) immediately before and immediately after giving effect to such payment, no Default or Event of Default exists or would result therefrom and (2) Borrower is in pro forma compliance with all of the financial covenants set forth in Section 11.12 (recomputed for the most recent Fiscal Quarter for which financial statements have been delivered);

(c) Borrower may make payments on account of Subordinated Debt to the extent such payments are permitted pursuant to the terms of the applicable Subordination Agreement;

(d) Borrower may make Restricted Payments in an aggregate amount not to exceed \$3,000,000 in any Fiscal Year during the term of this Agreement (1) to any officers to purchase or redeem Equity Interests of Borrower or Parent issued to officers, directors and employees, so long as no Event of Default exists or would result therefrom; (2) to any directors to pay the expenses of directors of Borrower or Parent and expenses for meeting of the board of directors subject to the limitations on amounts contained in this Agreement; and (3) to any officers in the ordinary course of business so long as such payment is to cover reimbursable expenses of such officer

(e) Borrower may make Restricted Payments that are Tax Distributions;

(f) Borrower and each Subsidiary may declare and make dividend payments or other Restricted Payments payable solely in the Equity Interests (other than Disqualified Equity Interests) of such Person (and, in the case of such a Restricted Payment by a non-wholly owned Subsidiary, to Borrower and any other Subsidiary and to each other owner of Equity Interests of such Subsidiary based on their relative ownership interests of the relevant class of Equity Interests);

(g) Borrower or any of its Subsidiaries (i) may repurchase Equity Interests if such Equity Interests represent a portion of the exercise price of any option or warrant upon the exercise thereof and (ii) may make cash payments in lieu of issuing fractional or "odd lot" Equity Interests in connection with any Permitted Acquisition or in lieu of issuing fractional shares in connection with the exercise of warrants, options or other securities convertible into or exchangeable for Equity Interests of Parent and direct or indirect parent thereof;

(h) Borrower or any of its Subsidiaries may make Restricted Payments in respect of the purchase price, the working capital adjustments, the purchase price adjustments and any holdback obligations in connection with any Permitted Acquisition or other permitted Investments;

(i) Borrower may make Restricted Payments to Parent, the proceeds of which shall be used solely:

(i) to pay franchise Taxes and other fees and Taxes (other than income Taxes) imposed upon it and expenses required to maintain its corporate existence;

(ii) to pay customary salary, bonus and other similar benefits payable to officers and employees of Parent or any direct or indirect parent thereof to the extent such salaries, bonuses and other benefits are attributable to the ownership or operation of Borrower and its Subsidiaries; or

(iii) to pay insurance premiums to the extent attributable to Parent or any direct or indirect parent thereof;

<u>74</u>

(j) so long as no Default or Event of Default shall have occurred and be continuing or would result therefrom, Loan Parties may make other Restricted Payments in an aggregate amount (x) prior to the EBITDA Trigger Date, not to exceed \$350,000 and (y) from and after the EBITDA Trigger Date not to exceed the lesser of \$10,000,000 and 10% of EBITDA for the Computation Period most recently ended, in the case of this clause (y) so long as the Total Debt to EBITDA Ratio on a pro forma basis immediately after giving effect to such Restricted Payment does not exceed 4.00:1.00;

(k) to make any payment on behalf of and for the benefit of Borrower or its Subsidiaries that is otherwise permitted to be made by Borrower or its Subsidiaries pursuant to the terms hereof (in lieu of Borrower or such Subsidiary making such payment); provided that any such payment shall be deemed to be made under the applicable basket that Borrower is relying on for purposes of making such payment and such payment shall be treated as if paid by Borrower or such Subsidiary for purposes of calculating Consolidated Net Income, EBITDA and Excess Cash Flow;

(1) any Subsidiary may make a Restricted Payment its equity holders on a pro rata basis (or greater than pro rata basis to any other Subsidiary of Borrower to Borrower);

(m) (x) Parent may pay (and/or Borrower may make Restricted Payments to Parent to enable Parent to pay) the Even Financial Dividend and (y) Parent may pay the Malka Make-Whole in the form of (x) shares of common stock par value \$0.0001 per share of Parent and/or (y) solely with the consent of the Administrative Agent, cash (in which case of this clause (y), Borrower may make Restricted Payments to Parent to pay such cash amount);

(n) so long as no Event of Default pursuant to Section 13.1.1, 13.1.4, 13.1.5(a) (solely as a result of failure to comply with Section 11.12) or 13.1.8(b) or (c) exists or would result therefrom, Borrower may make Restricted Payments to Parent to permit Parent to, and Parent may, (i) make interest payments owing to the holders of any Permitted Convertible Debt, so long as the interest rate on such Permitted Convertible Debt is not in excess of 7.00% per annum, and make other customary payments and/or deliveries required by the terms of any Permitted Convertible Debt (including, without limitation, payments due upon required repurchase thereof and/or payments and deliveries due upon conversion thereof), (ii) pay the premium in respect of, and otherwise perform its obligations under, any Permitted Bond Hedge Transaction and (iii) make any payments and/or deliveries required by the terms of, and otherwise perform its obligations under, any Permitted Warrant Transaction (including, without limitation, making payments and/or deliveries due upon exercise and settlement or termination thereof);

(0) Borrower or any of its Subsidiaries may make Junior Debt Payments in an aggregate principal amount not to exceed \$500,000 in any Fiscal Year so long as at the time of any such prepayment, no Event of Default exists or would result therefrom; and

(p) any refinancing, purchase, defeasance, redemption, repurchase, repayment or other acquisition or retirement of any Subordinated Debt made by exchange for, or out of the proceeds of, refinancing Debt permitted by Section 11.1.

<u>11.4Mergers, Consolidations, Sales</u>. Not, and not permit any of the Loan Parties and their Subsidiaries to, (a) be a party to any merger or consolidation; (b) sell, transfer, dispose of, convey, or lease any of its assets or Equity Interests (including the sale of Equity Interests of any Subsidiary); (c) sell or assign with or without recourse any receivables; or (d) purchase or otherwise acquire all or substantially all of the assets or any Equity Interests, or any partnership or joint venture interest in, any other Person or make any Acquisition, except the following:

<u>75</u>

(i) any such merger, consolidation, sale, transfer, acquisition, conveyance, lease, or assignment of or by Borrower or Subsidiary with and into Borrower so long as (A) no other provision of this Agreement would be violated thereby; (B) Borrower gives Administrative Agent at least five (5) Business Days' prior written notice of that merger or consolidation (or such shorter notice as Administrative Agent may agree); (C) no Default or Event of Default has occurred and is continuing either before or after giving effect to that transaction; and (D) the Lenders' rights in any Collateral, including the existence, perfection and priority of any Lien thereon, are not adversely affected by that merger or consolidation;

(ii) mergers, consolidations, or Acquisitions which are Permitted Acquisitions;

(iii) sales, leases, assignments, abandonment, or other dispositions of equipment that is substantially worn, damaged, or obsolete or no longer used or useful in the ordinary course of business and leases or subleases of Real Property not useful in the conduct of the business of the Loan Parties and their Subsidiaries;

(iv) the licensing, sublicensing or granting any other right to use on a non-exclusive basis (including the provision of software under an open source license) of Intellectual Property in the ordinary course of business;

 (\underline{v}) the sale or discount, in each case without recourse, of accounts receivable arising in the ordinary course of business, but only in connection with the compromise or collection thereof;

(vi) (i) the lapse of registered patents, trademarks, copyrights and other Intellectual Property of any Loan Party or any of its Subsidiaries to the extent no longer used or useful or not economically desirable in the conduct of its business, or (ii) the abandonment of patents, trademarks, copyrights, or other Intellectual Property rights in the ordinary course of business so long as (in each case under clauses (i) and (ii)), (A) with respect to copyrights, such copyrights are not material revenue generating copyrights, and (B) such lapse is not materially adverse to the interests of the Secured Parties;

(vii)so long as no Event of Default has occurred and is continuing or would immediately result therefrom, transfers of assets from any Subsidiary of any Loan Party that is not a Loan Party to any other Subsidiary of any Loan Party;

(viii)sales, leases, assignments, conveyances, other transfers for value and dispositions of assets, in each case, that constitute a Permitted Asset Disposition;

(ix) subject to the absence of any Default or Event of Default that is continuing or would immediately result therefrom, other Asset Dispositions of Borrower and its Subsidiaries for fair market value in an aggregate amount not to exceed \$2,500,000 in any Fiscal Year; provided that with respect to any single Asset Disposition or series of related Assets Dispositions involving assets having a fair market value in excess of \$500,000, at least 75% of the consideration for such Asset Disposition, shall consist of Cash or Cash Equivalent Investments; provided further, that the Net Cash Proceeds of such Asset Disposition shall be applied and/or reinvested as (and to the extent) required by this Agreement; and provided, further, that Asset Dispositions pursuant to this clause (ix) may not include any Intellectual Property material to the operation of Borrower's business;

<u>76</u>

 (\underline{x}) issuance by Parent or any of its Subsidiaries of its Equity Interests so long as no Change of Control will occur as a result thereof and such issuance is otherwise permitted hereunder;

(xi) dispositions of receivables to an SPV Financing Entity in connection with a receivables factoring, receivable sale or similar transaction (including any warehouse financing);

(<u>xii</u>)any termination or unwinding of any Permitted Bond Hedge Transaction solely to the extent the cash amount payable by Loan Parties as a result of such terminations does not exceed \$5,000,000 in the aggregate; and

(xiii)any disposition of Cash and Cash Equivalent Investments in connection with a transaction otherwise permitted by this Agreement-; and

(xiv)dispositions of Instacash Receivables pursuant to (and in accordance with) the SP Forward Flow Agreement, so long as (x) the proceeds of such dispositions are deposited into an account subject to a Control Agreement in accordance with Section 10.11(b) and (y) customer payments in respect of Instacash Receivables and related amounts collected by the Company (or a Subsidiary) pursuant to the Servicing Agreement are initially paid into the Instacash Servicer Account and are subject to (and applied in a manner consistent with) the terms of the SP Intercreditor Agreement, including sweeping Excluded Amounts (as defined in the SP Master Receivables Purchase Agreement), and other amounts to which the Company or its Subsidiaries are entitled, to account subject to a Control Agreement on a daily basis.

11.5Modification of Certain Documents; Organizational Form.

(a) Not permit the organizational documents or governing documents of any Loan Party to be amended or modified in any way that could reasonably be expected to materially adversely affect the interests of the Lenders.

(b) Not change, or allow any Loan Party to change, its state of formation or its organizational form except upon thirty (30) days' prior notice to Administrative Agent (or such shorter period as Administrative Agent may agree); provided that such Loan Party shall take all actions requested by the Administrative Agent in its sole discretion to maintain a Lien in favor of Administrative Agent (subject to Permitted Liens) on all of the assets of such Loan Party that constitute Collateral prior to any such change.

(c) Not amend, modify, waive or otherwise change (or permit the amendment, modification, waiver or other change in any manner of) any term of the SP Forward Flow Documents if such amendment, modification, waiver or other change would increase the frequency or percentage of repurchases permitted by the proviso to Section 10(b) of the SP Forward Flow Agreement as in effect on the Amendment No. 3 Effective Date, increase the Facility Amount, alter the Eligibility Criteria in a manner materially adverse to any Loan Party or Lender, change the Purchase Price or Repurchase Price (in each case, as such terms are defined in the SP Forward Flow Agreement as in effect on the Amendment No. 3 Effective Date) or related terms and obligations relating thereto or otherwise be adverse in any material respect to the interests any Loan Party or Lender, without the prior written consent of the Administrative Agent.



<u>11.6Transactions with Affiliates</u>. Not, and not permit any of the Loan Parties and their Subsidiaries to, enter into, or cause, suffer, or permit to exist any transaction, arrangement, or contract with any of its other Affiliates involving payments in excess of \$250,000 per transaction and \$750,000 in the aggregate which is on terms which are less favorable than are obtainable from any Person which is not one of its Affiliates, except:

(a) to the extent expressly permitted by Sections 11.3 and 11.4(i));

(b) transactions relating to agreements listed on Schedule 11.6 as of the Closing Date;

(c) any transaction between or among Parent, Borrower and/or one or more Subsidiaries (or any entity that becomes a Subsidiary as a result of such transaction) to the extent permitted or not restricted by this Agreement;

(d) any issuance, sale or grant of securities or other payments, awards or grants in cash, securities or otherwise pursuant to, or the funding of, employment arrangements, stock options and stock ownership plans approved by the board of directors (or equivalent governing body) of Parent or of Borrower or any Subsidiary;

(e) the declaration or payment of any Restricted Payment otherwise permitted hereunder;

(f) transactions with customers, clients, suppliers, licensees, joint ventures, purchasers or sellers of goods or services or providers of employees or other labor entered into in the ordinary course of business, which are (A) fair to Borrower and/or its applicable Subsidiary in the good faith determination of the board of directors (or similar governing body) of Borrower or the senior management thereof or (B) on terms not substantially less favorable to Borrower and/or its applicable Subsidiary as might reasonably be obtained from a Person other than an Affiliate,

(g) the payment of reasonable out-of-pocket costs and expenses related to registration rights and customary indemnities (as determined by Borrower in good faith) provided to shareholders under any shareholder agreement and the existence or performance by Borrower or any Subsidiary of its obligations under any such registration rights or shareholder agreement;

(h) any purchase by Parent of the Equity Interests of (or contribution to the equity capital of) Borrower;

(i) any transaction in respect of which Borrower delivers to Administrative Agent a letter addressed to the board of directors (or equivalent governing body) of Borrower from an accounting, appraisal or investment banking firm of nationally recognized standing stating that such transaction is fair to Borrower or such Subsidiary from a financial point of view or stating that the terms, when taken as a whole, are not substantially less favorable to Borrower or the applicable Subsidiary than might be obtained at the time in a comparable arm's length transaction from a Person who is not an Affiliate;

(j) with respect to any officer of a Loan Party, in connection with such officer's employment by such Loan Party or, with respect to any director of a Loan Party, in connection with such director acting in such capacity.

<u>11.7Inconsistent Agreements</u>. Not, and not permit any of the Loan Parties and their Subsidiaries to, enter into any agreement containing any provision that would (a) be violated or breached in any material respect by any borrowing by Borrower under this Agreement or by the performance by any

<u>78</u>

Loan Party of any of its Obligations under this Agreement or under any other Loan Document; (b) prohibit any Loan Party from granting to Administrative Agent and the Lenders a Lien on any of its assets (other than Excluded Property (as defined in the Guaranty and Collateral Agreement)); or (c) create or permit to exist or become effective any encumbrance or restriction on the ability of any Subsidiary to (i) pay dividends or make other distributions to Borrower or any other Subsidiary, or pay any Debt owed to Borrower or any other Subsidiary, (ii) make loans or advances to any Loan Party, or (iii) transfer any of its assets or properties to any Loan Party, other than:

(a) customary restrictions and conditions contained in agreements relating to the sale of all or a substantial part of the assets or Equity Interests of any Subsidiary pending any such sale, so long as those restrictions and conditions apply only to the Subsidiary or assets to be sold and that sale is permitted under this Agreement;

(b) restrictions by reason of customary provisions restricting assignments, subletting, licensing, sublicensing or other transfers (including the granting of any Lien) contained in leases, subleases, licenses, sublicenses, joint venture agreements, asset sale agreements, trading, netting, operating, construction, service, supply, purchase, sale or other agreements entered into in the ordinary course of business (each of the foregoing, a "<u>Covered Agreement</u>") (provided that such restrictions are limited to the relevant Covered Agreement and/or the property or assets secured by such Liens or the property or assets subject to such Covered Agreement);

(c) conditions imposed by any agreement relating to purchase money Debt, Capital Leases, and other secured Debt permitted by this Agreement, so long as those restrictions or conditions apply only to the property or assets securing that Debt;

(d) customary provisions in leases and other contracts restricting the assignment thereof;

(c) customary restrictions that arise in connection with any Lien permitted by clauses (b)(ii), (c), (d), (f), (g), (l), (q), (r) or (s) of Section 11.2 that limit the right of Borrower or any of its Subsidiaries to dispose of or encumber the assets subject to such Liens;

(f) restrictions contained in the Loan Documents;

(g) restrictions imposed by customary provisions in partnership agreements, limited liability company organizational governance documents, joint venture agreements and other similar agreements or agreement governing the Debt of any joint venture, (i) relating to the transfer of the assets of, or ownership interests in, the relevant partnership, limited liability company, joint venture or any similar Person (or any "shell company" parent with respect thereto), (ii) relating to such joint venture or its members and (iii) otherwise entered into in the ordinary course of business;

(h) restrictions on Cash or other deposits permitted hereunder and any net worth or similar requirements, including such restrictions or requirements imposed by Persons under contracts entered into in the ordinary course of business or for whose benefit such Cash or other deposits or net worth requirements exist;

(i) restrictions which exist on the Closing Date in the documents set forth on Schedule 11.7;

(j) restrictions contained in documents governing Debt of any Subsidiary that is not a Loan Party permitted hereunder;

<u>79</u>

(k) provisions restricting the granting of a security interest in Intellectual Property contained in in-bound licenses by Borrower and its Subsidiaries of such Intellectual Property, which in-bound licenses were entered into in the ordinary course of business (in which case such restriction shall relate only to such Intellectual Property); and

(1) restrictions arising under or as a result of applicable Requirements of Law or the terms of any license, authorization, concession or permit issued or granted by a Governmental Authority.

<u>11.8Business Activities</u>. Not, and not permit any of the Loan Parties and their Subsidiaries to, engage in any material line of business other than the businesses engaged (or proposed to be engaged in) in on the Closing Date and businesses reasonably related or reasonably complementary ancillary, synergistic or related thereto.

<u>11.9Investments</u>. Not, and not permit any of the Loan Parties and their Subsidiaries to, make or permit to exist any Investment in any other Person, except the following:

(a) contributions by Borrower or any Subsidiary to the capital of Borrower or any Loan Party (other than Parent) and contributions by any Excluded Subsidiary to any other Excluded Subsidiary;

- (b) Investments constituting Debt permitted by Section 11.1;
- (c) Contingent Liabilities constituting Debt permitted by Section 11.1 or Liens permitted by Section 11.2;
- (d) Cash Equivalent Investments;
- (e) subject to <u>Section 10.11</u>, bank deposits in the ordinary course of business;
- (f) Investments consisting of securities or instruments received pursuant to a disposition of assets permitted by this Agreement;
- (g) Investments constituting Permitted Acquisitions;
- (<u>h</u>) Investments acquired in a Permitted Acquisition;

(i) non-cash consideration received pursuant to the consummation of Asset Dispositions and Permitted Acquisitions, in each case permitted under this Agreement;

- (j) bank deposits established in accordance with the Loan Documents;
- (k) Investments listed on <u>Schedule 11.9</u> as of the Closing Date;

(1) advances to officers, directors and employees of and its Subsidiaries in an aggregate amount not to exceed \$500,000 at any time outstanding, for travel, entertainment, relocation and analogous ordinary business purposes;

(m) Investments by Parent and its Subsidiaries in Borrower and its Subsidiaries; provided that Investments by Loan Parties in Subsidiaries that are not Loan Parties outside of the ordinary course of business (i) shall not exceed \$5,000,000 at any time outstanding, (ii) shall not be permitted if a Default or an Event of Default has occurred and is continuing and (iii) may not include any Intellectual Property or other assets material to the operation of Borrower's business;

(n) Investments consisting of extensions of credit in the nature of accounts receivable or notes receivable arising from the grant of trade credit in the ordinary course of business, and Investments received in satisfaction or partial satisfaction thereof in connection with the settlement of delinquent accounts generated in the ordinary course of business or from financially troubled account debtors to the extent reasonably necessary in order to prevent or limit loss;

(0) (i) Investments consisting of notes receivable of, or prepaid royalties and other credit extensions to, customers and suppliers who are not Affiliates, in the ordinary course of business; provided that this clause (i) shall not apply to Investments of Borrower in any Subsidiary and (ii) to the extent not covered by clause (i) above, Investments consisting of consumer loans originated or acquired in the ordinary course of business by Borrower or any Subsidiaries;

(p) Investments consisting of joint ventures or strategic alliances in the ordinary course of business consisting of the licensing of technology, the development of technology or the providing of technical support, provided that any cash Investments pursuant to this clause (p) do not exceed \$2,500,000 in the aggregate in any Fiscal Year;

(g) Investments consisting of the licensing or Acquisition of Intellectual Property in the ordinary course of business;

(r) other Investments in an aggregate amount not to exceed \$5,000,000; provided any such Investment may not include any material Intellectual Property or other assets material to the operation of Borrower's business;

(s) deposits, prepayments and other credits to suppliers and deposits in connection with lease obligations, taxes, insurance, utility and workers' compensation and similar items, each in the ordinary course of business and securing contractual obligations of a Loan Party, in each case to the extent constituting a Permitted Lien;

(t) Investments in securities of account debtors received pursuant to any plan of reorganization or similar arrangement upon the bankruptcy or insolvency of such account debtors; and

(u) the purchase of any Permitted Bond Hedge Transaction by Parent and the performance of its obligations thereunder; and

 (\underline{v}) Investments in Subsidiaries that are not Loan Parties to fund ordinary and necessary operating expenses of such Subsidiary, such as payroll and rent expense, in the ordinary course of business and consistent with past practice.

11.10 Reserved Indemnification Payments under SP Forward Flow Agreement-

<u>Not</u>, and not permit any of the Loan Parties and their Subsidiaries to, make any indemnification payments (excluding, for the avoidance of doubt, any payments of "Repurchase Price") under the SP Forward Flow Documents, unless (a) Unrestricted Cash, after giving pro forma effect to any such payment, is at least \$75,000,000 and (b) after giving pro forma effect to any such payment, the aggregate amount of indemnification payments made under the SP Forward Flow Documents would not exceed an amount equal to the then applicable Group Outstanding Purchased Amount (as defined in the SP Forward Flow Agreement).

11.11Fiscal Year. Not, and not permit any of the Loan Parties and their Subsidiaries to, change its Fiscal Year.

<u>81</u>

11.12Financial Covenants.

<u>11.12.1 Minimum Revenue</u>. Not permit Adjusted Revenue for any Computation Period to be less than the amount set forth below for that Computation Period:

Computation <u>Period Ending</u>	Revenue
March 31, 2022	\$180,000,000
June 30, 2022	\$225,000,000
September 30, 2022	\$275,000,000
December 31, 2022	\$300,000,000
March 31, 2023	\$310,000,000
June 30, 2023	\$320,000,000
September 30, 2023	\$330,000,000
December 31, 2023	\$340,900,000
March 31, 2024	\$373,800,000
June 30, 2024	\$411,400,000
September 30, 2024	\$453,500,000
December 31, 2024	\$500,700,000
March 31, 2025	\$540,700,000
June 30, 2025	\$586,600,000
September 30, 2025	\$638,200,000
2025 and each Computation Period ending	\$696,500,000

thereafter

December 31,

11.12.2 Minimum EBITDA. Not permit EBITDA for any Fiscal Quarter to be less than the amount set forth below for that Fiscal Quarter:

<u>Fiscal Quarter</u> <u>Ending</u>	
March 31, 2022	

EBITDA

\$(30,000,000)

<u>82</u>

<u>Fiscal Quarter</u> <u>Ending</u>	<u>EBITDA</u>
June 30, 2022	\$(30,000,000)
September 30, 2022	\$(30,000,000)
December 31, 2022	\$(30,000,000)
March 31, 2023	\$(25,000,000)
June 30, 2023	\$(15,000,000)
September 30, 2023	\$(15,000,000)
December 31, 2023	\$(10,000,000)
March 31, 2024	\$0
June 30, 2024	\$5,000,000
September 30, 2024	\$10,000,000
December 31, 2024	\$15,000,000
March 31, 2025	\$20,000,000
June 30, 2025	\$25,000,000
September 30, 2025	\$30,000,000
December 31, 2025 and each Computation Period ending thereafter	\$35,000,000

<u>11.12.3 Minimum Liquidity</u>. Not permit the Loan Parties to have:

(a) Liquidity of less than \$60,000,000; or

(b) Unrestricted Cash of less than \$40,000,000.

<u>11.13Compliance with Laws</u>. Not, and shall not permit any of the Loan Parties and their Subsidiaries to, fail to comply with the laws, regulations and executive orders referred to in <u>Sections 9.30</u> and <u>9.31</u>, except where such failure could not reasonably be expected to have a Material Adverse Effect.

<u>11.14Permitted Activities of Parent</u>. Notwithstanding anything to the contrary contained herein, Parent shall not engage in any material business or activity or own any material assets other than (i) its ownership of the Equity Interests of Borrower and activities incidental thereto, including payment of dividends and other amounts in respect of its Equity Interests, in each case, solely as permitted pursuant to this Agreement, (ii) the maintenance of its legal existence (including the ability to incur fees, costs and expenses relating to such maintenance) and compliance with all applicable Laws, (iii) the performance of

its obligations as a holding company and as a Guarantor and as a guarantor of any Debt or obligations permitted to be incurred hereunder, (iv) any public offering of its common stock or any other issuance or sale of its Equity Interests or any merger permitted by Section 11.4, (v) if applicable, participating in tax, accounting and other administrative matters as a member of the consolidated group of Parent and Borrower, (vi) the making of any Restricted Payments or Investments permitted to be made, and the incurrence of, and performance of its obligations under, any Debt or other obligations permitted to be incurred, by Parent pursuant to this Agreement; provided that the only Debt for borrowed money Parent is permitted to borrow is Permitted Parent Debt and Permitted Convertible Debt, (vii) providing indemnification to officers and directors in the ordinary course of business, (viii) activities relating to employees of the Loan Parties (including its obligations under any stock option or stock purchase plan or benefit or compensation plan or other similar plan or under any employment agreement); (ix) [reserved]; (x) the issuance and sale of, and its obligations under, its Equity Interests as permitted hereunder, (xi) entering into the Loan Documents and the documents relating to any Permitted Parent Debt or Permitted Convertible Debt, (xii) its obligations under any acquisition agreement relating to any Permitted Acquisition or other Investment permitted hereunder or under any agreement related to any Asset Sale permitted hereunder, (xiii) holding cash and Cash Equivalent Investments, (xiv) maintaining, entering into and performing its obligations under any Permitted Bond Hedge Transaction or Permitted Warrant Transaction and (xvi) any activities incidental or reasonably related to the foregoing.

SECTION 12 EFFECTIVENESS; CONDITIONS OF LENDING, ETC.

The effectiveness of this Agreement and the obligation of each Lender to make its Loans are subject to the following conditions precedent:

12.1Conditions to Effectiveness. The effectiveness of this Agreement, and the obligation of the Lenders to make the Loans, are, in addition to the conditions precedent specified in <u>Section 12.2</u>, subject to satisfaction of the following conditions precedent (and the date on which all such conditions precedent have been satisfied or waived in writing by Administrative Agent and the Lenders is called the "<u>Closing Date</u>"):

12.1.1 Agreement, Notes, and other Loan Documents. Administrative Agent has received the following, each duly executed and dated as of the Closing Date (or any earlier date satisfactory to Administrative Agent), in form and substance satisfactory to Administrative Agent: (a) this Agreement; (b) to the extent requested by any Lender, one or more Notes made payable to that Lender; (c) the Guaranty and Collateral Agreement; together with all instruments, transfer powers, and other items required to be delivered in connection with the Guaranty and Collateral Agreement; and (d) all other Loan Documents.

12.1.2 Authorization Documents. For each Loan Party, Administrative Agent has received the following, each in form and substance satisfactory to Administrative Agent: (a) that Person's charter (or similar formation document), certified by the appropriate governmental authority; (b) good standing certificates in that Person's state of incorporation (or formation) and in each other state in which that Person is qualified to do business if reasonably requested by Administrative Agent; (c) that Person's bylaws (or similar governing document); (d) resolutions of its board of directors (or similar governing body) approving and authorizing that Person's execution, delivery, and performance of the Loan Documents to which it is party and the transactions contemplated thereby; and (e) signature and incumbency certificates of that Person's officers and/or managers executing any of the Loan Documents (which certificates Administrative Agent and each Lender may conclusively rely on until formally advised by a like certificate of any

changes in any such certificate), all certified by its secretary or an assistant secretary (or similar officer) as being in full force and effect without modification.

12.1.3 Consents, etc. Administrative Agent has received certified copies of all documents evidencing any necessary company action, consents and governmental approvals (if any) required for the execution, delivery, and performance by the Loan Parties of the documents referred to in this Section 12.

12.1.4 Letter of Direction. Administrative Agent has received a letter of direction containing funds flow information with respect to the proceeds of the Loans on the Closing Date, duly executed and dated as of the Closing Date, in form and substance reasonably satisfactory to Administrative Agent.

12.1.5 Collateral and Diligence Questionnaire. Administrative Agent has received a Collateral and Diligence Questionnaire completed and executed by each Loan Party, in form and substance satisfactory to Administrative Agent.

- 12.1.6 [Reserved].
- 12.1.7 [Reserved].
- <u>12.1.8</u> [Reserved].

12.1.9 Opinions of Counsel. Administrative Agent has received opinions of counsel for each Loan Party, including local counsel reasonably requested by Administrative Agent, each duly executed and dated as of the Closing Date (or any earlier date satisfactory to Administrative Agent), in form and substance reasonably satisfactory to Administrative Agent.

<u>12.1.10 Insurance</u>. Administrative Agent has received evidence of the existence of insurance required to be maintained pursuant to <u>Section</u> <u>10.3(b)</u>.

12.1.11 Payment of Fees. Administrative Agent has received evidence of payment by Borrower of all accrued and unpaid reasonable and documented fees, costs, and expenses to the extent then due and payable on the Closing Date (including, without limitation, fees under the Agent Fee Letter), together with all Attorney Costs of Administrative Agent to the extent invoiced prior to the Closing Date, <u>plus</u> all additional amounts of Attorney Costs that constitute Administrative Agent's reasonable estimate of Attorney Costs incurred or to be incurred by Administrative Agent through the closing proceedings (but no such estimate will preclude a final settling of accounts between Borrower and Administrative Agent in respect of those Attorney Costs).

12.1.12 Debt to be Repaid. Administrative Agent has received evidence, reasonably satisfactory to Administrative Agent, that all Debt to be Repaid has been (or concurrently with the initial borrowing will be) paid in full and that all agreements and instruments governing the Debt to be Repaid and that all Liens securing the Debt to be Repaid have been (or concurrently with the initial borrowing will be) terminated. Administrative Agent has received payoff letters evidencing repayment in full of all Debt to be Repaid, the termination of all agreements relating thereto, and the release of all Liens granted in connection therewith, with Uniform Commercial Code or other appropriate termination statements and documents effective to evidence the foregoing.

12.1.13 Solvency Certificate. Administrative Agent has received a solvency certificate, in form and substance satisfactory to Administrative Agent, executed by a Senior Officer of Borrower.

12.1.14 Search Results; Lien Terminations. Administrative Agent has received certified copies of Uniform Commercial Code search reports dated a date reasonably near to the Closing Date, listing all effective financing statements which name any Loan Party (under their present names and any previous names within the past five years) as debtors, together with (a) copies of all such financing statements; (b) Uniform Commercial Code termination statements pertaining to previously terminated financing, lease, and/or consignment relationships for which financing statements remain of record, in each case as Administrative Agent reasonably requests; and (c) all other Uniform Commercial Code termination statements as Administrative Agent reasonably requests.

12.1.15 Filings, Registrations, and Recordings. Administrative Agent has received each document (including Uniform Commercial Code financing statements) required by the Collateral Documents or under law or reasonably requested by Administrative Agent to be filed, registered, or recorded in order to create in favor of Administrative Agent, for the benefit of Administrative Agent and the Lenders, a perfected Lien on the collateral described therein (but only to the extent that perfection may be achieved by such a filing, registration, or recording), prior to any other Liens (subject only to Permitted Liens), in proper form for filing, registration, or recording.

12.1.16 Closing Certificate. Administrative Agent has received a certificate, in form and substance satisfactory to Administrative Agent, executed by a Senior Officer of Borrower certifying the matters set forth in Section 12.2.1 as of the Closing Date.

12.1.17 Financial Statements. Administrative Agent has received and is reasonably satisfied with the Financial Statements and all other historical and projected financial information of Loan Parties and their Subsidiaries requested by Administrative Agent.

<u>12.1.18 No Material Adverse Effect</u>. There has not occurred since December 31, 2021, any developments or events that, individually or in the aggregate with any other circumstances, has had or could reasonably be expected to have a Material Adverse Effect.

12.1.19 Capital Structure. Administrative Agent has received confirmation of ownership and capital structure of the Loan Parties.

12.1.20 Financial Condition.

(a) Administrative Agent has received evidence satisfactory to it that (a) Borrower has a trailing 12-month EBITDA of at least (\$63,962,130) as of January 31, 2022, determined on a pro forma basis after giving effect to (i) the funding of the initial Loans as provided under this Agreement, including the payment of all fees, costs and expenses as set forth above and (ii) year-end and other adjustments reasonably satisfactory to Administrative Agent; and (b) the Loan Parties have reasonably sufficient liquidity to operate their business plan and in an amount satisfactory to Administrative Agent (with no payables stretched beyond their customary payment practices).

(b) After giving effect to the initial Loans on the Closing Date and the payment by Borrower of all fees and expenses incurred in connection with this Agreement (with no payables stretched beyond their customary payment practices), the Liquidity of the Company and its Subsidiaries is not less than \$200,000,000.

12.1.21 Know-Your-Customer and Anti-Money Laundering. No later than two Business Days in advance of the Closing Date, Administrative Agent shall have received all documentation and other information reasonably requested in writing by Administrative Agent with respect to any Loan Party at least ten Business Days in advance of the Closing Date that is required by regulatory authorities under applicable "know-your-customer" and anti-money laundering rules and regulations, including, without limitation, IRS Form W-9, IRS Form W-8-BEN or IRS Form W-8BEN-E, as applicable, with results satisfactory to Administrative Agent.

For purposes of determining compliance with the conditions specified in this Section 12.1, Administrative Agent and each Lender that has signed this Agreement shall be deemed to have received, consented to, approved or accepted or to be satisfied with, each document or other matter required thereunder to be received, consented to or approved by or acceptable or satisfactory to Administrative Agent or a Lender.

12.2Conditions Precedent to all Loans. The obligation of each Lender to make each Loan is subject to the following further conditions precedent that:

12.2.1 Compliance with Warranties, No Default, etc. Both before and immediately after giving effect to any borrowing, the following statements are true and correct:

(a) the representations and warranties of each Loan Party set forth in this Agreement and the other Loan Documents are true and correct in all material respects (unless any such representation or warranty is by its terms qualified by concepts of materiality, in which case that representation or warranty is true and correct in all respects after giving effect to any such materiality qualifier) with the same effect as if then made (except to the extent stated to relate to a specific earlier date, in which case that representation or warranty is true and correct in all material respects or in all respects, as applicable, as of that earlier date); and

(b) solely with respect to the making of any Term B Loan, Enterprise Value shall have been greater than \$800,000,000 for no less than 40 of the immediately preceding 45 consecutive days (and no less than \$750,000,000 for any of the immediately preceding 45 consecutive days); and

(c) no Default or Event of Default has occurred and is continuing.

12.2.2 Confirmatory Certificate. If requested by Administrative Agent or any Lender, Administrative Agent has received (in sufficient counterparts to provide one to Administrative Agent and each Lender) a certificate dated the date of the requested Loan and signed by a duly authorized representative of Borrower as to the matters set out in Section 12.2.1 (it being understood that each request by Borrower for the making of a Loan will be deemed to constitute a representation and warranty by Borrower that the conditions precedent set forth in Section 12.2.1 will be satisfied at the time of the making of that Loan), together with such other documents as Administrative Agent or any Lender may reasonably request in support thereof.

SECTION 13 EVENTS OF DEFAULT AND THEIR EFFECT.

13.1Events of Default. Each of the following will constitute an Event of Default under this Agreement:

13.1.1 <u>Non-Payment of the Loans, etc.</u> Default in the payment when due of the principal of any Loan; or default, and continuance thereof for five or more days, in the payment when due

<u>87</u>

of any interest, fee or other amount payable by Borrower under this Agreement or under any other Loan Document.

13.1.2 Non-Payment of Other Debt. Any default occurs under the terms applicable to any Debt of any Loan Party or any Subsidiary of a Loan Party in an aggregate amount (for all such Debt so affected and including undrawn committed or available amounts and amounts owing to all creditors under any combined or syndicated credit arrangement) exceeding \$3,000,000 and that default (a) consists of the failure to pay that Debt when due, whether by acceleration or otherwise (beyond the grace period (if any) provided in the instrument or agreement under which such Debt was created), or (b) accelerates the maturity of that Debt or permits the holder or holders thereof, or any trustee or agent for any such holder or holders, with the giving of notice if required, to cause that Debt to become due and payable (or require any Loan Party to purchase or redeem that Debt or post cash collateral in respect thereof) prior to its expressed maturity; provided that clause (b) of this Section 13.1.2 shall not apply to secured Debt that becomes due as a result of the voluntary sale or transfer of the property securing such Debt if such sale or transfer is permitted hereunder and substantially concurrently with such voluntary sale or transfer of property such Debt is paid in full and otherwise fully discharged in accordance with the underlying agreement evidencing such Debt.

13.1.3 [Reserved]Repurchase of Defective Receivables.-

<u>The Loan Parties or their Subsidiaries effect repurchases of "Defective Receivables" as a result of a breach of any Eligibility Criteria set</u> forth in clauses 1 through 16 of Annex A of the SP Forward Flow Agreement in an aggregate amount in excess of (x) \$20,000,000 in any calendar year if, after giving proforma effect to any such repurchase, Unrestricted Cash is at least \$75,000,000 or (y) \$1,000,000 in any calendar year if, after giving proforma effect to any such repurchase, Unrestricted Cash is less than \$75,000,000.

<u>13.1.4</u> Bankruptcy, Insolvency, etc. Any of the following occurs: (a) any Loan Party becomes insolvent or generally fails to pay, or admits in writing its inability or refusal to pay, debts as they become due; (b) any Loan Party applies for, consents to, or acquiesces in the appointment of a trustee, receiver, or other custodian for that Loan Party or any property thereof, or makes a general assignment for the benefit of creditors; (c) in the absence of any such application, consent, or acquiescence, a trustee, receiver, or other custodian is appointed for any Loan Party or for a substantial part of the property of any thereof and is not discharged within sixty (60) days; (d) any Insolvency Proceeding, or any dissolution or liquidation proceeding, is commenced in respect of any Loan Party, and that Insolvency Proceeding or proceeding (i) is not commenced by that Loan Party, (ii) is consented to or acquiesced in by that Loan Party, or (iii) remains for sixty (60) days undismissed; or (e) any Loan Party takes any action to authorize, or in furtherance of, any of the foregoing.

13.1.5 Non-Compliance with Loan Documents.

(a) Failure by any Loan Party to comply with or to perform any covenant set forth in <u>Sections 10.1.6(a)</u>, <u>10.5 (solely with respect to</u> <u>Borrower)</u>, <u>10.6</u>, or <u>Section 11</u>.

(b) Failure by any Loan Party to comply with or to perform any covenant set forth in <u>Sections 10.1.1</u>, <u>10.1.2</u>, <u>10.1.3</u> or <u>10.1.4</u> and continuance of that failure described in this <u>Section 13.1.5(b)</u> for 5 or more Business Days.

<u>88</u>

(c) Failure by any Loan Party to comply with or to perform any other provision of this Agreement or any other Loan Document (and not constituting an Event of Default under any other provision of this <u>Section 13</u>) and continuance of that failure described in this <u>Section 13.1.5(c)</u> for 30 or more days.

13.1.6 <u>Representations; Warranties</u>. Any representation or warranty made by any Loan Party in this Agreement or any other Loan Document is breached or is false or misleading in any material respect, or any schedule, certificate, financial statement, report, notice or other writing furnished by any Loan Party to Administrative Agent or any Lender in connection with this Agreement is false or misleading in any material respect on the date as of which the facts therein set forth are stated or certified.

<u>13.1.7</u> Pension Plans. Any of the following events occurs and such event could reasonably be expected to result, individually or in the aggregate, in a Material Adverse Effect: (a) any member of the Controlled Group or the PBGC institutes steps to terminate a Pension Plan if as a result of that termination Borrower or any Subsidiary could be required to make a contribution to that Pension Plan, or could incur a liability or obligation to that Pension Plan, (b) a contribution failure occurs with respect to any Pension Plan sufficient to give rise to a Lien under Section 303(k) of ERISA with respect to Borrower or any Subsidiary; (c) the Unfunded Liability of all Pension Plans sponsored and maintained by Borrower or any Subsidiary exceeds 20% of the Total Plan Liability for those plans, (d) there occurs any withdrawal or partial withdrawal from a Multiemployer Pension Plan and Borrower or any Subsidiary incurs withdrawal liability (without unaccrued interest) to Multiemployer Pension Plans as a result of that withdrawal (including any outstanding withdrawal liability that Borrower or any member of the Controlled Group have incurred on the date of that withdrawal); or (e) the occurrence of a Termination Event.

<u>13.1.8</u> Judgments. (a) One or more final judgments which exceed an aggregate of \$1,000,000 in monetary damages are rendered against any Loan Party or a Subsidiary of a Loan Party (except to the extent covered by insurance (including D&O insurance) pursuant to which the insurer has been notified and has not denied coverage so long as that insurance is paid to the applicable Loan Party or Subsidiary within sixty (60) days of the rendering of those judgments) and have not been paid, discharged or vacated or had execution thereof stayed pending appeal within seventy-five (75) days after entry or filing of those judgments; or (b) one or more final non-appealable judgments or settlements (whether in respect of an action by a Governmental Authority, class action or substantially similar claim or otherwise) or one or more final non-appealable penalties or fines imposed by a Governmental Authority (in each case, except to the extent covered by insurance (including D&O insurance) pursuant to which the insurer has been notified within seven calendar days of such judgment, settlement, penalty or fine (and which Borrower believes in good faith is covered by such insurance at the time of such notification) and has not denied coverage so long as that insurance is paid to the applicable Loan Party or Subsidiary within sixty (60) days of the rendering of those judgments, entering into of those settlements or imposition of those penalties or fines), in each case, which exceed individually or in the aggregate the Threshold Amount in monetary damages, are rendered against, agreed by or imposed on any Loan Party or a Subsidiary of a Loan Party; or (c) any one or more permits or licenses of the Loan Parties in exceess of the Threshold Percentage of revenues for the most recently ended Computation Period, is suspended, revoked or terminated by a Governmental Authority (excluding, for avoidance of doubt, any voluntary termination or expiration of any such license or permit initiated by a Loan Party prior and unrelated to any such act

13.1.9 Invalidity of Loan Documents, etc. Any material Loan Document ceases to be in full force and effect, or any Loan Party (or any Person by, through or on behalf of any Loan Party) expressly contests in writing, any manner the validity, binding nature, or enforceability of any material Loan Document.

13.1.10 Charges or Proceedings. The felony indictment of any Loan Party or any of its Subsidiaries or any executive officer thereof under any criminal statute, or commencement of criminal proceedings against any Loan Party or any of its Subsidiaries or any senior officer thereof, in each case for fraud, misconduct, embezzlement, money laundering, insider trading, market manipulation, income tax evasion or any predicate crime to money laundering, insider trading, market was on or other similar illegality in respect of the business or operations of Parent and its Subsidiaries and pursuant to which statute or proceedings the penalties or remedies sought include forfeiture to any Governmental Authority of any material portion of the property of the Loan Parties and their Subsidiaries, or, in the case of any executive officer, solely to the extent such property is used in any material business of the Loan Parties and their Subsidiaries.

13.1.11 Change of Control. A Change of Control occurs.

13.1.12 Instacash. A Regulatory Trigger Event or an Event of Termination occurs under the SP Forward Flow Agreement and:

(i) following such Regulatory Trigger Event or Event of Termination, the Instacash Weekly Purchased Amount for any calendar week is less than 90% of the Baseline Average Weekly Volume ("Initial Reduction Event") and is not in excess of 90% of the Baseline Average Weekly Volume for any two consecutive calendar weeks of the six calendar weeks immediately following such Initial Reduction Event ("Extended Reduction Event"); and

(ii) if an Extended Reduction Event occurs under clause (i) above, within sixty (60) days after the occurrence of the applicable Initial Reduction Event, the Borrower or one of its Affiliates has not entered into a replacement forward flow purchase or warehouse financing reasonably satisfactory to the Administrative Agent.

<u>13.2Effect of Event of Default</u>. If any Event of Default described in <u>Section 13.1.4</u> occurs in respect of Borrower, then the Commitments will immediately terminate and the Loans and all other Obligations under this Agreement will become immediately due and payable, all without presentment, demand, protest, or notice of any kind. If any other Event of Default occurs and is continuing, then Administrative Agent may (and, upon the written request of the Required Lenders shall) declare, in a written notice to Borrower, the Commitments to be terminated in whole or in part and/or declare all or any part of the Loans and all other Obligations under this Agreement to be due and payable, whereupon the Commitments will immediately terminate (or be reduced, as applicable) and/or the Loans and other Obligations under this Agreement will become immediately due and payable (in whole or in part, as applicable), all without presentment, demand, protest, or notice of any kind (other than as expressly provided for above in this sentence). Administrative Agent shall promptly advise Borrower of any such declaration, but failure to do so will not impair the effect of any such declaration. If Administrative Agent and each Lender have received written notice from Borrower of Borrower's intent to cure a Specified Financial Covenant Default in accordance with Section 13.4, then Administrative Agent and the Required Lenders may not exercise any the foregoing remedies in this Section 13.2 with respect to that Specified Financial Covenant Default until the earlier of (a) ten (10) Business Days after the day on which financial statements are required to be delivered for the applicable Fiscal Quarter or Fiscal Year pursuant to Section

<u>90</u>

10.1.1 or Section 10.1.2, as applicable (but only to the extent such cure did not occur) and (b) the date that Administrative Agent receives notice Borrower will not cure that Specified Financial Covenant Default in accordance with Section 13.4.

<u>13.3Credit Bidding</u>. The Loan Parties and the Lenders hereby irrevocably authorize Administrative Agent, based upon the instruction of the Required Lenders, to Credit Bid and purchase (either directly or through one or more acquisition vehicles) all or any portion of the Collateral at any sale thereof conducted by Administrative Agent in accordance with applicable law, based upon the instruction of the Required Lenders, under any provisions of the Uniform Commercial Code, as part of any sale or investor solicitation process conducted by any Loan Party, any interim receiver, receiver, receiver and manager, administrative receiver, trustee, agent, or other Person pursuant or under any insolvency laws, in each case subject to the following limitations: (i) the Required Lenders may not direct Administrative Agent in any manner that does not treat each of the Lenders equally, without preference or discrimination, in respect of consideration received as a result of any Credit Bid; (ii) the acquisition documents must be commercially reasonable and contain customary protections for minority holders, such as, among other things, anti-dilution and tag-along rights; (iii) the exchanged debt or equity securities must be freely transferable, without restriction (subject to applicable securities laws); and (iv) reasonable efforts must be made to structure the acquisition obligations). For purposes of this <u>Section 13.3</u>, the term "<u>Credit Bid</u>" means an offer submitted by Administrative Agent (on behalf of the Lenders), based upon the instruction of the Required Lenders, to acquire the property of any Loan Party or any portion thereof in exchange for and in full and final satisfaction of all or a portion (as determined by Administrative Agent, based upon the instruction of the Required Lenders) of the claims and Obligations under this Agreement and other Loan Documents.

13.4Curative Equity.

(a) Subject to the limitations set forth in Section 13.4(d), Borrower may cure (and will be deemed to have cured) an Event of Default arising out of a breach of any of the financial covenants set forth in Sections 11.12.2 and 11.12.3 (each such financial covenant, a "Specified Financial Covenant"; each such Event of Default, a "Specified Financial Covenant Default") if Borrower receives the cash proceeds of Curative Equity within 10 Business Days after the earliest date on which the Compliance Certificate applicable to such Specified Financial Covenant is required to be delivered for the applicable Computation Period pursuant to this Agreement.

(b) Borrower shall promptly notify Administrative Agent of its receipt of any proceeds of Curative Equity and shall make a prepayment of the Term Loans in accordance with Sections 6.2 and 6.3.

(c) Upon delivery of a certificate by Borrower to Administrative Agent certifying as to the amount of the proceeds of any Curative Equity and that those proceeds have been applied in accordance with Section 13.4(b), then each applicable Specified Financial Covenant Default will be deemed cured with no further action required by the Required Lenders. Before the date of the delivery of that certificate, any Specified Financial Covenant Default that has occurred and is continuing will be deemed to be continuing, and, as a result, the Lenders will have no obligation to make additional loans or otherwise extend additional credit under this Agreement. If Borrower does not cure a Specified Financial Covenant Default as provided in this Section 13.4, then that Specified Financial Covenant Default will continue unless waived in writing by the Required Lenders in accordance with this Agreement.



(d) To the extent that proceeds of Curative Equity are received with respect to any Fiscal Quarter for the purpose of curing a breach of the financial covenant set forth in Section 11.12.2, those proceeds will be deemed to be EBITDA for purposes of determining compliance with the Specified Financial Covenants for that Fiscal Quarter and subsequent Computation Periods that include that Fiscal Quarter. Notwithstanding any provision of this Agreement to the contrary, (i) Borrower's rights under this Section 13.4 (A) may be exercised no more than three (3) times during the term of this Agreement; and (B) may be exercised no more than twice in any period of four Fiscal Quarters; (ii) the amount of proceeds of any Curative Equity (x) may not be less than \$2,500,000 and (y) except as otherwise required pursuant to the foregoing clause (x) may not be greater than the amount required to cause Borrower to be in compliance with each applicable Specified Financial Covenant as at the end of the applicable Computation Period; and (iii) the proceeds of Curative Equity will be disregarded for purposes of determining EBITDA for any pricing, financial covenant-based conditions, or baskets with respect to the covenants contained in this Agreement and there will be no pro forma reduction in Debt with the proceeds of any Curative Equity for determining compliance with the Specified Financial Covenants or for determining any pricing, financial covenant-based conditions, or baskets with respect to the covenants contained in this Agreement, in each case in such Fiscal Quarter and the next three Fiscal Quarters thereafter.

SECTION 14 AGENCY.

14.1Appointment and Authorization. Each Lender hereby irrevocably (subject to Section 14.10) appoints, designates, and authorizes Administrative Agent to take any action on its behalf under the provisions of this Agreement and each other Loan Document and to exercise any powers and perform any duties as are expressly delegated to it, as applicable, by the terms of this Agreement or any other Loan Document, together with all powers as are reasonably incidental thereto. Notwithstanding any provision to the contrary contained elsewhere in this Agreement or in any other Loan Document, Administrative Agent will not have any duty or responsibility except those expressly set forth in this Agreement, nor will Administrative Agent have or be deemed to have any fiduciary relationship with any Lender or participant, and no implied covenants, functions, responsibilities, duties, obligations, or liabilities are to be read into this Agreement or any other Loan Document or otherwise exist against Administrative Agent, as applicable. Without limiting the generality of the foregoing sentence, the use of the term "agent" in this Agreement and in other Loan Documents with reference to Administrative Agent is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any applicable law. Instead, that term is used merely as a matter of market custom, and is intended to create or reflect only an administrative relationship between independent contracting parties.

14.2[Reserved].

14.3Delegation of Duties. Administrative Agent may execute any of its duties under this Agreement or any other Loan Document by or through agents, employees, or attorneys-in-fact and is entitled to advice of counsel and other consultants or experts concerning all matters pertaining to those duties. Administrative Agent will not be responsible for the negligence or misconduct of any agent or attorney-in-fact that it selects in the absence of gross negligence or willful misconduct.

<u>14.4Exculpation</u>. None of Administrative Agent and its directors, officers, employees, and agents (a) will be liable for any action taken or omitted to be taken by any of them under or in connection with this Agreement or any other Loan Document or the transactions contemplated hereby (except to the extent resulting from its own gross negligence or willful misconduct in connection with its duties expressly set forth in this Agreement as determined by a final, non-appealable judgment by a court of competent jurisdiction), or (b) will be responsible in any manner to any Lender or participant for any recital, statement, representation or warranty made by any Loan Party or any Affiliate of Borrower, or any officer thereof,

<u>92</u>

contained in this Agreement or in any other Loan Document, or in any certificate, report, statement, or other document referred to or provided for in, or received by Administrative Agent under or in connection with, this Agreement or any other Loan Document, or the validity, effectiveness, genuineness, enforceability, or sufficiency of this Agreement or any other Loan Document (or the creation, perfection, or priority of any Lien or security interest therein), or for any failure of Borrower or any other party to any Loan Document to perform its Obligations under this Agreement or under any other Loan Documents. Administrative Agent is not and will not be under any obligation to any Lender to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, this Agreement or any other Loan Document or to inspect the properties, books, or records of any of the Loan Parties and their Subsidiaries and Affiliates.

14.5Reliance. Administrative Agent may rely, and will be fully protected in relying, upon any writing, communication, signature, resolution, representation, notice, consent, certificate, electronic mail message, affidavit, letter, telegram, facsimile, telex or telephone message, statement, or other document or conversation believed by it to be genuine and correct and to have been signed, sent or made by the proper Person or Persons, and upon advice and statements of legal counsel (including counsel to Borrower), independent accountants, and other experts selected by Administrative Agent. Administrative Agent will be fully justified in failing or refusing to take any action under this Agreement or any other Loan Document unless Administrative Agent first receives all advice or concurrence of the Required Lenders as it deems appropriate and, if it so requests, confirmation from the Lenders of their obligation to indemnify Administrative Agent against any and all liability and expense which might be incurred by Administrative Agent by reason of taking or continuing to take any such action. Administrative Agent will in all cases be fully protected in acting, or in refraining from acting, under this Agreement or any other Loan Document in accordance with a request or consent of the Required Lenders and each such request and any action taken or failure to act pursuant thereto will be binding upon each Lender. For purposes of determining compliance with the conditions specified in <u>Section 12</u>, each Lender that has signed this Agreement will be deemed to have consented to, approved, or accepted or to be satisfied with, each document or other matter required thereunder to be consented to or approved by or acceptable or satisfactory to a Lender unless Administrative Agent has received written notice from that Lender prior to the proposed Closing Date specifying its objection thereto.

<u>14.6Notice of Default</u>. Administrative Agent will not be deemed to have knowledge or notice of the occurrence of any Event of Default or Default except with respect to defaults in the payment of principal, interest and fees required to be paid to Administrative Agent for the account of the Lenders, unless Administrative Agent has received written notice from a Lender or Borrower referring to this Agreement, describing that Event of Default or Default and stating that that notice is a "notice of default." Administrative Agent shall promptly notify the Lenders of its receipt of any such notice. Administrative Agent shall take all such actions with respect to each such Event of Default or Default as requested by the Required Lenders in accordance with <u>Section 13</u>, but unless and until Administrative Agent has received any such request, Administrative Agent may (but will not be required to) take any action, or refrain from taking any action, with respect to any Event of Default or Default as Administrative Agent deems advisable or in the best interest of the Lenders.

14.7Credit Decision. Each Lender acknowledges that Administrative Agent has not made any representation or warranty to it, and that no act by Administrative Agent hereafter taken, including any consent and acceptance of any assignment or review of the affairs of the Loan Parties, will be deemed to constitute any representation or warranty by Administrative Agent to any Lender as to any matter, including whether Administrative Agent has disclosed material information in its possession. Each Lender represents to Administrative Agent that it has, independently and without reliance upon Administrative Agent and based on documents and information as it has deemed appropriate, made its own appraisal of, and investigation into, the business, prospects, operations, property, financial and other condition, and

<u>93</u>

creditworthiness of the Loan Parties, and made its own decision to enter into this Agreement and to extend credit to Borrower under this Agreement. Each Lender also represents to Administrative Agent that it will, independently and without reliance upon Administrative Agent and based on documents and information as it deems appropriate at the time, continue to make its own credit analysis, appraisals, and decisions in taking or not taking action under this Agreement and the other Loan Documents, and to make all investigations as it deems necessary to inform itself as to the business, prospects, operations, property, financial and other condition, and creditworthiness of Borrower. Except for notices, reports and other documents expressly required in this Agreement to be furnished to the Lenders by Administrative Agent, Administrative Agent will not have any duty or responsibility to provide any Lender with any credit or other information concerning the business, prospects, operations, property, financial or other condition or creditworthiness of Borrower which may come into the possession of Administrative Agent.

14.8Indemnification. Whether or not the transactions contemplated by this Agreement are consummated, each Lender shall indemnify upon demand Administrative Agent and its directors, officers, employees and agents (to the extent not reimbursed by or on behalf of Borrower and without limiting the obligation of Borrower to do so), according to its applicable Pro Rata Share, from and against any and all Indemnified Liabilities, except that no Lender will be liable for any payment to any such Person of any portion of the Indemnified Liabilities to the extent determined by a final, non-appealable judgment by a court of competent jurisdiction to have resulted from the applicable Person's own gross negligence or willful misconduct. No action taken in accordance with the directions of the Required Lenders will be deemed to constitute gross negligence or willful misconduct for purposes of this <u>Section 14.8</u>. Without limitation of the foregoing, each Lender shall reimburse Administrative Agent upon demand for its ratable share of any costs or out-of-pocket expenses (including Attorney Costs and Taxes) incurred by Administrative Agent in connection with the preparation, execution, delivery, administration, modification, amendment or enforcement (whether through negotiations, legal proceedings or otherwise) of, or legal advice in respect of rights or responsibilities under, this Agreement, any other Loan Document, or any document contemplated by or referred to in this Agreement, to the extent that Administrative Agent is not reimbursed for any such expenses by or on behalf of Borrower. The undertaking in this <u>Section 14.8</u> will survive repayment of the Loans, cancellation of the Notes, any foreclosure under, or modification, release or discharge of, any or all of the Collateral Documents, termination of this Agreement and the resignation or replacement of Administrative Agent.

14.9Administrative Agent in Individual Capacities. Monroe Capital and its Affiliates may make loans to, issue letters of credit for the account of, accept deposits from, acquire equity interests in and generally engage in any kind of banking, trust, financial advisory, underwriting or other business with the Loan Parties and Affiliates as though Monroe Capital were not Administrative Agent under this Agreement and without notice to or consent of any Lender. Each Lender acknowledges that, pursuant to those activities, Monroe Capital or its Affiliates might receive information regarding Borrower or their Affiliates (including information that is subject to confidentiality obligations in favor of Borrower or any such Affiliate) and acknowledges that Administrative Agent will be under no obligation to provide any such information to them. With respect to their Loans (if any), Monroe Capital and its Affiliates have the same rights and powers under this Agreement as any other Lender and may exercise the same as though Monroe Capital were not Administrative Agent, and the terms "Lender" and "Lenders" include Monroe Capital and its Affiliates, to the extent applicable, in their individual capacities.

14.10Successor Administrative Agent. Administrative Agent may resign as Administrative Agent upon 30 days' notice to the Lenders. If Administrative Agent resigns under this Agreement, the Required Lenders shall, with (so long as no Event of Default exists) the consent of Borrower (which may not be unreasonably withheld or delayed), appoint from among the Lenders a successor Administrative Agent for the Lenders. If no successor agent is appointed prior to the effective date of the resignation of Administrative Agent, Administrative Agent may appoint, after consulting with the Lenders and Borrower,

<u>94</u>

a successor agent from among the Lenders. Upon the acceptance of its appointment as successor agent under this Agreement, that successor agent will succeed to all the rights, powers, and duties of the retiring Administrative Agent and the term "Administrative Agent" will mean that successor agent, and the retiring Administrative Agent's appointment, powers and duties as Administrative Agent will be terminated. After any retiring Administrative Agent's resignation under this Agreement as Administrative Agent, the provisions of this Section 14 and Sections 15.5 and 15.17 will inure to its benefit as to any actions taken or omitted to be taken by it while it was Administrative Agent under this Agreement. If no successor agent has accepted appointment as Administrative Agent by the date which is 30 days following a retiring Administrative Agent's notice of resignation, the retiring Administrative Agent's resignation will nevertheless thereupon become effective and the Required Lenders shall perform all of the duties of Administrative Agent under this Agreement until such time, if any, as the Required Lenders appoint a successor agent as provided for above.

14.11Collateral Matters. Each Lender authorizes and directs Administrative Agent to enter into the other Loan Documents for the benefit of Lenders. Each Lender hereby agrees that, except as otherwise set forth in this Agreement, any action taken by Administrative Agent or Required Lenders in accordance with the provisions of this Agreement or the other Loan Documents, and the exercise by Administrative Agent or Required Lenders of the powers set forth in this Agreement or therein, together with all other powers as are reasonably incidental thereto, will be authorized by, and binding upon, all Lenders. Administrative Agent is hereby authorized on behalf of all Lenders, without the necessity of any notice to or further consent from any Lender to take any action with respect to any Collateral or Loan Documents which may be necessary to perfect and maintain perfected the Liens upon the Collateral granted pursuant to this Agreement and the other Loan Documents. The Lenders irrevocably authorize Administrative Agent, at its option and in its discretion, to do any and all of the following: (a) to release any Lien granted to or held by Administrative Agent under this Agreement or otherwise consented to by the Required Lenders (including the release of any Guarantor in connection with any such disposition); or (iii) subject to Section 15.1, if approved in writing by the Required Lenders; or (b) to subordinate its interest in any Collateral to any holder of a Lien on that Collateral which is permitted by Section 11.2(d)(ii) (it being understood that Administrative Agent may conclusively rely on a certificate from Borrower in determining whether the Debt secured by any such Lien is permitted by <u>Section 11.1(b)</u>). Upon request by Administrative Agent at any time, the Lenders will confirm in writing Administrative Agent's authority to release, or subordinate its interest in, particular types or items of Collateral pursuant to this <u>Section 14.11</u>.

14.12Restriction on Actions by Lenders. Each Lender shall not, without the express written consent of Administrative Agent, and shall, upon the written request of Administrative Agent (to the extent it is lawfully entitled to do so), set-off against the Obligations, any amounts owing by that Lender to a Loan Party or any deposit accounts of any Loan Party now or hereafter maintained with that Lender. Each Lender shall not, unless specifically requested to do so in writing by Administrative Agent, take or cause to be taken any action, including the commencement of any legal or equitable proceedings, to foreclose any loan or otherwise enforce any security interest in any of the Collateral or to enforce all or any part of this Agreement or the other Loan Documents. All enforcement actions under this Agreement and the other Loan Documents against the Loan Parties or any third party with respect to the Obligations or the Collateral may be taken by only Administrative Agent (at the direction of the Required Lenders or as otherwise permitted in this Agreement) or by its agents at the direction of Administrative Agent.

14.13Administrative Agent May File Proofs of Claim.

14.13.1 In case of the pendency of any receivership, insolvency, liquidation, bankruptcy, reorganization, arrangement, adjustment, composition, or other judicial proceeding relative to any

<u>95</u>

Loan Party (including any Insolvency Proceeding), Administrative Agent (irrespective of whether the principal of any Loan is then due and payable as expressed in this Agreement or by declaration or otherwise and irrespective of whether Administrative Agent has made any demand on Borrower) may, by intervention in any such proceeding or otherwise, do any and all of the following:

(a) file and prove a claim for the whole amount of the principal and interest owing and unpaid in respect of the Loans, and all other Obligations that are owing and unpaid and to file any other documents as are necessary or advisable in order to have the claims of the Lenders and Administrative Agent (including any claim for the reasonable compensation, expenses, disbursements and advances of the Lenders and Administrative Agent and their respective agents and counsel and all other amounts due the Lenders and Administrative Agent under Sections 5, 15.5, and 15.17) allowed in any such proceedings; and

(b) collect and receive any monies or other property payable or deliverable on any such claims and to distribute the same.

14.13.2 Any custodian, receiver, assignee, trustee, liquidator, sequestrator, or other similar official in any such proceeding is hereby authorized by each Lender to make all payments to Administrative Agent and, in the event that Administrative Agent consents to the making of such payments directly to the Lenders, to pay to Administrative Agent any amount due for the reasonable compensation, expenses, disbursements and advances of Administrative Agent and its agents and counsel, and any other amounts due Administrative Agent under Sections 5, 15.5, and 15.17.

14.13.3 Nothing contained in this Agreement will be deemed to authorize Administrative Agent to authorize or consent to or accept or adopt on behalf of any Lender any plan of reorganization, arrangement, adjustment or composition affecting the Obligations or the rights of any Lender or to authorize Administrative Agent to vote in respect of the claim of any Lender in any such proceeding.

14.14Other Agents; Arrangers and Managers. None of the Lenders or other Persons identified on the facing page or signature pages of this Agreement as a "syndication agent," "documentation agent," "co-agent," "book manager," "lead manager," "lead arranger," "lead arranger," or "co-arranger," if any, has any right, power, obligation, liability, responsibility, or duty under this Agreement other than, in the case of any Lender, those applicable to all Lenders as such. Without limiting the foregoing, none of the Lenders or other Persons so identified has or is deemed to have any fiduciary relationship with any Lender. Each Lender acknowledges that it has not relied, and will not rely, on any of the Lenders or other Persons so identified in deciding to enter into this Agreement or in taking or not taking action under this Agreement.

<u>14.15Protective Advances</u>. Administrative Agent may, from time to time at any time that an Event of Default has occurred and is continuing, make all disbursements and advances ("<u>Protective Advances</u>") that Administrative Agent, in its sole discretion, deems necessary or desirable to preserve, protect, prepare for sale or lease or dispose of the Collateral or any portion thereof, to enhance the likelihood or maximize the amount of repayment by the Loan Parties of the Loans and other Obligations or to pay any other amount chargeable to the Loan Parties pursuant to the terms of this Agreement and the other Loan Documents, including, without limitation, costs, fees and expenses as described in <u>Section 15.5</u>. Protective Advances are repayable on demand and will be secured by the Collateral and bear interest at a rate per annum equal to the rate then applicable to Term Loans. The maximum aggregate amount of Protective Advances that Administrative Agent may make is \$20,000,000. Protective Advances constitute Obligations under this Agreement. No Protective Advance made by Administrative Agent and charged to the Loan Account will be deemed to constitute a Loan and no Lender will have any obligation to fund any amount to Administrative Agent as a result thereof. Administrative Agent shall notify each Lender and Borrower

<u>96</u>

in writing of each Protective Advance made by Administrative Agent, which notice must include a description of the purpose of that Protective Advance.

SECTION 15 GENERAL.

15.1Waiver; Amendments.

(a) No amendment, modification, or waiver of, or consent with respect to, any provision of this Agreement or the other Loan Documents (other than an Incremental Facility Amendment pursuant to Section 2.1.2) will be effective unless it is in writing and acknowledged by Lenders having an aggregate Pro Rata Shares of not less than the aggregate Pro Rata Shares expressly designated in this Agreement with respect thereto or, in the absence of any such designation as to any provision of this Agreement, by the Required Lenders. Any amendment, modification, waiver, or consent will be effective only in the specific instance and for the specific purpose for which given.

(b) The Agent Fee Letter may be amended, waived, consented to, or modified by the parties thereto.

of that Lender.

(c) No amendment, modification, waiver, or consent may extend or increase the Commitment of any Lender without the written consent

(d) No amendment, modification, waiver, or consent may waive, or extend the date scheduled for, payment of any principal (in each case, excluding mandatory prepayments) of or interest on the Loans or any fees payable under this Agreement without the written consent of each Lender directly affected thereby.

(e) No amendment, modification, waiver, or consent may reduce the principal amount of any Loan, the rate of interest thereon, or any fees payable under this Agreement without the consent of each Lender directly affected thereby (except (i) for periodic adjustments of interest rates and fees resulting from a change in the Adjusted Term SOFR and the Base Rate as provided for in this Agreement, and (ii) that Required Lenders may rescind any increase in the interest rate under and in accordance with Section 4.1.2).

(f) No amendment, modification, waiver, or consent may do any of the following without the written consent of each Lender: (i) release Borrower or any Guarantor from its obligations, other than as part of or in connection with any disposition permitted under this Agreement; (ii) release all or any substantial part of the Collateral granted under the Collateral Documents (except as permitted by Section 14.11); (iii) change the definitions of Pro Rata Share or Required Lenders, any provision of this Section 15.1, any provision of Section 13.3, or reduce the aggregate Pro Rata Share required to effect an amendment, modification, waiver, or consent.

(g) No provision of <u>Sections 6.1.2, 6.2</u>, or <u>7.2.2(b)</u> with respect to the timing or application of mandatory prepayments of the Loans may be amended, modified, or waived without the consent of Lenders having a majority of the aggregate Pro Rata Shares of the Term Loans affected thereby.

(h) No provision of Section 14 or other provision of this Agreement affecting Administrative Agent in its capacity as such may be amended, modified, or waived without the consent of Administrative Agent.

(i) [Reserved]

<u>97</u>

(j) Notwithstanding the foregoing, this Agreement may be amended (or amended and restated) with the written consent of the Required Lenders, Administrative Agent and Borrower to do any of the following: (i) to add one or more additional credit facilities to this Agreement and to permit the extensions of credit from time to time outstanding thereunder and the accrued interest and fees in respect thereof to share ratably in the benefits of this Agreement and the other Loan Documents with the Term Loans and the accrued interest and fees in respect thereof and (ii) to include appropriately the Lenders holding any such additional credit facilities in any determination of the Required Lenders.

(k) If, in connection with any proposed amendment, modification, waiver or termination requiring the consent of all Lenders, the consent of the Required Lenders is obtained but the consent of other Lenders whose consent is required is not obtained (any such Lender whose consent is not obtained is referred to as a "<u>Non-Consenting Lender</u>"), then, so long as Administrative Agent is not a Non-Consenting Lender, Administrative Agent and/or one or more Persons reasonably acceptable to Administrative Agent may (but will not be required to) purchase from that Non-Consenting Lender, and that Non-Consenting Lenders shall, upon Administrative Agent's request, sell and assign to Administrative Agent and/or any such Person, all of the Loans and Commitments of that Non-Consenting Lender and all accrued interest, fees, expenses, and other amounts then due with respect thereto through the date of sale, which purchase and sale will be consummated pursuant to an executed Assignment Agreement.

15.2Confirmations. Borrower and each holder of a Note agree from time to time, upon written request received by it from the other, to confirm to the other in writing (with a copy of each such confirmation to Administrative Agent) the aggregate unpaid principal amount of the Loans then outstanding under that Note.

15.3Notices.

15.3.1 Generally. Except as otherwise provided in <u>Sections 2.2.2</u> and <u>2.2.3</u>, all notices under this Agreement must be in writing (including facsimile transmission) and must be sent to the applicable party at its address shown on <u>Annex B</u> or at any other address as the receiving party designates, by written notice received by the other parties, as its address for that purpose. Notices sent by facsimile transmission will be deemed to have been given when sent; notices sent by mail will be deemed to have been given three Business Days after the date when sent by registered or certified mail, postage prepaid; and notices sent by hand delivery or overnight courier service will be deemed to have been given when received. For purposes of <u>Sections 2.2.2</u> and <u>2.2.3</u>, Administrative Agent will be entitled to rely on telephonic instructions from any person that Administrative Agent in good faith believes is an authorized officer or employee of Borrower, and Borrower shall hold harmless Administrative Agent and each other Lender from any loss, cost, or expense resulting from any such reliance.

15.3.2 Electronic Communications.

(a) Notices and other communications to any Lender under this Agreement may be delivered or furnished by electronic communication (including e-mail and Internet or intranet websites) pursuant to procedures approved by Administrative Agent, but the foregoing does not apply to notices to any Lender pursuant to <u>Section 2.2</u> if that Lender has notified Administrative Agent and Borrower that it is incapable of receiving notices under <u>Section 2.2</u> by electronic communication. Administrative Agent or Borrower may, in its respective sole discretion, agree to accept notices and other communications to it under this Agreement by electronic communications

pursuant to procedures approved by it, and approval of any such procedures may be limited to particular notices or communications.

(b) Unless otherwise agreed by the sender and the intended recipient, (i) notices and other communications sent to an e-mail address will be deemed received upon the sender's receipt of an acknowledgement from the intended recipient (such as by the "return receipt requested" function, as available, return e-mail, or other written acknowledgement); (ii) notices or communications posted to an Internet or intranet website will be deemed received upon the deemed receipt by the intended recipient, at its e-mail address as described in the foregoing <u>clause (i)</u>, of notification that the notice or communication is available and identifying the website address therefor; and (iii) for both <u>clauses (i)</u> and (ii) of this <u>Section 15.3.2(b)</u>, any notice, e-mail or other communication that is not sent during the normal business hours of the intended recipient will be deemed to have been sent at the opening of business on the next Business Day for the intended recipient.

15.4Computations. Where the character or amount of any asset or liability or item of income or expense is required to be determined, or any consolidation or other accounting computation is required to be made, for the purpose of this Agreement, that determination or calculation will, to the extent applicable and except as otherwise specified in this Agreement, be made in accordance with GAAP, consistently applied, but if Borrower notifies Administrative Agent that Borrower wishes to amend any covenant in Section 10 or 11.12 (or any related definition) to eliminate or to take into account the effect of any change in GAAP on the operation of that covenant (or if Administrative Agent notifies Borrower that the Required Lenders wish to amend Section 10 or 11.12 (or any related definition) for that purpose), then Borrower's compliance with that covenant will be determined on the basis of GAAP in effect immediately before the relevant change in GAAP became effective, until either the applicable notice under this Section 15.4 is withdrawn or the applicable covenant (or related definition) is amended in a manner satisfactory to Borrower and the Required Lenders.

15.5Costs and Expenses. Borrower shall pay on demand all reasonable out-of-pocket costs and expenses of Administrative Agent (including, without limitation, Attorney Costs) in connection with the preparation, execution, syndication, delivery and administration (including perfection and protection of any Collateral and the costs of IntraLinks (or other similar service), if applicable) of this Agreement, the other Loan Documents, and all other documents provided for in this Agreement or delivered or to be delivered under or in connection with this Agreement (including any amendment, supplement, or waiver to any Loan Document), whether or not the transactions contemplated hereby or thereby are consummated, including, without limitation, all reasonable and documented out-of-pocket costs and expenses incurred pursuant to Section 10.2, and all reasonable out-of-pocket costs and expenses (including, without limitation, Attorney Costs) incurred by Administrative Agent and each Lender after an Event of Default in connection with the collection of the Obligations or the enforcement of this Agreement the other Loan Documents or any such other documents or during any workout, restructuring, or negotiations in respect thereof. In addition, each Loan Party shall pay, and shall save and hold harmless Administrative Agent and the Lenders from all liability for, any fees of Borrower's auditors in connection with any reasonable exercise by Administrative Agent and the Lenders of their rights pursuant to Section 10.2. All Obligations provided for in this Section 15.5 will survive repayment of the Loans, cancellation of the Notes and termination of this Agreement. This Section 15.5 shall not apply with respect to Taxes other than any Taxes that represent losses, claims, damages, etc. arising from any non-Tax claim.

15.6Assignments; Participations.

15.6.1 Assignments.

(a) Any Lender may at any time assign to one or more Persons (any such Person, an "Assignee") all or any portion of that Lender's (a) Loans and Commitments, with the prior written consent of Administrative Agent and, so long as no Event of Default exists, Borrower (which consent of Borrower may not be unreasonably withheld or delayed, except in the case of an assignment to a Disqualified Institution in which case consent of Borrower may be provided or withheld in Borrower's sole discretion and which consent, for avoidance of doubt, shall not be required from Borrower at any time while (x) an Event of Default pursuant to Sections 13.1.1, 13.1.2, or 13.1.4 has occurred and is continuing and (v) any other Event of Default has occurred and is continuing for a period of more than ten consecutive Business Days), but (i) no such consent of any kind is required for an assignment by a Lender to a Lender or an Affiliate of a Lender or an Approved Fund, and (ii) no assignment may be made to a Loan Party or an Affiliate of a Loan Party. Except as Administrative Agent otherwise agrees, any such assignment must be in a minimum aggregate amount equal to \$1,000,000 (which minimum will be \$250,000 if the assignment is to an Affiliate of the assigning Lender) or, if less, the remaining Commitment and Loans held by the assigning Lender. Borrower and Administrative Agent will be entitled to continue to deal solely and directly with the assigning Lender in connection with the interests so assigned to an Assignee until Administrative Agent has received and accepted an effective assignment agreement in substantially the form of Exhibit C (an "Assignment Agreement") executed, delivered, and fully completed by the applicable parties thereto and a processing fee of \$3,500. No assignment may be made to any Person if at the time of that assignment Borrower would be obligated to pay any greater amount under Section 7.6 or Section 8 to the Assignee than Borrower is then obligated to pay to the assigning Lender under that section (and if any assignment is made in violation of the foregoing, Borrower will not be required to pay any such greater amounts). Any attempted assignment not made in accordance with this Section 15.6.1 will be treated as the sale of a participation under Section 15.6.2, except for an assignment to a Disqualified Institution, which assignment to a Disqualified Institution will be null and void (unless such assignment is made during any time while an Event of Default has occurred and is continuing or with the consent of Borrower). Borrower will be deemed to have granted its consent to any assignment requiring its consent under this Agreement unless Borrower has expressly objected in writing to that assignment within five (5) Business Days after notice thereof.

(b) From and after the date on which the conditions described above have been met, (i) the Assignee will be deemed automatically to have become a party to this Agreement and, to the extent that rights and obligations under this Agreement have been assigned to that Assignee pursuant to the Assignment Agreement, will have the rights and obligations of a Lender under this Agreement, and (ii) the assigning Lender, to the extent that rights and obligations under this Agreement, will be released from its rights (other than its indemnification rights) and obligations under this Agreement. Upon the request of the Assignee (and, as applicable, the assigning Lender) pursuant to an effective Assignment Agreement, Borrower shall execute and deliver to Administrative Agent for delivery to the Assignee (and, as applicable, the assigning Lender) one or more Notes in accordance with Section 3.1 to reflect the amounts assigned to that Assignee and the amounts, if any, retained by the assigning Lender. Each such Note will be dated the effective date of the applicable assignment. Upon receipt by Administrative Agent of any such Note, the assigning Lender shall return to Borrower any applicable prior Note held by it.

(c) Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of that Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank, and this <u>Section 15.6.1</u> will not apply to any such pledge or assignment of a security interest. No such pledge or assignment of a security

<u>100</u>

interest will release a Lender from any of its obligations under this Agreement or substitute any such pledgee or assignee for that Lender as a party to this Agreement.

15.6.2 Participations. Any Lender may at any time sell to one or more Persons (other than (x) a Disgualified Institution unless expressly consented to by Borrower (which consent shall not be required from Borrower at any time while an Event of Default has occurred and is continuing) and (y) a natural person, or a holding company, investment vehicle or trust for, or owned and operated for the primary benefit of, a natural person, or Borrower or any of Borrower's Affiliates or Subsidiaries) participating interests in its Loans, Commitments or other interests under this Agreement (including all or a portion of its Commitment and/or the Loans owing to it) (any such Person, a "Participant"); provided that, in the event of a sale by a Lender of a participating interest to a Participant, (a) that Lender's obligations under this Agreement will remain unchanged for all purposes; (b) Borrower and Administrative Agent shall continue to deal solely and directly with that Lender in connection with that Lender's rights and obligations under this Agreement; and (c) all amounts payable by Borrower will be determined as if that Lender had not sold that participation and will be paid directly to that Lender. No Participant will have any direct or indirect voting rights under this Agreement except with respect to any event described in Section 15.1 expressly requiring the unanimous vote of all Lenders or, as applicable, all affected Lenders. Each Lender agrees to incorporate the requirements of the preceding sentence into each participation agreement which that Lender enters into with any Participant. Borrower agrees that if amounts outstanding under this Agreement are due and payable (as a result of acceleration or otherwise), each Participant will be deemed to have the right of set-off in respect of its participating interest in amounts owing under this Agreement to the same extent as if the amount of its participating interest were owing directly to it as a Lender under this Agreement, but that right of set-off is subject to the obligation of each Participant to share with the Lenders, and the Lenders shall share with each Participant, as provided in Section 7.5. Participant will be entitled to the benefits of Section 7.6 and Section 8 as if it were a Lender (but on the date of the participation no Participant will be entitled to any greater compensation pursuant to Section 7.6 or Section 8 than would have been paid to the participating Lender on that date if no participation had been sold, and each Participant must comply with Section 7.6.4 as if it were an Assignee). Each Lender that sells a participation shall, acting solely for this purpose as a non-fiduciary agent of Borrower, maintain a register on which it enters the name and address of each Participant and the principal amounts (and stated interest) of each Participant's interest in the Loans or other obligations under the Loan Documents (the "Participant Register"); provided that no Lender shall have any obligation to disclose all or any portion of the Participant Register (including the identity of any Participant or any information relating to a Participant's interest in any commitments, loans or its other obligations under any Loan Document) to any Person except to the extent that such disclosure is necessary to establish that such commitment, loan or other obligation is in registered form under Section 5f.103-1(c) of the United States Treasury Regulations. The entries in the Participant Register shall be conclusive absent manifest error, and such Lender shall treat each Person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary. For the avoidance of doubt, Administrative Agent (in its capacity as Administrative Agent) shall have no responsibility for maintaining a Participant Register.

15.7Register. Administrative Agent shall maintain, at its offices in the United States, and deliver a copy to Borrower upon written request, a copy of each Assignment Agreement delivered and accepted by it and register (the "Register") for the recordation of names and addresses of the Lenders and the Commitment of each Lender from time to time and whether that Lender is the original Lender or the Assignment will be effective unless and until the Assignment Agreement is accepted and registered in the Register. All records of transfer of a Lender's interest in the Register will be conclusive,

<u>101</u>

absent manifest error, as to the ownership of the interests in the Loans. Administrative Agent will not incur any liability of any kind with respect to any Lender with respect to the maintenance of the Register. It is the intention that the Loans and Commitments be treated as registered obligations and in "registered form" within the meaning of Sections 163(f), 871(h)(2) and 881(c)(2) of the Code, and that the right, title, and interest of the Lenders in and to those Loans and Commitments be transferable only in accordance with the terms of this Agreement.

15.8GOVERNING Law. This Agreement and each Note is a contract made under and governed by the internal laws of the State of New York applicable to contracts made and to be performed entirely within that state, without regard to conflict-of-laws principles.

15.9Confidentiality. As required by federal law and Administrative Agent's policies and practices, Administrative Agent may need to obtain, verify, and record certain customer identification information and documentation in connection with opening or maintaining accounts, or establishing or continuing to provide services. Administrative Agent and each Lender shall use commercially reasonable efforts (equivalent to the efforts Administrative Agent or that Lender applies to maintain the confidentiality of its own confidential information) to maintain as confidential all non-public information provided to them by any Loan Party in connection with this Agreement and the transactions contemplated thereby, except that Administrative Agent and each Lender may disclose any information as follows: (a) to Persons employed or engaged by Administrative Agent or that Lender or that Lender's Affiliates or Approved Funds in evaluating, approving, structuring, or administering the Loans and the Commitments so long as such Person is bound by the confidentiality provisions of this Agreement; (b) to any assignee or participant or potential assignee or participant that has agreed to comply with the covenant contained in this Section 15.9 (and any such assignee or participant or potential assignee or participant may disclose any such information to Persons employed or engaged by them as described in clause (a) of this Section 15.9 but in all cases subject to the confidentiality provisions of this Agreement); (c) as required or requested by any federal or state regulatory authority or examiner, or any insurance industry association, or as reasonably believed by Administrative Agent or that Lender to be compelled by any court decree, subpoena, or legal or administrative order or process, but Administrative Agent or that Lender, as applicable, shall (i) use reasonable efforts to give the applicable Loan Party written notice prior to disclosing the information to the extent permitted by that requirement, request, court decree, subpoena, or legal or administrative order or process, and (ii) disclose only that portion of the confidential information as Administrative Agent or that Lender reasonably believes, or as counsel for Administrative Agent or that Lender, as applicable, advises Administrative Agent or that Lender, that it must disclose pursuant to that requirement; (d) as Administrative Agent or that Lender reasonably believes, or on the advice of Administrative Agent's or that Lender's counsel, is required by law; (e) in connection with the exercise or potential exercise of any right or remedy under the Loan Documents pursuant to legal or judicial process or in connection with any litigation to which Administrative Agent or that Lender is a party; (f) to any nationally recognized rating agency that requires access to information about a Lender's investment portfolio in connection with ratings issued with respect to that Lender but only so long as such agency agrees to the confidentiality provisions hereof or is otherwise bound by customary confidentiality requirements; (g) to any Affiliate of Administrative Agent or any Lender who might provide bank products and cash management services to the Loan Parties but in all cases subject to the confidentiality provisions hereof; (h) to that Lender's independent auditors and other professional advisors as to which that information has been identified as confidential; or (i) if that information ceases to be confidential through no fault of Administrative Agent or any Lender. Notwithstanding the foregoing, Borrower consents to the publication by Administrative Agent or any Lender of a tombstone or similar advertising material relating to the financing transactions contemplated by this Agreement, in consultation with Borrower, and Administrative Agent reserves the right to provide to industry trade organizations information necessary and customary for inclusion in league table measurements. If any provision of any confidentiality agreement, non-disclosure agreement, or other similar agreement between Borrower and any Lender conflicts with or contradicts this Section 15.9 with

<u>102</u>

respect to the treatment of confidential information, then this <u>Section 15.9</u> will supersede all such prior or contemporaneous agreements and understandings between the parties.

15.10Severability. Whenever possible each provision of this Agreement is to be interpreted so as to be effective and valid under applicable law, but if any provision of this Agreement is prohibited by or invalid under applicable law, that provision will be ineffective to the extent of that prohibition or invalidity, without invalidating the remainder of that provision or the remaining provisions of this Agreement.

15.11Nature of Remedies. All Obligations of the Loan Parties and rights of Administrative Agent and the Lenders expressed in this Agreement or in any other Loan Document are in addition to and not in limitation of those provided by applicable law. No failure to exercise, and no delay in exercising, on the part of Administrative Agent or any Lender, any right, remedy, power, or privilege under this Agreement will operate as a waiver thereof, and no single or partial exercise of any right, remedy, power, or privilege under this Agreement will preclude any other or further exercise thereof or the exercise of any other right, remedy, power, or privilege.

15.12Entire Agreement. This Agreement, together with the other Loan Documents, embodies the entire agreement and understanding among the parties to this Agreement and supersedes all prior or contemporaneous agreements and understandings of all such Persons, verbal or written, relating to the subject matter hereof and thereof (except as relates to the fees described in <u>Section 5.1</u>) and any prior arrangements made with respect to the payment by the Loan Parties of (or any indemnification for) any fees, costs, or expenses payable to or incurred (or to be incurred) by or on behalf of Administrative Agent or the Lenders.

15.13Counterparts. This Agreement may be executed in any number of counterparts and by the different parties on separate counterparts. Each such counterpart will be deemed to be an original, but all such counterparts will together constitute but one and the same Agreement. Receipt of an executed signature page to this Agreement by facsimile or other electronic transmission will constitute effective delivery thereof. Electronic records of executed Loan Documents maintained by the Lenders will be deemed to be originals.

15.14Successors and Assigns. This Agreement binds Borrower, the Lenders, Administrative Agent, and their respective successors and assigns and will inure to the benefit of Borrower, the Lenders, and Administrative Agent and the successors and assigns of the Lenders and Administrative Agent. No other Person is or is intended to be a direct or indirect legal beneficiary of, or have any direct or indirect cause of action or claim in connection with, this Agreement or any of the other Loan Documents. No Loan Party may assign or transfer any of its rights or Obligations under this Agreement without the prior written consent of Administrative Agent and each Lender.

15.15Captions. Section captions used in this Agreement are for convenience only and do not affect the construction of this Agreement.

15.16Customer Identification—USA Patriot Act Notice. Each Lender and Monroe Capital (each for itself and not on behalf of any other party) hereby notifies the Loan Parties that, pursuant to the requirements of the USA Patriot Act, Title III of Pub. L. 107-56, signed into law October 26, 2001 (the "Patriot Act"), it is required to obtain, verify, and record information that identifies the Loan Parties, which information includes the name and address of the Loan Parties and other information that will allow that Lender or Monroe Capital, as applicable, to identify the Loan Parties in accordance with the Patriot Act.

15.17INDEMNIFICATION BY LOAN PARTIES. In consideration of the execution and delivery of this Agreement by Administrative Agent and the Lenders and the agreement to extend the

<u>103</u>

Commitments provided under this Agreement, Borrower hereby agrees to indemnify, exonerate, and hold harmless Administrative Agent, each Lender and each of the officers, directors, employees, Affiliates, agents, and Approved Funds of Administrative Agent and each Lender (each, a "Lender Party") from and against any and all actions, causes of action, suits, losses, liabilities, damages, and expenses, including Attorney Costs (collectively, the "Indemnified Liabilities"), incurred by the Lender Parties or any of them as a result of, or arising out of, or relating to (a) any tender offer, merger, purchase of capital securities, purchase of assets or other similar transaction financed or proposed to be financed in whole or in part, directly or indirectly, with the proceeds of any of the Loans; (b) the use, handling, release, emission, discharge, transportation, storage, treatment or disposal of any Hazardous Substance at any property owned or leased by any Loan Party, any Environmental Claim related to any Loan Party; (c) the execution, delivery, performance, or enforcement of this Agreement or any other Loan Document by any of the Lender Parties, in each case except for any such Indemnified Liabilities arising on account of the applicable Lender Party's gross negligence or willful misconduct as determined by a final, non-appealable judgment by a court of competent jurisdiction. If and to the extent that the foregoing undertaking is unenforceable for any reason, Borrower hereby agrees to make the maximum contribution to the payment and satisfaction of each of the Indemnified Liabilities which is permissible under applicable law. All obligations provided for in this Section 15.17 will survive repayment of the Loans, cancellation of the Notes, any foreclosure under, or any modification, release, or discharge of, any or all of the Collateral Documents and termination of this Agreement. This <u>Section 15.17</u> shall not apply with respect to Taxes other than any Taxes that represent losses, claims, damages, etc. arising fr

15.18Non-Liability of Lenders.

(a) The relationship between Borrower on the one hand and the Lenders and Administrative Agent on the other hand is solely that of borrower and lender. Neither Administrative Agent nor any Lender has any fiduciary relationship with or duty to any Loan Party arising out of or in connection with this Agreement or any of the other Loan Documents, and the relationship between the Loan Parties, on the one hand, and Administrative Agent and the Lenders, on the other hand, in connection herewith or therewith is solely that of debtor and creditor. Neither Administrative Agent nor any Lender undertakes any responsibility to any Loan Party to review or inform any Loan Party of any matter in connection with any phase of any Loan Party's business or operations. Borrower agrees, on behalf of itself and each other Loan Party, that neither Administrative Agent nor any Lender has any liability to any Loan Party (whether sounding in tort, contract or otherwise) for losses suffered by any Loan Party in connection with, arising out of, or in any way related to the transactions contemplated and the relationship established by the Loan Documents, or any act, omission, or event occurring in connection therewith, unless it is determined in a final non-appealable judgment by a court of competent jurisdiction that those losses resulted from the gross negligence or willful misconduct of the party from which recovery is sought.

(b) No Lender Party will be liable for any damages arising from the use by others of any information or other materials obtained through IntraLinks or other similar information transmission systems in connection with this Agreement. No Lender Party will have any liability with respect to, and Borrower, on behalf of itself and each other Loan Party, hereby waives, releases, and agrees not to sue for, any special, punitive, exemplary, indirect, or consequential damages relating to this Agreement or any other Loan Document or arising out of its activities in connection herewith or therewith (whether before or after the Closing Date).

(c) Each Loan Party acknowledges that it has been advised by counsel in the negotiation, execution and delivery of this Agreement and the other Loan Documents to which it is a party.

<u>104</u>

No joint venture is created hereby or by the other Loan Documents or otherwise exists by virtue of the transactions contemplated hereby among the Lenders or among the Loan Parties and the Lenders.

15.19FORUM SELECTION AND CONSENT TO JURISDICTION. Any litigation based hereon, or arising out of, under, or in connection with this Agreement or any other Loan Document, will be brought and maintained exclusively in the courts of the State of New York or in the United States District Court for the Southern District of New York, but nothing in this Agreement will be deemed or operate to preclude Administrative Agent from bringing suit or taking other legal action in any other jurisdiction. Borrower hereby expressly and irrevocably submits to the jurisdiction of the courts of the State of New York and of the United States District Court for the Southern District of New York for the purpose of any such litigation as set forth above. Borrower further irrevocably consents to the service of process by registered mail, postage prepaid, or by personal service within or without the State of New York. Borrower hereby expressly and irrevocably waives, to the fullest extent permitted by law, any objection that it now has or hereafter might have to the laying of venue of any such litigation brought in any such court referred to above and any claim that any such litigation has been brought in an inconvenient forum.

15.20WAIVER OF JURY TRIAL. Borrower, Administrative Agent, and each Lender hereby waives any right to a trial by jury in any action or proceeding to enforce or defend any rights under this Agreement, any Note, any other Loan Document, and any amendment, instrument, document, or agreement delivered or which might in the future be delivered in connection with this Agreement or therewith or arising from any lending relationship existing in connection with any of the foregoing, and agrees that any such action or proceeding will be tried before a court and not before a jury.

15.21Acknowledgement and Consent to Bail-In of EEA Financial Institutions. Notwithstanding anything to the contrary in any Loan Document or in any other agreement, arrangement, or understanding among any such parties, each party hereto acknowledges that any liability of any EEA Financial Institution arising under any Loan Document, to the extent such liability is unsecured, may be subject to the write-down and conversion powers of an EEA Resolution Authority and agrees and consents to, and acknowledges and agrees to be bound by (a) the application of any Write-Down and Conversion Powers by an EEA Resolution Authority to any such liability, including, if applicable: (i) a reduction in full or in part or cancellation of any such liability; (ii) a conversion of all, or a portion of, such liability into shares or other instruments of ownership in that EEA Financial Institution, its parent undertaking, or a bridge institution that may be issued to it or otherwise conferred on it, and that such shares or other instruments of ownership will be accepted by it in lieu of any rights with respect to any such liability under this Agreement or any other Loan Document; or (iii) the variation of the terms of such liability in connection with the exercise of the write-down and conversion powers of any EEA Resolution Authority.

[Signature pages follow]

<u>105</u>

[*****] = Certain identified information has been excluded from the exhibit because it is both not material and is the type that the Registrant treats as private or confidential

Execution Version

MASTER RECEIVABLES PURCHASE AGREEMENT

among

SOUND POINT CAPITAL MANAGEMENT, LP, as Purchaser Agent,

SP MAIN STREET FUNDING I LLC

as Initial Purchaser THE ADDITIONAL PURCHASERS TIME TO TIME PARTY HERETO,

and

ML PLUS LLC, as Seller

dated as of June 30, 2024

[MoneyLion] Master Receivables Purchase Agreement

TABLE OF CONTENTS

		Page
Section 1.	Definitions.	1
Section 2.	Sale of Eligible Receivables; Related Covenants.	15
Section 3.	Sale Procedures; Seller Delivery Obligations.	16
Section 4.	Purchase Price; Payment; Unused Fee.	17
Section 5.	Seller's Representations and Warranties.	18
Section 6.	Purchasers' Representations and Warranties.	18
Section 7.	Repurchase Rights and Obligations.	18
Section 8.	Conditions to Purchase and Sale.	20
Section 9.	Covenants of the Seller.	21
Section 10.	Events of Termination.	22
Section 11.	Termination.	24
Section 12.	Backup Servicing Agreement	24
Section 13.	Miscellaneous Provisions.	25
Section 14.	Confidentiality	29
Section 15.	Joinder of Additional Purchasers.	30
Section 16.	Increase in Facility Limit.	31

i

SCHEDULE I COMMITMENTS SCHEDULE II TARGET SALE AMOUNTS SCHEDULE III PURCHASE PRICE CALCULATIONS SCHEDULE IV CALCULATION OF REVISED PERCENTAGE SCHEDULE V SELLER INFORMATION SCHEDULE VI LIST OF COMPETITORS

[MoneyLion] Master Receivables Purchase Agreement

<u>Page</u>

- ANNEX A ELIGIBILITY CRITERIA AND CONCENTRATION LIMITATIONS
- ANNEX B SELLER'S REPRESENTATIONS AND WARRANTIES
- ANNEX C PURCHASER'S REPRESENTATIONS AND WARRANTIES
- EXHIBIT A FORM OF CONFIRMATION
- EXHIBIT B FORM OF BACK-UP UCC-1
- EXHIBIT C FORM OF REPURCHASE CONFIRMATION
- EXHIBIT D FORM OF JOINDER
- EXHIBIT E ELIGIBILITY GUIDELINES
- EXHIBIT F FORM OF COMPLIANCE CERTIFICATE

-ii-

MASTER RECEIVABLES PURCHASE AGREEMENT

This Master Receivables Purchase Agreement (this "Agreement") is made and entered into as of June 30, 2024, by and among Sound Point Capital Management, LP, a Delaware limited partnership, as purchaser agent (the "Purchaser Agent"), SP Main Street Funding I LLC, a Delaware limited liability company (the "Initial Purchaser"), each additional Purchaser that may from time to time become party hereto by execution of the attached joinder supplement substantially in the form of <u>Exhibit D</u> (each, an "Additional Purchaser" and, together with the Initial Purchaser, each individually, a "Purchaser" and collectively, the "Purchasers") and ML Plus LLC, a Delaware limited liability company, as seller (the "Seller").

WITNESSETH:

WHEREAS, the Purchasers desire to acquire certain Assets (as defined below) from the Seller from time to time, and the Seller desires to sell such Assets to the Purchasers from time to time, in each case in accordance with the terms and conditions of this Agreement; and

WHEREAS, the Purchasers desire to appoint the Purchaser Agent to act on its or their behalf, in each case in accordance with the terms and conditions of this Agreement.

NOW, THEREFORE, in consideration of the premises and mutual covenants herein contained, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties to this Agreement hereby agree as follows:

Section 1. Definitions.

"**3-Month Average Unpaid Rate (30 Days)**" means, as of any Measurement Date, the average Unpaid Rate (30 Days) for the Monthly Vintage Pool ending on the last day of the month that is two (2) calendar months prior to such Measurement Date and the two immediately preceding Monthly Vintage Pools.

"**3-Month Average Unpaid Rate (60 Days)**" means, as of any Measurement Date, the average Unpaid Rate (60 Days) for the Monthly Vintage Pool ending on the last day of the month that is three (3) calendar months prior to such Measurement Date and the two immediately preceding Monthly Vintage Pools.

"Account Bank" means Silicon Valley Bank, a division of First Citizens Bank and its successors and assigns.

"Account Control Agreement" means, with respect to the Collection Account, an account control agreement, in form and substance satisfactory to the Purchaser Agent, among the Purchaser Agent, the Account Bank and the Seller, as amended, supplemented or otherwise modified from time to time.

1

"Additional Purchaser" has the meaning set forth in the preamble to this Agreement.

"Advance Collections" means, with respect to any Receivable, all Collections thereon classified as a repayment of the Instacash Advance Amount of such Receivable.

"Affiliate" means, with respect to any Person, any other Person which, directly or indirectly, controls, is controlled by, or is under common control with, such Person. For purposes of this definition, a Person shall be deemed to control another Person if the controlling Person possesses, directly or indirectly, the power to direct or cause the direction of the management and policies of the other Person, whether through the ownership of voting securities, by contract or otherwise.

"Agreement" has the meaning set forth in the preamble to this Agreement, and includes all exhibits, annexes and schedules attached hereto, and any addenda, supplements or amendments hereto.

"Applicable Law" shall mean any and all federal, state, local and/or applicable foreign statutes, ordinances, rules, regulations, court orders and decrees, administrative orders and decrees, and other legal requirements of any and every conceivable type applicable to the Receivables, Seller or Servicer.

"Asset" means a Receivable and Related Rights (including, without limitation, the MoneyLion Instacash Terms and Conditions and all rights of the Seller thereunder), together with any other property and any other rights related thereto.

"Backup Servicer" means Vervent Inc. or such other Person mutually selected by the Servicer and the Purchaser Agent.

"Backup Servicing Agreement" means the Backup Servicing Agreement (however styled) to be entered into by and among the Backup Servicer, the Servicer and the Purchaser Agent after the Closing Date, as amended, supplemented or otherwise modified from time to time.

"Bankruptcy Code" means The Bankruptcy Reform Act of 1978, as amended from time to time, and as codified as 11 U.S.C. Section 101 et seq.

"Business Day" means any day excluding Saturday, Sunday and any day which is a legal holiday under the laws of the State of New York or is a day on which banking institutions located in New York are authorized or required by law or other governmental action to close.

"Calculation Tenor" means, for each Monthly Vintage Pool, a period of 30, 60, 90, 180, 270 or 360 days, as applicable, in each case beginning on the latest Instacash Repayment Date for any Receivable contained in such Monthly Vintage Pool; provided, that, for purposes of calculating the Purchase Price, so long as the Receivables contained in any Monthly Vintage Pool with the related Instacash Repayment Date greater than 30 days following the related Instacash Disbursement Date (such Receivables, the "**30 Day Plus Due Date Receivables**") represent less than or equal to 1% of all Receivables included in such Monthly Vintage Pool, the Instacash Repayment Date for the 30 Day Plus Due Date Receivables shall be deemed to be 30 days following the related Instacash Disbursement Date.

[MoneyLion] Master Receivables Purchase Agreement

"Change of Control" means, at any time, any of the following events shall occur:

(a) with respect to the Servicer, Parent shall cease to Beneficially Own and control 100% on a fully diluted basis of the economic and voting interest in the Stock of the Servicer;

(b) with respect to the Seller, the Servicer shall cease to Beneficially Own and control 100% on a fully diluted basis of the economic and voting interest in the Stock of Seller;

(c) with respect to the Parent, any Unrelated Person or any Unrelated Persons, acting together, that would constitute a Group together with any Affiliates or Related Persons thereof (in each case also constituting Unrelated Persons) other than one or more Permitted Holders shall at any time Beneficially Own more than 50% of the aggregate voting power of all classes of Voting Stock of the Parent; or

(d) the Seller merges or consolidates with any other Person and after giving effect to such merger or consolidation, the Seller is not the surviving entity, or the Seller sells all or substantially all of its assets to any Person.

As used herein, (a) "Beneficially Own" shall mean "beneficially own" as defined in Rule 13d-3 of the Exchange Act, or any successor provision thereto; provided, however, that, for purposes of this definition, a Person shall not be deemed to Beneficially Own securities tendered pursuant to a tender or exchange offer made by or on behalf of such Person or any of such Person's Affiliates until such tendered securities are accepted for purchase or exchange; (b) "Group" shall mean a "group" for purposes of Section 13(d) of the Exchange Act; (c) "Unrelated Person" shall mean at any time any Person other than the Seller or any of its Subsidiaries and other than any trust for any employee benefit plan of the Seller or any of its Subsidiaries; (d) "Related Person" shall mean any other Person owning (1) 5% or more of the outstanding common stock of such Person, or (2) 5% or more of the Voting Stock of such Person; and (e) "Voting Stock" of any Person shall mean the capital stock or other indicia of equity rights of such Person which at the time has the power to vote for the election of one or more members of the Board of Directors (or other governing body) of such Person.

"Closing Date" means June 30, 2024.

"Code" means the Internal Revenue Code of 1986, as amended.

"Concentration Limitations" means the criteria set forth on <u>Annex A</u>.

"Collection Account" means the deposit account held at the Account Bank in the name of the Servicer on behalf of the Purchasers with account number ended 2168, which at all times shall be subject to the Account Control Agreement.

"Collections" means, with respect to a Purchased Receivable, all payments made by or on behalf of the MoneyLion Accountholders in respect of any Purchased Receivable and the Related Rights, and any proceeds thereof, whether in the form of cash, checks, wire transfers, electronic transfers or any other form of cash payment. For the avoidance of doubt, any ACH return fees or

[MoneyLion] Master Receivables Purchase Agreement

debit card chargebacks in respect of any Purchased Receivable and the Related Rights shall not constitute Collections.

"Commitment" means, with respect to each Purchaser, the aggregate maximum Purchase Price amount such Purchaser is obligated to pay (without utilizing any Recycled Collection) and keep outstanding hereunder on account of all Purchases as set forth opposite such Purchaser's name on <u>Schedule I</u> attached to this Agreement, as such amount may be modified in accordance with the terms hereof.

"Competitor" means any company or business that operates in any line of business, or offers any product or service offering, of the type performed or offered by Parent in the consumer finance or embedded finance marketplace industries and listed on Schedule VI hereto, as the same may be amended by notice from the Seller to the Purchaser Agent from time to time.

"Compliance Certificate" means the form of Compliance Certificate attached hereto as <u>Exhibit F</u>, executed by an authorized signatory of the Seller and delivered in accordance with Section 9(c).

"Confirmation" means an electronic notification from the Seller to a Purchaser, in a form substantially similar to the form attached as **Exhibit A**, notifying such Purchaser of the intended sale of certain Receivables on a specified Purchase Date.

"Controlled Group" means all members of a controlled group of corporations, all members of a controlled group of trades or businesses (whether or not incorporated) under common control and all members of an affiliated service group which, together with the Seller or any Subsidiary of the Seller (or (x) in the case of the Purchaser Agent, together the Purchaser Agent or any Subsidiary of the Purchaser Agent and (y) in the case of a Purchaser, together such Purchaser or any Subsidiary of such Purchaser), are treated as a single employer under Section 414 of the Code or Section 4001(b) of ERISA.

"Debt" of any Person means, without duplication: (a) all indebtedness for borrowed money; (b) all obligations issued, undertaken or assumed as the deferred purchase price of property or services (other than trade payables entered into in the ordinary course of business); (c) the face amount of all letters of credit issued for the account of such Person and without duplication, all drafts drawn thereunder and all reimbursement or payment obligations with respect to letters of credit, surety bonds and other similar instruments issued by such Person; (d) all obligations evidenced by notes, bonds, debentures or similar instruments, including obligations so evidenced incurred in connection with the acquisition of property, assets or businesses; (e) all indebtedness created or arising under any conditional sale or other title retention agreement, or incurred as financing, in either case with respect to property acquired by such Person; (f) all obligations in respect of any lease of, or other arrangement conveying the right to use, any property by such Person as lessee that has been or should be accounted for as a capital lease on a balance sheet of such person prepared in accordance with GAAP; (g) the principal balance outstanding under any synthetic lease, off-balance sheet loan or similar off balance sheet financing product; (h) all obligations, whether or not contingent, to purchase, redeem, retire, defease or otherwise acquire for value any shares of capital stock, equity interests, beneficial, partnership or membership

4

interests, joint venture interests, participations or other profit interests in or equivalents of any Person (other than an individual), or any securities convertible into or exchangeable for any of the foregoing or warrants, options or other rights to purchase, subscribe for or otherwise acquire any of the foregoing; (i) all indebtedness referred to in clauses (a) through (h) above secured by (or for which the holder of such Debt has an existing right, contingent or otherwise, to be secured by) any lien or security interest upon or in property (including accounts and contracts rights) owned by such Person, even though such Person has not assumed or become liable for the payment of such indebtedness; and (j) all contingent obligations, including, without limitation, liabilities under any guarantees.

"Debtor Relief Laws" means the Bankruptcy Code, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief laws of the United States or other applicable jurisdictions from time to time in effect and affecting the rights of creditors generally.

"Defective Receivable" has the meaning set forth in Section 7.

"Dollars" and the sign "\$" mean the lawful money of the United States.

"Eligibility Criteria" means the criteria specified in <u>Annex A</u> hereto, subject to any changes agreed to in writing by the Purchaser Agent and the Seller after the date hereof.

"Eligibility Guidelines" means the Eligibility Guidelines of the Servicer for Instacash in effect on the Closing Date, a copy of which is attached hereto as <u>Exhibit E</u>, as amended, supplemented or otherwise modified from time to time in accordance with the terms of Section 9(c).

"Eligible MoneyLion Accountholder" means, with respect to any Receivable, a MoneyLion Accountholder that (a) is a natural person at least eighteen (18) years of age and (b) is not an employee, or affiliated with any employee of, the Servicer or any of its respective Affiliates.

"Eligible Receivable" means a Receivable with respect to which the Eligibility Criteria are satisfied as of the applicable date of determination.

"Eligible Receivables Schedule" means the list of Eligible Receivables offered for sale to a Purchaser on any Purchase Date.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended to the date hereof and from time to time hereafter, and any successor statute as in effect from time to time. References to sections of ERISA also refer to any successor sections.

"Excluded Amounts" means with respect to Instacash, any voluntary tips paid by the applicable MoneyLion Accountholder and any fees charged and collected in connection therewith including, without limitation, any Turbo Fees, wire transfer fees, payment processing fees and other similar fees and charges. For the avoidance of doubt, tips and fees are separate financial instruments and not part of the sale transaction contemplated by this Agreement.

5

"Experiment Tier" means an alternative underwriting flow for eligibility and rule testing (it being understood that MoneyLion Accountholder may transition from the Experiment Tier to other tiers based on certain actions (positive performance and/or addition of direct deposit)).

"Facility Limit" means \$175,000,000 which amount may be increased by up to \$75,000,000 pursuant to Section 16.

"FATCA" means Sections 1471 through 1474 of the Internal Revenue Code, as of the date of this Agreement (or any amended or successor version that is substantially comparable and not materially more onerous to comply with), any current or future United States Treasury Regulations promulgated thereunder or interpretations thereof, any applicable agreement entered into pursuant to Section 1471(b)(1) of the Internal Revenue Code, and any fiscal or regulatory legislation, rules or practices adopted pursuant to any intergovernmental agreements, treaty or convention among Governmental Authorities and implementing such Sections of the Internal Revenue Code.

"Funding Account" means the deposit account designated by the Seller in writing to the Purchaser Agent into which the Seller has directed each Purchaser to deposit its pro rata portion of the applicable Purchase Price.

"Group Outstanding Purchased Amount" means, as of any date of determination, an amount equal to the aggregate Individual Outstanding Purchased Amount of all Purchasers as of such date.

"GAAP" means the generally accepted accounting principles in the United States as in effect from time to time.

"Governmental Authority" means the United States, any State, any political subdivision of a State and any court, governmental agency, authority, instrumentality or body of the United States or any State or political subdivision thereof and any entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government.

"Immediate Family Members" shall mean, with respect to any individual, such individual's child, stepchild, grandchild or more remote descendant, parent, stepparent, grandparent, spouse, former spouse, qualified domestic partner, sibling, mother-in-law, father-in-law, son-in-law and daughter-in-law (including adoptive relationships), the estates of such individual and such other individuals above and any trust, partnership or other bona fide estate-planning vehicle the only beneficiaries of which are any of the foregoing individuals or any private foundation or fund that is controlled by any of the foregoing individuals or any donor-advised fund of which any such individual is the donor.

"Indemnified Party" has the meaning set forth in Section 13(b).

"Individual Outstanding Purchased Amount" means, with respect to any Purchaser as of any date of determination, the aggregate amounts of Purchase Price paid by such Purchaser (including, for the avoidance of doubt, any portion of the Purchase Price paid utilizing the

[MoneyLion] Master Receivables Purchase Agreement

Recycled Collections) on or prior to such date on account of all Purchases *less* any Recycled Collections applied toward the Purchase Price as described in Section 4(c) and *less* all Advance Collections (other than Yield Collections) actually received by such Purchaser hereunder from time to time pursuant to clause 3. of Section 2(a)(i) of the Servicing Agreement; provided that at no time shall the Individual Outstanding Purchased Amount be less than \$0.00.

"Insolvency Event" means, with respect to any Person, (a) the filing of a decree or order for relief by a court having jurisdiction in the premises with respect to such Person or any substantial part of its assets or property in an involuntary case under any applicable Insolvency Law, or appointing a receiver, liquidator, assignee, custodian, trustee, sequestrator or similar official for such Person or for any substantial part of its assets or property, or ordering the winding-up or liquidation of such Person's affairs, and such decree or order shall remain unstayed and in effect for a period of thirty (30) days, (b) the commencement by such Person of a voluntary case under any applicable Insolvency Law, (c) the consent by such Person to the entry of an order for relief in an involuntary case under any Insolvency Law, (d) the consent by such Person or for any substantial part of its assets or property, (e) the making by such Person of any general assignment for the benefit of creditors, (f) the admission by such Person in a legal proceeding of the inability of such Person to pay its debts generally as they become due, (g) the inability by such Person generally to pay its debts as they become due, or (h) the taking of action by such Person in furtherance of any of the foregoing.

"Insolvency Law" means in respect of any Person, any bankruptcy, reorganization, moratorium, delinquency, arrangement, insolvency, readjustment of debt, dissolution or liquidation law of any jurisdiction in effect at any time during the term of this Agreement applicable to such Person.

"Instacash" means a transaction in which a MoneyLion Accountholder (i) receives an advance from the Seller; (ii) maintains a RoarMoney Account (as defined in the MoneyLion Instacash Terms and Conditions) with the Seller or has linked an external eligible bank account; and (iii) has authorized the Seller to debit such RoarMoney Account (as defined in the MoneyLion Instacash Terms and Conditions) or external eligible bank account and associated debit card, as applicable, for the amount of repayment on the scheduled repayment date, all in accordance with the MoneyLion Instacash Terms and Conditions.

"Instacash Advance Amount" means, with respect to each Instacash, the amount of cash that the MoneyLion Accountholder receives in the form of an advance from the Seller in accordance with the MoneyLion Instacash Terms and Conditions.

"Instacash Application Time" means, with respect to each Instacash, the time on which the related MoneyLion Accountholder applies for the related Instacash.

"Instacash Disbursement Date" means, with respect to each Instacash, the date on which the Seller disburses such Instacash.

"Instacash Repayment Amount" means, with respect to each Instacash, the total dollar amount expected to be received from the applicable MoneyLion Accountholder in connection with such Instacash, including, without limitation, repayment of the Instacash Advance Amount and Excluded Amounts made in connection with such Instacash.

"Instacash Repayment Date" means, with respect to each Instacash, the date on which the MoneyLion Accountholder is expected to remit the Instacash Repayment Amount. It is understood and agreed that other than Qualified Deferrals as such term is defined in the Servicing Agreement, the Seller may not permit MoneyLion Accountholders to defer their respective Instacash Repayment Dates without the Purchaser Agent's consent.

"Joinder Agreement" means a Joinder Agreement in substantially the form of Exhibit D hereto.

"Lien" means, with respect to any property or asset of any Person, any mortgage, lien, pledge, charge or other security interest or encumbrance of any kind in respect of such property or asset.

"Lite Tier" means an underwriting classification applied to a MoneyLion Accountholder without a RoarMoney Account (as defined in the MoneyLion Instacash Terms and Conditions).

"Material Adverse Effect" means, with respect to any event or circumstance and any Person, a material adverse effect on: (i) the business or financial condition of such Person and its consolidated Subsidiaries, if any, taken as a whole; (ii) the ability of such Person to perform its material obligations under the Transaction Documents; or (iii) the validity or enforceability of any Transaction Document to which such Person is a party.

"Material Underperformance Event" means the occurrence of any of the following:

(a) For each Monthly Vintage Pool, as of any Measurement Date, (i) the aggregate Instacash Advance Amount that has been deferred of all MoneyLion Accountholders who have deferred their Instacash Repayment Date more than once on any particular Instacash Advance Amount divided by (ii) the aggregate Instacash Advance Amount for such Monthly Vintage Pool, exceeds 10%; or

(b) For each Monthly Vintage Pool, as of any Measurement Date, (i) the aggregate Instacash Advance Amount that has been deferred of all MoneyLion Accountholders who have deferred their Instacash Repayment Date for the first time divided by (ii) the aggregate Instacash Advance Amount for such Monthly Vintage Pool, exceeds 3%.

"Measurement Date" means the last day of each calendar month, commencing with July 31, 2024.

"MoneyLion Instacash Terms and Conditions" means "MoneyLion InstacashSM Terms and Conditions" posted on Parent's website and agreed to by a MoneyLion Accountholder.

8

"MoneyLion Accountholder" means a Person who created and maintains an account with the Seller in accordance with the MoneyLion Instacash Terms and Conditions.

"Monthly Period" means the period from and including the first day of a calendar month to and including the last day of such calendar month, provided, however, that the first Monthly Period hereunder will commence on the Closing Date and end on the last day of the calendar month in which the Closing Date occurs.

"Monthly Vintage Pool" means each pool of Receivables advanced by the Seller and sold to the Purchasers here under (and not repurchased pursuant to the terms hereof) whose Instacash Disbursement Date occurs during the same calendar month.

"Multiemployer Pension Plan" means a multiemployer plan, as defined in Section 4001(a)(3) of ERISA, to which the Seller or any other member of the Controlled Group contributes or has any liability, whether direct or indirect or absolute or contingent.

"**Optional Repurchased Receivable**" means any Receivable repurchased by the Seller, solely at the Seller's option, pursuant to Section 10(a) or Section 10(b) of this Agreement.

"Parent" means MoneyLion Inc., a Delaware corporation.

"**Parties**" and "**Party**" means the Seller, Purchaser Agent and the Purchasers party to this Agreement from time to time, which shall include, for the avoidance of doubt, the Initial Purchaser and each Additional Purchaser that may from time to time become party hereto by execution of a Joinder Agreement pursuant to <u>Section 14</u> hereof.

"PBGC" means the Pension Benefit Guaranty Corporation and any entity succeeding to any or all of its functions under ERISA.

"Pension Plan" means a "pension plan," as that term is defined in Section 3(2) of ERISA, which is subject to Title IV of ERISA or the minimum funding standards of Section 302 of ERISA (other than a Multiemployer Pension Plan), and as to which the Seller, any Subsidiary of the Seller or any member of the Seller's Controlled Group (or (x) in the case of the Purchaser Agent, as to which the Purchaser Agent, any Subsidiary of Purchaser Agent or any member of Purchaser Agent's Controlled Group and (y) in the case of a Purchaser, as to which such Purchaser, any Subsidiary of such Purchaser or any member of such Purchaser's Controlled Group), has any liability, whether direct or indirect or absolute or contingent, including any liability by reason of having been a substantial employer within the meaning of Section 4063 of ERISA at any time during the preceding five years, or by reason of being deemed to be a contributing sponsor under Section 4069 of ERISA.

"**Permitted Liens**" means (i) Liens for taxes, assessments and governmental charges not yet due or the payment of which is being contested in good faith and by appropriate proceedings and for which adequate reserves are maintained in accordance with GAAP, and in the aggregate securing taxes, assessments and governmental charges of \$10,000 or less at any one time, (ii) Liens arising by operation of law or agreement in favor of landlords, warehousemen, carriers,

9

mechanics or materialmen, custodians or depository banks incurred in the ordinary course of business and not in connection with the borrowing of money that are either not yet due or being contested in good faith and by appropriate proceedings, (iii) any right of set-off granted in favor of any depository institution in respect of deposit accounts opened and maintained pursuant to the requirements of this Agreement, (iv) Liens in favor of the Purchaser Agent and (v) Liens existing on any Purchased Receivables and Related Rights so long as any such Liens are released automatically and immediately upon the sale of any such Purchased Receivables and Related Rights to the Purchasers in accordance with the terms of this Agreement.

"**Permitted Holder**" means (a) each of Rohit D'Souza and Diwakar Choubey, (b) any Immediate Family Member of any such individual referred to in clause (a), and (c) any of Edison Partners Management LLC, FinTech Collective Management LLC and their respective Affiliates.

"Person" means any natural person, corporation, business trust, joint venture, association, limited liability company, partnership, joint stock company, corporation, trust, unincorporated organization or Governmental Authority.

"**Performance Guaranty**" means that certain Performance Guaranty, dated as of the Closing Date, by the Servicer in favor of the Purchasers, as amended, supplemented or otherwise modified from time to time.

"Purchase Date" means each date a sale of Receivables is consummated hereunder; provided, that in any weekly period, there shall not be more than two Purchase Dates.

"Purchase Price" has the meaning set forth in Section 4(a).

"**Purchased Receivables**" means, as of any date of determination, each Receivable purchased by the Purchasers hereunder (and not repurchased pursuant to the terms hereof); provided, that upon any repurchase of a Purchased Receivable pursuant to the terms hereof, such Receivable shall cease to be a Purchased Receivable.

"Purchaser" has the meaning set forth in the preamble to this Agreement, and includes such Person's permitted successors and assigns.

"Purchaser Indemnified Party" has the meaning set forth in Section 13(a).

"**Receivable**" means with respect to each Instacash made by the Seller in accordance with the MoneyLion Instacash Terms and Conditions, 100% of the right to receive payments from the related MoneyLion Accountholder, including the repayment of the Instacash Advance Amount and excluding any Excluded Amounts in connection with such Instacash.

"Recycled Collections" has the meaning set forth in Section 4(c).

"**Regulatory Trigger Event**" means (a) the commencement by any Governmental Authority in a jurisdiction of any formal inquiry or investigation (which for the avoidance of doubt excludes any routine inquiry or investigation), legal action or proceeding, against the Seller, the

[MoneyLion] Master Receivables Purchase Agreement

Servicer, any Subsidiary of the Servicer or any third party then engaged as a sub-servicer, challenging such Person's authority to advance, hold, own, service, pledge or enforce any Purchased Receivable with respect to the residents of such jurisdiction, or otherwise alleging any noncompliance by such party with such jurisdiction's Applicable Laws related to advancing, holding, pledging, servicing or enforcing such Purchased Receivable or otherwise related to such Purchased Receivable, which inquiry, investigation, legal action or proceeding (i) is not released or terminated in a manner acceptable to the Purchaser Agent in its reasonable good faith discretion within one hundred and twenty (120) calendar days of the earlier of Seller or the Servicer's knowledge or receipt of written notice thereof and (ii) could reasonably be expected to have a Material Adverse Effect, as determined by the Purchaser Agent in its reasonable good faith discretion, or (b) the issuance or entering of any stay, order, judgment, cease and desist order, injunction, temporary restraining order, or other judicial or non-judicial sanction, order or ruling against the Seller, the Servicer, any Subsidiary of the Servicer or any third party then engaged as a sub-servicer related in any way to the advancing, holding, pledging, servicing or enforcing of any Purchased Receivable or rendering this Agreement unenforceable in such jurisdiction, the effect of which could reasonably be expected to have a Material Adverse Effect, as determined by the Purchaser Agent in its reasonable good faith discretion: provided, in each case, that upon the favorable resolution of such inquiry, investigation, action or proceeding (whether by judgment, withdrawal of such action or proceeding or settlement of such action or proceeding), as determined by the Purchaser Agent in its reasonable good faith discretion and confirmed by written notice from the Purchaser Agent, such Regulatory Trigger Event for such jurisdiction shall cease to exist immediately upon such determination by the Purchaser Agent. It is understood and agreed that the jurisdiction of a Regulatory Trigger Event is the entire United States if the applicable Governmental Authority is a federal authority. For the avoidance of doubt, a Regulatory Trigger Event shall only affect those Purchased Receivables with respect to which the applicable Governmental Authority or court has jurisdiction.

"**Related Rights**" means, with respect to any Receivable, (a) all payment rights relating thereto under the MoneyLion Instacash Terms and Conditions and all indemnities, warranties, insurance (and proceeds and premium refunds thereof) or other agreements or arrangements of any kind from time to time supporting payment of such Receivable, whether pursuant to the MoneyLion Instacash Terms and Conditions or otherwise, (b) all Advance Collections in respect of such Receivable, and (c) all products and proceeds of any of the foregoing items (a) through (b). Notwithstanding the foregoing, Related Rights shall not include any Excluded Amounts.

"**Reportable Event**" means a reportable event as defined in Section 4043 of ERISA and the regulations issued thereunder as to which the PBGC has not waived the notification requirement of Section 4043(a), or the failure of a Pension Plan to meet the minimum funding standards of Section 412 of the Code or under Section 302 of ERISA.

"Repurchase Confirmation" means a Confirmation (Repurchase) substantially in the form of <u>Exhibit C</u> hereto relating to the repurchase by the Repurchase Purchase repurchase from time to time of a Defective Receivable or an Underperforming Receivable pursuant to Section 7 or an Optional Repurchased Receivable pursuant to Section 10.

"**Repurchase Price**" means, as of any date of determination, with respect to the applicable Receivable that the Repurchase Purchaser is repurchasing from the applicable Repurchase Seller, an amount equal to (x) the Purchase Price previously paid for such Receivable plus (y) solely in the event that (A) the aggregate Instacash Advance Amount of any Monthly Vintage Pool in which such Receivable is advanced exceeds \$2,500,000 and (B) any such Receivable's Instacash Disbursement Date is more than five (5) calendar days prior to the related Repurchase Date, the amount of income the Repurchase Seller expected to earn to from such Receivable from the related Purchase Date (as mutually determined by the Repurchase Seller and the Seller) to the related Repurchase Date less (z) all Advance Collections received by the applicable Repurchase Seller with respect to such Receivable on or prior to the related Repurchase Date.

"Repurchase Purchaser" has the meaning set forth in Section 7.

"Repurchase Seller" has the meaning set forth in Section 7.

"Requirements of Law" means, with respect to any Person or any of its property, the certificate of incorporation or articles of association and by-laws or other organizational or governing documents of such Person, and any law, treaty, rule or regulation, or determination of any arbitrator or Governmental Authority, in each case applicable to or binding upon such Person or any of its property or to which such Person or any of its property is subject, whether federal, state or local (including, without limitation, usury laws and retail installment sales acts).

"Scheduled Termination Date" means the date that is the earliest to occur of (i) the two (2) year anniversary of the Closing Date, which date may be extended as provided in Section 11, (ii) the date on which the "Scheduled Termination Date" is declared and deemed to have occurred pursuant to Section 10 or (iii) this Agreement is terminated pursuant to Section 11.

"SEC" means the Securities and Exchange Commission or any other governmental authority succeeding to any of the principal functions thereof.

"Seller" has the meaning set forth in the preamble to this Agreement.

"Seller Indemnified Party" has the meaning set forth in Section 10(b).

"Servicer" means MoneyLion Technologies Inc., a Delaware corporation, in its capacity as Servicer under the Servicing Agreement.

"Servicer Event of Default" has the meaning set forth in the Servicing Agreement.

"Servicing Agreement" means that certain Servicing Agreement, dated as of the Closing Date, among the Initial Purchaser, Purchaser Agent, Servicer and any Additional Purchasers from time to time party thereto and hereto, as amended, supplemented or otherwise modified from time to time.

"Solvent" means, with respect to any Person, (a) such Person will own assets for which the sum of the value of the assets, both at fair market valuation and at fair valuations, exceeds the

[MoneyLion] Master Receivables Purchase Agreement

sum of such Person's debts and other obligations, both at fair market valuation and at fair valuations; (b) such Person is and shall be able to pay its debts as such debts become due; (c) such Person does not intend to, does not believe, and should not have reasonably believed, that it would incur debts beyond its ability to pay those debts as they become due, or that the debts are beyond its ability to pay as the debts mature; and (d) such Person has assets and property that are sufficient and adequate, and not unreasonable small, for the business(es) or transaction(s) in which such Person is engaged or is about to be engaged.

"Stock" means all shares of capital stock (whether denominated as common stock or preferred stock), equity interests, beneficial, partnership or membership interests, joint venture interests, participations or other ownership or profit interests in or equivalents (regardless of how designated) of or in a Person (other than an individual), whether voting or non-voting.

"Subsidiary" means, with respect to any Person, any corporation, partnership, limited liability company, association, joint venture or other business entity of which more than 50% of the total voting power of shares of stock or other ownership interests entitled (without regard to the occurrence of any contingency) to vote in the election of the Person or Persons (whether directors, managers, trustees or other Persons performing similar functions) having the power to direct or cause the direction of the management and policies thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof; provided, in determining the percentage of ownership interests of any Person controlled by another Person, no ownership interest in the nature of a "qualifying share" of the former Person shall be deemed to be outstanding.

"**Taxes**" means any and all present and future taxes, duties, levies, imposts, deductions, assessments, charges, similar fees or withholdings (including backup withholdings) imposed under applicable law and/or by any governmental authority that are in the nature of a tax, and any and all liabilities (including interest and penalties and other additions to taxes) with respect to any of the foregoing.

"Term" means the period of time from and including the Closing Date through the Scheduled Termination Date (or as otherwise extended by the Parties).

"Termination Event" means, with respect to a Pension Plan that is subject to Title IV of ERISA, the following: (a) a Reportable Event; (b) the withdrawal of the Seller or any other member of the Controlled Group from that Pension Plan during a plan year in which that the Seller or other member of the Controlled Group was a "substantial employer" as defined in Section 4001(a)(2) of ERISA; (c) the termination of that Pension Plan, the filing of a notice of intent to terminate the Pension Plan or the treatment of an amendment of that Pension Plan as a termination under Section 4041 of ERISA; (d) the institution by the PBGC of proceedings to terminate that Pension Plan; or (e) any event or condition that could reasonably constitute grounds under Section 4042 of ERISA for the termination of, or appointment of a trustee to administer, that Pension Plan.

"Total Assets" means, as of any date with respect to any Person, all assets of such Person and its consolidated Subsidiaries that in accordance with GAAP would be classified as assets upon a balance sheet of such Person prepared as of such date.

[MoneyLion] Master Receivables Purchase Agreement

"Total Book Value" means, as of any date with respect to any Person, Total Assets minus Total Liabilities.

"Total Liabilities" means, as of any date with respect to any Person, all liabilities of such Person and its consolidated Subsidiaries that in accordance with GAAP would be classified as liabilities upon a balance sheet of such Person prepared as of such date.

"Total Plan Liability" means, at any time, the present value of all vested and unvested accrued benefits under all Pension Plans, determined as of the then most recent valuation date for each Pension Plan, using PBGC actuarial assumptions for single-employer plan terminations.

"Transaction Document" means this Agreement, the Servicing Agreement, the Account Control Agreement, the Performance Guaranty, each Joinder Agreement, and the Backup Servicing Agreement.

"Treasury Regulations" means such provisions of the income tax regulations of the United States Department of the Treasury or any successor provisions promulgated under the Code.

"Turbo Fee" means a service fee charged by the Seller upon a MoneyLion Accountholder's request to expedite disbursement of Instacash.

"UCC" means the Uniform Commercial Code as in effect from time to time in the applicable jurisdiction.

"Unfunded Liability" means the amount (if any) by which the present value of all vested and unvested accrued benefits under all Pension Plans exceeds the fair market value of all assets allocable to those benefits, all determined as of the then most recent valuation date for each Pension Plan, using PBGC actuarial assumptions for single employer plan terminations.

"Unpaid Rate (30 Days)" means, as of any date of determination with respect to any Monthly Vintage Pool, a percentage equal to (i) the aggregate Instacash Advance Amount of Receivables in such Monthly Vintage Pool that has not been repaid within thirty (30) calendar days of the Instacash Repayment Date relating to each such Receivable, divided by (ii) the aggregate Instacash Advance Amount for such Monthly Vintage Pool.

"Unpaid Rate (35 Days)" means, as of any date of determination with respect to any Monthly Vintage Pool, a percentage equal to (i) the aggregate Instacash Advance Amount of Receivables in such Monthly Vintage Pool that has not been repaid within thirty (35) calendar days of the Instacash Repayment Date relating to each such Receivable, divided by (ii) the aggregate Instacash Advance Amount for such Monthly Vintage Pool.

"Unpaid Rate (40 Days)" means, as of any date of determination with respect to any Monthly Vintage Pool, a percentage equal to (i) the aggregate Instacash Advance Amount of Receivables in such Monthly Vintage Pool that has not been repaid within thirty (40) calendar

[MoneyLion] Master Receivables Purchase Agreement

days of the Instacash Repayment Date relating to each such Receivable, divided by (ii) the aggregate Instacash Advance Amount for such Monthly Vintage Pool.

"Unpaid Rate (60 Days)" means, as of any date of determination with respect to any Monthly Vintage Pool, a percentage equal to (i) the aggregate Instacash Advance Amount of Receivables in such Monthly Vintage Pool that has not been repaid within sixty (60) calendar days of the Instacash Repayment Date relating to each such Receivable, divided by (ii) the aggregate Instacash Advance Amount for such Monthly Vintage Pool.

"Unpaid Rate (65 Days)" means, as of any date of determination with respect to any Monthly Vintage Pool, a percentage equal to (i) the aggregate Instacash Advance Amount of Receivables in such Monthly Vintage Pool that has not been repaid within sixty (65) calendar days of the Instacash Repayment Date relating to each such Receivable, divided by (ii) the aggregate Instacash Advance Amount for such Monthly Vintage Pool.

"Unpaid Rate (70 Days)" means, as of any date of determination with respect to any Monthly Vintage Pool, a percentage equal to (i) the aggregate Instacash Advance Amount of Receivables in such Monthly Vintage Pool that has not been repaid within sixty (70) calendar days of the Instacash Repayment Date relating to each such Receivable, divided by (ii) the aggregate Instacash Advance Amount for such Monthly Vintage Pool.

"Unused Portion of the Facility Limit" means, at any time of determination, an amount equal to (x) the Facility Limit at such time, minus (y) the Group Outstanding Purchased Amount at such time.

"Yield Collections" means, with respect to any Receivable, all Collections thereon in excess of, and that are not classified as a repayment of, the Purchase Price of such Receivable.

Section 2. Sale of Eligible Receivables; Related Covenants.

(a) Subject to the terms and conditions of this Agreement and in reliance upon the representations, warranties and agreements set forth herein, each Purchaser shall, on a committed basis, from time to time purchase from the Seller and make cash payments to Seller of the related Purchase Price in respect of the Eligible Receivables and all Related Rights pertaining thereto (each such cash payment, a "**Purchase**"), and the Seller shall take commercially reasonable measures to offer from time to time for sale to the Purchasers, in each case against delivery of the applicable Purchase Price, Eligible Receivables and all Related Rights pertaining thereto with an anticipated minimum sale volume as described on <u>Schedule II</u> hereto. Notwithstanding the foregoing, under no circumstances shall any Purchaser be obligated to purchase from the Seller if, after giving effect to such Purchase:

- (i) The aggregate Group Outstanding Purchased Amount would exceed the Facility Limit; or
- (ii) the aggregate Individual Outstanding Purchased Amount of such Purchaser would exceed its Commitment.

15

(b) The Eligible Receivables and Related Rights sold on each Purchase Date shall be as described in the applicable Eligible Receivables Schedule as further described in <u>Section 3</u> and shall be sold, transferred, assigned, set over and otherwise conveyed to the applicable Purchaser, absolutely and not as collateral security, pursuant to the applicable Confirmation pursuant to <u>Section 3(b)</u> below.

(c) Although the Parties hereto intend that any sale of Eligible Receivables hereunder shall constitute an absolute assignment and transfer of ownership of all of the Seller's right, title and interest in and to the Eligible Receivables sold, and not a loan, if any such assignment is deemed to be a loan, the Parties hereto intend that the rights and obligations of the Parties hereto to such loan shall be established pursuant to the terms of this Agreement. The Parties also intend and agree that if and to the extent that the transfer of Purchased Receivables and Related Rights is for any purpose recharacterized by a court or other Governmental Authority as a collateral transfer for security or the transactions contemplated by this Agreement are characterized as financing transactions or loans, the Seller shall be deemed to have granted to the Purchaser Agent (on behalf of the Purchasers), and the Seller's right, title and interest in, to and under such Purchased Receivables and Related Rights and that this Agreement shall constitute a security agreement under applicable law and the Purchasers will have all the rights and remedies of a secured party under the UCC and other applicable law. The Seller hereby consents to the filing by the Purchaser Agent or its assignee of the UCC-1 financing statement set forth as **Exhibit B** hereto in connection with the granting of security interest pursuant to this Section 2(c).

(d) The Seller shall not take any action inconsistent with the Purchaser's ownership in Purchased Receivables and shall not claim any ownership interest in any such Purchased Receivables (in each case other than with respect to any such Defective Receivables, Underperforming Receivables and Optional Repurchased Receivables repurchased by the Repurchase Purchaser pursuant to this Agreement).

(e) The Seller shall not organize under the law of any jurisdiction other than the state under which it is organized as of the date hereof (whether changing its jurisdiction of organization or organizing under an additional jurisdiction) without giving at least ten (10) days prior written notice of such action to the Purchaser Agent. Before effecting such change, the Seller shall prepare and file in the appropriate filing office any financing statements or other statements necessary to continue the perfection of the Purchaser Agent's interests in the Purchased Receivables.

(f) At any time, and from time to time hereafter, upon the reasonable request of the Purchaser Agent, and without payment of further consideration to the Seller, the Seller will do, execute, acknowledge and deliver, and will cause to be done, executed, acknowledged and delivered, all such further acts, deeds, assignments, transfers, conveyances, powers of attorney, recordings and assurances as may be reasonably required in order to better assign, transfer, grant, convey, assure and confirm to the applicable Purchasers ownership of the Assets sold hereunder (if any) (other than any such Defective

Receivables, Underperforming Receivables and Optional Repurchased Receivables repurchased by the Repurchase Purchaser in accordance with this Agreement).

(g) The Purchaser Agent and the Purchasers shall not notify any MoneyLion Accountholder of any sale, assignment, set-over, transfer and conveyance to the Purchaser(s) hereunder of any Receivable.

Section 3. Sale Procedures; Seller Delivery Obligations.

(a) On each Purchase Date, the Seller shall deliver to the Purchaser Agent, on behalf of the Purchasers, a Confirmation with respect to those Eligible Receivables offered by the Seller for sale to the Purchasers on such date. Such Confirmation shall (i) be accompanied by an Eligible Receivables Schedule; (ii) include the aggregate Purchase Price for such Eligible Receivables to be paid on such proposed Purchase Date; and (iii) not specify any Purchaser, it being understood that Purchaser Agent shall specify the applicable Purchaser party to such Confirmation.

(b) One or more Purchasers shall purchase such Eligible Receivables on any Purchase Date in accordance with <u>Section 2</u> by remitting the aggregate Purchase Price listed on the applicable Confirmation on the Purchase Date set forth therein, and such payment shall for all purposes be deemed an acceptance of the offer of sale of the Receivable described therein (and the Related Rights), and upon such acceptance the Confirmation shall serve for all purposes as a legal, binding and enforceable bill of sale and assignment for each Receivable described therein (and the Related Rights).

(c) In connection with each sale of Eligible Receivables hereunder, the Seller shall, at its own expense, on and after such Purchase Date, other than with respect to (x) any Defective Receivables and Underperforming Receivables repurchased by the Repurchase Purchaser after such Purchase Date pursuant to Section 7 below and (y) any Optional Repurchased Receivables repurchased by the Repurchase Purchaser after such Purchase Date pursuant to Section 10 below, indicate in its computer files that the Eligible Receivables transferred on such Purchase Date have been sold to a Purchaser (and shall specify such Purchaser) pursuant to this Agreement. Notwithstanding the foregoing, and without prejudice to the absolute sale (if any) intended hereby, the Seller may retain data relating to each Asset sold hereunder to be used by the Seller or its Affiliates for any lawful purpose.

(d) Notwithstanding anything in this Agreement to the contrary, under no circumstances shall any Purchaser be obligated to purchase from the Seller until:

(i) the earlier to occur of (x) July 19, 2024 or (y) the date on which Initial Purchaser has entered into a credit facility that provides debt financing to purchase the Eligible Receivables on terms and conditions satisfactory to the Purchaser Agent in its sole discretion (for the avoidance of doubt, if the Initial Purchaser does not enter into a credit facility that provides debt financing to purchase the Eligible Receivables on terms and conditions satisfactory to the

[MoneyLion] Master Receivables Purchase Agreement

Purchaser Agent in its sole discretion by July 19, 2024, each Purchaser's obligation to purchase Receivables from the Seller shall no longer be subject to the satisfaction of the condition set forth in this clause (y)); and

(ii) the Purchaser Agent shall have received a duly executed Account Intercreditor Agreement among the Purchaser Agent, Monroe Capital Management Advisors, LLC and the Servicer along with executed consents and releases with to the Collection Account and the transactions contemplated hereunder.

Section 4. Purchase Price; Payment; Unused Fee.

(a) From the Closing Date until the end of the first Monthly Period to occur after the Closing Date, the "**Purchase Price**" to be paid by a Purchaser in respect of Eligible Receivables on any Purchase Date shall be the Initial Purchase Price, which shall be calculated in accordance with the definition of such term set forth on Schedule III attached hereto.

(b) After the end of the first Monthly Period to occur after the Closing Date, the "**Purchase Price**" to be paid by a Purchaser in respect of Eligible Receivables on any Purchase Date shall be the Adjusted Purchase Price, which shall be calculated in accordance with the definition of such term set forth on Schedule III attached hereto.

(c) On each Purchase Date, all payments of the Purchase Price shall be made by wire transfer (or as otherwise mutually agreed by the Seller and the applicable Purchaser), in immediately available funds, to the Funding Account or an account designated by the Seller to the Purchaser Agent in writing. Notwithstanding the foregoing, if on any Purchase Date there are Advance Collections in the Collection Account, the Purchasers may utilize such Advance Collections (any such Advance Collections so utilized, "**Recycled Collections**") to pay all or a portion, as applicable, of the Purchase Price on such Purchase Date and on such Purchase Date the applicable Purchaser shall only be required to wire to the Funding Account any Purchase Price in excess of the Recycled Collections so applied. In the event any Purchaser utilizes the Recycled Collections to pay all or a portion of the Purchase Price, on the applicable Purchase Date, the Purchaser Agent shall direct the Servicer to transfer the Recycled Collections to the Funding Account.

(d) The Seller agrees to pay to the Purchasers, through the Purchaser Agent, a non-refundable unused fee (the "Unused Fee"), payable on a monthly basis, on the date that is five (5) Business Days after the end of each calendar month, commencing with July 8, 2024, in an amount equal to two percent (2%) per annum times the daily average Unused Portion of the Facility Limit, which shall be calculated on the basis of a 360-day year for the actual number of days elapsed during each calendar month; provided, that, the aggregate amount of the Unused Fees payable to the Purchasers during the Term shall in no event exceed \$1,750,000.

Section 5. Seller's Representations and Warranties.

(a) The Seller makes, as of the date hereof and as of each Purchase Date, each of the representations and warranties set forth in <u>Annex B</u> hereto.

(b) The representations and warranties set forth in this <u>Section 5</u> shall survive the date hereof and the sale of Eligible Receivables to a Purchaser on each Purchase Date.

Section 6. Purchasers' Representations and Warranties.

(a) The Purchaser Agent and the Purchasers make, as of the date hereof and as of each Purchase Date, each of the representations and warranties set forth in <u>Annex C</u> hereto.

(b) The representations and warranties set forth in this <u>Section 6</u> shall survive the date hereof and the purchase of Eligible Receivables by a Purchaser on each Purchase Date.

Section 7. Repurchase Rights and Obligations.

(a) Upon discovery by the Seller or Servicer, or if the Seller or Servicer receives notice and related documentation from any of the other Parties to the Transaction Documents, that any Receivable sold to a Purchaser hereunder which was represented to be an Eligible Receivable, was not an Eligible Receivable as of the related Purchase Date (each such Receivable, a "Defective Receivable"), the Seller (in its capacity as a purchaser of a Receivable pursuant to this Section 7 or Section 10, the "Repurchase Purchaser") shall have four (4) Business Days to cure such breach (the date on which the Repurchase Purchaser pays the applicable Repurchase Seller the Repurchase Price, the "Repurchase Date") by repurchasing from the applicable Purchaser (in its capacity as a seller of a Receivable pursuant to this Section 7 or Section 10, a "Repurchase Seller") such Defective Receivable and all Related Rights at the Repurchase Price.

(b) Upon the occurrence of a Material Underperformance Event, the Repurchase Purchaser may (but shall not be obligated to) cure such Material Underperformance Event within ten (10) Business Days from the occurrence of such Material Underperformance Event by repurchasing from the applicable Repurchase Seller at the Repurchase Price certain of the Receivables (as determined by the Repurchase Purchaser) whose related MoneyLion Accountholders have deferred their Instacash Repayment Date (each such Receivable repurchased by the Seller, solely at the Seller's option, in connection with the occurrence of a Material Underperformance Event, an "Underperforming Receivable") until the applicable Material Underperformance Event no longer exists.

(c) Upon the repurchase by the Repurchase Purchaser of a Defective Receivable pursuant to Section 7(a), an Underperforming Receivable pursuant to or

[MoneyLion] Master Receivables Purchase Agreement

Section 7(b) and an Optional Repurchased Receivable pursuant to Section 10, the applicable Repurchase Seller does hereby:

(i) sell, transfer, assign, and otherwise convey to the Repurchase Purchaser all right, title and interest of such Repurchase Seller in and to the applicable Defective Receivable, the applicable Underperforming Receivable, or the applicable Optional Repurchased Receivable and Related Rights thereto;

(ii) release (or cause the release of) any Liens in respect of such Defective Receivable, such Underperforming Receivable or such Optional Repurchased Receivable and represent and warrant to the Repurchase Purchaser that such Repurchase Seller has full and marketable title to such Defective Receivable, such Underperforming Receivable or such Optional Repurchased Receivable, free and clear of all Liens; and

(iii) represent and warrant to the Repurchase Purchaser that such Repurchase Seller has the full power and authority to so convey such rights, titles, and interests to the Repurchase Purchaser.

(d) The representations and warranties set forth in Section 7(c)(ii) and Section 7(c)(iii) shall survive the date of repurchase and the termination of this Agreement.

(e) Upon any repurchase of a Defective Receivable, an Underperforming Receivable or an Optional Repurchased Receivable, the applicable Repurchase Seller shall execute any documents, instruments, or assignments necessary to effect the transfer and assignment of all of such Repurchase Seller's right, title and interest in and to such Defective Receivable, such Underperforming Receivable or such Optional Repurchased Receivable to the Repurchase Purchaser, including, without limitation, a Repurchase Confirmation, and shall release or terminate (or caused to be released or terminated) any Liens then covering such Defective Receivable, such Underperforming Receivable or such Optional Repurchased Receivable. Any reasonable and documented out-of-pocket expenses incurred by the applicable Repurchase Seller in connection with any such transfer and assignment, release or termination shall be paid for by the Repurchase Purchase Purchase Purchase Seller.

(f) The Repurchase Purchaser and each Repurchase Seller agree to act in good faith in connection with any repurchase, or requested repurchase, under this <u>Section 7</u> or <u>Section 10</u>. In case of any dispute regarding a requested repurchase, each of the Repurchase Purchaser and the applicable Repurchase Seller agrees to cooperate in good faith in attempting to resolve such dispute.

Section 8. Conditions to Purchase and Sale.

(a) The Seller's obligation to sell and deliver the applicable Eligible Receivables hereunder as of any Purchase Date shall be subject to each of the following conditions precedent:

(i) all representations, warranties and statements by or on behalf of the Purchasers contained in this Agreement shall be true and correct in all material respects on the applicable Purchase Date (or such other date as is specifically set forth therein); and

(ii) the applicable Purchase Price shall have been paid as described herein.

(b) The Purchasers' obligation to purchase the Eligible Receivables hereunder as of any Purchase Date shall be subject to each of the following conditions precedent:

(i) all representations, warranties and statements by or on behalf of the Seller contained in this Agreement shall be true and correct in all material respects on the applicable Purchase Date (or such other date as is specifically set forth therein);

(ii) Seller shall be in compliance in all material respects with the applicable covenants under the Transaction Documents;

(iii) such Purchaser's receipt of the applicable Confirmation (including the applicable Eligible Receivables Schedule);

(iv) no Servicer Event of Default shall have occurred and be continuing;

(v) the Purchaser Agent has received the Transaction Documents (other than the Backup Servicing Agreement pursuant to clause (vi) below), each dated the Closing Date or as of the Closing Date (other than such Transaction Documents that are executed in connection with any Joinder Agreement, which Transaction Documents shall be dated as of the date of such Joinder Agreement) duly executed and delivered by each party thereto and being in full force and effect;

(vi) with respect to any Purchase Date occurring after the date that is sixty (60) days (or such later date that the Purchaser Agent (on behalf of the Purchasers) may reasonably agree to) after the Closing Date, the Purchaser Agent shall have received a duly executed and delivered Backup Servicing Agreement;

(vii) no Event of Termination or Material Underperformance Event has occurred and is continuing; and

(viii) unless otherwise waived by the Purchasers, the Seller shall have been in compliance with the Concentration Limitations on a pro forma basis after giving effect to such Purchase.

Section 9. Covenants of the Seller.

At all times from the Closing Date until the Scheduled Termination Date.

[MoneyLion] Master Receivables Purchase Agreement

(a) Financial Reporting. The Seller shall furnish to the Purchaser Agent:

(i) within sixty (60) days after the end of the first three fiscal quarters of each fiscal year, commencing with the fiscal quarter ending June 30, 2024, unaudited consolidated balance sheets of the Servicer and its Subsidiaries as of the end of that fiscal quarter, together with consolidated statements of earnings and a consolidated statement of cash flows for that fiscal quarter and for the period beginning with the first day of that fiscal year and ending on the last day of that fiscal quarter, together with a comparison with the corresponding period of the previous fiscal year.

(ii) on or prior to the date that is the earlier of (x) one hundred twenty (120) days after the end of each fiscal year (commencing with the fiscal year ending on December 31, 2024) and (y) the date following the end of a fiscal year (commencing with the fiscal year ending December 31, 2024) by which Parent is required to file Form 10-K with the SEC, (i) the consolidated balance sheet of Servicer as at the end of such fiscal year and the related consolidated statements of operations, stockholders' equity and cash flows of Servicer for such fiscal year and setting forth, in reasonable detail, in comparative form the corresponding figures for the previous fiscal year and (ii) with respect to such consolidated financial statements, a report thereon of an independent certified public accountant of recognized national standing or another accounting firm reasonably acceptable to the Purchaser Agent (which report shall not be subject to a "going concern" or scope of audit qualification (except for any such qualification pertaining to, or disclosure of an exception or qualification resulting from, (x) the maturity (or impending maturity) of any Debt or (y) any breach or anticipated breach of any financial covenant in any agreement governing Debt of Parent or any Subsidiary), and shall state that such consolidated financial statements fairly present, in all material respects, the consolidated financial position of the Servicer as at the dates indicated and its income and cash flows for the periods indicated in conformity with GAAP in all material respects.

(iii) Notwithstanding the foregoing, the obligations in Section 9.1(a)(i) and Section 9.1(a)(i) may be satisfied with respect to financial information of the Servicer and its Subsidiaries by furnishing the Form 10-K or 10-Q of Parent filed with the SEC; provided that, (x) if (1) such financial statements relate to such other Person, such financial statements or the Form 10-K or Form 10-Q, as applicable, shall be accompanied by consolidating information (which need not be audited) that summarizes in reasonable detail the differences between the information relating to such Person, on the one hand, and the information relating to Servicer and its consolidated Subsidiaries on a standalone basis, on the other hand, which consolidating information shall be certified by an officer of Servicer as having been fairly presented in all material respects and (y) to the extent such statements are in lieu of statements required to be provided under Section 9.1(a)(ii), such statements shall be accompanied by an audit report that would satisfy the applicable

[MoneyLion] Master Receivables Purchase Agreement

requirements set forth in Section 9.1(a)(ii) as if the references to "Servicer" therein were references to such Person;

(b) Event of Termination; Regulatory Trigger Events. The Seller will deliver to the Purchaser Agent (on behalf of the Purchasers), promptly, and in any event within five (5) Business Days of any authorized officer of the Seller obtaining knowledge (i) of any condition or event that constitutes an Event of Termination or that after notice or lapse of time or both would constitute an Event of Termination or that notice has been given to the Seller or any of its subsidiaries with respect thereto; (ii) of the occurrence of any event or change that has caused or evidences, either in any case or in the aggregate, a materially adverse effect on the business, a material portion of the Purchased Receivables, properties or financial condition of the Seller to perform the services provided for herein or (iii) except for any Regulatory Trigger Event which has been disclosed in Parent's public filings with the SEC, the occurrence of any of the events described in the definition of Regulatory Trigger Event, whether or not Seller has determined that such events will be material, a certificate of such authorized officer specifying the nature and period of existence of such condition, event or change, or specifying the notice given and action taken by any such Person and the nature of such claimed Event of Termination, potential Regulatory Trigger Event, and what action the Seller or any of its Affiliates, as applicable, has taken, is taking and proposes to take with respect thereto.

(c) Monthly Compliance Reporting. Within three (3) Business Days following the last day of each calendar month commencing with the calendar month ending on July 31, 2024, the Seller shall deliver a Compliance Certificate signed by an officer of the Seller, stating that no Event of Termination has occurred during the most recently completed Monthly Period.

(d) Sales and Liens. Except as otherwise provided herein and Permitted Liens, the Seller shall not sell, assign or otherwise dispose of, or permit any Lien (except for Permitted Liens) upon or with respect to any Purchased Receivables and the Related Rights.

(e) Eligibility Guidelines. The Seller shall not, and shall not permit Servicer to, amend or otherwise modify the Eligibility Guidelines in any manner without the prior written consent of the Purchasers.

Section 10. Events of Termination.

If any of the following events (each an "Event of Termination") shall occur:

(a) As of the last day of any Monthly Period, with respect to any Monthly Vintage Pool, if the Unpaid Rate (30 Days) shall exceed 13.00% (the "Level One Unpaid Rate Repurchase Trigger" and any such Monthly Vintage Pool whose Unpaid Rate (30 Days) exceeds 13.00%, a "Level One Monthly Vintage Pool"); provided, that such occurrence shall not be deemed an occurrence of an Event of Termination under this

Section 10(a) if either (x) the Unpaid Rate (35 Days) of each such Level One Monthly Vintage Pool, measured as of the last day of such Monthly Period, does not exceed 13.00% and the Unpaid Rate (40 Days) of each such Level One Monthly Vintage Pool, measured as of the last day of such Monthly Period, does not exceed 13.00% and the Unpaid Rate (40 Days) of each such Level One Monthly Vintage Pool, measured as of the last day of such Monthly Period or as of, or prior to the date that is ten (10) calendar days following the last day of such Monthly Period, does not exceed 13.00% (such Level One Monthly Vintage Pool, a "**Cured Level One Monthly Vintage Pool**") or (y) in the event the Seller, at its sole option, within ten (10) Business Days after the last day of such Monthly Period repurchases at the Repurchase Price certain of the Receivables (as determined by the Seller) in such Monthly Vintage Pool that have not been repaid within thirty (30) calendar days of the Instacash Repayment Date relating to each such Receivable until the Unpaid Rate (30 Days) for the remaining Receivables in such Monthly Vintage Pool is equal to or not greater than the Level One Unpaid Rate Repurchase Trigger; further provided that it shall be an Event of Termination under this Section 10(a) if the Monthly Vintage Pool immediately succeeding a Cured Level One Monthly Vintage Pool is Level One Monthly Vintage Pool;

(b) As of the last day of any Monthly Period, with respect to any Monthly Vintage Pool, if the Unpaid Rate (60 Days) shall exceed 11.00% (the "Level Two Unpaid Rate Repurchase Trigger" and any such Monthly Vintage Pool whose Unpaid Rate (60 Days) exceeds 11.00%, a "Level Two Monthly Vintage Pool"); provided, that such occurrence shall not be deemed an occurrence of an Event of Termination under this Section 10(b) if either (x) the Unpaid Rate (65 Days) of each such Level Two Monthly Vintage Pool, measured as of the last day of such Monthly Period or as of, or prior to the date that is five (5) calendar days following the last day of such Monthly Period, does not exceed 11.00% and the Unpaid Rate (70 Days) of each such Level Two Monthly Vintage Pool, measured as of the last day of such Monthly Period or as of, or prior to the date that is ten (10) calendar days following the last day of such Monthly Period, does not exceed 11.00% (such Level Two Monthly Vintage Pool, a "Cured Level Two Monthly Vintage Pool") or (y) in the event the Seller, at its sole option, within ten (10) Business Days after the last day of such Monthly Period repurchases at the Repurchase Price certain of the Receivables (as determined by the Seller) in such Monthly Vintage Pool that have not been repaid within sixty (60) calendar days of the Instacash Repayment Date relating to each such Receivable until the Unpaid Rate (60 Days) for the remaining Receivables in such Monthly Vintage Pool is equal to or not greater than the Level Two Unpaid Rate Repurchase Trigger; further provided that it shall be an Event of Termination under this Section 10(b) if the Monthly Vintage Pool immediately succeeding a Cured Level Two Monthly Vintage Pool is Level Two Monthly Vintage Pool

provided that, notwithstanding the foregoing, as of any date the aggregate amount of repurchases that may be effected by the Seller under this Section 10 or Section 7(b), in each case, solely at the Seller's option, on a combined basis, shall not exceed an amount equal to 10% of the Group Outstanding Purchased Amount as of the date of such repurchase; *further provided* that, Seller shall only be entitled to exercise a repurchase under this Section 10 or Section 7(b) once per calendar year.

[MoneyLion] Master Receivables Purchase Agreement

(c) As of the last day of any Monthly Period beginning with the Monthly Period ending on July 31, 2024, if the 3-Month Average Unpaid Rate (30 Days) shall exceed 8.25%;

(d) As of the last day of any Monthly Period beginning with the Monthly Period ending on July 31, 2024, if the 3-Month Average Unpaid Rate (60 Days) shall exceed 6.75%; or

(e) The occurrence of any Change of Control which would reasonably be expected to impair the Servicer's ability to perform its obligations under the Servicing Agreement;

(f) At all times, unrestricted and unencumbered cash and cash equivalents of the Parent and its Subsidiaries shall be not less than \$30,000,000;

(g) As of the last day of each fiscal quarter, commencing with the fiscal quarter ending June 30, 2024, the Total Book Value of the Parent and its consolidated Subsidiaries shall not decrease by greater than 50% from the last day of the immediately preceding fiscal quarter; and

(h) Failure by the Seller to repurchase any Defective Receivable in accordance with Section 7(a).

then, in any such Event of Termination, the Purchaser Agent may (and at the direction of all Purchasers, shall) by notice to the Seller declare the Scheduled Termination Date to have occurred (in which case the Scheduled Termination Date shall be deemed to have occurred).

Section 11. Termination.

This Agreement shall terminate on the Scheduled Termination Date unless the Seller and the Purchaser Agent (on behalf of the Purchasers) mutually agree to extend the Scheduled Termination Date for an additional one-year period. Notwithstanding the foregoing, at any time prior to the Scheduled Termination Date, this Agreement may be terminated by mutual agreement between the Seller and the Purchaser Agent (on behalf of the Purchasers) or in accordance with Section 10. The expiration or termination of this Agreement shall not terminate or otherwise affect any obligations of the Seller, Purchaser Agent or the Purchasers from time to time under this Agreement relating to events prior to such termination or expiration. Without limiting the foregoing, the Seller's repurchase obligations under <u>Section 7</u>, and the indemnification obligations of the Parties under <u>Section 13(a)</u> and <u>Section 13(b)</u>, shall expressly survive the expiration or termination of this Agreement.

Section 12. Backup Servicing Agreement

Each of the Purchaser Agent and the Purchasers agrees that neither the Seller nor the Servicer shall have any obligation to pay or reimburse the Backup Servicer, the Purchaser Agent or any Purchaser for fees, costs and expenses incurred by the Backup Servicer, the Purchaser Agent

25

or any Purchaser in connection with the preparation, negotiation, execution, delivery and administration of the Backup Servicing Agreement, including, without limitation, any fees or other amounts payable under the Backup Servicing Agreement.

Section 13. Miscellaneous Provisions.

(a) The Seller hereby indemnifies and holds harmless the Purchaser Agent, each Purchaser, their permitted assignees and their respective Affiliates and their respective officers, directors, employees, counsel and other agents (each, a "**Purchaser Indemnified Party**") from and against any and all liabilities, losses, damages, claims, costs and expenses, interest, awards, judgments and penalties (including reasonable attorneys' fees and expenses) actually suffered or incurred by such Purchaser Indemnified Party, arising out of or resulting from (i) the breach or inaccuracy of any representation, warranty, covenant or other agreement made by the Seller hereunder (other than with respect to any representation as to any Defective Receivable being an Eligible Receivable, provided such Defective Receivable is timely repurchased pursuant to the terms hereof) and (ii) any Regulatory Trigger Event occurring with respect to any Purchased Receivable.

(b) Purchaser Agent and each Purchaser hereby indemnifies and holds harmless the Seller and its Affiliates and its officers, directors, employees, counsel and other agents (each, a "Seller Indemnified Party," and collectively with each Purchaser Indemnified Party, an "Indemnified Party") from and against any and all liabilities, losses, damages, claims, costs and expenses, interest, awards, judgments and penalties (including reasonable attorneys' fees and expenses) actually suffered or incurred by such Seller Indemnified Party, arising out of or resulting from the breach or inaccuracy of any representation, warranty, covenant or other agreement made by a Purchaser hereunder.

(c) Notwithstanding any other provision of this Agreement (including Section 13(a) and Section 13(b) above), to the fullest extent permitted by applicable law, each Party hereto, and each other Indemnified Party, shall not assert, and by its acceptance of rights hereunder hereby waives, and acknowledges that no other Person (including any assignee of any Eligible Receivable) shall have, whether pursuant to Section 13(a) or Section 13(b) or otherwise, any claim against the other Party hereto, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement or any agreement or instrument contemplated hereby or the transactions contemplated hereby or thereby. Notwithstanding any other provision of this Agreement (including Section 13(a) and Section 13(b) above), no Party shall be liable for any indemnification to an Indemnified Party to the extent that any such indemnified liabilities (a) result from such Indemnified Party's gross negligence or willful misconduct, as finally determined by a court of competent jurisdiction, or (b) constitute recourse for uncollectible Purchased Receivables.

(d) NOTWITHSTANDING ANY OTHER PROVISION OF THIS AGREEMENT, UNTIL THE DATE ON WHICH INITIAL PURCHASER HAS ENTERED INTO A CREDIT FACILITY THAT PROVIDES DEBT FINANCING TO PURCHASE THE ELIGIBLE RECEIVABLES ON TERMS AND CONDITIONS

[MoneyLion] Master Receivables Purchase Agreement

SATISFACTORY TO THE PURCHASER AGENT IN ITS SOLE DISCRETION, PURCHASHER AGENT AND EACH PURCHASER'S MAXIMUM LIABILITY, IN THE AGGREGATE, UNDER THIS AGREEMENT SHALL NOT EXCEED THE UNUSED FEES REQUIRED TO BE PAID TO PURCHASHER AGENT AND EACH PURCHASER PURSUANT TO THE TERMS OF THE PRIOR TO ANY CLAIM (SUCH LIMIT ON THE MAXIMUM LIABILITY OF THE PURCHASER AGENT AND EACH PURCHASER, THE "LIABILITY LIMIT"). THE LIABILITY LIMIT SET FORTH IN THIS SECTION SHALL APPLY CUMULATIVELY TO ALL MULTIPLE LIABILITY EVENTS AND SHALL NOT BE MULTIPLIED BY THE NUMBER OF INDIVIDUAL EVENTS OR CLAIMS.

(e) This Agreement and the other Transaction Documents set forth the entire understanding among the Parties hereto as to the subject matter set forth herein. The provisions of this Agreement cannot be waived or modified unless such waiver or modification be in writing and signed by Parties hereto. Inaction or failure to demand strict performance shall not be deemed a waiver.

THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER SHALL BE (f)GOVERNED BY, AND SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK WITHOUT REGARD TO CONFLICT OF LAWS PRINCIPLES (OTHER THAN SECTIONS 5-1401 AND 5-1402 OF THE NEW YORK GENERAL OBLIGATIONS LAW) THEREOF. ALL JUDICIAL PROCEEDINGS BROUGHT ARISING OUT OF OR RELATING HERETO, OR ANY OF THE OBLIGATIONS HEREUNDER, MAY BE BROUGHT IN ANY STATE OR FEDERAL COURT OF COMPETENT JURISDICTION IN THE STATE, COUNTY AND CITY OF NEW YORK. BY EXECUTING AND DELIVERING THIS AGREEMENT, EACH OF THE PARTIES HEREUNDER, FOR ITSELF AND IN CONNECTION WITH ITS PROPERTIES, IRREVOCABLY (i) ACCEPTS GENERALLY AND UNCONDITIONALLY THE EXCLUSIVE JURISDICTION AND VENUE OF SUCH COURTS; (ii) WAIVES ANY DEFENSE OF FORUM NON CONVENIENS; (iii) AGREES THAT SERVICE OF ALL PROCESS IN ANY SUCH PROCEEDING IN ANY SUCH COURT MAY BE MADE BY REGISTERED OR CERTIFIED MAIL, RETURN RECEIPT REQUESTED, TO SUCH PARTY'S ADDRESS SET FORTH IN Section 13(h) OF THIS AGREEMENT, IS SUFFICIENT TO CONFER PERSONAL JURISDICTION OVER SUCH PARTY IN ANY SUCH PROCEEDING IN ANY SUCH COURT, AND OTHERWISE CONSTITUTES EFFECTIVE AND BINDING SERVICE IN EVERY RESPECT; AND (iv) AGREES THAT THE OTHER PARTIES HEREUNDER RETAIN THE RIGHT TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY LAW OR TO BRING PROCEEDINGS IN THE COURTS OF ANY OTHER JURISDICTION.

EACH OF THE PARTIES HERETO HEREBY AGREES TO WAIVE ITS RESPECTIVE RIGHTS TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING HEREUNDER OR ANY DEALINGS BETWEEN THEM RELATING TO

[MoneyLion] Master Receivables Purchase Agreement

THE SUBJECT MATTER OF THIS AGREEMENT. THE SCOPE OF THIS WAIVER IS INTENDED TO BE ALL-ENCOMPASSING OF ANY AND ALL DISPUTES THAT MAY BE FILED IN ANY COURT AND THAT RELATE TO THE SUBJECT MATTER OF THIS AGREEMENT, INCLUDING CONTRACT CLAIMS, TORT CLAIMS, BREACH OF DUTY CLAIMS AND ALL OTHER COMMON LAW AND STATUTORY CLAIMS. EACH PARTY HERETO ACKNOWLEDGES THAT THIS WAIVER IS A MATERIAL INDUCEMENT TO ENTER INTO A BUSINESS RELATIONSHIP, THAT EACH HAS ALREADY RELIED ON THIS WAIVER IN ENTERING INTO THIS AGREEMENT, AND THAT EACH WILL CONTINUE TO RELY ON THIS WAIVER IN ITS RELATED FUTURE DEALINGS. EACH PARTY HERETO FURTHER WARRANTS AND REPRESENTS THAT IT HAS REVIEWED THIS WAIVER WITH ITS LEGAL COUNSEL AND THAT IT KNOWINGLY AND VOLUNTARILY WAIVES ITS JURY TRIAL RIGHTS FOLLOWING CONSULTATION WITH LEGAL COUNSEL. THIS WAIVER IS IRREVOCABLE, MEANING THAT IT MAY NOT BE MODIFIED EITHER ORALLY OR IN WRITING (OTHER THAN BY A MUTUAL WRITTEN WAIVER SPECIFICALLY REFERRING TO THIS <u>Section 13(e)</u> AND EXECUTED BY EACH OF THE PARTIES HERETO), AND THIS WAIVER SHALL APPLY TO ANY SUBSEQUENT AMENDMENTS, RENEWALS, SUPPLEMENTS OR MODIFICATIONS HERETO OR TO ANY OTHER DOCUMENTS OR AGREEMENTS RELATING TO THIS AGREEMENT. IN THE EVENT OF LITIGATION, THIS AGREEMENT MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.

(g) This Agreement may be executed in one or more counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same instrument. Delivery of an executed counterpart of a signature page by facsimile or electronic transmission (*i.e.* ".pdf" or ".tif") shall be effective as delivery of a manually executed counterpart of this Agreement. Each of the Parties hereto agrees that this Agreement and any other documents to be delivered in connection herewith may be electronically signed, and that any electronic signatures appearing on this Agreement or such other documents are the same as handwritten signatures for the purposes of validity, enforceability, and admissibility.

(h) If any provision of this Agreement shall be held, deemed to be or shall, in fact, be inoperative or unenforceable as applied in any particular situation, such circumstances shall not have the effect of rendering the provision in question inoperative or unenforceable in any other situation or of rendering any other provision or provisions herein contained invalid, inoperative or unenforceable to any extent whatsoever. The invalidity of any one or more phrases, sentences, clauses or paragraphs herein contained shall not affect the remaining portions of this Agreement or any part hereof.

(i) All notices, requests, demands or other instruments which may or are required to be given by a Party to another hereunder to a Party's address set forth in this Section 13(h) and may be delivered or furnished by electronic communication (including e-mail) pursuant to procedures mutually approved by the Seller and the Purchaser Agent.

[MoneyLion] Master Receivables Purchase Agreement

All notices, requests, demands or other instruments given by any Party to another hereunder shall be effective and be deemed to have been received (i) if delivered by hand, upon personal delivery, (ii) if delivered by overnight courier service, one (1) Business Day after delivery to such courier service, (iii) certified mail, upon receipt or refusal thereof, and (iv) if delivered by electronic mail, upon sender's receipt of confirmation of proper transmission.

If to the Purchaser Agent or the Purchasers:

c/o Sound Point Capital Management, LP, as Purchaser Agent 375 Park Avenue, 33rd Floor New York, NY 10152 Attn: [*****] Tel: [*****] Email: [*****]

If to the Seller:

ML Plus LLC c/o MoneyLion Technologies Inc. 249-245 West 17th Street, 4th Floor New York, New York 10011 Attention: [*****] E-mail: [*****]

Any Party may change the address and name of the addressee to which subsequent notices are to be sent to it by notice to the others given as aforesaid, but any such notice of change, if sent by mail, shall not be effective until the fifth day after it is mailed.

(j) All forms specified by the text hereof or by reference to exhibits attached hereto shall be substantially as set forth herein, subject to such changes agreed to between the Purchaser Agent and the Seller, or as may be required by applicable laws hereafter enacted.

(k) Information provided by Seller or the Servicer to a Purchaser, Purchaser Agent or any of their respective Affiliates shall not include confidential information that meets the definition of non-public personally identifiable information ("NPI") regarding a MoneyLion Accountholder as defined by Title V of the Gramm-Leach-Bliley Act of 1999 and implementing regulations (collectively, the "GLB Act"), or protected by other Requirements of Law regarding privacy (collectively, "Privacy Laws"). Seller and Servicer agree that NPI will not be disclosed or made available to any Purchaser, Purchaser Agent or any of their third party, agent or employee for any reason whatsoever, other than upon written request by Purchaser Agent and:

29

(i) on a "need to know" basis in order for a Purchaser or Purchaser Agent to perform its obligations under this Agreement, any other Transaction

Document and any other agreement related to the Receivables; <u>provided</u>, that such agents or representatives receiving NPI are subject to a confidentiality obligation which shall be consistent with and no less restrictive than the provisions of this <u>Section</u> <u>13(j)</u>; or

(ii) as required by Requirements of Law or Governmental Authorities or as otherwise expressly permitted by this Agreement (subject to Privacy Laws); provided, that, prior to any disclosure of NPI to Purchaser Agent or a Purchaser as required by Requirements of Law, the applicable Purchaser(s) or Purchaser Agent shall, if permitted by Requirements of Law:

(A) not disclose any such information until it has notified Seller or Servicer in writing of all actual or threatened legal compulsion of disclosure, and any actual legal obligation of disclosure promptly upon becoming so obligated; and

disclosure.

(B) cooperate to the fullest extent possible with all lawful efforts by Seller and Servicer to resist or limit

Upon receipt of any NPI pursuant to <u>Section 13(j)(i)-(ii)</u>, the receiving Purchaser(s) and Purchaser Agent shall: (i) promptly notify Seller of any unauthorized access to that NPI or any breach in security measures or systems for the protection of that NPI and take appropriate action to prevent further unauthorized access or cure such breach, (ii) cooperate with Seller and Servicer with respect to their investigation or inquiry as to any such unauthorized access or breach, provide any notices regarding such unauthorized access or breach to appropriate law enforcement agencies and government regulatory authorities and affected consumers as Servicer reasonably deems appropriate and pay reasonable expenses related thereto.

Each of the Parties hereto shall comply with all Requirements of Law regarding use, disclosure and safeguarding of any and all consumer information and will maintain a comprehensive written information security program, in compliance with Requirements of Law, which shall include all necessary measures, including the establishment and maintenance of appropriate policies, procedures and technical, physical, and administrative safeguards, to (w) ensure the security and confidentiality of the NPI, (x) protect against any foreseeable threats or hazards to the security or integrity of NPI, (y) protect against unauthorized access to or use of such information, and (z) ensure appropriate disposal of NPI.

(1) This Agreement shall be binding on, and shall inure to the benefit of, the Seller, each Purchaser and the Purchaser Agent and may not be assigned by Party without the consent of the other Parties. Notwithstanding the foregoing, any Receivables sold hereunder and any related rights of the Purchasers with respect thereto under this Agreement and the other Transaction Documents may be assigned by a Purchaser to (i) Persons that are Affiliates and not Competitors, or (ii) in connection with a pledge under a private debt facility to a lender. In connection with any such permitted assignment, any additional reasonably customary fees and expenses incurred by the Seller as a result of any

[MoneyLion] Master Receivables Purchase Agreement

such assignment shall be promptly reimbursed by the applicable Purchaser(s) upon demand by the Seller. The Seller shall provide commercially reasonable assistance to the Purchasers or Purchaser Agent in connection with any such assignment of Eligible Receivables, *provided* that the Seller shall not be required to expend any of its own funds or incur any expenses in connection with such assistance. Notwithstanding the foregoing, assignments may not be made to any entity (including any Affiliates) for purposes of effecting or participating in a securitization transaction or other similar capital markets transaction (including but not limited to 144A/Regulation S private placements or registered offerings).

(m) Notwithstanding anything to the contrary herein, the Purchaser Agent and Seller may from time to time, in their sole discretion, amend the Eligibility Criteria set forth on <u>Annex A</u> (which shall also include any corresponding amendments to the definition of "Eligible MoneyLion Accountholder", as necessary), and any such amendments shall not require the execution of an amendment to this Agreement and shall be deemed effective upon Purchaser Agent's and Seller's written agreement thereof (which may be via email).

Section 14. Confidentiality

Each of the Purchaser Agent, Purchasers and the Seller (each, a "**Receiving Party**") agrees to maintain the confidentiality of all information furnished or delivered to it pursuant to or in connection with the Transaction Documents (the "Information"). Such Information may be used by the Receiving Parties only for the purpose for which it was disclosed to them and may be disclosed only for the purpose of or in connection with the transactions contemplated by the Transaction Documents to:

(a) such Receiving Party's Affiliates or such party's or its Affiliates' directors, officers, employees, agents, accountants, auditors, legal counsel and other representatives (collectively, "**Receiving Party Representatives**"), in each case, who need to know such Information for the purpose of assisting in the negotiation, completion and administration of such Transaction Documents, provided that any such Receiving Party Representative is made aware of the Receiving Party's obligations under this Section 14 prior to such disclosure being made;

(b) such Receiving Party's permitted assigns, transferee, successors and participants to the extent such disclosure is made pursuant to a written agreement to hold such information upon substantially the same terms as this Section 14 or such other terms as may be agreed by the Servicers and the Purchaser Agent;

- (c) any Person who is a party to a Transaction Document;
- (d) to the extent required by Applicable Law or by any Governmental Authority;

(e) the extent that such Receiving Party needs to disclose the same for the exercise, protection or enforcement of any of its rights under any of the Transaction Documents or in connection with any action or proceeding relating to any Transaction Document; and

(f) if the Seller, the Purchaser Agent or the applicable Purchaser, as applicable, shall have consented, in writing, to such disclosure.

Section 15. Joinder of Additional Purchasers.

At any time during the Term:

(a) without the written consent of the Seller or Servicer, one or more Additional Purchasers that (i) is an Affiliate of Purchaser Agent and/or a Purchaser and (ii) is not a Competitor may join this Agreement and the Servicing Agreement as a Purchaser in all respects by delivering an executed Joinder Agreement to the Seller and Servicer and such delivered Joinder Agreement shall be executed by Seller and Servicer; and

(b) with the written consent of the Seller, one or more Additional Purchasers that (i) is not an Affiliate of Purchaser Agent and/or a Purchaser and (ii) is not a Competitor may join this Agreement and the Servicing Agreement as a Purchaser in all respects by delivering an executed Joinder Agreement to the Seller and Servicer along with such other approvals, resolutions, certificates, legal opinions and other documents as the Seller may reasonably request, in each case, in form and substance reasonably acceptable to the Seller. Seller's written consent pursuant to this Section 15(b) shall be evidenced by Seller and Servicer's execution of such Joinder Agreement delivered by such Additional Purchaser *provided*, that, notwithstanding the foregoing, the Seller shall be entitled to deny its consent to assignments made pursuant to this Section 15(b) to more than five (5) other Persons in the aggregate hereunder.

Upon execution of any Joinder Agreement, the Additional Purchaser thereunder shall become a Purchaser hereunder and under the Servicing Agreement, subject to the rights, duties and obligations of a Purchaser in all respects.

Section 16. Increase in Facility Limit.

Upon notice to the Purchaser Agent and each Purchaser, the Seller may request at any one or more times after the Closing Date that each of the Purchasers ratably increase their respective unused Commitments (an "Incremental Commitment"). The total amount of the Incremental Commitments shall not exceed \$75,000,000. At the time of sending such notice with respect to the Purchasers and the Purchaser Agent, the Seller shall specify (x) the time period within which the Purchasers are requested to respond to the Seller's request (which shall in no event be less than ten (10) days from the date of delivery of such notice to the Purchaser Agent) and (y) the amount of the Incremental Commitment being requested, which shall be in a minimum amount of \$1,000,000. Each of the Purchasers shall notify the Purchaser Agent and the Seller and the Servicer within the

[MoneyLion] Master Receivables Purchase Agreement

applicable time period whether or not such Purchaser agrees, in its sole discretion, to make such ratable increase to such Purchaser's Commitment or otherwise agrees to any lesser increase in its Commitment. Any Purchaser not responding within such time period shall be deemed to have declined to consent to an increase in such Purchaser's Commitment. In the event that one or more Purchasers fails to consent to all or any portion of any such request for an increase in its Commitment, the Seller may request that any unaccepted portion of the requested increases in Commitments be allocated to one or more willing Purchasers as agreed in writing among the Seller and such willing Purchasers (in each case, in their sole discretion), such that such Purchasers increase in their Commitment exceeds such Purchaser's ratable share. Any such Purchaser may agree, in its sole discretion, to such increase in its Commitment. If the Commitment of any Purchaser is increased in accordance with this Section 16, the Purchaser Agent, the Purchasers, the Seller and the Servicer shall determine the effective date with respect to such increase and shall enter into such documents as agreed to by such parties to document such increase in the Facility Limit. On the effective date of such increase, the Seller shall pay to each Purchaser that has increased its Commitment, through the Purchaser Agent, a non-refundable upsize fee (the "Upsize Fee") in an amount equal to the product of (x) 1.00% and (y) the amount by which the Purchaser has increased its Commitment.

[SIGNATURE PAGES FOLLOW]

IN WITNESS WHEREOF, each Party hereto has caused this Agreement to be duly executed and delivered by its officer thereunto duly authorized as of the date first above written.

Sound Point Capital Management, LP, as the Purchaser Agent

By: <u>/s/ Kevin Gerlitz</u> Name: Kevin Gerlitz Title: Chief Financial Officer

SP Main Street Funding I LLC, as an Initial Purchaser

By: <u>/s/ Kevin Gerlitz</u> Name: Kevin Gerlitz Title: Authorized Signatory

ML PLUS LLC, as the Seller

By: <u>/s/ Richard Correia</u> Name: Richard Correia Title: Chief Financial Officer

SCHEDULE I

COMMITMENTS

[*****] [<u>Schedule I]</u>

SCHEDULE II

TARGET SALE AMOUNTS

[*****]

[Schedule II]

SCHEDULE III

PURCHASE PRICE CALCULATIONS

[*****]

[MoneyLion] Master Receivables Purchase Agreement

[Schedule IV]

SCHEDULE IV

CALCULATION OF REVISED PERCENTAGE

[*****] [<u>Schedule IV]</u>

SCHEDULE V

SELLER INFORMATION

[*****] [Schedule V]

SCHEDULE VI

LIST OF COMPETITORS

[*****]

[MoneyLion] Master Receivables Purchase Agreement

[Schedule VI]

Annex A <u>Eligibility Criteria</u>

[*****]

[Annex A]

Annex B Seller's Representations and Warranties

[*****]

[Annex B]

Annex C <u>Purchaser Agent and Purchasers' Representations and Warranties</u>

[*****]

[Annex C]

Exhibit A Form of Confirmation

[*****]

[Exhibit A]

Exhibit B Form of Back-up UCC-1

[*****] [<u>Exhibit B]</u>

Exhibit C <u>Form of Confirmation (Repurchase)</u>

[*****]

[MoneyLion] Master Receivables Purchase Agreement

[Exhibit C]

Exhibit D Form of Joinder Agreement to Master Receivables Purchase Agreement

[*****] [<u>Exhibit D</u>]

Exhibit E <u>Eligibility Guidelines</u>

> [*****] [<u>Exhibit E</u>]

Exhibit F <u>Form of Compliance Certificate</u>

[*****]

[Exhibit F]

[*****] = Certain identified information has been excluded from the exhibit because it is both not material and is the type that the Registrant treats as private or confidential

Execution Version

SERVICING AGREEMENT

THIS SERVICING AGREEMENT (this "**Agreement**") is made and entered into as of June 30, 2024, by and MoneyLion Technologies Inc., a Delaware corporation (the "**Servicer**"), Sound Point Capital Management, LP, a Delaware limited partnership, as purchaser agent (the "**Purchaser Agent**"), SP Main Street Funding I LLC, a Delaware limited liability company (the "**Initial Purchaser**"), and the Additional Purchasers that may from time to time become party hereto (together with the Initial Purchasers, each individually, a "**Purchaser**" and collectively, the "**Purchasers**").

RECITALS

WHEREAS, the Purchaser Agent and the Purchasers desire to appoint the Servicer, and the Servicer desires to accept such appointment, to service certain Purchased Receivables which may be transferred to the applicable Purchaser from time to time pursuant to the terms of the Master Receivables Purchase Agreement, dated as of June 30, 2024 (as it may be amended, supplemented or otherwise modified from time to time in accordance with the terms thereof, the "Purchase Agreement"), by and among the Purchaser Agent, the Purchasers party thereto and ML Plus LLC, a Delaware limited liability company, as seller; and

WHEREAS, the Purchaser Agent, the Purchasers and the Servicer desire to execute this Agreement to define the rights, duties and obligations of each party hereto with respect to any such Purchased Receivables.

NOW, THEREFORE, in consideration of the mutual covenants and agreements hereinafter set forth, the parties hereto hereby agree as follows:

1. Definitions and Interpretations.

(a) <u>Definitions</u>. All capitalized terms used herein, but not otherwise defined herein, shall have the respective meanings assigned to such terms in this <u>Section 1(a)</u> or in the Purchase Agreement:

"Account Intercreditor Agreement" means that certain Account Intercreditor Agreement dated as of June 30, 2024, as amended, restated, amended and restated, supplemented or otherwise modified from time to time, by and among the Servicer, the Purchaser Agent and Monroe Capital Management Advisors, LLC as Monroe Agent.

"Agreement" has the meaning set forth in the introductory statement hereto, as amended, supplemented or otherwise modified from time to time.

"Authorized Officer" means, as applied to any Person, any individual holding the position of chairman of the board (if an officer), chief executive officer, president, chief financial officer, general counsel, treasurer or controller (or, in each case, the equivalent thereof).

"Bankruptcy Code" shall mean Title 11 of the United States Code, 11 U.S.C. §§ 101 et. seq., as amended from time to time.

"Code" means the Internal Revenue Code of 1986, as amended from time to time.

"Indemnified Liabilities" has the meaning set forth in Section 12 hereto.

"Indemnified Person" has the meaning set forth in Section 12 hereto.

"Net Collections Amount" means, for any Monthly Period, the aggregate cash payments made by or behalf of the MoneyLion Accountholders in respect of Purchased Receivables, whether in the form of cash, checks, wire transfers, electronic transfers or any other form of cash payment but excluding in all cases any Excluded Amounts.

"Permitted Receivable Modification" means, with respect to any Receivable, Qualified Deferrals, any waiver, modification or variance of any term of any Receivable or any consent to the postponement of strict compliance with any such term or any other grant of an indulgence or forbearance to the related MoneyLion Accountholder, in each case, granted by the Servicer either (i) in accordance with the Servicing Standard and is determined by Servicer at the time of such modification to be a practical manner to obtain a reasonable recovery from such Receivable based upon its prior servicing experience for similar receivables or (ii) is required by Applicable Law.

"Purchase Agreement" has the meaning set forth in the recitals hereto.

"Purchased Receivable" has the meaning set forth in the Purchase Agreement.

"Purchased Receivables File" means an electronic file, in a computer readable format reasonably satisfactory to the Purchaser Agent containing the data files listed on the Data Tape with respect to the Purchased Receivables and the Monthly Servicing Reports.

"Qualified Deferral" means a deferral of an Instacash Repayment Date by up to two weeks as long as the related MoneyLion Accountholder's current outstanding Instacash Advance Amount has not been deferred before and there are no pending payments in respect of the related Instacash.

"Servicer" has the meaning set forth in the introductory statement hereto.

"Servicer Account" has the meaning set forth in the Account Intercreditor Agreement.

"Servicer Event of Default" has the meaning set forth in Section 10(a) hereto.

"Servicing Fee" has the meaning set forth in Section 6 hereto.

"Servicing Fee Rate" means three percent (3.00%).

"Servicing Guidelines" means the written servicing policies and procedures of the Servicer, as in effect on the Closing Date, as modified from time to time in accordance with the terms of Section 9(f).

"Servicing Standard" has the meaning set forth in Section 2(a) hereto.

"Settlement Date" has the meaning set forth in Section 6 hereto.

"Successor Servicer" has the meaning set forth in Section 10(a) hereto.

"Successor Servicing Agreement" has the meaning set forth in Section 9(a) hereto.

"Termination Notice" has the meaning set forth in Section 10(a) hereto.

"Treasury Regulations" means such provisions of the income tax regulations of the United States Department of the Treasury or any successor provisions promulgated under the Code.

(b) Other Definitional Provisions.

(i) Unless otherwise specified, references to any amount as on deposit or outstanding on any particular date shall mean such amount at the close of business on such day.

(ii) Any of the terms defined herein may, unless the context otherwise requires, be used in the singular or the plural, depending on the reference. References herein to any Section, Schedule or Exhibit shall be to a Section, a Schedule or an Exhibit, as the case may be, hereof unless otherwise specifically provided. The use herein of the word "include" or "including," when following any general statement, term or matter, shall not be construed to limit such statement, term or matter to the specific items or matters set forth immediately following such word or to similar items or matters, whether or not no limiting language (such as "without limitation" or "but not limited to" or words of similar import) is used with reference thereto, but rather shall be deemed to refer to all other items or matters that fall within the broadest possible scope of such general statement, term or matter.

(iii) In the event that any reports are not available to any Person on the date on which such Person is required to make a determination of whether a computational test has been satisfied pursuant hereto, such determination shall be made using the most current available information.

2. Appointment of Servicer and Servicing Duties.

(a) <u>Servicer Servicing Duties</u>. The Purchasers and the Purchaser Agent hereby appoint Servicer, and Servicer hereby accepts such appointment to act as the contractual representative to service the Purchased Receivables on behalf of the Purchasers pursuant to the terms and conditions of this Agreement. The Servicer agrees to service the Purchased Receivables in accordance with the standard of care set forth in <u>Section 5</u> (the "Servicing Standard"). Servicer shall take all lawful actions and procedures which Servicer deems necessary or advisable to service, and collect on, the Purchased Receivables in accordance with the Servicing Guidelines. The foregoing notwithstanding, the Servicer shall not institute judicial proceedings on behalf of the applicable Purchaser without such Purchaser's prior consent, which consent shall be deemed given on all ordinary course matters relating to collection when due of all amounts owing under the terms and conditions of any Purchased Receivable. In furtherance of the foregoing, the Servicer shall perform the following services until it is no longer required to do so in accordance with <u>Section 11</u>:

(i) The Servicer shall:

1. In a manner consistent with the Servicing Guidelines and the Servicing Standard, collect all payments due under each Purchased Receivable when the same shall become due and payable. It is understood and agreed that all payments received from a MoneyLion Accountholder in respect of Instacash Repayment Amounts shall be applied (i) first, as a repayment of all outstanding and due Instacash Advance Amounts of such MoneyLion Accountholder until all such Instacash Advance Amounts have been repaid in full, and (ii) thereafter, to any outstanding Excluded Amounts relating thereto.

2. (A) Use commercially reasonable efforts to cause all Collections with respect to Purchased Receivables to be deposited in a Servicer Account by each third-party payment processor utilized by the Servicer and (B) promptly, but in any event within one (1) Business Day after the Servicer's receipt of such Collections, identify such Collections that are Advance Collections and transfer such Advance Collections to the Collection Account in accordance with the terms of the Account Intercreditor Agreement.

3. On each Business Day, direct the Account Bank to transfer all Advance Collections (net of any Servicing Fee solely if such Business Day is a Settlement Date) in the Collection Account to an account designated by the Purchaser Agent to the Servicer in writing.

(ii) With respect to any Purchased Receivable which remains unpaid past the related Instacash Repayment Date, or in which such a payment default is imminent, the Servicer shall undertake, in accordance with the Servicing Standard, such servicing actions set forth in the Servicing Guidelines.

(iii) The Servicer shall not permit the commingling of Collections with respect to the Purchased Receivables at any time with its funds or the funds of any other Person, other than (x) in a Servicer Account, until timely transferred to the Collection Account as described in Section 2(a)(i)2 above, and (y) with respect to the funds related to a Defective Receivable, an Underperforming Receivable or any Receivable repurchased by the Seller in accordance with Section 7(b) or Section 10 of the Purchase Agreement, so long as the Servicer clearly indicates that any such funds are held for the benefit of the Seller and any such funds are transferred out of the Collection Account within three (3) Business Days following the deposit thereof; provided, that, inadvertent and non-reoccurring errors by Servicer in applying proceeds that are promptly cured, shall not be considered a breach of this covenant. In the event that the Servicer deposits into the Collection Account any amount not required to be deposited therein, it shall be entitled to deduct such amount from deposits to the Collection Account.

(iv) Such other duties as the Servicer and the Purchasers may mutually agree to.

(b) <u>No Ownership Rights</u>. Notwithstanding the foregoing, the Servicer acknowledges and agrees that the Servicer shall have no ownership rights in and to the Purchased Receivables.

(c) <u>Modification of Purchased Receivables</u>. Other than (i) any Permitted Receivable Modifications or (ii) with the prior consent of the Purchaser Agent, the Servicer may not amend, modify or waive any term or condition of any Purchased Receivable.

(d) <u>Delegation</u>. Without the prior written consent of the Purchaser Agent, the Servicer may delegate its responsibilities to any subservicer in the ordinary course of the Servicer's performing its obligations hereunder and consistent with Servicer's customary practice, subject to such third parties being under the control, supervision and direction of the Servicer; provided, however, that any such delegation shall not affect the Servicer's obligations and liabilities hereunder.

(e) <u>Maintenance of and Access to Records</u>. The Servicer, in its capacity as custodian, shall keep safely all electronic files of the Purchased Receivables Files in its possession as necessary to conduct collection and other servicing activities in accordance with its customary practices and procedures. The Servicer, in its capacity as custodian, shall provide to the Purchaser Agent or its duly authorized representatives, attorneys or auditors electronic access to view the Purchased Receivables File maintained by the Servicer as the Purchaser Agent shall reasonably request which do not unreasonably interfere with the Servicer's normal operations or customer or employee relations.

(f) <u>Release of Documents</u>. Upon written instruction from the Purchaser Agent, the Servicer, in its capacity as custodian, shall release (or provide online access to view the Purchased Receivables Files posted on an electronic portal maintained by the Servicer or any third-party service provider) or cause to be released any Purchased Receivables File to the Purchaser Agent, the Purchaser Agent's agent or its designee, as the case may be, at such place or places as the Purchaser Agent may designate, as soon as

practicable (but in no event more than seven (7) days after the date of such request). Notwithstanding the foregoing, the Purchaser Agent shall provide, or cause its agent to provide, access to such Purchased Receivables File to the Servicer for the purpose of carrying out its duties and responsibilities with respect to the servicing of Purchased Receivables hereunder.

3. Reporting and Other Information.

(a) The Servicer will provide, as reasonably requested by the Purchaser Agent, a back-up servicing tape or disk or data file acceptable to Purchaser Agent, such that upon the occurrence of the election by Purchaser Agent following a Servicer Event of Default, the Backup Servicer or any other third-party servicer selected by Purchaser Agent pursuant to the terms of the Purchase Agreement can commence the immediate servicing of the Purchased Receivables.

(b) Not later than 5:00 p.m. (Eastern Standard Time) on each Business Day, the Servicer shall provide to the Backup Servicer and the Purchaser Agent a copy of the Daily Reconciliation Report (as defined in the Account Intercreditor Agreement) prepared on such Business Day in accordance with the Account Intercreditor Agreement.

(c) Not later than 5:00 p.m. (Eastern Standard Time) on each Purchase Date, the Servicer shall provide to the Backup Servicer and the Purchaser Agent online access to view a data tape in the form attached hereto as Exhibit A that consists of most recent originations, Advance Collections, and advance status update, in each case, setting forth information relating to the Purchased Receivables that took place on the immediately preceding Purchase Date (the "**Data Tape**") posted on an electronic portal maintained by the Servicer or any third-party service provider.

(d) Not later than 5:00 p.m. (Eastern Standard Time) on the fifth Business Day of each calendar month, the Servicer shall provide to the Backup Servicer and the Purchaser Agent online access to view a monthly servicing report in form and substance consistent in all material respects with similar reports provided to the Purchaser Agent's Affiliate prior to the Closing Date setting forth information relating to the Servicer's servicing Report") and a data tape which includes the data to support the calculations in the Monthly Servicing Reports posted on an electronic portal maintained by the Servicer or any third-party service provider. Upon discovery of any error or receipt of notice of any error in any Monthly Servicing Report, the Servicer and the Purchaser Agent shall arrange to confer and shall agree upon any adjustments necessary to correct any such errors.

4. Inspections and Examinations.

(a) The Servicer will permit or cause to be permitted, as applicable, any authorized representatives designated by the Purchaser Agent (which such authorized representatives may be unaffiliated third parties) to visit and inspect any of the properties of the Servicer at any time, and from time to time upon reasonable advance notice and during normal working hours, to inspect, copy and take extracts from its books, records, financial statements, credit and collection policies, legal and regulatory compliance, operating and reporting procedures and information systems and to discuss its affairs, finances and accounts with any employees of the Servicer. Servicer agrees to pay the Purchaser Agent's reasonable, customary and documented out-of-pocket expenses for any such field examination and audit and the preparation of reports thereof performed or prepared only once in any calendar year in an aggregate amount not to exceed thirty five thousand Dollars (\$35,000) in any calendar year.

(b) Servicer shall hold and maintain all electronic documents evidencing or pertaining to the Purchased Receivables representing the same in trust for the exclusive benefit of the Purchaser Agent (on behalf of the Purchasers) during the term of this Agreement.

5. The Servicer's Standard of Care and Diligence.

The Servicer agrees to diligently perform the services contemplated hereunder in good faith and in accordance with industry standards customary for servicing financial transactions of the type which comprise the portfolio of Purchased Receivables, in accordance with the specific terms of such consumer receivables, with the same care, skill and diligence with which prudent receivables servicers service comparable receivables, all in accordance with all Applicable Law and in accordance with the Servicing Guidelines.

6. The Servicer's Fees and Expenses.

With respect to each Monthly Period, the Servicer shall receive as full compensation for its services hereunder a monthly servicing fee equal to the Servicing Fee Rate *times* the Net Collections Amount for such Monthly Period (the "Servicing Fee"). The Servicer shall be responsible for all reasonable out-of-pocket costs and expenses of the Servicer due and payable in connection with the performance of its duties hereunder. The Servicing Fee for any Monthly Period shall be due and payable within five (5) Business Days immediately following such Monthly Period (each such day, a "Settlement Date"), commencing with July 8, 2024. The Servicing Fee shall, on each Settlement Date, be deducted by the Servicer from Advance Collections on deposit in the Collection Account.

7. Servicer Representations and Warranties.

To induce the Purchasers and the Purchaser Agent to enter into this Agreement, the Servicer represents and warrants to the Purchasers and the Purchaser Agent as of the Closing Date and as of each Purchase Date that:

(a) Existence. It is (i) duly incorporated, validly existing and in good standing under the laws of the State of Delaware, (ii) has all requisite corporate power and has all governmental licenses, authorizations, consents and approvals, necessary to own its assets and carry on its business as now being or as proposed to be conducted, except as would not reasonably be expected (either individually or in the aggregate) to result in a Material Adverse Effect, (iii) is qualified to do business and is in good standing in all other jurisdictions in which the nature of the business conducted by it makes such qualification necessary, except where failure to so qualify could not reasonably be expected (either individually or in the aggregate) to result in a Material Adverse Effect, and (iv) is in compliance with all Requirements of Law, except as would not reasonably be expected (either individually or in the aggregate) to result in a Material Adverse Effect.

(b) <u>Power and Authority; Enforceability</u>. This Agreement has been duly executed and delivered by it and constitutes a legal, valid and binding obligation of it, enforceable against it in accordance with its terms, except as such enforcement may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors' rights generally and except that the equitable remedy of specific performance and other equitable remedies are subject to the discretion of the courts.

(c) Due Authorization. The execution, delivery and performance by it of this Agreement has been duly authorized by all necessary action and does not require any additional approvals or consents or other action by or any notice to or filing with any Person other than any that have heretofore been obtained, given or made.

(d) <u>No Consents</u>. Except (x) as would not reasonably be expected to result in a Material Adverse Effect and (y) approvals, consents, exemptions, authorizations, actions, notices and filings

which have been duly obtained, taken, given or made and are in full force and effect, no authorizations, approvals or consents of, and no filings or registrations with, any Governmental Authority, or any other Person, are necessary for the execution, delivery or performance by it of this Agreement or for the legality, validity or enforceability thereof, except for filings and recordings in connection with this Agreement and the security interest created pursuant to this Agreement.

(e) <u>Litigation</u>. Except for any Regulatory Trigger Event which has been disclosed in Parent's public filings with the SEC, there are no actions, suits, arbitrations, investigations or proceedings pending or, to its knowledge, threatened in writing against it affecting any of its property before any Governmental Authority that, individually or in the aggregate, could reasonably be expected to result in a Material Adverse Effect.

(f) Regulatory Action. Except for any Regulatory Trigger Event which has been disclosed in Parent's public filings with the SEC, to its knowledge, it is not currently under investigation by any Governmental Authority the result of which would be reasonably likely to result in a Material Adverse Effect. Except for any Regulatory Trigger Event which has been disclosed in Parent's public filings with the SEC, it has not been the subject of any government investigation which has resulted in the voluntary or involuntary suspension of a license, a cease and desist order, or such other action as could reasonably be expected to result in a Material Adverse Effect on its business.

(g) Solvency. It is Solvent. It is not contemplating the commencement of an Insolvency Event in respect of it or any of its assets.

(h) <u>Ability to Perform</u>. The Servicer has the requisite facilities, procedures, and experienced personnel necessary for the sound servicing of the Purchased Receivables in accordance with this Agreement. The Servicer does not believe, nor does it have any reason or cause to believe, that it cannot perform the duties and obligations of the Servicer contained in this Agreement.

(i) <u>Reports Accurate</u>. All reports (including, Monthly Servicing Reports) information, exhibits, financial statements, documents, books, records or reports furnished or to be furnished by the Servicer to the Purchasers, the Purchaser Agent or the Backup Servicer in connection with this Agreement are, taken as a whole, accurate, true and correct in all material respects as of the date specified therein or the date so furnished, as applicable.

(j) <u>Compliance with the Servicing Guidelines</u>. With respect to the servicing of the Purchased Receivables, the Servicer has complied in all material respects with the Servicing Guidelines.

(k) Investment Company Act. It is not an "investment company" within the meaning of the Investment Company Act of 1940, as

amended.

(1) Information Security Program. The Servicer maintains an information security program that is reasonable and in compliance with applicable law and the Servicer's collection, storage, processing, use, transmission and disclosure of any "non-public personal information" (as defined in the regulations under the Gramm-Leach-Bliley Act) or other sensitive information regarding individuals, their employment, family, health, racial or financial status, comply with all privacy policies, contractual obligations, and applicable law, except where failure to comply or to have complied with applicable law could not reasonably be expected to have a material adverse effect on the value or collectability of the affected Purchased Receivable.

(m) Location of Principal Place of Business; Chief Executive Office. As of the Closing Date, the Servicer's principal place of business and chief executive office, and the offices where it keeps

tangible records concerning the Purchased Receivables is 249-245 West 17th Street, 4th Floor, New York, NY 10011.

8. Affirmative Covenants.

The Servicer covenants and agrees that during the term of this Agreement, unless the Purchaser Agent shall otherwise consent in writing, the Servicer will:

(a) <u>Existence</u>; <u>Licenses</u>, <u>etc.</u> Perform all actions reasonably necessary to preserve and keep in full force and effect its corporate existence, and qualifications, permits and licensing as and where required to perform its obligations under this Agreement and comply with all laws applicable to the Servicer and its activities where the failure to so comply would reasonably be expected to result in a Material Adverse Effect.

(b) Personnel. Maintain adequate and qualified personnel to perform the duties undertaken herein with respect to the Purchased Receivables.

(c) <u>Cooperation</u>. Reasonably cooperate with the Backup Servicer (once engaged) or, if applicable, any Successor Servicer engaged by the Purchaser Agent (on behalf of the Purchasers) in accordance herewith.

(d) Notice of Default or Regulatory Trigger Event. The Servicer will deliver to the Purchaser Agent (on behalf of the Purchasers), promptly, and in any event within three (3) Business Days of any Authorized Officer of the Servicer obtaining knowledge (i) of any condition or event that constitutes a Servicer Event of Default or that after notice or lapse of time or both would constitute a Servicer Event of Default or that notice has been given to the Servicer or any of its subsidiaries with respect thereto; (ii) of the occurrence of any event or change that has caused or evidences, either in any case or in the aggregate, a materially adverse effect on the business, a material portion of the Purchased Receivables, properties or financial condition of the Servicer to perform the services provided for herein or (iii) except for any Regulatory Trigger Event which has been disclosed in Parent's public filings with the SEC, the occurrence of any of the events described in the definition of Regulatory Trigger Event, whether or not Servicer has determined that such events will be material, a certificate of such Authorized Officer specifying the nature and period of existence of such condition, event or change, or specifying the notice given and action taken by any such Person and the nature of such claimed Servicer Event of Default, potential Regulatory Trigger Event, and what action the Servicer or any of its Affiliates, as applicable, has taken, is taking and proposes to take with respect thereto.

(e) Other Covenants of Servicer.

(i) At any time or from time to time upon the request of the Purchaser Agent, the Servicer will, at its expense, promptly execute, acknowledge and deliver such further documents and do such other acts and things as the Purchaser Agent may reasonably request in order to effect fully the purposes of this Agreement.

(ii) The Servicer shall duly satisfy all obligations on its part to be fulfilled under or in connection with each Purchased Receivable and the MoneyLion Instacash Terms and Conditions, as applicable, and will comply in all respects with all Requirements of Law in connection with servicing each Receivable and the MoneyLion Instacash Terms and Conditions, as applicable, the failure to comply with which would reasonably be expected to result in a Material Adverse Effect.

(iii) In addition, and without limiting any other provision herein, the Servicer agrees to cooperate and use its commercially reasonable efforts at all times in providing, at the Servicer's expense, the Backup Servicer (or any other applicable Successor Servicer) with reasonable access during normal working hours to Servicer's employees and to any and all of the books, records (in electronic or other form) or other information reasonably requested by the Backup Servicer (or any other applicable Successor Servicer) to enable the Successor Servicer to assume the servicing in respect of the Purchased Receivables. Subject to the following sentence, the Servicer's obligation to provide such access shall be applicable at all times (regardless of whether (x) the Servicer's obligations have been terminated hereunder or (y) a Servicer Event of Default has occurred). Unless a Servicer Event of Default has occurred and is continuing, the Backup Servicer (or any other applicable Successor Servicer) shall provide reasonable prior notice to the Servicer of any request for access pursuant to this paragraph, and such access shall be limited to four (4) times per year.

(f) <u>Keeping of Records and Books of Account</u>. The Servicer will maintain and implement administrative and operating procedures (including an ability to recreate records evidencing Purchased Receivables, including the Purchased Receivables Files, in the event of the destruction of the originals thereof), and keep and maintain all documents, books, records and other information reasonably necessary or advisable for servicing of Purchased Receivables.

(g) Merger, Consolidation or Sale of Assets. Any Person (i) into which the Servicer may be merged, amalgamated or consolidated, (ii) resulting from any merger, amalgamation or consolidation to which the Servicer is a party and in which the Servicer is not the surviving entity, (iii) that acquires by conveyance, transfer or lease substantially all of the assets of the Servicer or (iv) succeeding to the business of the Servicer, which Person shall in each case execute an agreement of assumption to perform every obligation of the Servicer under this Agreement, shall be the successor to the Servicer under this Agreement without the execution or filing of any paper or any further act on the part of any of the parties to this Agreement.

(h) Tax Returns. The Servicer will pay all material Taxes imposed upon it or any of its properties or assets or in respect of any of its income, businesses or franchises before any penalty or fine accrues thereon, and all claims (including claims for labor, services, materials and supplies) for sums that have become due and payable and that by law have or may become a Lien upon any of its properties or assets, prior to the time when any penalty or fine shall be incurred with respect thereto; provided that, no such Tax or claim need be paid if it is being contested in good faith by appropriate proceedings promptly instituted and diligently conducted, so long as adequate reserve or other appropriate provision, as shall be required in conformity with GAAP shall have been made therefor.

9. Negative Covenants.

(a) Servicer shall not resign from the obligations and duties hereby imposed on it except (i) by mutual consent of the Servicer and the Purchaser Agent, (ii) upon the determination evidenced by an opinion of counsel to the Servicer, in form and substance reasonably satisfactory to the Purchaser Agent, that its servicing duties hereunder are no longer permissible under Applicable Law and such incapacity cannot be cured by the Servicer, in which event the Servicer may resign as servicer or (iii) upon execution at the direction of the Purchaser Agent by the Successor Servicer of a successor servicing agreement (the "Successor Servicer Servicer shall have assumed the Servicer's responsibilities and obligations hereunder in accordance with the terms hereof.

(b) Notwithstanding anything to the contrary herein, Servicer will not sell, assign or otherwise transfer or dispose of any Purchased Receivables, without the prior written consent of the Purchaser Agent.

(c) The Servicer shall take no action which, nor omit to take any action the omission of which, would impair the rights of a Purchaser in any Purchased Receivables.

(d) <u>Collection Account</u>. Except as otherwise set forth herein or in any other Transaction Document, the Servicer shall not create or participate in the creation of, or permit to exist, any Liens other than Permitted Liens with respect to the Collection Account.

(e) Change of Name or Location of Chief Executive Office. The Servicer shall not change its name or its state of organization, move the location of its principal place of business and chief executive office, and the offices where it keeps tangible records concerning the Purchased Receivables from the location referred to in Section 7(m), unless the Servicer has given at least ten (10) days' (or such shorter period as agreed by the Purchaser Agent) prior written notice to the Purchaser Agent.

(f) <u>Servicing Guidelines</u>. The Servicer shall not amend or otherwise modify the Servicing Guidelines in any manner that would reasonably be expected to have a material adverse effect on the interests of the Purchasers without the prior written consent of the Purchasers. It is acknowledged and agreed that any modification to the Servicing Guidelines that would not be reasonably be expected to have a material adverse effect on the interests of the Purchasers (as determined by the Servicer in a commercially reasonable manner) may be made by the Servicer without consent from any Purchaser or the Purchaser Agent and without notice to any Purchaser or the Purchaser Agent.

10. Servicer Event of Default.

(a) Each of the following shall constitute a Servicer Event of Default (each, a "Servicer Event of Default"):

(i) failure of the Servicer to perform or comply with any term or condition contained in Section 2(a)(i)(2) and such failure shall remain unremedied for three (3) Business Days after receipt by the Servicer of notice from the Purchaser Agent of such default;

(ii) The Servicer shall default in any material respect in the performance of or compliance with any term contained herein (provided however, with respect to any terms contained herein that already include a material qualifier, this clause shall not be deemed to include a material qualifier), other than any such term referred to in any other Section of this <u>Section 10</u>, and such default shall not have been remedied or waived within ten (10) days after receipt by the Servicer of notice from the Purchaser Agent of such default;

(iii) any representation, warranty, certification or other statement made or deemed made by the Servicer or in any statement or certificate at any time given by the Servicer in writing pursuant hereto or thereto or in connection herewith or therewith shall be false in any material respect as of the date made or deemed made and such incorrect representation or warranty has (x) a material and adverse effect on the Purchasers (taken as a whole) and (y) is not remedied or waived within thirty (30) days after receipt by the Servicer of notice from the Purchaser Agent of such default;

(iv) the occurrence of an Insolvency Event with respect to the Servicer;

(v) Richard Correia shall cease to employed by the Servicer and a successor thereto appointed by Parent's board of directors (or similar body) is not employed in such Person's place within 180 days (or such later date as may be agreed by the Purchasers in their sole discretion) thereafter;

(vi) The occurrence of any Change of Control which would reasonably be expected to impair the Servicer's ability to perform its obligations under this Agreement.

then, and in each and every such case, so long as such Servicer Event of Default shall not have been waived by the Purchaser Agent, the Purchaser Agent may, in its sole discretion, in addition to whatever rights the Purchasers or Purchaser Agent may have at law or in equity to damages, including injunctive relief and specific performance, by notice in writing to the Servicer (such notice, a "**Termination Notice**"), terminate all the rights and obligations of the Servicer as servicer under this Agreement, effective as of the date specified in the Termination Notice (the "**Effective Termination Date**"). On or after the receipt by the Servicer of a Termination Notice or resignation of the Servicer pursuant to <u>Section 10(a)</u>, all authority and power of the Servicer to service the Purchased Receivables under this Agreement shall on the date set forth in such notice pass to and be vested in a successor servicer (the "**Successor Servicer**") appointed by the Purchaser Agent. On or prior to the Effective Termination Date, the Successor Servicer, the Purchasers and the Purchaser Agent shall enter into a successor servicing agreement.

(b) Servicing Transfer.

(i) Upon receipt of a Termination Notice, the Servicer shall terminate its activities as servicer hereunder on the Effective Termination Date. After the receipt by the Servicer of a Termination Notice, the Servicer shall continue to perform all servicing functions under this Agreement until the Effective Termination Date. The Servicer agrees to cooperate with the Purchasers, Purchaser Agent and any Successor Servicer in effecting the termination of the responsibilities and rights of the Servicer to conduct servicing hereunder, including the transfer to such Successor Servicer of all authority of the Servicer to service the Purchased Receivables provided for under this Agreement, including all authority over all Collections which shall thereafter be received with respect to the Purchased Receivables. The Servicer shall promptly and in any event within three (3) Business Days after appointment of a Successor Servicer prepare, execute and deliver any and all documents and other instruments reasonably requested by the Successor Servicer or the Purchaser Agent in connection with the succession by the Successor Servicer, place in the Successor Servicer's possession (or provide online access to view the Purchased Receivables Files posted on an electronic portal maintained by the Servicer or any third-party service provider) all records related to the Purchased Receivables, including the Purchased Receivables Files (other than any such records held by the Purchaser Agent or the Purchaser Agent's custodian) and do or accomplish all other acts or things reasonably necessary to enable the Successor Servicer, to assume the servicing in respect of the Purchased Receivables.

(ii) Without limiting the forgoing, in connection with the delivery of a Termination Notice under <u>Section 10(a)</u>, following as soon as possible (but, in any event within three (3) Business Days) after receipt of the Termination Notice, Servicer shall provide to Successor Servicer all Purchased Receivables Files (or provide online access to view the Purchased Receivables Files posted on an electronic portal maintained by the Servicer or any third-party service provider) relating to the Purchased Receivables and Servicer shall use commercially reasonable efforts to provide to Successor Servicer reasonable access to and transfer of the Purchased Receivables Files and reasonable access by Successor Servicer to its personnel, software, hardware, facilities, and employees in each case reasonably necessary for Successor Servicer to perform the duties of the Successor Servicer, including the initial delivery of the reports as set forth in the Successor

Servicing Agreement.

11. Term.

The Servicer shall commence servicing each Purchased Receivable on the date such Purchased Receivable is acquired by a Purchaser and shall continue, for so long as the Servicing Fee (or other fees or amounts due to the Servicer hereunder) continues to be paid in accordance with this Agreement, servicing such Purchased Receivable until the earliest of (a) the payment in full (or charge off in accordance with the Servicing Guidelines) of the amount outstanding under each such Purchased Receivable, (b) the date specified in <u>Section 10(b)</u> following the receipt by the Servicer of a Termination Notice, or (c) by written agreement of the Purchaser Agent and the Servicer.

12. Indemnification.

(a) Subject to <u>Section 12(b)</u>, but without limiting any other rights that the Purchasers, the Purchaser Agent or any of their respective Affiliates, officers, directors and employees (each, an "**Indemnified Person**") may have hereunder or under applicable law, Servicer hereby agrees to indemnify and hold harmless each Indemnified Person from and against any and all suits, actions, proceedings, claims, damages, losses, liabilities and expenses (including reasonable attorneys' fees and disbursements and other costs of investigation or defense, including those incurred upon any appeal, and whether any such damages, losses, liabilities or expenses are incurred by a Indemnified Person as an actual or potential party, witness or otherwise and including those relating to or arising from such Indemnified Person's enforcement of the Servicer's indemnification obligations hereunder) that may be instituted or asserted against or incurred by any such Indemnified Person arising out of or incurred in connection with the following (collectively, "**Indemnified Liabilities**"):

(i) reliance on any representation or warranty made or deemed made by the Servicer (or any of its officers acting in its capacity as such) under or in connection with this Agreement;

- (ii) the failure by the Servicer to comply with any term, provision or covenant contained in this Agreement;
- (iii) the occurrence of a Servicer Event of Default;
- (iv) any claim brought by any Person arising from any activity by Servicer in servicing, administering or collecting any Purchased Receivables; and
- (v) the gross negligence, bad faith or willful misconduct of the Servicer in the performance of its duties under this Agreement;

provided, that Servicer shall not be liable for any indemnification to an Indemnified Person to the extent that any such Indemnified Liabilities (a) result from such Indemnified Person's gross negligence or willful misconduct, as finally determined by a court of competent jurisdiction, or (b) constitute recourse for uncollectible Purchased Receivables.

(b) <u>Limitation of Liability</u>. IN NO EVENT SHALL ANY PARTY HERETO OR ANY OF THEIR RESPECTIVE AFFILIATES, BENEFICIARIES, ASSIGNEES OR SUCCESSORS (BY ASSIGNMENT OR OTHERWISE) BE LIABLE TO THE OTHER PARTIES HERETO OR TO ANY OTHER ENTITY FOR SPECIAL, INDIRECT, PUNITIVE OR CONSEQUENTIAL LOSS OR DAMAGE OF ANY KIND WHATSOEVER (INCLUDING, BUT NOT LIMITED TO, LOSS OF PROFIT), UNDER THIS AGREEMENT INCURRED OR CLAIMED BY ANY PARTY OR ENTITY (OR SUCH PARTY

OR ENTITY'S OFFICERS, DIRECTORS, STOCKHOLDERS, MEMBERS OR OWNERS), HOWEVER CAUSED, ON ANY THEORY OF LIABILITY, EVEN IF SUCH PARTY HAS BEEN ADVISED OF THE LIKELIHOOD OF SUCH LOSS OR DAMAGE AND REGARDLESS OF THE FORM OF ACTION.

13. Miscellaneous.

(a) Information Rights, Statements, Reports; Tax Matters.

(i) The Servicer shall provide the Purchaser Agent with such information concerning the Purchased Receivables as is reasonably necessary for the Purchaser Agent and the Purchasers to prepare their federal income tax return and comply with any reporting or audit requirements of Governmental Authorities.

(ii) Each party hereto acknowledges that payments on the Purchased Receivables may be subject to withholding taxes, including, but not limited to, backup withholding under Section 3406 of the Code and FATCA, if applicable, if the recipient of any payment under such Receivables fails to furnish certain documentation and information, such as its taxpayer identification number, or otherwise fails to establish an exemption from such applicable withholding taxes. The Servicer shall be entitled to deduct and withhold from any payments on the Purchased Receivable any amounts that the Servicer is required to deduct and withhold under the Code, Treasury Regulations or any other provision of applicable law. The Servicer shall not be required to pay any additional amounts on any payments under the Purchased Receivables with respect to withholding taxes that are withheld or deducted from such payments. The Purchased Receivables, shall furnish to the Servicer a duly executed and properly completed Internal Revenue Service Form W-9 or applicable W-8 (including any required statements or attachments thereto) and such other documentation as reasonably requested by the Servicer from time to time.

(b) All notices and other communications provided for herein shall be in writing and shall be delivered by hand or overnight courier service, mailed by certified or registered mail or sent by facsimile, portable document format ("**PDF**"), tagged image file format ("**TIFF**") or other electronic format sent by electronic transmission, as follows:

If to the Purchaser Agent or a Purchaser, to:

c/o Sound Point Capital Management, LP, as Purchaser Agent 375 Park Avenue, 33rd Floor New York, NY 10152 Attn: [*****] Tel: [*****] Email: [*****]

If to Servicer to:

MoneyLion Technologies Inc. 249-245 West 17th Street, 4th Floor New York, New York 10011 Attention: [*****] E-mail: [*****] Any party hereto may change its address, facsimile number or e-mail address for notices and other communications hereunder by notice to all of the other parties hereto. All notices and other communications given to any party hereto in accordance with the provisions of this Agreement shall be effective and be deemed to have been received (i) if delivered by hand, upon personal delivery, (ii) if delivered by overnight courier service, one (1) Business Day after delivery to such courier service, (iii) certified mail, upon receipt or refusal thereof, and (iv) if delivered by electronic mail, upon sender's receipt of confirmation of proper transmission.

(c) This Agreement shall be binding upon and inure to the benefit of the parties hereto and their successors and assigns; <u>provided</u> that, other than as set forth herein, the Servicer may not assign any of its obligations hereunder without the written consent of the Purchaser Agent. This Agreement may not be modified, amended, waived or supplemented unless any such modification, amendment, waiver or supplement is in writing and signed by the Servicer and the Purchaser Agent.

(d) THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER SHALL BE GOVERNED BY, AND SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK WITHOUT REGARD TO CONFLICT OF LAWS PRINCIPLES (OTHER THAN SECTIONS 5-1401 AND 5-1402 OF THE NEW YORK GENERAL OBLIGATIONS LAW) THEREOF.

(e) The descriptive headings of the various parts of this Agreement are for convenience only and do not constitute a part of this Agreement and shall not affect the meaning or construction of any of the provisions hereof.

(f) This Agreement may be executed simultaneously in any number of counterparts. Each counterpart shall be deemed to be an original, and all such counterparts shall constitute one and the same instrument. The parties agree that this Agreement, any documents to be delivered pursuant to this Agreement and any notices hereunder may be transmitted between them by email and/or by facsimile. The parties intend that faxed signatures and electronically imaged signatures such as .pdf files shall constitute original signatures and are binding on all parties. The original documents shall be promptly delivered, if requested.

(g) No failure to exercise and no delay in exercising, on the part of the Purchasers, any right, remedy, power or privilege under this Agreement shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege under this Agreement preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges provided under this Agreement are cumulative and not exhaustive of any rights, remedies, powers and privileges provided by law.

(h) <u>Confidentiality</u>. Each party hereto shall comply with the confidentiality provisions set forth in Section 14 of the Purchase Agreement. Further to the foregoing, Servicer agrees that it shall not use the name of Purchaser Agent or any Purchaser in the performance of its duties and obligations hereunder without the prior written consent of the Purchaser Agent.

14. Joinder of Additional Purchasers. Each Additional Purchaser shall join this Agreement as a Purchaser in all respects upon execution of a Joinder Agreement under and pursuant to Section 15 of the Purchase Agreement.

IN WITNESS WHEREOF, Servicer, Purchasers and Purchaser Agent have caused this Agreement to be executed and delivered by their respective officer or officers thereunto duly authorized.

SERVICER:

MONEYLION TECHNOLOGIES INC.

By: <u>/s/ Richard Correia</u> Name: Richard Correia Title: Chief Financial Officer

PURCHASER:

SP MAIN STREET FUNDING I LLC

By: <u>/s/ Kevin Gerlitz</u> Name: Kevin Gerlitz Title: Authorized Signatory

PURCHASER AGENT:

SOUND POINT CAPITAL MANAGEMENT, LP

By: <u>/s/ Kevin Gerlitz</u> Name: Kevin Gerlitz Title: Chief Financial Officer

EXHIBIT A

FORM OF DATA TAPE

[On file with Purchaser Agent] [Exhibit A to Servicing Agreement]

CERTIFICATION OF CHIEF EXECUTIVE OFFICER PURSUANT TO EXCHANGE ACT RULES 13a-14(a) AND 15d-14(a) AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Diwakar Choubey, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of MoneyLion Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 6, 2024

By: /s/ Diwakar Choubey

Diwakar Choubey Chief Executive Officer (Principal Executive Officer)

CERTIFICATION OF CHIEF FINANCIAL OFFICER PURSUANT TO EXCHANGE ACT RULES 13a-14(a) AND 15d-14(a) AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Richard Correia, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of MoneyLion Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 6, 2024

By: /s/ Richard Correia

Richard Correia President, Chief Financial Officer and Treasurer (Principal Financial Officer)

Certification of Chief Executive Officer Pursuant To Rule 18 U.S.C. Section 1350

In connection with the Quarterly Report on Form 10-Q of MoneyLion Inc. (the "Company") for the quarterly period ended June 30, 2024, as filed with the U.S. Securities and Exchange Commission (the "Report"), I, Diwakar Choubey, Chief Executive Officer of the Company, hereby certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to my knowledge:

- 1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: August 6, 2024

/s/ Diwakar Choubey

Diwakar Choubey Chief Executive Officer

Certification of Chief Financial Officer Pursuant To Rule 18 U.S.C. Section 1350

In connection with the Quarterly Report on Form 10-Q of MoneyLion Inc. (the "Company") for the quarterly period ended June 30, 2024, as filed with the U.S. Securities and Exchange Commission (the "Report"), I, Richard Correia, Chief Financial Officer of the Company, hereby certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to my knowledge:

- 1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: August 6, 2024

/s/ Richard Correia

Richard Correia President, Chief Financial Officer and Treasurer